Early Detection of Microvascular Dysfunction in Asymptomatic Individuals with Different Cardiovascular Diseases Risk Factors

HALA SHOKR

Doctor of Philosophy

Aston University

September 2021

© Hala Shokr, 2021

Hala Shokr, 2021 asserts her moral right to be identified as the author of this thesis.

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and that no quotation from the thesis and no information derived from it may be published without proper acknowledgement.

Thesis Summary

In health, the endothelium maintains the balance between opposing dilator and constrictor influences, while in disease, it is the common ground on which cardiovascular risk factors act and a site at which cardiovascular disease begins. Consequently, as such, the endothelial acts as a barometer of an individual's likely future cardiovascular health; however, the currently used cardiovascular diseases risk assessments scores tend to overlook the importance of endothelial dysfunction in the progress of cardiovascular diseases. Moreover, the presence and severity of endothelial dysfunction in apparently healthy individuals suggest considerable variability in pre-clinical risk that could be identified well before the onset of the disease. The earlier detection of cardiovascular diseases can improve the effectiveness of treatments and avoid long-term complications and may lead to longer survival.

This possibility has led to public health programs to recommend periodic screening examinations for detecting specific chronic diseases, for example, cancer, diabetes, and cardiovascular disease.

Therefore, this thesis aimed to investigate microvascular dysfunction as an early culprit in the pathophysiology of cardiovascular diseases in clinically healthy individuals with different cardiovascular risk factors.

As cardiovascular diseases are clearly multifactorial with risk factors that tend to cluster and interact, in an individual to determine the level of future disease risk and to increase the clinical accuracy of the conducted investigations, influences of systemic circulatory oxidative stress biomarkers on retinal microvascular function were also evaluated.

The principal sections and findings of this work are:

1. Microvascular Function and Oxidative Stress in Adult individuals with Early Onset of Cardiovascular Disease

 Prehypertension patients showed abnormal retinal vascular response to flicker light stimuli throughout the entire functional response curve for arteries. Systemic blood pressure and impairments in microvascular function at the retinal level correlated with established plasma markers for oxidative stress. 2. European Society of Cardiology/European Society of Hypertension versus the American College of Cardiology/American Heart Association guidelines on the cut-off values for early hypertension: A Microvascular Perspective

 Microvascular dysfunction was present in ESC/ESH grade 1 hypertension patients and not in stage 1 hypertension in the form of altered peripheral and retinal responses to systemic stimuli (blood occlusion, flickering light).

3. Dry Eye Disease is Associated with Retinal Microvascular Dysfunction and Possible Risk for Cardiovascular Disease

• Individuals with a positive diagnosis of Dry Eye Disease exhibit a higher risk for cardiovascular disease than age, sex-matched normal individuals. These alterations were correlated to circulatory blood cholesterols.

4. Novel Composite Early Risk Markers for Vascular Ageing and Risk for Cardiovascular Disease

• Study findings confirmed the systemic association between plasma oxysterol levels and gradual microvascular endothelial dysfunction at the retinal arterial level after the age of 30. This association was also correlated to decreased systemic antioxidant capacity without any significant change in NO concentrations.

5. Prediction of Cardiovascular Risk in Asymptomatic Individuals: A Symbolic Regression-Based Analysis of Single and Composite Vascular-Omics

• Study findings confirmed the predictive power of the telomere and retinal arteries function as a composite biomarker for predicting cardiovascular risk factors such as blood pressure, while the assessment of retinal vascular function is better predictor of chronological ageing than relative telomere length.

Key words: Endothelial dysfunction, dynamic retinal vessels analysis, oxidative stress, cardiovascular diseases, oxysterols, telomeres, Aging.

Dedication

This thesis is dedicated to my family and friends For their endless love, support, and encouragement

To my dad Abdelmohaimen Shokr, the man who have taught me everything I stand for and managed to change me, without ever trying to do so. You have raised me to be strong and independent, however, I will always be your little girl, and I will always need you. Thank you dad for teaching me to smile in the face of adversity and to believe in myself no matter what anyone says. I can only hope to one day become half the man you are. You are the person I will always aspire to be, love you dad. Thank you for everything.

To my mother Sanaa Neanaa, who has been a source of motivation and strength during moments of despair and discouragement. I know I challenge you a lot with my views and I don't always tell you how much I care about you, but these past four years would have been impossible without you and your prayers. I am forever thankful for and to you. I love you with everything I have, and more. Thank you for being my mom.

To my brothers Yasser, Ahmed, and Ayman, you always were the biggest support in my life and my ultimate saviour.

Through countless memories, laughs, and arguments, you have always been there for me. From saying "I hate you" growing up to "I miss you" to "Thank you" right now, I wouldn't be who I am without you. Thank you for being my rock.

To Didi, thank you for making me feel supported, loved, and cared for. Every single word of yours inspired me and gave me new motivation to continue. You already know it, but still, I will say it. You are the best gift life has gifted to me. Thank you, bestie.

To Lamia, thanks for walking with me when I needed support, thanks for walking ahead of me when I needed guidance and thanks for walking behind me when I needed someone to watch my back. Thanks for understanding me better than me.

To Hagar thanks for sharing an awesome and crazy friendship that always makes me smile and never let me down. Your existence is invaluable to me, and I appreciate everything about you. Thanks for being the shoulder I can always depend on.

Acknowledgements

First and foremost, I would like **to praise Allah the Almighty**, the Most Gracious, and the Most Merciful for His blessing given to me during my study and in completing this thesis.

I would like to sincerely thank my supervisor Dr. Doina Gherghel for her patient guidance, encouragement, and advice. I have been extremely lucky to have a supervisor who cared so much about my work, and who responded to my questions and queries so promptly. She helped me achieving a dream by allowing this thesis to be my own work while doing her best to put me on the right track. I also want to sincerely thank her for the innumerable opportunities she offered beyond the domain of my PhD work, which enriched me both personally and professionally, my gratitude is beyond words.

I would also like to express my deepest sense of gratitude to Dr. Irundika Dias for bringing the weight of her experience and knowledge to this thesis. It was only because of her keen interest and continuous supervision that gave my work this extent form. She introduced me to laboratory research, helped me in overcoming what seemed like insurmountable obstacles throughout my PhD.

It has been a great enriching experience to work under Dr. Dias authoritative guidance, my appreciation is immeasurable.

To Dr. Mandeep Marwah thank you for being my support system. You always provided the words of love and encouragement that I needed to get through the rough patches. You are such a great blessing and every day I thank God for your mere existence. The journey would not have been possible without you.

I am highly indebted to Dr. Lorena Diaz for her constant help and willingness to share her vast knowledge in biomedical research. Lori your kindness and generosity know no bounds I am forever grateful to have you in my life.

To Dr. Yolanda Correia, thank you for always being just a phone call away any time of the day and for being a walking judgement-free zone. You were with me every step of the way and I appreciate everything that you have done for me, your genuine love and care is truly rare.

Many thanks and appreciations also go to all my friends in Aston Medical School. Thank you for your understanding and encouragement in my moments of crisis. Your friendship made my PhD a wonderful experience. Thank you for being my home away from home.

Table of Contents

1. Introduction	23
1.1. Anatomy and Physiology of the Cardiovascular System	24
1.1.1. The Heart	25
1.1.2. The Conduction System and the Cardiac Cycle	26
1.1.3. The Vascular System	28
1.1.3.1. Basic Structural Features of Blood Vessels	28
1.1.3.2. Arterial and Venous Components of the Circulatory System	30
a. Arteries and Arterioles	30
b. The Capillaries	30
c. Veins and Venules	31
1.1.4. Physiology of Circulation	33
1.1.4.1. Systemic Blood Pressure	33
1.1.4.2. Arterial Blood Pressure	34
1.1.4.3. Arterial Blood Pressure Regulation	35
1.2. Anatomy and Physiology of the Ocular System	37
1.2.1. The Ocular Circulation	37
1.2.2. Physiology of the Ocular Circulation	38
1.2.3. The Eyes: A Window to the Heart	38
1.3. The Endothelium	39
1.3.1. Nitric Oxide	40
1.3.1.1. Generation and Regulation of NO	41
1.3.1.2. Nitric Oxide Function	43
a. Modulation of the Vascular Tone	43
b. Regulation of Myocardial Contractility	43
c. Anti-Thrombotic Effect	45
d. Anti-Inflammatory Effect	45
e. Regulation of Endothelial-Leukocyte Interactions and	16
1.3.2. Endothelial Dysfunction	40
-	

1.4. Cardiovascular Diseases Predisposing Factors	48
1.4.1. Ageing	48
1.4.1.1. Telomere Length	49
1.4.1.2. Telomere Length and Oxidative Stress	50
1.4.1.3. Telomere Length and Cardiovascular Risk Factors	51
1.4.2. Blood Pressure and Shear Stress	52
1.4.3. Serum Blood Lipids	52
1.4.4. Oxidative Stress	52
1.4.4.1. Radical Formation	52
1.4.4.2. Radical Scavenging	54
1.4.4.3. Oxidative Stress Biomarkers	56
a. Glutathione Redox Buffer	56
b. Glutathione Redox Homeostasis and its Relation to	
Cardiovascular Diseases	57
c. Protein Carbonyl	59
d. Protein Oxidation and Cardiovascular Diseases	59
e. Lipid Oxidation	59
f. Oxysterols and Cardiovascular Diseases	60
1.4.5. Dry Eye Syndrome	62
1.5. Assessment of Cardiovascular Diseases Risk	65
1.5.1. Assessment of Endothelial Function	65
1.5.1.1. Invasive Techniques	65
1.5.1.2. Non-Invasive Techniques	65
1.5.1.3. Peripheral Vascular Reactivity	67
1.5.1.4. Endothelial Dysfunction as a Marker of Future Cardiovascular Risk	69
1.5.2. Cardiovascular Events Risk Scores	69
1.5.2.1. Non-Invasive Endothelial dysfunction Assessment Techniques Versus Ris	sk
Scores	71
1.5.3. Assessment of the Ocular Circulation	72
1.5.2.1. Ocular Blood Flow Assessment	72
1.5.2.2. Retinal Vascular Analysis	74
a. Background	74
b. Structural Retinal Changes	76
H.A. Shokr, PhD Thesis, Aston University 20	/ 21

c. Limitations of Structural Retinal Vascular Measurements	76
d. Functional Retinal Changes	76
e. Retinal Vascular Functionality and Hypertension	77
f. Retinal Vascular Functionality and Dyslipidaemia	78
g. Retinal Vascular Functionality and Ageing	78
2. Research Rational	78
2.1 Aims	79
3. Subjects and Methods	80
3.1. Ethical Approval	80
3.2. Patient Recruitment	80
3.2.1. General Inclusion/Exclusion Criteria	80
3.3. Methods	81
3.3.1. Experimental Protocol	82
3.3.2. Preliminary Assessments	84
3.3.3. Blood Pressure Measurement	84
a. Clinic Blood Pressure Monitoring	84
b. Ambulatory Blood Pressure Monitoring	84
3.3.4. Intraocular Pressure Measurements	85
3.3.5. Ocular Vascular Assessment	85
3.3.5.1. Dynamic Retinal Vessel Analysis	85
3.3.5.2. Device Setup	85
3.3.5.3. Technical Specifications	87
3.3.5.4. Advantages and limitations	88
3.3.5.5. Flicker light Stimulation	88
3.3.5.6. Measurement protocol	89
3.3.5.7. Normal Vascular Response	90
3.3.5.8. Reaction Mechanism	91
3.3.5.9. Data Analysis	92
3.3.5.10. Previous Methods of Analysis	93
3.3.5.11. Sequential and Diameter Response Analysis (SDRA)	94
	Q

3.3.5.12. Novel Analysis	. 95
3.3.6. Systemic Vascular Assessment (Digital thermal monitoring (DTM))	. 99
3.3.6.1. Measurement Protocol	. 99
3.3.7. Assessment of anterior ocular structure and function	100
3.3.7.1. Keratography	100
3.3.7.2. Procedure	101
3.3.7.3. Tear Meniscus Height	101
3.3.7.4. Non-Invasive Keratograph Break Up Time (NIKBUT)	102
3.3.7.5. Dynamic Evaluation of the Lipid Layer	102
3.3.7.6. Redness	103
3.3.7.7. Meibography	104
3.3.7.8. Data Analysis	104
3.3.8. Blood Sampling	106
3.3.9. Blood Analysis	106
3.3.10. Systemic circulatory markers – Biochemical assays	107
3.3.10.1. Glutathione/Glutathione Disulphide Redox System	107
a. Glutathione	107
b. Glutathione Recycling Assay Principle	107
c. GSH Assay	108
d. GSSG Assay	109
e. Analyte Concentration Calculations	110
3.3.10.2. Griess Assay for Nitrite	112
a. Protocol	112
3.3.10.3. Telomere Relative Expression	113
3.3.10.4. Extraction and liquid chromatography-tandem mass spectrometry of	
plasma oxysterols	114
3.3.11. Symbolic Regression	115
3.3.12. Statistical Analysis	118
4. Study 1: Microvascular Function and Oxidative Stress in Adult individuals With	
Early Onset of Cardiovascular Disease	119
4.1. Abstract	119
4.2. Introduction	120
4.3. Subjects and Methods	121
4.3.1. Study Participants	121

4.3.2. General Assessment	121
4.3.3. Blood Pressure Assessment and Patients Grouping	121
4.3.4. Dynamic Retinal Vessel Analysis	121
4.3.5. Biomarkers Assays	122
4.3.6. Sample Size	122
4.3.7. Statistical Analysis	122
4.4. Results	122
4.4.1. Correlations between Vascular and Systemic Circulatory Parameters	123
4.5. Discussion	128
4.6. Conclusion	129
5. Study 2: European Society of Cardiology/European Society of Hyperten	sion versus
the American College of Cardiology/American Heart Association guideline	s on the
cut-off values for early hypertension: A Microvascular Perspective	
5.1. Abstract	130
5.2. Introduction	131
5.3. Subjects and Methods	132
5.3.1. Study Participants	132
5.3.2. General Assessment	132
5.3.3. Biomarkers Assay	132
5.3.4. Blood Pressure Assessment and Patients Grouping	133
5.3.5. Dynamic Retinal Vessel Analysis	133
5.3.6. Digital Thermal Monitoring (DTM)	133
5.3.7. Statistical Analysis	133
5.4. Results	133
5.4.1. Retinal Vascular Response	134
5.5. Discussion	139
5.6. Conclusion	140
6. Study 3: Dry Eye Disease is Associated with Retinal Microvascular Dysf	unction and
Possible Risk for Cardiovascular Disease	141
6.1. Abstract	141
	10

H.A. Shokr, PhD Thesis, Aston University 2021

6.2. Introduction
6.3. Subjects and Methods 142
6.3.1. Study Participants 142
6.3.2. General Assessment 143
6.3.3. Biomarkers Assays
6.3.4. DED Diagnosis
6.3.5. Dynamic Retinal Vessel Analysis 143
6.3.6. Digital Thermal Monitoring (DTM)144
6.3.7. Sample Size
6.3.8. Statistical Analysis
6.4. Results
6.5. Discussion 150
6.6. Conclusion
7. Study 4: Novel Composite Early Risk Markers for Vascular Ageing and Risk for
Cardiovascular Diseases
7.1. Abstract
7.2. Introduction 153
7.3. Materials and Methods 154
7.3.1. Sample Size
7.3.2. General Investigations 154
7.3.3. Dynamic Retinal Vessel Analysis 154
7.3.4. Biomarkers Assays
7.3.5. Measurement of GSH and Oxidized Glutathione (GSSG)
7.3.6. Measurement of Nitric Oxide (NO) 155
7.3.7. Extraction and Liquid Chromatography-Tandem Mass Spectrometry (LC-MS/MS)
of Plasma Oxysterols 155
7.3.8. Statistical Analysis
7.4. Results
7.4.1. Clinical characteristics
7.4.2. Retinal microvascular function 155
11 H.A. Shokr, PhD Thesis, Aston University 2021

7.4.3. Oxidative Stress markers	160
7.5. Discussion.	162
7.6. Conclusion	163
8. Study 5: Prediction of Cardiovascular Risk in Asymptomatic Individuals: A	
Symbolic Regression-Based Analysis of Single and Composite Vascular-Omics	164
	104
	165
8.3. Subjects and Methods	166
8.3.1. Study Participants	166
8.3.2. General Investigations	167
8.3.3. Blood Analysis	167
8.3.4. Measurement of GSH and Oxidized Glutathione (GSSG)	167
8.3.5. Relative Telomere Length	167
8.3.6. Dynamic Retinal Microvascular Function Vessel Analysis	167
8.3.7. Sample Size	167
8.3.8. Statistical Analysis	168
8.3.9. Symbolic Regression	168
8.4. Results	168
8.4.1. Differences in Retinal Vascular Function	168
8.4.2. Symbolic Regression	173
8.4.3. Oxidative Stress markers	173
8.4.2.1. Age Prediction	173
8.4.2.2. Blood Pressure Prediction	176
8.5. Discussion.	180
9. General Summary and Discussion	182
9.1. Microvascular Function and Oxidative Stress in Adult individuals With Early Onset	of
Cardiovascular Disease	182
9.2. European Society of Cardiology/European Society of Hypertension versus the	
American College of Cardiology/American Heart Association guidelines on the cut-	off
values for early hypertension: A Microvascular Perspective	183
9.3. Dry Eye Disease is Associated with Retinal Microvascular Dysfunction and Possible	e

Risk for Cardiovascular Diseases1	83
9.4. Novel Composite Early Risk Markers for Vascular Ageing and Risk for Cardiovascula	ar
Disease1	84
9.5. Prediction of Cardiovascular Risk in Asymptomatic Individuals: A Symbolic Regression	n-
Based Analysis of Single and Composite Vascular-Omics	85
10. General Conclusion & Limitations 1	86
11. Future Direction 1	87
11.1. Individuals-focused Studies 1	87
11.2. Improving the Predictive Power of Currently Used Risk Scores	87
11.3. Expansion of Preliminary Data 1	87
12. References 1	88
13. Appendices	226
13.1. Appendix a. Ocular Surface Disease Index (OSDI) 2	226
13.2. Appendix b. Dry Eye Questionnaire (DEQ-5) 2	227
13.3. Appendix c. Patient Examination Sheet 2	228
13.4. Publications 2	230
13.4.1. Study 1 2	230
13.4.2. Study 2 2	231
13.4.3. Study 3 2	232
13.4.4. Study 4 2	233

List of Figures

Figure 1.1: Anatomy of the Heart and the Myocardium	25
Figure 1.2: Intersection of the Heart Wall Showing the Three Cardiac Wall Layers	26
Figure 1.3: The Cardiac Conduction System	27
Figure 1.4: The Cardiac Cycle	28
Figure 1.5: The Vascular System	29
Figure 1.6: The Vasa Vasorum	30
Figure 1.7: Percentage Blood Volume in the Circulatory System	31
Figure 1.8: Blood Vessels Anatomy	32
Figure 1.9: Blood Pressure Gradient in the Systemic Circulation	33
Figure 1.10: Factors Affecting Arterial Blood Pressure	36
Figure 1.11: The Ocular Circulation	37
Figure 1.12: Endothelium-Derived Factors	40
Figure 1.13: Structural Design of Nitric Oxide Synthases	42
Figure 1.14: Structural Design of Soluble Guanylyl Cyclase	44
Figure 1.15: Endothelial Dysfunction and Atherothrombotic and Cardiovascular Diseases	47
Figure 1.16: Oxidative Stress and Endothelial Dysfunction	49
Figure 1.17: Glutathione Redox Cycle	50
Figure 1.18: Main Products of Enzymatic or Auto-Oxidation of Cholesterol	55
Figure 1.19: Mechanisms Involved in the Formation of Advanced Lipoxidation, End-Produc	cts,
Relevant Targets and Their Involvements in Cardiovascular Diseases	57
Figure 1.20: Biological versus Chronological Vascular Ageing	61
Figure 1.21: Telomere Structure	62
Figure 1.22: Precorneal Tear Film	64
Figure 1.23: Schematic Figure Depicting Digital Thermal Monitoring Patient Setup	68
Figure 3.1: Flowchart of the Experimental Protocol	83
Figure 3.2: Diagrammatic Representation of the Dynamic Retinal Vessel	
Analyser Set-up	86
Figure 3.3: Example of Retinal Vessel Selection before Analysis	89
Figure 3.4: Breakdown of the 350 Second DVA Testing Period	90
Figure 3.5: Diagrammatic Representation of a Typical Retinal Vessel Response to Flick	ker
Light on Dynamic Retinal Vessel Analysis	91
Figure 3.6: Local Temporal Course Representation of Retinal Vessel Analysis	93
Figure 3.7: Diagrammatic Representation of the SDRA Parameters.	97
Figure 3.8: Graphical Representation of The Digital Thermal Monitor Software Analysis. 1	00
Figure 3.9: Example of TMH Measurement. 1	01
Figure 3.10: Example of NIKBUT Measurement	02
H.A. Shokr, PhD Thesis, Aston University 20	14)21

Figure 3.11: Example Measurement of Dynamic Evaluation of Lipid Layer.	103
Figure 3.12: Measurement Example of Bulbar Redness	103
Figure 3.13: Example of Infrared Superior and Inferior Meibography.	104
Figure 3.14: Measurement Image representing Area of Loss of the Superior Tarsus	105
Figure 3.15: Standard Curve for the GSH assay	109
Figure 3.16: Glutathione Recycling Assay Principle.	109
Figure 3.17: Standard Curve for the GSSG Assay	110
Figure 3.18: Nitrite Standard Curve Reference	113
Figure 3.19: Symbolic Regression tree for age group 1	117
Figure 4.1: Comparison Of Retinal Arterial Response Profile Across Groups	127
Figure 5.1: Comparison of Retinal Arterial and Venous Response Profile across Groups	138
Figure 6.1: Comparison of Retinal Arterial Response Profile across Groups	148
Figure 6.2: Correlation between Retinal Arterial Responses and Systemic Blood Lipids	149
Figure 7.1: Comparison of Retinal Arterial Response Profile across Groups	158
Figure 8.1: Comparison of Retinal Arterial Response Profile across Groups	172
Figure 8.2: Predicted Age vs Actual Age Using Models Based on Artery Data, Telon	nere
Relative Expression and Artery and Telomere Combined Data	175
Figure 8.3: Predicted Systolic Blood Pressure vs Actual Systolic Blood Pressure U	sing
Models Based on Artery Data, Telomere Relative Expression and Artery and Telon	nere
Combined Data	177
Figure 8.4: Predicted Diastolic Blood Pressure vs Actual Diastolic Blood Pressure U	sing
Models Based on Artery Data, Telomere Relative Expression and Artery and Telon	nere
Combined Data	179

List of Tables

Table 1.1: The ESC/ESH versus the ACC/AHA Hypertension Guidelines	35
Table 1.2: Tissue Distribution of NOS Isoforms in the Body	43
Table 1.3: Different Types of ROS and RNS Produced in the Cell	54
Table 1.4: Enzymatic and Nonenzymatic Antioxidants that Protect Against ROS/ RNS	
Generation	. 56
Table 1.5: Advantages and Limitations of the Most Common Noninvasive Techniques U	sed
to Assess Endothelial Function	58
Table 1.6: Relationship between GSH levels and Cardiovascular Diseases	67
Table 1.7: Comparison of Frequently Used Cardiovascular Risk Prediction Tools	70
Table 1.8: Techniques of Ocular Blood Flow Measurement and Assessment	73
Table 1.9: Measurement of Retinal Vascular Caliber	75
Table 3.1: Overview of Main Investigative Techniques/Clinical Parameters Measured	. 81
Table 3.2: Technical Specifications of Retinal Vessel Analyser	87
Table 3.3: Advantages and Limitations of Retinal Vessel Analyser	88
Table 3.4: Summary of DVA Parameters Calculated and Used for Analysis	98
Table 3.5: Telomere Primers Genomic Sequence	114
Table 3.6: Artery Measurements and Telomere Encodings for the Symbolic Regression	
Formulas	116
Table 4.1: Summary of the Systemic Characteristics of the Study Participants	124
Table 4.2: Summary of Retinal Arterial Vascular Function Parameters	125
Table 4.3: Summary of Retinal Venous Vascular Function Parameters	126
Table 5.1: General Characteristics of the Study Population	135
Table 5.2: Summary of Retinal Arterial Vascular Function Parameters	136
Table 5.3: Summary of Retinal Venous Vascular Function Parameters	137
Table 6.1: General Characteristics of the study Population	145
Table 6.2: Summary of Retinal Arterial Vascular Functional Parameters	146
Table 6.3: Summary of Retinal Venous Vascular Function Parameters	. 147
Table 7.1: Summary of the Systemic Characteristics of the Study Participants	. 156
Table 7.2: Summary of Retinal Arterial Vascular Functional Parameters	. 157
Table 7.3: Summary of Retinal Venous Vascular Function Parameters	. 159
Table 7.4: Summary of Normalized Oxidative Stress and Inflammatory Systemic Circula	tory
markers	161
Table 8.1: General Characteristics of the Study Population	170
Table 8.2: Summary of Retinal Arterial Vascular Functional Parameters	171
Table 8.3a: Mean Absolute Errors for age predictions (Age group 1)	174
Table 8.3b: Mean Absolute Errors for age predictions (Age group 2).	174
H.A. Shokr, PhD Thesis, Aston Universitv	16 2021

Table 8.3c: Mean Absolute Errors for age predictions (Age group 3)	174
Table 8.4a: Mean Absolute Errors for Systolic BP predictions (Age group 1)	176
Table 8.4b: Mean Absolute Errors for Systolic BP predictions (Age group 2)	176
Table 8.4c: Mean Absolute Errors for Systolic BP predictions (Age group 3)	176
Table 8.5a: Mean Absolute Errors for Diastolic BP predictions (Age group 1)	178
Table 8.5b: Mean Absolute Errors for Diastolic BP predictions (Age group 2)	178
Table 8.5c: Mean Absolute Errors for Diastolic BP predictions (Age group 3)	178

List of Equations

Equation 1.1: Calculation of Mean Arterial Blood Pressure	34
Equation 1.2: The Poiseuille Relationship	35
Equation 1.3: Ocular Perfusion Pressure Calculation	
Equation 3.1: Body Mass Index	84
Equation 3.2: Baseline Corrected Flicker Response	
Equation 3.3: Dilation Amplitude	
Equation 3.4: Dilation Slope	
Equation 3.5: Constriction Slope	
Equation 3.6: Friedewald Equation	106
Equation 3.7: Glutathione Oxidation	107
Equation 3.8: Oxidized Glutathione Reduction	108
Equation 3.9: Absorbance Change	111
Equation 3.10: Net Rate Calculation	111
Equation 3.11: GSH and GSSG Concentration	111
Equation 3.12: Total GSH Calculation	111
Equation 3.13: Redox Index Calculation	111

Abbreviations

ABPM	Ambulatory blood pressure monitoring				
AHA	American Heart Association				
ACC	American College of Cardiology				
ACCF	American College of Cardiology Foundation				
ACH	Acetylcholine				
ADMA	Asymmetric dimethylarginine				
aTR	Adjusted temperature rebound				
AU	Arbitrary Units				
	Area under the curve temperature rebound				
AVR	Arteriovenous ratio				
BCFR	Baseline Corrected Flicker Response				
BDF	Baseline diameter fluctuation				
BF	Blood Flow				
BFR	Baseline corrected flicker response				
BH4	Tetrahydrobiopterin				
BMI	Body mass index				
BP	Blood pressure				
bpm	Beat per minute				
CAD	Coronary Artery Disease				
cGMP	Cyclic guanosine monophosphate				
cNOS	Constitutive nitric oxide synthase				
CRAE	Central arteriolar equivalent				
CRVE	Central venular equivalent				
СОР	Cardiac Output				
CVDs	Cardiovascular Diseases				
DA	Dilation amplitude				
DBP	Diastolic blood pressure				
DED	Dry Eye Disease				
DEQ-5	Dry Eye Questionnaire				
DTM	Digital thermal monitoring				
DVA	Dynamic retinal vessel analysis				
EEG	Electrocardiogram				
EDCF	Endothelial derived constricting factor				
EDRF	Endothelial derived relaxing factor				
EDVF	Endothelial derived vasoactive factors				

EndoPAT	Endothelial Peripheral Arterial Tonometry			
eNOS	Endothelial nitric oxide synthase			
EPCS	Endothelial progenitor cells			
ESC	European Society of Cardiology			
ET-1	Endothelin-1			
FH	Family History			
FDA	Food and Drug Administration			
FMD	Flow mediated dilation			
GLUC	Glucose			
GSH	Reduced glutathione			
GSSG	Oxidised glutathione			
GSR	Glutathione Reductase			
GST	Glutathione Transferase			
GMP	Guanosine monophosphate			
H⁺	Hydrogen			
HDL-C	High-density lipoprotein cholesterol			
HF	High frequency			
HO-1	Heme oxygenase			
HR	Heart rate			
HRV	Heart rate variability			
IDEEL	Impact of Dry Eye on Everyday Life			
IL-6	Interleukin-6			
ІМТ	Intima-media thickness			
iNOS	Inducible nitric oxide synthase			
IOP	Intraocular pressure			
Kg	Kilogram			
L	Vessel length			
LDL-C	Low-density lipoprotein cholesterol			
L-FMC	Low flow-mediated constriction			
LTL	Leukocyte Telomere length			
m²	Square meter			
MAE	Mean Absolute Errors			
MAP	Mean arterial blood pressure			
MC	Maximum constriction			
MC%	Maximum constriction percentage			
MD	Maximum dilation			

MD%	Maximum dilation Percentage			
mg/dL	Milligrams per deciliter			
МІ	Myocardial Infarction			
M/L	Media-to Lumen ratio			
mmHg	Millimeters of mercury			
µmol	Micromole			
μM	Micromole			
mmol/L	Mill moles per liter			
n	Blood Viscosity			
NADPH	Adenine dinucleotide phosphate			
nmol	Nanomoles			
nNOS	Neuronal nitric oxide synthase			
NO	Nitric oxide			
NO ₂	Nitrite			
NO ₃	Nitrate			
NOS	Nitric oxide synthase			
O ₂	Oxygen			
OBF	Ocular blood flow			
ост	Optical coherence tomography			
OHT	Ocular Hypertension			
ONH	Optic Nerve Head			
OPP	Ocular perfusion pressure			
OSDI	Ocular Surface Disease Index			
Oxy-LDL	Oxidized Low-density lipoprotein cholesterol			
ΡΑΤ	Peripheral Arterial Tonometry			
PDE	Phosphodiesterases			
PET	Positron emission tomography			
PP	Perfusion pressure			
PVR	Peripheral vascular resistance			
QCA	Quantitative coronary angiography			
R	Vessel resistance			
r	Internal radii			
RBCs	Red blood cell			
RNS	Reactive nitrogen species			
ROS	Reactive oxygen species			
RT	Reaction time			
I				

RTL	Relative Telomere Length			
RVA	Retinal vascular analyzer			
S1P	Sphingosine-1			
SA node	Sinoatrial node			
SBP	Systolic blood pressure			
SDRA	Sequential and diameter response analysis			
SD	Standard deviation			
sGC	Soluble guanylyl cyclase			
SEE	Salisbury Eye Evaluation Questionnaire			
Slope _{AC}	Arterial constriction slope			
SlopeAD	Arterial dilation slope			
Slopevc	Venous constriction slope			
Slopevd	Venous dilation slope			
SVR	Systemic vascular resistance			
T-CHOL	Total Cholesterol			
TG	Triglycerides			
t-GSH	Total glutathione			
TL	Telomere Length			
tMD	Time to maximum dilation			
tMC	Time to maximum constriction			
TR	Temperature rebound			
UM	Units of measurement			
VWF	von Willebrand factor			
хо	Xanthine oxidase			
7-KC	7-Ketocholesterol			
25-OHC	25-hydroxycholesterol			
27-OHC	27-hydroxycholesterol			
4β-ОНС	4 beta-hydroxycholesterol			
7β-ОНС	7 beta-hydroxycholesterol			
7,27-OHC	7,27-hydroxycholesterol			
7,25-OHC	7,25-hydroxycholesterol			

1. Introduction

Cardiovascular diseases (CVDs) are a broad group of heart and blood vessels disorders that include coronary artery diseases, heart failure, cerebrovascular diseases, rheumatic heart diseases and many other conditions.¹ They are the main leading cause of death worldwide and responsible for over 18 million deaths each year (representing 31% of all global deaths, including approximately 35% of deaths in developed countries and 32% in developing ones).² CVDs are directly linked to several risk factors, including diabetes, advanced age, obesity, dyslipidaemia ethnicity, family history and genetic predisposition.³ In the past decade, early identification of those at risk gained considerable scientific attention due to its significant role in decreasing CVDs- associated mortalities and clinical complications.⁴ Given this, several risk prediction models were developed; however, the reliability of these risk scores was always a constant cause of concern.⁵ The majority of their validation data was based on small studies conducted in developed countries⁶ and, it was found that most of them either over or underestimate the future risk of developing CVDs.⁷ Moreover, these scores were mainly tested in White populations; therefore, using them to classify CVD risks in different races was found clinically misleading, increased unnecessary medical interventions or led to missing vulnerable cases.^{8,9} Accordingly, in the past few years, many techniques have been developed to facilitate the assessment of cardiovascular function; however, it was challenging to accurately predict the risk of CVDs due to the difficulties associated with the invasive procedures used in visualising the heart and the blood vessels.^{10,11} Subsequently, finding less invasive and more accessible ways to examine the cardiovascular system became a must.⁵ Microvascular endothelial function and structure measures were of particular interest in this regard, as they have recently emerged as a potential surrogate marker of future cardiovascular events. Additionally, small arteries remodelling was found in most of the cardiovascular aetiologies and represented a relevant factor in terms of the development of target organ damage or events.¹²

Of all the examined microvascular beds, retinal vessels were found to exhibit very close features comparable to the cardiovascular system. One of the most prominent features shared by the ocular and cardiac units is the nature of their vascular supply, with the microcirculation in particular demonstrating a large number of anatomical and physiological similarities.¹³ These similarities have raised the question of whether the two systems may share similar aetiologies. Subsequently, functional and structural retinal vascular analysis techniques were used to assess the endothelial function in both systems.¹⁴ One of the most advanced techniques used is the dynamic retinal vascular analysis (RVA), which allowed direct and precise CVDs complications, and had a vital prognostic significance in individuals with established cardiac diseases.¹⁵ However, little was published regarding its importance in predicting disease development in low risk and clinically asymptomatic individuals. Given that, ²³

H.A. Shokr, PhD Thesis, Aston University 2021

this thesis aims to explore this possible association and additionally address the question of whether functional assessment at the ocular level could prove effective as an indirect measure of systemic cardiac and vascular functions and hence whether the retinal vessels could effectively be used as a surrogate marker for subclinical cardiovascular system function.

Of particular interest to this thesis is the involvement of cofounder factors in the aetiology of cardiovascular diseases and endothelial dysfunction. Although the participation of vascular factors in the aetiology of both cardiac and vascular diseases has been increasingly realised, the nature of this involvement is still somewhat uncertain, and many questions remain. As such, this thesis additionally aims to explore the presence and aetiological relevance of physiological disorders at both the ocular and systemic levels in apparently healthy individuals to enhance our understanding of the pathological mechanisms involved in this population. Age-related vascular physiological changes and oxidative stress are of particular interest to this thesis.

To provide a foundation for the research conducted in this thesis, the following sections will first outline the normal cardiac, vascular and ocular anatomy and physiology, which is necessary to aid in interpreting this work. It will then discuss and outline the Background and current aetiological thinking, firstly for endothelial function, and secondly for the factors leading to its dysfunction and how it is connected to the development of CVDs.

1.1. Anatomy and Physiology of the Cardiovascular System

The circulatory system is a complex system that consists of two main components: the **cardiovascular** and the **lymphatic** system.¹⁶ The cardiovascular system consists of the heart, the blood vessels and the blood, while the lymphatic system consists of lymphatic microvessels (capillaries) and the larger lymph vessels.¹⁷

Blood compromises approximately 7% of the body's volume; regardless of its small volume, it plays a fundamental role in controlling all body functions while distributed through the vascular system. In the last centuries, many scientists, such as the ancient Greek physician Galen, proposed that blood moves through the body like an ocean tide, first moving out from the heart and then ebbing back in the same vessels.¹⁸ In the 1620s, after the massive developments in vascular imaging techniques, an English physician (William Harvey) was the first to describe the vascular system as a circulatory system.¹⁸

Using Dr Malpighi visualising method, he described the vascular system as an interconnecting network spreading all over the body.¹⁹ Later, scientists discovered that the vascular system consists of different vessels (arteries, veins, and capillaries), forming a closed circle that begins and ends in the heart.²⁰

Another wrong description of blood vessels is a system of pipes. Unlike rigid pipes, blood vessels are dynamic; they dilate, constrict and pulsate.²¹ They are also the key factor in

24

maintaining and balancing all living cells' vital functions in the human body, e.g. reproduction, metabolism, and growth.²² To balance all these functions, vessels must exhibit similar cellular structures yet be different enough to fulfil different physiological roles. For example, the coronary arteries in the heart are responsible for supplying oxygenated blood to the heart, while those in the kidney are required for the selective filtration of blood waste products.¹⁶ An overview of the cardiovascular system with relevance to this thesis is given in the following sections.

1.1.1. The Heart

The heart is a muscular organ that weighs almost 350 grams and is considered the strongest muscle in the human body. It is the main organ in the cardiovascular system that contracts rhythmically and autonomously.²³

The heart is located between the lungs in the middle of the mediastinal cavity of the thorax, and its end extends downwards to the left between the second and the fifth intercostal space. Anatomically, its cavity divided into four chambers, two atria and two ventricles separated by the cardiac valves that regulate the blood flow direction (Figure 1.1).²⁴



The cardiac muscles consist of three layers; the **myocardium**, which is the thickest middle layer of the heart wall. This layer lies between the single-cell endocardium layer, which lines the inner chambers. The outer epicardium, also known as the visceral pericardium, forms the

H.A. Shokr, PhD Thesis, Aston University 2021

pericardium's inner layer that surrounds and protects the heart from the outside. The primary function of the pericardium is to protect the heart and prevent its over-expansion inside the thorax cavity.²²

The inner layer of the heart wall is the **endocardium;** it lines the inner heart chambers, heart valves and is continuous with the endothelium of the large blood vessels (Figure 1.2).²⁴



Figure 1.2: Intersection of the Heart Wall Showing the Three Cardiac Wall Layers Adapted from the Atlas of Human Anatomy ²⁶

1.1.2. The Conduction System and the Cardiac Cycle

One of the unique characteristics of the cardiac muscles is their conduction system which gives them the ability to depolarise and repolarise continuously. The right atrium's sinoatrial node (SAN) is the heart pacemaker; it generates electrical impulses that start a sequence of electrical excitation all over the heart and stimulate its contraction. The electrical impulses generated by the SAN stimulates the atria to contract, then it travels to the interatrial septum and stimulates the atrioventricular node (AVN).²⁷ The AVN is a tissue located between the right atrium and the left atrium, and it provides a pathway of electrical conduction between the atria and the ventricles.²⁸ The fibres in the AVN is smaller than the SAN, which cause a 0.1-second delay in conduction and give the atria the time to contract and empty its blood content before the next ventricular contraction. The impulses are then transmitted through the bundle

26

of His branches to the two ventricles. The Purkinje fibres then continue down to the inferior aspect of the heart before looping upwards and travelling in the lateral aspects of the right and left ventricles (Figure 1.3).^{27,29,30}



Adapted from the EMS education Website ³¹

The contraction phase of the heart chambers is called systole, and the relaxation phase is referred to as diastole. One complete cycle of these events is referred to as the cardiac cycle.³² The change of the blood pressure inside the cardiac chambers during the cardiac cycle affects the opening and closure of the interchamber valve, thereby regulating blood flow between the chambers. Although the same volume of blood is pumped per each cardiac beat, the blood pressure in the left side of the heart is almost five times the right side.³³ These changes of the intercardiac blood pressure areas in the heart (Figure 1.4).³²



Figure 1.4: The Cardiac Cycle **Top diagram** represents ECG cardiac signals, pressures and volumes. **Bottom diagram** illustrates blood circulation in heart cavities at each phase. Adapted from the Atlas of Human Anatomy ²⁶

1.1.3. The Vascular System

1.1.3.1. Basic Structural Features of Blood Vessels

The vascular system, with few exceptions, share common physiological and histological features. Generally, blood vessels consist of three layers; each layer is named a `tunic', a Greek word meaning a membrane covering an organ. ^{34,35} These layers are different in the amount of smooth muscle and elastin they contain.

Anatomically inside out these layers are (Figure 1.5):

- The tunica intima
- The tunica media
- The tunica adventitia



The first layer (**tunica intima**) is the thinnest layer of the three, consists of a single endothelial cell layer based on a basal lamina (basement membrane).³⁷ Underneath is the sub-endothelial layer, a fibro-connective layer attached to an internal elastic layer that gives the endothelial cells elasticity and flexibility. The second layer in the vascular wall is the **tunica media**, consisting mainly of elastin fibres and smooth muscle cells. The tunica media is supported by the external lamina that provides its structure. The third and outermost layer of blood vessels is the **tunica adventitia**, which is entirely fibro-elastic connective tissue.³⁸

In large blood vessels, simple diffusion of oxygen is minimal, and the endothelium is the only layer that can receive oxygen and nutrition from the blood carried by the vessels.²⁹ Arteries and veins have their own blood supply network through small thin-walled blood vessels located outside the adventitia called the vasa vasorum (vessels of the vessels) (Figure 1.6).^{35,39}



Figure 1.6: The Vasa Vasorum Adapted from the clinical anatomy associates ⁴⁰

1.1.3.2 Arterial and Venous Components of the Circulatory System a. Arteries and Arterioles

Arteries carry oxygenated blood ejected from the left ventricles; the only exception is the pulmonary artery that carries deoxygenated blood from the right ventricle to the lungs.¹⁶ Based on cell histology, arteries can be classified into elastic arteries and muscular arteries.⁴¹ Elastic arteries are also known as **conducting** arteries; they are large arteries placed near the heart and can carry large amounts of blood (e.g. the aorta). Anatomically these arteries' walls have several layers of perforated elastic membrane (Figure 7); therefore, they are more flexible to expand, and relax rapidly to accommodate changes in blood volume during the cardiac systole and diastole.⁴² The large population of arteries in the body are muscular (**distributive** arteries). Their walls mainly consist of a massive layer of smooth muscles and few elastic fibres. Their primary function is to distribute an adequate amount of blood to the organs according to their needs.⁴²

Arterioles are smaller in diameter blood vessels extending and branching out from the arteries leading to the capillaries. The function of these arterioles is to deliver oxygen and nutrition to the organs and reduce blood flow reaching the capillaries to prevent their damage.⁴³

b. The Capillaries

The arterial vascular system transitions to the venous system through the capillary network. Capillaries are the thinnest vessels (about 0.25 mm thick) in the vascular system; they connect the arterial system to the venous tree and regulate the release of neurotransmitters at the sympathetic synapses.⁴³ They also regulate the microcirculation via series of complex interactions with autacoids, hormones, and neurotransmitters released at the sympathetic synapses leading to alteration of the smooth muscle tone in the arteriolar wall.⁴⁴

Walls of capillaries are marked histologically by a lack of smooth muscle, and instead, they have a single layer of endothelial cells surrounded by a basement membrane. ^{45,46}

c. Veins and Venules

After circulating in the arteries, the deoxygenated blood return to the heart via the venous network. The blood in the post-capillary venules moves to the larger veins then to the heart's right atrium.¹⁶ Veins size can vary from 1 to 10 mm in diameter. The transition from the capillaries to the veins characterised by the reappearance of muscle cells in the tunica media and elastic fibres in the adventitia. Small and medium-sized veins contain a high percentage of muscle cells in their tunica media, while larger ones have greater connective tissue levels.^{38,47} Compared to corresponding arteries, veins have thinner walls and larger lumens (Figure 1.8). With their large lumens, veins can accommodate around 60% of the body's total blood volume, and that is why they are named blood reservoirs (Figure 1.7).³⁸



Figure 1.7: Percentage Blood Volume in the Circulatory System

The second component of the venous system is the **venules**. They are similar to veins in function; however, they are formed from united capillaries. The post-capillary venules are the most miniature and consist of a single layer endothelium. Larger venules have one or two layers of smooth muscles and a thin externa. Unlike veins, venules are highly porous, allowing

free water movement and other molecules from the blood to the tissues, including white blood cells.³⁸



1.1.4. Physiology of Circulation

1.1.4.1. Systemic Blood Pressure

In engineering, it is well known that any fluid pumped in a closed circuit operates under pressure; the closer the liquid to the pump, the higher this pressure.⁴⁹ Blood circulation in the cardiovascular system is not an exception, as mentioned above, the heart is the pump responsible for generating the blood flow in the body; the resistance the blood face while circulating in the vessels generates the pressure and blood follow pressure gradient (flow from the higher pressure to the lower pressure areas.⁵⁰

As explained in Figure 1.9, the highest systemic blood pressure is in the aorta, and it drops throughout its path until it reaches 0 mm Hg in the right atrium. The sharpest drop happens in the arterioles as they are the most resistant to blood flow; nevertheless, as long as there is a pressure gradient, blood continues to circulate until it completes an entire cycle back to the heart.^{51,52}



Figure 1.9: Blood Pressure Gradient in the Systemic Circulation

Adapted from Physical Characteristics of the Circulation ⁵³

1.1.4.2. Arterial Blood Pressure

Arterial blood pressure is a parameter that reflects the elasticity of the arterial walls and blood volume. It is a function of the cardiac output (COP) and the peripheral vascular resistance (PVR). COP depends on the myocardium contractility and end-diastolic volume, while the PVR is the arterial resistance to the blood flow, and it depends on the artery diameter, length and blood viscosity.⁵⁴ As the left ventricle contracts and pumps blood to the aorta, the aortic pressure increases until it reaches its peak. This peak is termed the "systolic blood pressure" (SBP), and its average is 120 mm Hg in healthy young adults.⁵⁵ Following pressure gradient, blood moves towards the distal arteries, and during the diastole, the walls of the elastic arteries and the aorta recoil to sustain enough pressure to keep the blood moving forward into smaller vessels. During this period, the aortic valve closes to prevent the backflow of blood into the left ventricle; sequentially, the aortic pressure drops to its lowest level (70-80 mm Hg), which is called the diastolic blood pressure (DBP).⁵⁶ The difference between the systolic and diastolic pressures is called pulse pressure (PP).

As seen in Figure 9, aortic blood pressure fluctuates with each heartbeat. A mean value of the aortic pressure is used to reflect an accurate estimation of mean arterial blood pressure (MAP), and it is used as a mirror of the tissue pressure. As diastole lasts longer than the systole; consequently, MAP is not directly the mean of the SBP and DBP. Instead, it is DBP plus one-third PP (Equation 1.1).

MAP = DBP + 1/3 (PP) $MAP = DBP + \frac{1}{3} X (SBP - DBP)$ MAP = mean arterial blood pressureDBP = Diastolic blood pressureSBP = Systolic blood pressure

Equation 1.1: Calculation of Mean Arterial Blood Pressure

According to the SBP and the DBP patients are classified into different hypertension classes. The most recent classification of hypertension according to the European Society of Cardiology/ European Society of Hypertension ⁵⁷ and the American College of Cardiology classification of Arterial Hypertension ⁵⁸ are outlined in Table 1.1.

ESC/ESH 2018		ACC/AHA 2017			
Category	Systolic	Diastolic	Category	Systolic	Diastolic
	(mmHg)	(mmHg)		(mmHg)	(mmHg)
Optimal	<120	<80	Normal	<120	<80
Normal	120-129	80-84	Elevated BP	120-129	<80
High Normal	130-139	85-89	Stage 1	130-139	80-89
Grade 1	140-159	90-99	Stage 2	≥140	≥90
Grade 2	160-179	100-109	Hypertensive	≥180	≥120
			Crisis		
Grade 3	≥ 180	≥110			•

Table 1.1: The ESC/ESH versus the ACC/AHA Hypertension Guidelines

1.1.4.3. Arterial Blood Pressure Regulation

Arterial BP is controlled by changes in COP and systemic vascular resistance (SVR). SVR is determined by the anatomical features of the vascular system. As we explained before, the vascular structure remains relatively unchanged; however, some pathological conditions such as atherosclerosis, vascular thrombosis can affect the number of perfused blood vessels in the organs.⁵⁹ It was proven that hypertensive patients have a decrease in the anatomical number of the arterioles and the capillaries (rarefaction).

The most important factor affecting the SVR is the changes in vascular luminal diameter. According to the Poiseuille relationship, the SVR is inversely related to the vessel radius (Equation 1.2). ⁶⁰





Other vascular factors such as nitric oxide (NO), endothelin and prostacyclin also play an essential role in altering vessel diameter, discussed in detail in the coming sections. Similarly, some tissue factors such as adenosine and histamine can significantly affect organ blood flow when secreted from the parenchymal cells surrounding the vascular system.⁶¹

Finally, neurohumoral mechanisms regulated mainly by the baroreceptor and, to a lesser extent, the chemoreceptor play a significant role in regulating systemic vascular resistance and arterial pressure, particularly in certain forms of secondary hypertension.⁶²

The following diagram summarises the main factors regulating blood pressure by altering systemic vascular resistance and blood volume (Figure 1.10).



Figure 1.10: Factors Affecting Arterial Blood Pressure Abbreviation: SV, stroke volume; CO, cardiac output; MAP, mean arterial pressure; HR, heart rate; SVR, systemic vascular resistance; Na, sodium; H2O, water.

Adapted from CV Physiology 708
1.2. Anatomy and Physiology of the Ocular System

1.2.1. The Ocular Circulation

The eye has four different blood circuits (Figure 1.11): the ciliary body and the anterior part of the eye circulation; the retinal circulation, which lacks autonomic control similar to the brain circulation; the optic never head circulation; and the choroidal vasculature, which has the greatest autonomic innervations.⁶³ The retinal blood flow is completely auto-regulated and is not dependent on the blood perfusion pressure. Its blood flow is mainly dependant on the endothelium, the neural and neuroglia cells function (neurovascular coupling).⁶⁴ Unlike retinal circulation, choroidal circulation is mainly affected by psychological and physical stressors and temperature change, with cold increasing the choroid blood flow. ⁶⁵ The optic nerve circulation is affected mainly by the neurovascular coupling and some other mediators diffused from the choroid. ^{63,66,67}





1.2.2. Physiology of the Ocular Circulation

Similar to the systemic circulation, blood flow in the eyes depends on the perfusion pressure (OPP) and flow resistance generated by the vessels walls (R). Following the Hagen-Poiseuille pipe flow law, the blood flow (BF) equal PP / R; however, it is difficult to apply this law to microvessels due to the changes in red blood cells (RBCs) velocity profile, local haematocrit and the difference in the shear stress between branches, junctions and others.⁶⁹

Due to these challenges, scientists introduced another approach calculating blood flow in the ocular microcirculation based on Murray's Law.⁷⁰ According to this approach, the mean ocular PP driving blood through the eye is the mean blood pressure in the ophthalmic artery minus the pressure in the veins leaving the eye. The venous blood pressure was approximately equal to the intraocular pressure (IOP) ⁷¹, and the ocular PP is almost two-thirds of the mean arterial blood pressure. The factor 2/3 is due to the natural drop between the heart and the ophthalmic artery see Equation 1.3.^{72,73}

$$OPP = \frac{2}{3}MAP - IOP$$

OPP = Ocular perfusion pressure
MAP = Mean arterial blood pressure
IOP = Intraocular pressure

Equation 1.3: Ocular Perfusion Pressure Calculation

1.2.3. The Eyes: A Window to the Heart

For years scientists dealt with the eye and the heart as two different organs with no association. However, with advances in technology and research methods, it was discovered that these two organs have more common than expected. Vessels in the eye have their own uniqueness; nevertheless, they share many features with the heart and the circulatory vessels.³⁴⁷ Both vessels (retinal and cardiovascular) are exposed to the same intrinsic and extrinsic influences, and their response is the same in both organs. Hence, the vasculature of the eye is accessible and easily assessed; many scientists used it to evaluate the cardiovascular system functions.^{13,74}

1.3. The Endothelium

The endothelium is the largest organ in the human body. It is a layer of simple squamous cells lining the blood/ lymphatic vessels and the heart. Its weight is around 1.5 kg in average size individuals and covers a total surface area of 4000 to 7000 square meters.⁶⁹ In the beginning, it was thought it served only as a barrier between the vascular wall and the blood. However, later it has been shown to have a pivotal role in homeostasis in micro and macrocirculation through response to humoral, neural, and mechanical stimuli. It also helps synthesise and release vasoactive substances in the body (Figure 1.12).^{75,76}

As an autocrine, endocrine and paracrine organ, the endothelium interacts merely with every system in the human body. It helps supplying oxygen, nutrition, receives active metabolites and delivers them back to the circulation. Additionally, it has the ability to sense any stimuli on the vascular system (mechanical or hormonal), and in response, it regulates the vasomotor function by releasing some mediators that affect haemostasis, dilate or constrict the vessels, stimulate the inflammatory process or affect coagulation.⁷⁷

Examples of the vasodilatory substances produced by the endothelial cells are NO, prostacyclin and endothelium-derived hyperpolarisation factors such as c-natriuretic peptides, histamine, prostaglandins, vasoactive intestinal peptide, bradykinin and nicotinic acid. ^{78,79} Several vasoconstrictors are also secreted from the endothelium, e.g. angiotensin II, endothelin-1, thromboxane A2 and reactive oxygen species.^{80,81} Similarly, inflammatory mediators such as intercellular adhesion molecule-1 and vascular adhesion molecule-1 are produced by the endothelium tissue. Modulators of haemostasis also include plasminogen activator, tissue factor inhibitor, Von-Willebrand factor, NO, prostacyclin, thromboxane A2, plasminogen-activator inhibitor-1, and fibrinogen.⁸² It also has an essential role in angiogenesis and body fluids balance.

Under normal circumstances, the total of these endothelial-dependent factors is to maintain vasoconstriction, adequate blood flow, control vascular inflammatory mediators and muscular cell proliferation.⁷⁵ Nevertheless, when vascular risk factors are present, the endothelium may adopt a phenotype that facilitates inflammation, thrombosis, vasoconstriction, and atherosclerotic lesion formation.⁸³



Figure 1.12: Endothelium-Derived Factors ANG-II, angiotensin II; EDHF, endothelium-derived hyperpolarising factor; ET-1, endothelin 1; NO, nitric oxide; PGI 2, prostacyclin; TXA 2, thromboxane A 2.

1.3.1 Nitric Oxide

NO is a free radical diatomic gas synthesised and expressed in various tissues in the body, especially the cardiovascular and nervous systems.^{84,85} Its properties as a free radical and gas allow it to pass easily between the cells and the tissues and react with many molecules in the body.^{86–88} These biological and chemical properties enabled NO to be an ideal signalling molecule between tissues and cells and be a stimulator of many biological functions, especially in the cardiovascular system. Although the endothelium produces many other mediators, NO is considered the key molecule mediating all the endothelial functions.⁸²

Its effects on the cardiovascular system, e.g. vasodilatory, anti-thrombotic and antiinflammatory effects ⁸⁹ made it gain a huge interest over the years and became the core of many pharmaceutical treatments for various vascular diseases, e.g. hypertension, preeclampsia and peripheral vascular disorders.^{90,91}

It is also produced in various human body cells, including platelets, macrophages, and neuronal cells. Different concentrations of NO can activate different pathways in the human body; for example, low levels activate soluble guanylyl cyclase (sGC), and intermediate concentration (5-300 nM) activates wound healing and oncology pathways.⁹² Concentrations higher than 1Um cause oxidative, nitrative and nitrosative stress.^{92–94} Additionally, with

40

reactive oxygen species, it can cause modification of many human proteins, lipids and DNA strands.^{92,95}

1.3.1.1 Generation and Regulation of NO

The endothelial production of NO is a crucial step in regulating vascular blood flow, maintaining enough vascular tone and resistance to help circulating blood around the body.^{95–98} Over the years, scholars identified various mechanisms involved in the production of NO, yet until now, it has been agreed that the activity of NO synthase (NOS) is the key regulating step in NO production.^{99–102} The substrate for NO synthesis is the terminal guanidine nitrogen of the amino acid L-arginine. Catalysed by the NOS enzyme family and using one molecule of oxygen(O₂), iron protoporphyrin-IX (heme complex), oxidative cofactors (reduced glutathione and tetrahydrobiopterin (BH₄)), flavin adenine dinucleotide (FAD), flavin mononucleotide (FMN), and NADPH as cofactors; arginine undergoes 5-electron oxidation to form L-citrulline and a free NO radical.¹⁰³

NOS enzyme has three domains: **The oxygenase domain** binds the heme molecule, the BH4, and in which the formation of NO from arginine and O₂ takes place. **The reductase domain** uses NADPH to reduce the FAD, and FMN then transfers the free electron produced to **the oxygenase domain** through the calmodulin (CaM) binding domain (Figure 1.13).^{104,105} According to the physiological and biological properties, the NOS enzyme family, can be classified into two isomers:

- The constitutive isomers (cNOS) present mainly in the vascular endothelial (eNOS or NOS III) and neuronal cells (nNOS or NOS I).^{106,107}
- The inducible isomers (iNOS or Nos II) are found mainly in the macrophages ¹⁰⁸ and the neutrophils (Figure 13). The Inducible isomers are mainly activated immunologically by exposure to bacterial endotoxin or cytokines such as interleukin-6 (IL-6) and interferons.^{106,107}

Similarly, calcium ions, calmodulin and endothelial agonists such as acetylcholine and bradykinin control the stimulation of the two constructive isoforms.^{109,110}



Figure 1.13: Structural Design of Nitric Oxide Synthases

A: the oxygenase domain containing the heme and BH4 and connected to the reductase domain by a calmodulin link.

The reductase domain containing FMN, FAD and NADPH as an electron source. **B:** calmodulin electron transfer from the FMN domain to the heme domain.

C: three-dimensional structure model of NOS.

Adapted from Tejero J et.al. Sources of vascular nitric oxide and reactive oxygen species and their regulation.⁷⁰⁹

Table 1.	2. Tissue	Distribution	of NOS I	lsoforms	in the	Body
----------	-----------	--------------	----------	----------	--------	------

	Tissues	Cell Types
nNOS (NOS1)	Heart, lungs, and blood vessels	ECs, VSMCs, Adventitial fibroblasts, Cardiomyocytes
iNOS (NOS2)	Heart, lungs, and blood vessels	Macrophages and other leukocytes ECs, VSMCs, Adventitial fibroblasts
eNOS (NOS3)	Heart, lungs, and blood vessels	ECs, VSMCs, Erythrocytes, Platelets, Cardiomyocytes

Table 1.2: EC, endothelial cell; eNOS, endothelial NOS; iNOS, inducible NOS; NOS, nitric oxide synthase; nNOS, neuronal NOS; VSMC, vascular smooth muscle cell.

1.3.1.2 Nitric Oxide Function

a. Modulation of the Vascular Tone

NO's effects on the vascular wall is mediated by the soluble guanylate cyclase (sGC) receptor activation. ^{111–113} sGC is a heterodimeric cytosolic protein composed of two subunits, α and β . Each subunit has two isoforms (α 1, α 2 and β 1, β 2) which determine the tissue distribution of the protein ¹¹⁴. Each protein monomer consists of four domains the heme-binding domain, which is the N-terminal, the Per-Arnt-Sim (PAS) domain, a helical/coiled-coil motif, and a C-terminal guanylyl cyclase domain that catalyses the formation of cGMP from GTP (Figure 1.14). Once NO binds to the heme domain, it stimulates the activation of sGMP synthesis through the interaction with the C-terminal domain.¹¹⁵ NO activation of sGC increases cGMP synthesis, which consequently activates phosphodiesterases (e.g. PDE5) and cGMP-dependent gated ion channels and kinases, leading to vasodilation. ^{116–118}

b. Regulation of Myocardial Contractility

NO is directly involved in the control of myocardial contractility. Many scientists showed that NO reduces the contractility of the cardiac muscles via cGMP dependant mechanism.^{120,121} Both cNOS and iNOS are expressed in the ventricular muscles of the heart. eNOS and nNOS are constitutively expressed; they produce a minimal amount (less than 100 nM) of NO in the heart ^{122,123} with eNOS, particularly associated with the maintenance of basal, physiological cardiac function.¹²⁴

Myocardium contractility depends on the amount of NO produced; at lower levels (submicromolar), it causes a positive inotropic effect. However, at higher levels (micromolar or above), it causes negative inotropic effects.^{121,125} Hence, the impact of NO in the heart depends on which NOS isoform is activated and the amount of NO produced.

NO also helps maintaining adequate blood supply to the myocytes. In ischemia, the level of nNOS is upregulated, which inhibits xanthine oxidoreductase and prevents the formation of peroxide molecules.^{126–128} In addition, nNOS overexpression in ischemia downregulates the L-type calcium channels, decreasing the ischemic calcium overload and protecting the heart from reperfusion injury and progression of heart failure.^{129–131}



Figure 1.14: Structural Design of Soluble Guanylyl Cyclase **Top:** soluble guanylyl cyclase domains in the α and β subunits. **Bottom:** the interaction of the N-termini domains of the α and β subunits of soluble guanylyl cyclase.

Adapted from Childers KC. Structure/function of the soluble guanylyl cyclase catalytic domain.¹¹⁹

c. Anti-Thrombotic Effect

NO plays a vital role in endothelial protection against unusual vasoconstrictors and thrombus formation risks. The protective role of NO is also extending to the prevention of platelet aggregation and adhesion of foreign molecules to the endothelial cells. In hypercoagulopathy states, platelet aggregation stimulates the release of serotonin (5HT) and adenosine diphosphate (ADP), which augment NO production from the endothelial cells and, consequently, vasodilation.¹³² The relaxation of the blood vessels increases the blood flow to the affected area and stop thrombus formation.¹³³ NO also potentiate the effect of prostacyclin and prevent platelet aggregation stimulates vasospasm by the release of thromboxane A2, serotonin and haemostasis is initiated.^{133,136–139}

d. Anti-Inflammatory Effect

In recent years, increasing evidence supported the importance of NO in acute and chronic inflammation. They are also proved that treatment with NOS enzyme inhibitors reduces inflammation, especially in acute episodes where L-arginine enhances it.^{140,141} It has also been shown that NOS enzyme inhibitors can attenuate immune induced vasculitis in the lung and skin vasculature.¹⁴² Similarly, the colon expression of the NOS enzyme is augmented in inflammatory bowel diseases as ulcerative colitis and chronic ileitis.¹⁴³ In addition, scientists have reported that once the level of NO decreases, inflammatory responses leading to atherosclerosis is activated.^{144,145}

Despite all these proved anti-inflammatory/inflammatory actions of NO, its exact origin in the inflammatory process is still unclear; however, scholars have suggested it might come from the endothelial cells of the blood vessels, neutrophils or macrophages.¹⁴⁶ NO also plays an important role in tissue damage, and it is believed to be a cytotoxic and cytoprotective agent at the same time.

It is cytotoxic by itself in normal conditions, and even in some situations, its effect is enhanced by direct interaction with oxygen free radicals to produce more cytotoxic compounds. Even though NO's cytotoxicity is well established, some scientists also proved its cytoprotective effect in some tissue injuries.^{146,147} This dual functional nature of NO was explained by its vasodilator effect, which increases blood flow to the tissues and, consequently, protects it from harm.^{148–151} All these properties gave NO its multifaceted role in the inflammatory cascade, from the vasodilation effects to increased permeability, oedema formation, leukocyte stimulation, to tissue cytotoxicity.¹⁴⁶

iNOS was also found to be highly expressed in inflamed tissues, and its correlation with disease activity has been reported. Additionally, high doses of NO can trigger the necrotic and apoptotic pathways of cell death.¹⁵²

e. Regulation of Endothelial-Leukocyte Interactions and Vascular Permeability

NO inhibits the neutrophil aggregation and adhesion to the endothelium. It has also been reported that it increases microvascular permeability and vascular protein leakage, which are prevalent features of acute inflammatory response.¹⁵³ The antiadhesive effects of NO on the neutrophils are related to the superoxide interaction.¹⁵⁴ Superoxide enhances leukocytes adhesions to the vascular endothelium, and in the absence of NO, it activates the mast cells, causing degranulation that enhance leukocyte adhesion to the endothelium. Thus, a relative imbalance in superoxide anion and/or NO levels may promote mast cell degranulation, leukocyte adherence, and leukocyte emigration, thereby inducing acute inflammation.¹⁵⁴

1.3.2. Endothelial Dysfunction

Dysfunction of the endothelium has been implicated in the pathophysiology of different forms of CVDs, including hypertension, coronary artery disease, chronic heart failure, peripheral artery diseases, diabetes, and chronic renal failure.⁷⁷

After discovering NO and its importance, endothelial dysfunction was defined as the inability of the endothelial cells to release NO and the enhancement of the endothelium-derived contracting factors (EDCF) effects.¹⁵⁵ Later, endothelial dysfunction was identified as the hallmark of many CVDs and the first step in a chain of events that leads to atherosclerosis and coronary artery disease.¹⁵⁶

In the past decade, a plethora of studies showed that impaired vascular coagulation function, inflammatory properties, vascular growth and vascular remodelling are pathologically associated with endothelial dysfunction, with loss of NO bioavailability as the main feature in all these pathologies.^{157,158} Additionally, localised endothelial dysfunction was proven to lead to induction of the tissue factor, increased release of von Willebrand factor (vWF) factor, homeostatic shift towards coagulation and inflammation.¹⁵⁹ All these changes were linked directly to cardiovascular diseases and complications (Figure 1.15).¹⁵⁶ Another feature of endothelial dysfunction that was highlighted in cardiac patients is the impairment of the endothelium barrier; this dysfunction manifested as oedema, coupled with leukocyte migration, which are the fundamental signs of inflammation.¹⁶⁰



Figure 1.15: Endothelial Dysfunction and Atherothrombotic and Cardiovascular Diseases

1.4. Cardiovascular Diseases Predisposing Factors

1.4.1. Ageing

Ageing is one of the main risk factors of CVDs.¹⁶¹ Vascular ageing is a new terminology used to describe vascular dysfunction, structural changes and degradation, which is the ultimate cause of end-organ damage, especially in highly perfused organs, e.g. the heart, the kidneys and the brain.¹⁶² Although there is high inter-individual variability in disease onset and associated mortality, scholars have shown that age-dependent arterial injury typically manifests in the early fifties or sixties.¹⁶³

Over the years, scientists proved that individuals do not age at the same pace; many factors were identified as contributors to the ageing progress, such as lifestyle and exercise ¹⁶⁴, disease/health condition^{165,166}, genetic factors ¹⁶⁷ and family history.¹⁶⁵ This observation led to the concept of biological ageing or physiological ageing, which describes the decline in organs function due to the advanced age effect, whereas chronological ageing is the number of years since birth (Figure 1.16). ¹⁶⁸

Biologically, ageing reduces the endothelium ability to dilate and contract as It decreases the endothelial ability to respond to NO and EDH.^{169–171} Many researchers linked this dysfunction to the reduced production of NO, which is due to:

- Augmented arginase enzyme activity, competing with eNOS for the common substrate arginine ^{172–174}
- Increase the production of oxygen free radicals, which reduce the bioavailability of nitric oxide ^{175–177}
- Reduce the activity of eNOS ¹⁷⁸

Ageing was also found to affect the release of endothelium dependant prostanoids and can reduce the expression of sGC in the vascular smooth muscles.¹⁷⁹ Similarly, vascular contractions were reported as more prominent with any increase in age.^{180,181} The upregulation of cyclooxygenase enzymes explained this by increased oxidative stress ^{182–184} and the augmented expression of the prostacyclin synthase gene.^{184–186}





1.4.1.1. Telomere Length (TL)

Telomeres are repetitive non-signalling DNA components at the end of the circulating peripheral blood leukocytes' chromosomes (TTAGGG hexamer in the vertebrates) (Figure 1.17).¹⁸⁷ Its main function is coding sequence protection from degradation. A shelterin protein complex caps the telomere to protect it from DNA damage response and several forms of double-strand break repair. This complex function depends on both the stability of the nucleoprotein structure and length. ^{188–190}

The telomeric sequences are lengthened by DNA exchange during the mitosis phase in cell division by a reverse transcriptase enzyme named telomerase, which elongates the telomeres to maintain their function.^{191,192}

Most body cells (somatic cells) do not express telomerase; however, germ cells in the human stem cells, immune cells, and embryogenesis have active telomerase enzymes.¹⁹³ The absence or insufficient amount of telomerase results in TL attrition. Additionally, the destruction of the shelterin results in cellular senescence.^{190,194}

TL shorten in proliferating cells with each physiological cell division due to the inability of DNA polymerase to complete the chromosome termini, which is reflected in age-dependent

49

telomere attrition.^{192,195–197} Similarly, exposure to abnormal physiological conditions can also cause direct damage to the telomeric DNA, e.g. oxidative stress and inflammation, and some studies linked it to telomere attrition.¹⁹⁸

1.4.1.2. Telomere Length and Oxidative Stress

Oxidative damage to the telomeric DNA is mediated by the formation of guanine, 8-oxodG, which leads to disturbances in the maintenance of telomere length. Furthermore, ROS can induce breaks in the DNA and weaken base repair.¹⁹⁹ As discussed before, telomeres cannot repair breaks in single strand DNA ²⁰⁰ making it very sensitive to the accumulation of the guanine oxide.^{201–203} Additionally, non-matched bases in the telomeric sequence make it difficult for the replication mechanisms to maintain structural integrity, and oxidative stress can produce early telomere shortening regardless of age.²⁰⁴

Another critical point is the close correlation between oxidative stress and the inflammatory process in the human body.^{205,206} In this context, several studies reported a negative correlation between the telomerase enzyme, TL and exposure to tumour necrosis factoralpha.^{207,208} Following the same concept, proinflammatory produced cytokines can cause telomere shortening directly.^{209,210}





50

1.4.1.3. Telomere Length and Cardiovascular Diseases

Independent of the age effect on the TL, studies found a possible involvement of telomeres in the initiation and progression of CVDs. It was noticed that patients with established CVDs or even early stages of myocardial infarction had shorter TL than age and sex-matched controls.^{211–213} A direct correlation between TL in circulating cells such as leukocytes and TL in vascular tissues such as the aorta was also reported by some studies showing that leukocyte TL reflects telomere dynamics in tissues affected by CVDs, e.g. myocardium and coronary vessels.²¹⁴ This relation was also confirmed in coronary artery disease patients (CAD), although it was not possible to determine if it is a consequence or the cause of CAD.^{213,215} However, in most cases, CAD was in the context of atherosclerosis/thrombosis and cellular senescence²¹⁶, which made scholars suggest that shorter TL can cause a certain type of senescence in both endothelial and smooth muscles cells.^{215,216}

A correlation between CAD, TL and family history was also reported in a case-control study suggesting that the presence of short telomeres is a principal anomaly in atherosclerotic coronary diseases.²¹⁷ Similar correlation was found between shorter TL and increased mortality in CAD patients independent of any other risk factors.²¹⁸ Likewise, studies investigating the correlation between TL and HTN showed shorter telomere length in HTN patients compared to normotensive individuals.^{198,219,220}

In addition, various CVDs risk factors were found to be strongly associated with leukocyte TL. For example, a significantly negative correlation was reported between Leukocyte TL (LTL) and the body mass index (BMI) (r = -0.077) and smoking history (r = -0.087) in a cohort of 1122 women.²²¹ The negative effect of smoking on LTL was confirmed later in many studies and was further correlated to early onset of chronic obstructive pulmonary disease ^{222–224}, survival rate in smoker individuals ²²⁵ and airflow limitation in non-smokers.²²⁶ Similarly, as a metabolic risk factor, homocysteine was found to negatively correlate to the LTL in a cohort of 1319 subject 236 and the association between BMI and decrease in LTL association was confirmed in the same study (r = -0.106).²²⁷ Other risk factors such as atherosclerosis and elevated C-reactive protein were also reported to strongly correlate to shortened LTL.²¹⁸ To conclude, currently, it is still difficult to define the telomere role in the pathophysiology of CVDs, and further research is needed to confirm the ability of its use as a prognostic factor of CVDs.

1.4.2. Blood Pressure and Shear Stress

As discussed in section 1.1.4, blood vessels are under constant mechanical loading due to the blood pressure and flow, leading to increased internal stresses, e.g., endothelial shear stress and circumferential wall stress.

These mechanical forces cause not only morphological changes of endothelium and blood vessel wall but also trigger biochemical and biological events.²²⁸ There is considerable evidence that the increased frictional force of the blood on the vascular endothelium (wall shear stress) increases the blood flow and the pressure on the vascular endothelium, and this consequently stimulates endothelial nitric oxide synthase activity and NO production.^{229–232} Additionally, it was proven that disturbed blood flow and uncontrolled blood pressure patterns cause epigenetic DNA, RNA methylation and other changes that modify gene expression, increase oxidative stress, blunt all endothelium dependant responses, and enhance atherosclerosis.^{233–237}

Similarly, prolonged exposure to high blood pressure raises angiotensin I and II signalling and oxidative stress ^{238,239}, leading to endothelial dysfunction.²⁴⁰

1.4.3. Serum Blood Lipids

Scientists have proved that high-density lipoproteins (HDL) bind to a bioactive lipid mediator named sphingosine 1-phosphate (S1P) and its endothelial receptor promoting the action of eNOS. This action induces the antioxidant enzyme heme oxygenase-1 (HO-1), leading to increased release of NO, explaining the beneficial effects of high HDL levels on the endothelial function.^{241–243} Therefore, the agreement that higher HDL cholesterol levels are protective against endothelial dysfunction and CVDs. However, unlike other patients, higher HDL levels in CAD patients inhibit the action of eNOS and lose of its protective effect and stimulate platelet aggregation.^{244–246}

Additionally, Hypercholesterolaemia, particularly high concentrations of low-density lipoprotein (LDL) and triglycerides (TG), were found to associate with reduced endothelial dilation.^{247–250} Both the oxidised (Oxy LDL) and the carbamylated forms of LDL increase oxidative stress, which reduces the bioavailability of NO, attenuate its vasodilation effect ^{251–254}, impair the turnover rate of eNOS ^{255,256} and increase asymmetric dimethylarginine (ADMA), causing vessel stiffness.^{257,258}

1.4.4. Oxidative Stress

1.4.4.1. Radical Formation

Oxidative stress is the imbalance between the production of oxidant and antioxidant substances in the body. In various pathophysiological states, activation of nicotinamide– adenine dinucleotide phosphate (NADPH) oxidase, xanthine oxidase (XO), cyclooxygenase, eNOS when uncoupled with tetrahydrobiopterin (BH4) or lack of its substrate L-arginine and mitochondrial electron transport as well as inactivation of the antioxidant system, including superoxide dismutase, glutathione peroxidase, and catalase, lead to an increase in reactive oxygen species (ROS) production and decrease in ROS degradation.^{259–265} The reduction of oxygen through the addition of electrons leads to the formation of several reactive species, including (Table 1.3):

- Free radical species, which are usually highly reactive and have one or more unpaired electrons.
- **Non-radical species** that can be derived from either oxygen or nitrogen.

In normal physiological conditions, oxygen free radicals can activate eNOS and stimulate the production of NO and vasodilation. Some of these superoxide molecules can also undergo dismutation by the superoxide dismutase enzyme to form hydrogen peroxide (H₂O₂), which can act as endothelium-derived hydrogen peroxide and stimulate vascular endothelium relaxation.²⁶⁶ Additionally, the normal physiological amount of oxidative free radicals produced by the mitochondria can stimulate eNOS and help in the endothelial relaxation process.^{267–270} However, under oxidative stress, the natural physiological process in the biological systems where the presence of free oxygen radicals overpowers the radical scavenging mechanisms is impaired.²⁷¹ Thus, creating an imbalance between the oxidants and the antioxidants.

Reactive ROS binds to NO molecules forming peroxynitrite, which decrease the bioavailability of NO and impair its effect on the vascular endothelial.^{261,272–274} Adding to this, ROS can cause inactivation of eNOS by S-glutathionylation of the enzyme, subsequently, leads to endothelial dysfunction.^{275,276} Overproduction of ROS is associated with developing various human diseases (including cancer, cardiovascular, neurodegenerative, and metabolic disorders), inflammation, and ageing (Figure 1.18).

Radicals		Non-Radicals					
Reactive Oxygen Species							
0 ₂ -	Superoxide	H ₂ O ₂	Hydrogen peroxide				
ОН	Hydroxyl	HOCL ⁻	Hypochlorous acid				
RO ₂	Peroxyl	O ₂	Single oxygen				
RO	Alkoxyl	ONOO-	Peroxynitrite				
HO ₂	Hydroperoxyl						
Reactive Nitrogen Species							
NO	Nitric Oxide	ONOO ⁻	Peroxynitrite				
NO ₂	Nitrogen dioxide	ROONO	Alkyl peroxynitrites				
		N_2O_3	Dinitroge trioxide				
		NO ⁻	Nitroxyl anion				
		NO ₂ CL	Nitryl chloride				

 Table 1.3: Different Types of ROS and RNS Produced in the Cell

1.4.4.2. Radical Scavenging

The human body is equipped with a variety of antioxidants that serve to counterbalance the effect of oxidants.²⁷⁷ They can be classified into two categories: enzymatic and nonenzymatic (Table 1.4).

The body's ability to respond to oxidative stress is limited. Under conditions of sustained oxidative stress or excessive ROS production, the damage induced may exceed the capacity of the antioxidants and the bodies repair mechanisms leading to more permanent cell, DNA and protein damage.²⁷⁸

Oxidative stress can be measured directly by measuring the concentration of the reactive oxygen species and plasma oxidant/antioxidant balance, or indirectly by measuring the levels of DNA/RNA damage, lipid peroxidation, and protein oxidation/nitration.²⁷⁹



Figure 1.18: Oxidative Stress and Endothelial Dysfunction

Table 1.4: Enzymatic and Nonenzymatic Antioxidants that Protect Against ROS/ RNS

 Generation

Enzymatic antioxidants	Nonenzymatic antioxidants			
Thioredoxin	 Vitamin C, E, A 			
 Peroxiredoxins 	Thiols			
Glutaredoxin	 B- Carotene 			
 Glutathione peroxidase 	 Polyphenols 			
 Reduced glutathione (GSSG) 	 NAC 			
 Oxidised glutathione (GSH) 	 Zinc, selenium 			
 Glutathione reductase 	Glutathione			
Catalase	 Uric acid 			
 Peroxidase 	Lycopene			
 Superoxide dismutase 	 Allyl sulfide 			
	 Indoles 			
	Gallic acid			
	 Hesperidin 			
	Catechin			
	Chrysin			

1.4.4.3. Oxidative Stress Biomarkers

a. Glutathione Redox Buffer

One of the most commonly used methods to assess oxidative stress status in the body is measuring glutathione in the plasma. Glutathione has two forms; the reduced form (GSH) and the oxidized glutathione disulphide (GSSG). ^{280–282} The GSH/GSSG is the most efficient plasma and tissue oxidant/antioxidant system in the human body. Usually, this system is in balance; however, under oxidative stress conditions, it can be disrupted. As an antioxidant, GSH can prevent the devastating effects of ROS either directly through its oxidisation or indirectly by maintaining other cellular oxidants in a functional state.²⁸³

The product of GSH oxidation is glutathione disulfide. Within the cells, GSH is regenerated from GSSG by glutathione reductase using NADPH as an electron donor (Figure 1.19).^{284,285}

Under normal physiological conditions, GSSG represents only around 10% of the total glutathione pool of a healthy cell.²⁸⁶ However, under conditions of oxidative stress, ROS levels are raised, and hence the production of GSSG is increased significantly. As such, determining

the GSH: GSSG ratio or glutathione status of a cell is considered a good indicator of cellular redox imbalance and oxidative stress, especially as maintaining an optimal GSH: GSSG ratio is critical for cell survival. A reduced GSH: GSSG ratio would be considered indicative of oxidative stress.²⁸⁷





b. Glutathione Redox Homeostasis and its Relation to Cardiovascular Diseases

The first study to evaluate the relation between GSH and CVDs was conducted in Japan, results from this study showed that increased GSH levels decrease systolic and diastolic blood pressure and decrease the incidence of diabetes in cardiac patients. The same authors reported decreased plasma GSH levels in subjects with CVDs compared to individuals with no cardiac history.²⁸⁹

The effect of antihypertensive drugs was also evaluated in a study conducted by Chaves et al., GSH levels were decreased while GGSG increased in hypertensive patients compared to controls.²⁹⁰ Additionally, GSSG levels decreased, and GSH increased after three months of the hypertension treatment, highlighting the importance of oxidative stress in the onset and progression of hypertension. ^{290,291} Similarly, lower plasma and right atrial appendages levels of GSH were found in CAD patients and were consistent with the severity of left ventricular dysfunction.²⁹² The same study also reported higher GSH levels in asymptomatic CVDs

patients compared to symptomatic patients, which was linked to more severe cardiac abnormalities in CVDs patients.²⁹²

Higher intraoperative GHS levels were also associated with higher prevalence of myocardial infarction, which aligns with the findings of other studies reporting the association between Glutathione S-transferase polymorphism and the incidence of myocardial infarction during cardiac surgeries.²⁹³ The same extent, decreased glutathione peroxidase activity was inversely associated with increased risk of stroke and coronary heart diseases (Table 1.5). ^{294,295}

Condition Paper **GSH** association Haruki Shimizu et Stroke & Myocardial Infarction Lower plasma levels of glutathione. al.²⁸⁹ Viktória Kovacs et Myocardial Infarction after cardiac Glutathione Transferases polymorphism. al.²⁹³ surgery Myocardial Infarction/Cardiac death Decreased level of erythrocyte Glutathione Stefan Peroxidase-1. Blankenberg et al.²⁹⁵ Thibaud Damy et Heart transplantation/coronary Decreased level of glutathione. al.²⁹² artery bypass grafting/ aortic valve replacement Kiyotaka Cardiac catheterization Vasodilator effects on coronary arteries and Kugiyama et al.296 increased blood flow. J. Robaczewska Elderly people with hypertension Disturbed level of glutathione and enzymes et al.297 involved in the synthesis of glutathione. Decreased level of glutathione; increased Josep Redón et Hypertension al.²⁹¹ level of glutathione disulphide. Felipe J. Chaves Non-treated hypertension Decreased level of OS and glutathione et al. 290 disulfide; increased level of glutathione Saleem Ullah Decreased level of glutathione; increased Diabetes/coronary heart disease Shahid et al.298 level of glutathione disulphide. Tohru Hamanishi Increased values of intima-media thickness in Type 2 diabetes mellitus/some et al.299 forms of CVD Pro/Leu glutathione peroxidase-1 genotype. J. Rybka et al. ³⁰⁰ Elderly on antihypertensive drugs Increased level of glutathione and glutathione-disulfide reductase. Josep Redón et Hypertension Decreased level of glutathione; increased al.²⁹¹ level of glutathione disulphide.

Table 1.5: Relationship between GSH levels and cardiovascular diseases

c. Protein Carbonyl

As discussed above, increased oxidative stress leads to proteins, lipids and DNA damage; however, protein is considered the immediate vehicle imposing oxidative stress on the cells as they are catalysts rather than stoichiometric mediators; hence, the effect of the damage to one molecule is more remarkable than stoichiometric.

During increased oxidative stress, the carbonyl groups either aldehydes or ketones, are formed on the side chains of the proteins.

The formed Carbonyl proteins are very stable, which facilitated their detection and measurements in the human plasma.³⁰¹ Although there are many potential oxidative protein modifications, only few have been studied in clinical practice. Of them, protein carbonyl was the most used, and its accumulation was linked to several diseases such as diabetes, Parkinson's disease, Alzheimer's, sclerosis and inflammatory arthritis.^{302–304}

d. Protein Oxidation and Cardiovascular Diseases

Few studies evaluated the relationship between protein carbonyl and cardiovascular diseases. Descamps et al. evaluated advanced oxidation protein to predict the development of atherosclerosis in non-diabetic patients. During the follow-up period (7 years), oxidized protein products were associated with a higher incidence of ischemic heart diseases, ischemic stroke, peripheral artery diseases, and aortic aneurysms.³⁰⁵ On the other hand, another two studies conducted by Mocatta et al. and Semba et al. evaluated the relation between protein carbonyl and myocardial infarction in advanced aged women. These studies supported the CVDs predictive power of protein carbonyl; however, no correlation was found regarding all causes of mortality over a five-year follow-up period. ^{306,307}

e. Lipid Oxidation

Lipid oxidation can be described generally as a process under which oxidants such as free radicals attack lipids containing carbon-carbon double bond(s), especially polyunsaturated fatty acids. Over the last four decades, an extensive body of literature regarding lipid peroxidation has shown its important role in cell biology and human health and different exogenous stimuli, such as the ionizing radiation, ultraviolet rays, tobacco smoke, pathogen infections, environmental toxins, and exposure to herbicide/insecticides, are sources of in vivo ROS production.³⁰⁸

The two most prevalent ROS that can affect profoundly the lipids are mainly hydroxyl radical (HO[•]) and hydroperoxyl (HO[•]²).

Human sterols are prone to oxidation as they all have an unsaturated double bond between the carbon number 5 and 6 in their structure. Under the influence of reactive oxygen species and free radicals, cholesterol molecules bind to hydroxyl, epoxide, and ketone groups to form a more polar and easily absorbed moieties.³⁰⁸

Oxidation of cholesterol is usually via two mechanisms, autoxidation or enzymatic metabolism Figure 1.20. This oxidation produces an epimeric mixture of hydroperoxides which is further disintegrated to 7 α - or 7 β -hydroxysterol and 7-ketocholesterol ³⁰⁹ while epoxidation produce 5,6 α -epoxy-5 α - and 5,6 β -epoxy-5 β -cholestan-3 β -ols and hydration of epoxysterols produces 5 α -cholestane-3 β ,5,6 β -triol.³¹⁰ Additionally, auto-oxidation of crystalline cholesterol produces the 20, 24, 25, and 26-hydroperoxides.³¹¹

The most common oxysterols found in human plasma are 27OH, 24OH, 3 β OH and 7 α OH cholesterol produced by the enzymatic reaction of the mitochondria and the endoplasmic reticulum hydroxylases.

The activity of oxysterols (pathophysiological and biomedical) is more than twice that of cholesterol, and molecules produced by the enzymatic reaction are always present in equilibrium with those formed by the non-enzymatic ones. ³⁰⁸ This equilibrium is disturbed in certain aetiologies and has been used as a biomarker for diagnosing and monitoring of diseases progression.

f. Oxysterols and Cardiovascular Diseases

Hypercholesterolemia and atherosclerosis are part of the primary risk factors of CVDs. Of particular note oxysterols were found to contribute to the vascular cellular dysfunction by promoting oxidative stress, inflammation and atheroma development Figure 1.21.³¹³

Since the end of the 90s, efforts were exerted to evaluate oxysterols in human blood and blood vessels. In this consideration most of the investigations confirmed oxysterols presence in CVDs; specifically, 7-ketocholesterol, cholestane- 3β , 5α , 6β -triol, 7-HC, and 5α , 6α - and the 5β , 6β -epoxycholesterol isomers were found in the carotid artery atherosclerotic plaques.³¹⁴

Similarly, mass spectrometry techniques identified high plasma concentrations of cholesterol oxides (24S-Hydroxycholesterol, and 27-hydroxycholesterol) in peripheral artery diseases patients. Additionally, the tissue concentrations of 24S-Hydroxycholesterol, and 27-hydroxycholesterol were significantly correlated to the plasma levels of C-reactive protein.³¹⁵ Elevated levels of 24S-Hydroxycholesterol were also found in the serum of individuals with genetic hypercholesterolemia and were positively associated with higher carotid intima-media thickness.³¹⁶

Preeclampsia (PE) is another CVD associated with pregnancy; altered cholesterol metabolism and pregnancy-induced oxidative stress were confirmed as one of this disease's aetiologies.^{317,318}

Higher plasma levels of cytochrome P450 and ROS derived oxysterols (7 α -hydroxycholesterol, 7 β -hydroxycholesterol,4 β -hydroxycholesterol, 20 α -hydroxycholesterol

60

and 27-hydroxycholesterol) were confirmed in PE patients compared to normotensive pregnant women and was directly correlated to preeclampsia.³¹⁹

Recently scientists started to study the potential application of oxysterols as indicators of the efficacy of lifestyle interventions and therapies. Preliminary animal results also showed that aerobic exercises and anti-dyslipidemic drugs (e.g. Simvastatin) decreased oxysterols plasma levels, thus suggesting antioxidant properties of the drugs and the applied interventions besides their cholesterol-lowering action.^{320–323}



Figure 1.20: Main Products of Enzymatic or Auto-Oxidation of Cholesterol Adapted from Leonarduzzi G et al. Oxidized products of cholesterol: dietary and metabolic origin, and proatherosclerotic effects.³¹²





1.4.5. Dry Eye Syndrome

Dry eye disease (DED) is a multifactorial disease of the ocular surface (cornea, conjunctiva, accessory lacrimal glands, meibomian glands) with loss of tear film homeostasis and development of ocular symptoms. The disease is also named keratoconjunctivitis sicca, dry eye syndrome, and dysfunctional tear syndrome.³²⁵

The prevalence of DED ranges from five to 50% globally. This prevalence was estimated to be 2.7% in those between 18 to 34 years old and increases to 19% in those above 75 years old and was higher in women 8.8% than men 4.5%.³²⁶

Disease Risk factors ^{327–330}:

- Advanced age
- Female sex

- Decreased androgens
- Medications (antihistaminic, anticholinergics, hormonal therapy, antidepressants, antiarrhythmic drugs, ocular medications).
- Diseases (diabetes mellitus, Parkinson disease, Sjögren's syndrome)
- Contact lens
- vitamin A deficiency
- Corneal refractive surgery
- Dry environments

Severe DED can decrease visual acuity, which can limits individuals physical functioning and productivity.^{331–333}

DED has a complex and multifactorial aetiology. The tear film of the eye consists of three components:

- Aqueous component
- Mucous component
- Lipid component

A healthy tear film depends on the balanced interaction between the eyelid, ocular surface and the lacrimal glands (Figure 1.22).³³⁴

It is also classified into two groups ³²⁵ :

- **Decreased tears production**, resulting in aqueous component deficiency DED
- Abnormal meibomian gland physiology, resulting in evaporative DED.

Tear film hyperosmolarity, inflammation of the ocular surface and subsequent activation of the sensory nerves of the ocular surface leads to signs and symptoms of the disease.^{335,336} As the clinical evaluation findings of DED are variable, clinicians base their diagnosis on validated questionnaires.³³⁷ These questionnaires can also be used to monitor disease progress and treatment follow-up.³³⁸

The most commonly used questionnaires are:

- Ocular Surface Disease Index (OSDI): Useful in patients with more severe symptoms to monitor the response to therapy and variability in symptoms over time (Appendix 1).³³⁹
- Dry Eye Questionnaire (DEQ-5): a five-item questionnaire reduced from the Dry Eye Questionnaire to determine DED symptom severity (Appendix 2).³⁴⁰
- Impact of Dry Eye on Everyday Life (IDEEL): Fifty-seven questions in three modules validated in patients with DED.³⁴¹

 Salisbury Eye Evaluation Questionnaire (SEE): a six-item questionnaire used in selfreported, population-based prevalence surveys to determine visual impairment among older adult subjects.³⁴²

Details of the diagnosis of DED is provided in section 3.3.9.

Recently, the relationship between DED and inflammation produced systemic illnesses such as hypertension, diabetes and dyslipidaemia was evaluated in several studies; however, further investigation is needed to assess and confirm this association and identify the possible correlation mechanisms.^{344–346}



1.5. Cardiovascular Diseases Risk Assessment

1.5.1. Assessment of Endothelial Function

Endothelial function testing can be divided into peripheral (systemic) and central (coronary) vascular assessment. Most clinically used techniques depend on the pharmacological/ physiological stimulation of NO and other vasoactive mediators release and measuring the vascular response to the endothelial dependant stimuli, e.g., increased shear stress, hyperaemia, or systemic administration of vasoactive stimulants.

The endothelial function testing technique can also be divided into invasive and non-invasive procedures.³⁴⁷

1.5.1.1. Invasive Techniques

Invasive techniques include **quantitative coronary angiography** (QCA), which measures coronary arteries luminal diameter in response to intracoronary administrated endothelial dependant vasodilator substances such as Acetylcholine (Ach). The NO-mediated vasodilation is measured by angiography or intravascular ultrasound.^{348,349} Coronary microcirculation can also be assessed by **intracoronary Doppler** using a pharmacological stimulator, e.g. adenosine or nitroprusside or physiological stimuli.³⁵⁰

A similar method but dependant on thermodilation and index of microcirculatory resistance is also used to assess coronary microcirculation. This technique is identical to pharmacological and pressure-based approaches but uses **intracoronary temperature** measurements to approximate flow.^{351,352}

1.5.1.2. Non-Invasive Techniques

The gold standard non-invasive technique used to assess macrovascular endothelial function is the NO-driven **flow-mediated dilation (FMD).** Using brachial artery ultrasound, it is possible to assess peripheral changes in the blood flow rate in response to increased shear stress in the arm.^{353,354} Other ways of stimulating vasodilation are also used, e.g. reactive hyperaemia, exercise, psychological stress and cold presser sympathetic activation (Table 1.6).

A newer method of testing macrovascular function is the **low flow-mediated constriction (L-FMC)** which measures the endothelial response to reduced blood flow and shear stress.³⁵⁵ L-FMC is a better measure of resting and basal arterial tone. The exact mechanisms of the arterial response to decreased shear stress are not fully elucidated; however, scientists believe it is partially dependent on NO.³⁵⁶

Plethysmography of the forearm circulation is one of the leading techniques used to assess peripheral endothelial dysfunction. It examines the changes in forearm blood flow in response to IV administered vascular stimuli. Because it is a semi-invasive technique (IV

administration of vasoactive substances), it is not a preferred method in clinical practice and is only used for research purposes.³⁵⁷

Microvascular endothelial function is also used to measure vascular reactivity to the same stimuli; however, typically utilising peripheral tonometry or pulse wave amplitude to approximate blood flow.

Peripheral Arterial Tonometry (PAT) is another technique that depends on measuring reactive hyperaemia response in finger pulse wave amplitude after the inflation/deflation of a blood pressure cuff to supra-systolic pressures.^{358–361}

Traditional Imaging techniques are also used to assess microvascular endothelial function; for example, myocardial positron emission tomography (PET) is used in diabetic patients ³⁶² and was verified in other patients cohorts.³⁶³ Similarly, cardiac magnetic resonance perfusion has been used to assess endothelial function in patients without overt CAD and normal angiograms angina patients.³⁶⁴

There are other measures of endothelial function that are currently in various stages of use and design, mainly in the research realm, e.g., measures of endothelial progenitor cells (EPCs) and subtypes that assist in endothelial repair and function. These measures has been associated with the Framingham scores³⁶⁵ and various measures of CVD ³⁶⁶, Statins ³⁶⁷, erythropoietin ³⁶⁸ and exercise. ³⁶⁹

Nucleotide polymorphisms, microRNAs, urinary markers and genome sequencing were also linked to endothelial dysfunction and atherosclerosis; however, they are not clinically used.

Table 1.6: Advantages and Limitations of the Most Common Non-invasive Techniques Used toAssess Endothelial Function

Technique	Vascular bed	Advantages	Limitations	Stimulus
Flow-mediated dilation ³⁷⁰	Brachial artery	Easy access	Several measurement	Reactive Hyperaemia
	Conduit artery	Strong correlation with invasive epicardial	protocols	
		vascular function	Not standardised	
		Used to assess other parameters (flow, baseline		
		arterial diameters, FMC) Inexpensive		
Venous occlusion	Forearm	Assess dose-response	Invasive	Different
Plethysmography ³	vasculature	relation	Time-	vasoactive
71		Easy access	consuming	substances, e.g.,
	Microvasculature			acetylcholine
Peripheral Arterial	Finger	Easy access	Influenced by	Reactive
Tonometry ³⁷²	Microcirculation	Easy to perform Automated so decreases interpersonal variability	non- endothelial variables	Hyperaemia

1.5.1.3. Peripheral Vascular Reactivity

Vascular reactivity is the response of blood vessels to external stimuli. It is a simple, nonimaging and non-invasive method used to assess endothelial dysfunction in primary clinical settings.

Digital thermal monitoring (DTM) is one of the most commonly used techniques for evaluating peripheral reactivity. It measures changes in finger temperature during and after arm-cuff-induced ischemia (Figure 1.23).^{373,374} The Arm- cuff occlusion induces reactive hyperaemia, which correlates to brachial artery reactivity.³⁷⁴

A vital advantage of DTM is that it measures both microvascular and macrovascular reactivities.³⁷⁵ Studies have shown that cutaneous micro and macrocirculation are similar to the systemic circulation a function of NO ³⁷⁶, and temperature rebound measured by DTM was found to correlate directly with the presence and extent of CAD.³⁷⁷ It was also correlated to the Framingham risk scores and coronary artery calcification in asymptomatic populations.³⁷⁸ In another study, it was directly associated with myocardial perfusion independent of cardiovascular risk factors.³⁷⁹

Additionally, it was also used to evaluate the effect of the pharmacological influences of some vasoactive drugs.³⁸⁰

Although large prospective trials are needed to establish the clinical value of fingertip DTM, the clinical implications are very promising.



Figure 1.23: Schematic Figure Depicting Digital Thermal Monitoring

1.5.1.4. Endothelial Dysfunction as a Marker of Future Cardiovascular Risk

Does it provide prognostic information beyond commonly used risk scores?

As discussed above, many studies linked endothelial dysfunction to most CVDs' predisposing factors, e.g., systemic hypertension ^{381–383}, dyslipidemia ^{384,385}, diabetes ^{386–389}, smoking ^{390,391}, obesity ^{392,393}, hyperhomocysteinemia ^{394,395} and age.^{396,397} Similarly, extensive literature evidence confirmed the systemic nature of endothelial dysfunction and showed its strong correlation to the coronary arteries functions.^{353,358,398} Additionally, the pathophysiology of the impaired endothelium was found to relate to vascular structural changes like the carotid intimamedia thickness in individuals free from CVDs ^{399,400} and in less healthy individuals as hypertensive postmenopausal women.⁴⁰¹

From the above, we can see that endothelial dysfunction is a confirmative link to CVDs and CVDs' risk factors. However, it is essential to ask whether endothelial dysfunction provides more clinical information beyond the routinely used assessment scores in clinical settings.

1.5.2. Cardiovascular Events Risk Scores

Over the years, many invasive methods to assess endothelial function in the coronary arteries were used as a prognostic factor for future risk of development of CVD, even in healthy individuals.^{402,403} However, in primary prevention, the risk of using invasive techniques overweighted their benefit. Thus, most clinicians preferred using disease risk scores or peripheral ways to assess the future risk of developing cardiovascular diseases. The Framingham, Reynolds, PROCAM, QRISK, SCORE, JBS3, MESA, and the pooled cohort ASCVD risk calculators are examples of the most commonly used CVDs estimation risk scores in the clinical settings (Table 1.7).^{404,405}

Regardless of clinicians' wide use of these risk calculators, studies showed many drawbacks associated with their use. For example, the Framingham risk score was found to overestimate CVDs risk by 37% and up to 67% in some populations.^{404,405} Similarly, the Reynolds Risk Score was found to overestimate the risk by 9% to 21%, and the AHA-ACC-ASCVD risk calculator did not show improved calibration or discrimination compared with four other scores used in clinical practice.^{404,405}

Results from the evaluation studies pushed clinicians to use non-invasive peripheral endothelial dysfunction evaluation methods as risk evaluators for future cardiovascular events.

Table 1.7: Comparison of Frequently Used Cardiovascular Risk Prediction Tools

	FRS	Pooled Cohort Equ.	WHO/ ISH Risk Charts	JBS3	SCORE CVD	QRISK	Reynolds	MESA
Age	✓	✓	✓	✓	✓	✓	\checkmark	✓
Gender	✓	✓	✓	\checkmark	√	✓	×	✓
Ethnicity	×	√	×	~	×	×	×	✓
Region	×	×	×	~	√	✓	×	×
Diabetes	~	√	✓	~	×	×	✓	✓
Smoking	~	√	✓	~	√	✓	✓	✓
FH	×	×	×	~	×	✓	✓	✓
AF	×	×	×	~	×	×	×	×
CKD	×	×	×	~	×	×	×	×
RA	×	×	×	~	×	×	×	×
BP treatment	✓	✓	×	\checkmark	×	✓	×	✓
SBP	✓	√	✓	~	✓	✓	✓	✓
BMI	×	×	×	~	×	✓	×	×
Apolipoprotein	×	×	×	×	×	×	×	×
T-Chol	\checkmark	✓	✓	✓	\checkmark	✓	\checkmark	✓
HDL	✓	✓	×	✓	✓	✓	\checkmark	✓
Lipid-lowering treatment	×	*	×	×	×	×	×	~
CRP	×	×	×	×	×	×	✓	×
CAC score	×	×	×	×	×	×	×	✓
Psychological assessment	×	×	×	×	×	×	×	×
			End Point	s Asses	sed			
CHD Death	✓	✓	×	✓	\checkmark	✓	✓	✓
МІ	~	√	×	✓	×	✓	✓	✓
Angina	✓	×	×	✓	×	✓	×	×
Revascularisation	×	×	×	~	×	√	~	~
Arrhythmia	×	×	×	×	✓	×	×	×
Heart failure	✓	×	×	×	\checkmark	×	×	×
Stroke	✓	✓	×	✓	\checkmark	✓	~	×
TIA	✓	×	×	 ✓ 	×	✓	×	×
Intermittent Claudication	✓	×	×	~	\checkmark	✓	×	×
Aortic aneurysm	×	×	×	×	\checkmark	×	×	×
Cardiac resuscitation	×	×	×	×	×	×	×	~

Table 1.7: FH, family history; AF, atrial fibrillation; CKD, chronic kidney disease; RA, rheumatic arthritis; BP, blood pressure; SBP, systolic blood pressure; BMI, body mass index; T-Cho, total cholesterol; HDL, high-density lipoprotein; CRP, C reactive protein; MI, myocardial infarction; TIA, transit ischemic attack.

1.5.2.1. Non-Invasive Endothelial Dysfunction Assessment Techniques versus Risk Scores in the Prediction of Cardiovascular Events

FMD was one of the fundamental ways used for this purpose. It was found to predict cardiovascular events beyond traditional risk scores in some special populations, e.g., postoperative assessment of patients who underwent elective vascular surgeries ⁴⁰⁶, healthy postmenopausal women ⁴⁰⁷, patients who suffered from chest pain ⁴⁰⁸ and patients admitted for invasive evaluation of chest pain.⁴⁰⁹ Additionally, many large-scale studies recognized the value of the use of FMD in primary prevention settings. In two studies (435 and 268 healthy subjects), brachial FMD assessment enhanced the long-term adverse cardiovascular events prediction capability of the traditional risk factor assessment.^{410,411} Similarly, in the 2007 cardiovascular health study (2700 apparently healthy 72 years old participants), a five-year follow-up period showed the association of endothelial dysfunction measured by FMD and event-free survival even after adjusting for traditional risk factors.⁴¹²

In 2009, the multi-ethnic study of atherosclerosis (3000 participants) confirmed the predictive power of FMD in White, Black, Hispanic and Chinese subjects even after adjusting for the Framingham risk factors.⁴¹³ Similarly, many other studies have confirmed the additional prognostic information the FMD provides over other traditional risk scores.^{414,415} However, all these studies showed that adjustment to conventional risk factors weakened the correlation between FMD and the assessed outcomes. Additionally, many studies showed that the vascular bed in which the assessment is performed could affect the results to the extent of no correlation at all with CVD risk.⁴¹⁶ Similar to FMD, scholars found that adding hyperemic velocity to the Framingham Risk score led to a risk-reclassification improvement.^{400,417} Acetylcholine-associated peripheral microvascular endothelium dysfunction was also associated with CV events in the elderly while FMD was not and its addition to the Framingham risk score improved risk discrimination.⁴¹⁸ Likewise, EndoPAT helped predict non-coronary atherosclerosis ⁴¹⁹ and adverse cardiac events in outpatients ⁴²⁰ that the Framingham risk score could not predict.

From the previous discussion, we can say that macrovascular assessment is beneficial in patients with established diseases, and microvascular function is a better indicator for future risk. However, and even with all the data indicating the importance of endothelial dysfunction as a predictive value of future risk of CV events, routine use of endothelial function measurements assessment is not yet recommended by neither the European (ESC)^{421,422} nor the American (AHA/ACCF) guidelines ⁴²³ and received a lower recommendation than carotid IMT measurements and coronary calcium score.

The reason against the routine use of these techniques despite their high sensitivity is that their specificity is not well established and yet remains a concern. Another limitation is the poorly standardized methodology (except EndoPAT).⁴²⁴ Additionally, these techniques are a 71

static way to assess the endothelial, so if the endothelium function is appreciated as a dynamic process, these methods are not the best to reflect the real state of the endothelium.⁴²⁴ Furthermore, the conflicting reports regarding the agreement between the FMD and the PAT results made the situation more confusing. Though some scientists found an association between the two tests, others highlighted the discordance between them.^{425–427} Likewise, although IMT and calcium scores can be performed in a more standardized manner and are less impacted by transient abnormalities than endothelial function, these measures give information about the vascular structure (more established disease) rather than the function. Additionally, these measurements do not change rapidly with interventions, which is an invaluable advantage of endothelial function measurements.^{428,429} Finally, cost and limited resources are other reasons limiting the wide adaptation of these modalities.

1.5.3. Assessment of the Ocular Circulation

1.5.3.1. Ocular Blood Flow Assessment

Over the years, several methods were developed to assess the ocular blood flow depending on the area of interest. Colour Doppler ultrasound imaging is one of the most used methods to assess retro-ocular vessels blood flow ⁴³⁰, while intra-ocular vessels can be visualised and evaluated by direct methods such as ophthalmoscope, fluorescence or indocyanine green angiography.⁴³¹ Laser Doppler velocimetry is used to quantify the blood velocity ⁶⁶, and laser-flowmetry and laser-speckling are used to quantify blood flow in the capillary bed, such as the optic nerve. Similarly, thermography is used to assess the bulk flow to the eye ⁴³², and retinal vascular analyser (RVA) is used to observe dynamic changes in blood flow over time. A summary of the main characteristics of these methods is provided in Table 1.8.
Table 1.8: Techniques of Ocular Blood Flow Measurement and Assessment

Technique	Measured Variables	Vascular bed	Measurement Location	Advantages	Limitation
Fluorescein angiography ^{433–435}	Mean Circulation time, Arteriovenous passage time	Retina	Main retinal vessels	Simple, painless procedure	Exact relation to retinal blood flow unclear
Fluorescein Angiography with scanning laser 436–438	Erythrocyte and Leucocytes velocity	Retina	Perimacular retinal capillaries	Provides detailed information on ocular perfusion	No information on vessel diameter
Laser Doppler Velocimetry combined with fundus photography ⁴³⁹	Blood flow	Retina	Main retinal vessels	Capable of calculating the blood velocity, cross- sectional area, vessel diameter, and hence the total blood flow. Non-invasive	A highly skilled operator is mandatory
Laser Doppler flowmetry ^{440,441}	Blood flow	Optic nerve or choroid	Capillaries	One of the most widely utilised techniques. Non-invasive	Sampling depth unknown
Scanning Doppler flowmetry ^{442,443}	Blood flow	Retina or optic nerve	Capillaries	Non-invasive, does not require dilation, quick, direct assessment of ocular blood flow	Sampling depth unknown, exact relation to blood flow unclear
Colour Doppler imaging ^{444,445}	Blood velocities	Central retinal artery, posterior ciliary arteries, ophthalmic artery, Retina	Extraocular vessels	High reproducibility Fast procedures	No information on vessel diameter, can't be used for small vessels
Blue field entopic technique 446-449	Leucocyte velocity and density	Retina	Perimacular retinal capillaries	Allows assessment of foveal perfusion.	Leucocyte movement is blood flow subjective
Pneumotonometry ⁴⁵⁰	Changes in intra- ocular pressure during cardiac cycle	Choroid	Global	A non-invasive method of estimating the intraocular pressure	Only pulsatile flow component, standard ocular rigidity assumed
Laser interferometric measurement of fundus pulsation ⁴⁵¹	Changes in corneo-retinal distance during cardiac cycle	Choroid	High topical resolution	Simple, non-invasive	Only pulsatile flow component
Dynamic retinal vessel analysis ⁴⁵²	Retinal vascular Diameter and reactivity	Retina	Retinal vessels	Measurement of the diameter of retinal vessels in relation to time and location. Noncontact & non- invasive method	Requires clear media, good fixation and pupil dilation. More aimed at assessing retinal vascular function than OBF
Retinal Oximetry ^{453–}	Retinal vessels	Retina	Retinal metabolism	Reliable, non-invasive technology for retinal oxygen measurements	Influenced by many external factors e.g. light
Laser Speckle Flowgraphy ⁴⁴³	Blood flow velocity	Retinal, choroidal vessels, circulation of the ONH	Global	A noncontact & non- invasive method	Not direct assessment of OBF

1.5.3.2. Retinal Vascular Analysis

a. Background

For years, structural changes in peripheral vessels were frequently associated with numerous CV risk factors, e.g. hypertension, diabetes^{456,457}, obesity ^{458–460}, chronic kidney disease ⁴⁶¹ and smoking. ^{462,463} It was also linked to the progress and development of cardiovascular events ^{464,465} and end-organ damage in CVD patients.^{465,466}

Increased media-to lumen ratio (M/L) is the most frequently used technique to evaluate the resistance of small peripheral arteries. Wire or pressure micromygraphy was highlighted as the best approach to demonstrate an increased M/L ratio in high and low to moderate risk patients.⁴⁶⁷ Still, unfortunately, the invasive nature of this technique prevented the wide and routine application in clinical settings. To develop alternative non-invasive approaches for evaluating the microvascular structure and function, the interest of many researchers shifted towards the retinal vascular district. The rapid development of retinal vessels investigational methodologies allowed scientists to quantify geometrical and topological changes in the arteriolar and venular trees in many cardiovascular diseases, e.g., hypertension, coronary artery disease and cardiovascular risk factors such as diabetes mellitus, ageing, smoking and obesity.

On the other hand, microcirculation in the retina can be imaged directly with plenty of noninvasive methods. Thus, it provides a window to visualize any microvascular changes that lead to the development of cardiovascular diseases.⁴⁶⁸ Although data from retinal vascular analysis provide information about the blood vessels' structure and function; its clinical application has only recently gained attention.⁴⁶⁹

b. Structural Retinal Changes

Systemic CVDs and cardiovascular risk factors like arterial hypertension^{470,471}, coronary heart disease⁴⁷², stroke ^{473,474} or preeclampsia⁴⁷⁵, as well as obesity⁴⁷⁶ are all associated with structural vascular changes in the retina. These changes include arterioles narrowing, veins dilation and arteriovenous ratio alterations (AVR).

The development in ophthalmoscopy techniques in the 19th century opened a window to investigate the retina and its microvasculature in a simple and non-invasive way. Robert Marcus Gunn was the first to detect retinal vascular changes in systemic CVDs.^{477,478} Later, Keith, Wagener and Barker associated retinal changes to the survival of hypertensive patients, and this was used later for the classification of hypertensive retinopathy (The KWB scale).^{479–482} Similarly, Hubbard and his colleagues developed formulas to calculate the central retinal arteriolar/ venular equivalent (CRAE/CRVE) using retinal photography.⁴⁸³ Based on these formulas, the arteriovenous ratio (AVR) was calculated in many studies as it is not dependent on the optical properties of the eye or the used camera.⁴⁸⁴ However, the introduction of more

74

sophisticated fundoscopic and optical coherence tomography (OCT) techniques offered a way for early detection of retinal structural changes and future susceptibility to develop CVDs.^{485–} ⁴⁸⁸ Later several studies reported a strong association between larger venular calibre and higher risk of future development of CVDs.⁴⁸⁹ Additionally, the M/L ratio was found to be independently associated with CVDs in high, moderate and low-risk populations.^{465,490–494}

Similarly, CRAE and AVR were found to precede hypertension's clinical appearance even in normotensive individuals and children.^{495–501}

Based on the findings of these studies, the 2011 British society of hypertension guidelines recommended routine fundus examination for all hypertension/prehypertensive patients to assist in the evaluation of the future risk of CVDs.⁵⁰² However, regardless of the data supporting its importance, fundus examination value in clinical assessment was questioned. ^{503,504} and the following European society of cardiology guidelines recommended against its routine use in the evaluation of hypertensive patients.⁵⁰⁵

A list of the most commonly used measurements of retinal vascular caliber is provided in Table 1.9.

Retinal	Definition		Limitations
vaccular	Demitton	Advantages	Elimations
vascular			
parameters			
Arteriovenous	Ratio of the caliber of	Dimensionless, controls for	Nonspecific.
Ratio	arterioles to venules.	magnification differences	Changes in AVR can't
		from camera lenses and	specify if the changes in
		refractive error.	arteriolar or venular
			diameter or in both.
Central Retinal	An index that reflects	Reflects distinct systemic	Not highly accurate.
Artery	the average width of	vascular disease pathways	May underestimate the true
Equivalent	retinal arterioles.	that tend to target the	internal vascular diameter.
		arterial system.	
Central Retinal	An index that reflects	Reflects distinct systemic	Not highly accurate.
Vein Equivalent	the average width of	vascular disease pathways	May underestimate the true
-	retinal venules.	that tend to target the	internal vascular diameter.
		venous system.	
		-	

Table 1.9. Measurement of Retinal Vascular Caliber

c. Limitations of Structural Retinal Vascular Measurements

Regardless of scientists support for the routine use of retinal imaging in clinical settings, the technique has many limitations that limit its clinical applicability. Firstly, the used equations to calculate summarized indices from individual calibers is based on empirical models. The Knudtson-parr-Hubbard formulas were derived from analysis of population sample retinal images, and they used a root mean square deviation model that best fits the observed data.483,506,507 Later, Knudtson et al. developed revised formulas for summarizing vascular caliber. The new formulas gained wide acceptance from scientists and demonstrated more accuracy in calculating the CRAE and CRVE.⁴⁸⁴ Still, all these formulas were an estimated summary, not the actual value of the retinal vascular caliber. The second issue faced these formulas was that all the researchers used them to compare different groups. However, to use these measures for risk stratification in clinical practices, it should be optimised to measure individual patients.⁵⁰⁸ That brought the magnification effect from retinal image, either by incorporating an adjusted measurement to compensate for this effect or using dimensionless measurements.⁵⁰⁸ Ocular biometric data can be used to adjust for magnification whoever its accuracy in digital retinal images is not well established. The third limitation that was addressed by many scientists is retinal caliber measurements are usually obtained from one image, and it has been shown it vary by up to 15% depending on the point of the pulse cycle when the shoot was taken.⁵⁰⁹ The Fourth and the most critical limitation is the lack of normal population data. Although the technique was widely used in different populations, there is still no clear identification of what should be considered normal measurement and what is abnormal. Finally, the retinal vascular caliber measurements do not reflect the 3-dimensional architecture or the functional changes of the retinal microvasculature. Therefore, the full potential of retinal image analysis in relation to the prediction of cardiovascular diseases remains undetermined.480

d. Functional Retinal Changes

In the last decade, dynamic retinal vascular analysis (DVA) grabbed all the attention as an innovation in endothelial function evaluation techniques. The technique and its application in clinical practice greatly impacted the assessment and risk stratification of many diseases.⁴⁵² DVA is a closed system consists of a fundus camera attached to a real-time monitor and a video camera with special software to record and analyse the vascular response to external stimulation. Scientists used many techniques to stimulate the endothelial reaction in the vascular vessels; oxygen or meduna's mixture inhalation was the first technique used to provoke retina vascular response; however, due to its side effect, scientists moved to using systemic vasoactive drugs or flickering light to activate the retinal vascular response.^{510,511} Of

all the vascular stimulation methods, light stimulation was the most accepted due to its non-invasive nature and the absence of systemic side effects.⁵¹²

Flicker-light induced vasodilation in the retinal artery was proved to be endothelium and nitric oxide dependent ^{513–515}, and as it precedes the development of structural vascular changes, it played an important role in the diagnosis, prognosis evaluation and treatment monitoring diabetic, cardiovascular and dyslipidaemic patients.^{159,516,517}

While reduced flicker light-induced vasodilation has already been demonstrated in patients with established cardiovascular diseases, ^{452,518,519} few studies used it to evaluate patients' future risk of developing CVDs.^{513,515} Compared to static retinal imaging, DVA offers a more precise way to assess microcirculatory behaviour and identify any impairments even before developing clinical signs.

e. Retinal Vascular Functionality and Hypertension

Similar to structural retinal vascular changes, there is a huge number of evidence reporting reduced retinal arterial vasodilation in response to flicker light in people with untreated hypertension.⁵²⁰ One of the first studies to evaluate retinal vessel response to flicker light stimulation was Nagel and his colleagues in 2004.⁵²⁰ They reported a significant reduction in retinal vascular response to flicker stimulation in hypertensive patients compared to healthy individuals. The untreated hypertensive patients had a mean baseline-corrected flicker response of $2.2 \pm 2.5\%$ compared to $6.4 \pm 2.7\%$ with a P value of 0.001.⁵²⁰ However, because of the small sample size (45 subjects) and as there was no follow-up period, researchers could not prove if these reported differences were an individual variation or due to the untreated pathology.⁵²⁰ Nonetheless, another study performed by the same group showed that treatment of hypertensive is directly correlated to improvement in arterial baseline diameter, however not to flicker response after one and a half year follow-up period. These results suggested that vascular baseline diameter is a better indicator of vascular reactivity than vasodilation in hypertensive patients.⁵²¹

Later, many studies reported impairments in other retinal vascular parameters (arterial and venous) in hypertensive patients. Machalińska and her colleagues reported a decreased arterial dilation in hypertensive patients compared to healthy controls (mean 1.31, P=0.001) and a lower vein dilation by an average of 1.32, p=0.002 after adjustment of all influential factors.⁵²² Additionally, the same study reported a negative correlation between arterial response to flicker and plasma C reactive protein (Rs = -0.29, p = 0.07), plasma tissue necrotising factor α (TNF α) concentrations and arterial response to flickering light stimulation (Rs = -0.39, p = 0.02).⁵²² There is also growing evidence suggesting a decreased maximum dilation and baseline corrected flicker response in middle age untreated hypertensive patients.⁵²³

f. Retinal Vascular Functionality and Dyslipidaemia

A strong association between dyslipidemia and wider venular caliber was found in many researches.⁵²⁴ Scientists also reported improvement in visual acuity with LDL apheresis and control of refractory dyslipidemia.⁵²⁵ Although systematic studies on flicker-induced retinal vasodilatation in hypercholesterolemia are still scarce, a reduced response to flicker has been reported in patients with hypercholesterolemia.⁵²⁶ Similarly, scientists showed that LDL apheresis significantly improved flicker light-induced retinal venular dilation, indicating improved retinal venular response, endothelial function, or both.⁵²⁶

g. Retinal Vascular Functionality and Ageing

Most cardiovascular and cerebrovascular diseases are directly connected to ageing and have been linked to progressive endothelial dysfunction in the macrovessels.⁵²⁷

Microcirculation analysis using DVA showed impaired flicker light response in the elderly, suggesting a decline in the regulation of the retinal vessel diameter.⁵²⁸ An abnormal constriction response was also noticed, with young individuals showing an arterial constriction response to amplitudes below baseline diameter post flicker light-induced vasodilation.^{529,530} This constriction response did not appear in older individuals.⁵²³ Similarly, in individuals with low cardiovascular risk, age-related differences in flicker-induced retinal vessel diameter change throughout the entire functional response curve for arteries and veins were reported, where gender differences mainly affected the arterial dilatory phase and are only present in young individuals.⁵³¹

2. Research Rationale

While various biomarkers offer a disease-specific individual biological profile, none of them enables non-invasive assessments in primary care settings while being sufficiently sensitive to allow early detection and prevention.⁵³²

Similarly, although quantification of vascular and endothelial dysfunction is under investigation as a promising early indicator for CVD risk, assessing it implies complex techniques, high costs, and specialised settings.⁵³³

The Dynamic Vessel Analyser (DVA, Imedos GmbH, Jena, Germany) is one such device that enables the assessment of retinal vascular function through a non-invasive method. The use of DVA in early CVDs screening was pioneered by the supervisory team at Aston University, who was also the first to demonstrate that dynamic vascular behaviour is unique to each individual and their age group and existing pathologies.⁵³⁴

Using DVA for risk stratification of CVDs patients will also provide a method of obtaining easyto-understand data for practitioners to use for follow-up on disease onset progression and treatment effectiveness, including personalised reports. Benefits to patients are expected over the immediate, medium- to long-term, as the solution will provide specific and personalised warnings of CVD risk and opportunities for immediate advice regarding lifestyle changes, preventive treatments, and disease and management follow-up.

Assessing individual patients rather than the whole population will give more insight into specific risk factors and will open the door for tailored therapy plans. CVDs prevention is imperative, and the most effective path to it should include actions primarily at an individual level.

2.1. Research Aims:

Study I: To investigate the presence and impact of ocular and systemic vascular alterations in prehypertensive patients and to explore the concept of using retinal vascular analysis to identify people at risk of developing future CVDs.

Study II: To analyse retinal and peripheral microvascular function in subjects that fall into different BP groups when using either the ESC/ESH or the ACC/AHA guidelines to identify the diagnostic cut-off values for early hypertension.

Study III: To explore the presence of microvascular endothelial dysfunction (as a measure for early CVD) in individuals diagnosed with dry eye disease and its correlation with systemic lipids.

Study IV: To investigate the correlation between lipid peroxidation markers, glutathione redox ratio and retinal microvascular functional reactivity in apparently healthy individuals of various age groups.

Study V: To assess the performance of using either leucocytes telomere length or retinal microvascular dysfunction or the combination of both for predicting age and systemic blood pressure, two of the most known risk factors for CVD.

3. Subjects and Methods:

This chapter outlines the recruitment procedure, inclusion, and exclusion criteria for all the participants involved in this research. It then outlines the study protocol and investigative techniques used throughout this thesis for the assessment of ocular and systemic vascular function.

3.1. Ethical Approvals

Ethical approval was sought and received from the relevant research ethics committee(s) (National Health Service (NHS) and/or Aston University College of Health and Life Sciences). Written informed consent was received from all subjects prior to study enrolment, and all procedures were designed and conducted in accordance with the tenets of the Declaration of Helsinki.

3.2. Patient Recruitment

3.2.1. General Inclusion/Exclusion Criteria

Volunteers were recruited through advertisements at the Health and Vision Sciences Clinic and Vascular Research Laboratory at Aston University (Birmingham, UK).

General **study inclusion criteria** were defined as age \geq 18 years; no known history of cardio-, cerebro- or related vascular disease; not to be taking any vasoactive medications; absence of symptoms or history of any other diseases that had not been resolved before study inclusion; and willingness and capacity to freely provide informed consent.

Study exclusion criteria were positive diagnosis of cardio- or cerebrovascular diseases such as CAD, heart failure, arrhythmia, stroke, or transient ischaemic attacks; peripheral vascular disease; severe dyslipidaemia (plasma triglycerides > 6.00 mmol/L or cholesterol levels > 7.00 mmol/L); diabetes; smoking; inflammatory conditions such as rheumatoid arthritis; as well as other metabolic disorders or chronic diseases that required treatment, and frequent use of vasoactive medications such as dietary supplements containing vitamins or antioxidants and bronchodilators. Potential participants were also screened for ocular diseases and excluded from the study if they had a refractive error of more than \pm 3DS and more than \pm 1DC equivalent (to address minification or magnification, which can cause under-or overestimation of the retinal vessel diameter), elevated intraocular pressures (>21 mmHg), retinal disease, a history of intraocular surgery, cataract or any other media opacities preventing adequate examination with the retinal vessel analyser, as well as, any retinal or neuro-ophthalmic disease affecting the ocular vascular system. Further study specific recruitment and inclusion/criteria are defined in each chapter as appropriate.

3.3 Methods

Prior to study initiation, the author was trained to use each investigative technique in this thesis. For techniques that required more than one training session, a series of supervised preliminary examinations on 10 to 15 volunteers was undertaken to ensure that an adequate level of competency was achieved. See Table 3.1 below for a summary of the main investigative techniques used for each of the studies (1-5) in this thesis.

Table 3.1: Overview of I	Vain Investigative Technique	s/Clinical Parameters Measured

	Technique/Parameter	Purpose		Study				
			1	2	3	4	5	
Ocular	-Non-contact tonometry	-IOP	✓	~	~	✓	~	
	-DVA	-Endothelial Function	✓	~	✓	✓	✓	
	-Non-invasive Keratograph	-Tear film and corneal assessment	-	-	✓	-	-	
	- Ocular Osmolarity	- Osmolarity	-	-	\checkmark	-	-	
Clinical	-Anthropometry	-Body composition	✓	~	✓	✓	✓	
	-Sphygmomanometry	-Blood Pressure	✓	✓	✓	✓	✓	
	-Ambulatory blood pressure	- 24 blood pressure	✓	✓	✓	✓	✓	
Biomarkers	-Redox Index (GSH/GSSG)	- Oxidative stress	✓	✓	-	✓	✓	
	-Inflammatory Markers (IL-6)	- Endothelial dysfunction	-	-	-	\checkmark	-	
	-Telomere Length	- Biological ageing	-	-	-	\checkmark	-	
	-Oxidized lipids (Oxysterol)	- Endothelial function	-	-	-	-	✓	
	-Oxidized proteins	- Endothelial Function	-	-	-	-	✓	
	-Lipid Panel	- Circulatory blood lipids	✓	✓	✓	✓	✓	
	-Glucose	- Diabetes	✓	✓	\checkmark	✓	✓	
	-Venepuncture	- Blood withdrawal	✓	~	✓	\checkmark	~	
Other	-General questionnaire	- Medical history/diet/lifestyle	~	~	✓	✓	✓	
subjective	- Framingham Risk Score	- CVDs risk	✓	✓	-	-	-	
	- DEQ-5 item score	- Dry eye severity	-	-	✓	-	-	
	- OSDI Scores	- Ocular surface disease	-	-	~	-	-	

3.3.1. Experimental Protocol

All measurements were performed between 8 am and 11 am following a 12 hour overnight fast, which included refining from alcohol or caffeine. All procedures were conducted in a consistent order outlined below and detailed in the following sections (Figure 3.1):

- 1. Suitable participant identified, approached, and provided with the study information pack
- 2. Procedures and risks explained, concerns addressed
- 3. Consent form read, understood, completed, and signed by participants
- 4. Preliminary assessments:
 - Demographic questionnaire
 - IOP measurement
 - Height and weight measurement
 - Baseline BP readings
 - DEQ-5 and OSDI questionnaires (DED study participants)
- 5. 1% Tropicamide inserted into a randomly selected eye
- 6. Peripheral vascular reactivity assessment
- 6. Fasting venous blood sample obtained by venepuncture
- 7. Ambulatory BP and heart rate variability (HRV) monitor fitted (Study 1&2 participants)
- 8. Assessment of retinal vessel reactivity (DVA)
- 11. Dry Eye Disease examination (DED study participants)
- 12. Final BP measurement

Appointment Flowchart



Figure 3.1: Flowchart of the Experimental Protocol

3.3.2. Preliminary Assessments

Participants who met the inclusion criteria and had provided informed consent were requested to complete a demographic and general health history questionnaire detailing their age, gender, ethnicity, personal and family history of illness, medication, daily diet, tobacco and alcohol consumption, and physical activity routine, as well as, menstrual cycle details for women. A template of the general health questionnaire is provided in Appendix C. Anthropometric measures of height and weight were then recorded, and body mass index (BMI) was calculated as per (Equation 3.1).

BMI= Weight (*Kg*)/Height² (*m*)

Equation 3.1: Body Mass Index

3.3.3. Blood Pressure Measurement

Measurements of BP were performed on two separate occasions, one in-clinic and one outof-clinic.^{535,536}

a. Clinic Blood Pressure Monitoring

Following a short acclimatization period where participants were requested to sit quietly and comfortably for 5 minutes, a baseline BP reading was obtained using an automated BP monitoring device (UA-767, A & D Instruments Ltd., Oxford UK). The procedure involved a BP cuff fastened snuggly around the upper arm above the elbow with the forearm elevated to approximately heart level and supported. The automated device was then initiated, and three subsequent readings of the SBP, DBP, and heart rate (HR) were recorded.

b. Ambulatory Blood Pressure Monitoring

The BP values were confirmed by a second measurement using a computer-operated ambulatory BP and electrocardiogram (ECG) monitor (Cardiotens-01, Meditech Ltd, Budapest, Hungary). All subjects maintained their normal activity and were carefully instructed to complete a diary each time their activities changed, or when any chronic medication was taken. The 24-h BP were later downloaded and analysed using the 'Medibase' software program (Meditech). SBP as well as diastolic DBP measurements were calculated for the daytime (6 am to 10 pm) and night-time (10 pm to 6 am) intervals. At least 80% of the programmed recordings were required for a diurnal curve to be considered in the present analysis.^{514,515}

Using the daytime SBP and DBP values, study participants were stratified into three subcategories: "normal", "high normal", and "Grade I" as recommended by the 2018 ESC/ESH Guidelines. They were then also further classified into three other subcategories: "normal", "elevated", and "stage I", as recommended by the 2017 ACC/AHA guidelines. Subjects classed as "optimal" according to the 2018 ESC/ESH guidelines were excluded, and their values were included in our pool of "normal" data.⁵¹⁴

3.3.4. Intraocular Pressure Measurements

IOP readings were then obtained by way of non-contact tonometry using the Pulsair system (Keeler Ltd., UK) and in accordance with the manufacturer's guidelines. Briefly, the patient was positioned seated comfortably. The patient's eye was located through the device's eyepiece at a distance of approximately 25 cm. The handheld device was then moved closer to the patient, with the other operator hand rested on the patient's forehead (to prevent the device from inadvertently touching the eye). Upon proper alignment, the device automatically fires a puff of air and takes a reading. Following an average of at least three satisfactory readings, the procedure was repeated on the alternate eye.

3.3.5. Ocular Vascular Assessment

Vascular dysregulation in the retina is implicated in the development of vascular diseases⁵³⁷ and typically refers to an inadequate vascular adaptation to the particular needs of the organ; usually characterized as insufficient dilation/constriction or excessive dilation/constriction in response to a physiological stressor. DVA using flicker-light stimulation is one of the most widely used methods for assessing retinal microvascular reactivity.^{538–542} A detailed description of this technique is provided in the following sections.

3.3.5.1. Dynamic Retinal Vessel Analysis

As described in section 1.5.3.2, the RVA system (IMEDOS GmbH, Jena, Germany) enables real-time recording of retinal vessel diameters, and can be coupled to a variety provocative stimulus to evaluate endothelial functioning of the vasculature. Flicker-light stimulation was selected as the provocation method of choice in this thesis. Typically, studies using the RVA system with an integrated flicker simulator are referred to as dynamic retinal vessel analysis (DVA) studies.

3.3.5.2. Device Setup

The device setup comprised of a fundus camera (FF450, Carl Zeiss, Germany), a charged coupling device (CCD) camera, a high-resolution video recorder, a real-time monitor, and a dedicated personal computer with analysis software. The device is also equipped with an optoelectronic shutter placed in the optical pathway of the camera to generate flickering light

85

at a sampling rate of 12.5 Hz, which lies within the optimum flicker frequency range and has previously been shown to induce appropriate retinal stimulation, as illustrated in Figure 3.2.^{512,529,538}





The DVA is capable of measuring retinal vessel diameter continuously along a selected vessel segment over a specified time. This is achieved via the analysis of the brightness profile of the retinal vessels. The brightness profile is based on the absorbing properties of the red blood cells (RBCs) within the vessel. With regard to retinal vessels, maximum absorption of light occurs at a wavelength of 400-620nm and by comparing the brightness profile of the RBC column within the vessel with that of the surrounding tissues, a continuous assessment of vessel diameter can be made. DVA, therefore, measures vessel diameter as the width of the red blood cell column within the selected vessel.

It achieves this as follows:

- The illumination light of the fundus camera enters the eye through a dilated pupil and is reflected by the different layers of the retina and retinal vessels.
- It is then delivered via the observation pathway to the CCD camera.

• The brightness profile data is then analysed by the computer system and simultaneously recorded by a high-quality video recorder so that offline analysis can be conducted at a later date if necessary.

To ensure optimal alignment and set-up, the operator can observe an image of the fundus on the computer display (Figure 3.3) and to ensure optimal contrast for vessel visualisation, a green filter is inserted into the illumination pathway of the fundus camera.

Furthermore, the device is equipped with a series of adaptive algorithms to compensate for any disruption of the vessel brightness profile by the presence of either shadowing structures from the background or reflections from the vessel surface. It also has the capability to correct for slight eye movements during assessment and can continuously monitor image quality, according to image contrast and then automatically remove any inadequate measurements from the analysis.^{534,543}

3.3.5.3. Technical Specifications

The technical specifications of the RVA are summarized in Table 3.2. The resolution of the device limits the accurate measurement of vessel diameters to vessels with diameters greater than 90 μ m, and the temporal resolution of the device is 40 ms, such that 25 video frames are captured per second (i.e., sampling rate = 25 Hz) for the duration of a recording. For the studies in the present thesis, the image field or camera angle was set at a 30° angle, and all size related measurements are expressed in 'units of measurement' (UM), whereby 1 UM is equivalent to 1 μ m in a normal emmetropic eye.

Fable 3.2: Technical Specificat	ions of Retinal Vessel Analyser
---------------------------------	---------------------------------

Parameter	Value	
Measurement range	90 µm	
Measurement resolution	<1 µm	
Temporal resolution	≥ 40ms	
Image field angle	30°	
Recording time	350 seconds (can be up to 10 minutes)	
The maximum length of the vessel segment	3 mm	
Spatial resolution (along vessel segment)	180 µm	

3.3.5.4. Advantages and limitations

With a growing number of studies citing the use of the RVA system, some of the inherent advantages and limitations of its use are important and accordingly summarized below (Table 3.3).

Advantages	Limitations		
Non-invasive	Media opacities compromise image quality		
Optimal spatial and temporal resolution to	Uses standardized measurement or relative		
measure provocation responses.	units rather than the absolute vessel		
	diameter		
Simultaneous assessments of multiple	Heavily reliant on steady patient fixation over		
vessels and vessel segments	recording duration		
High reproducibility 544	Assumes no refractive error		
Low variability ⁵⁴⁵	Requires full pupil dilation		

Table 3.3. Advantages and Limitations of Retinal Vessel Analyser

As mentioned previously, in order to assess the dynamic behaviour and autoregulatory capacity of the retinal vessels, the vessels need to be provoked. The DVA can be operated in conjunction with a variety of provocation devices, including suction cup IOP enhancement, pure oxygen breathing, inhalation of CO2, and flicker light stimulation.⁵⁴³ For this thesis, flicker light stimulation has been used to assess dynamic retinal vessel reactivity, and this will be discussed in more detail in the following section.⁵⁴⁶

3.3.5.5. Flicker Light Stimulation

Flicker light can be considered the most natural provocation method for assessing dynamic retinal vessel behaviour and has the advantage of stimulating the retina exclusively, without the involvement of any other vascular bed. The normal vascular response to flickering light has been widely studied ^{523,529}, and there is plentiful evidence to indicate that, under normal circumstances, flicker stimulation should lead to an increase in vessel diameter, retinal blood flow and ONH blood flow in humans.⁵⁴⁷

Flicker light can be defined as illumination, which alternates in brightness or colour at a frequency of approximately 1-50 Hz.⁵³⁸ Electrophysiological studies have shown that the maximum sensitivity of the human visual system to flicker stimulation is obtained with a flicker frequency of between 10-20 Hz.⁵³⁸ The DVA device used throughout this thesis was equipped

to generate flickering light at a sampling rate of 12.5 Hz via an optoelectronic shutter placed in the optical pathway of the camera. A sampling rate of 12.5 Hz lies within the optimum flicker frequency range and has been shown to provide appropriate retinal stimulation by numerous studies.^{512,529,538}

The measurement protocol used to assess retinal microvascular reactivity to flicker light in this thesis is in accordance with that introduced by Nagel et al.⁵²³ This protocol is widely used and recommended ⁵³⁴ and is outlined in the following section.

3.3.5.6. Measurement Protocol

- The fundus camera is positioned so as to obtain a uniformly illuminated fundus image through the fully dilated pupil without unwanted reflections, and the brightness is adjusted to ensure optimal contrast.
- The patient's fixation is directed, with the use of a fixation needle, so that the measurement area of interest lies in the centre of the fundus picture A region of interest is defined on the retina by the user selecting a rectangular area on the real-time monitor, which outlines the region to be studied (usually inferior to ONH).
- From within this area, a section of the inferior temporal retinal artery and a section of the inferior temporal retinal vein, located approximately 1.5-disc diameters from the ONH and approximately 0.5-1-disc diameters in length and a reasonable distance apart, are selected for analysis and monitoring (Figure 3.3).



Figure 3.3: Example of Retinal Vessel Selection before Analysis

Measurement then starts automatically, and vessel diameters are continuously calculated along the length of the selected vessels over a 350 second testing period. This consists of 50

seconds of baseline measurements under still illumination (25Hz), followed by three cycles of 20-second flicker stimulation (optoelectronically generated at 12.5 Hz), each interrupted by 80 seconds of still illumination (recovery), as outlined in Figure 3.4.



All measurements are performed in a quiet, temperature-controlled room (22°C) following full dilation of one unselected eye (1% tropicamide, Chauvin Pharmaceuticals Ltd). The analysis is conducted in a dark room 30 mins after dilation to allow dark adaptation.

The use of a 350 second testing period with three sequential flicker cycles was initially introduced so that an averaged vessel response could be calculated and analysed to ensure stable results could be obtained over a testing period of tolerable length.^{521,523} More recently, new analysis techniques have been introduced, which also consider the vessel responses to each flicker cycle individually and the overall average.⁵⁴⁸ This is discussed in more detail in the following sections.

3.3.5.7. Normal Vascular Response

The normal retinal vessel response profile to flicker light stimulation by the DVA is illustrated in figure 3.5. Previous studies have shown that the maximum vessel response to flickering light typically occurs within 20 seconds.⁵¹² After this time period only small increases in diameter occur. Once the flicker stimulation ends, dilation ceases immediately but rather than simply returning to baseline, the baseline is usually overshot and a vasoconstriction occurs. This overshoot has been found to start within approximately 6-10 seconds following the end of the flicker period, reaching its minimum diameter between 10-40 seconds following the end of the flicker.^{512,529} The vessel diameter then returns to its baseline level (Figure 3.5). Both retinal arteries and veins respond to flicker light stimulation; however, arteries tend to show a more pronounced diameter change in comparison to veins and whilst arteries start to react immediately, veins have been shown to have approximately a 5-second delay before a response is seen.^{512,538}

Deviation away from this normal vascular response profile can be indicative of vascular disease, and indeed numerous studies have already provided evidence of altered vascular response to flickering light in both ocular and systemic disease, including POAG ⁵⁴⁹, ARMD ⁵⁵⁰, diabetes ^{551–553} and hypertension⁵²⁰. In order to understand the implications of an impaired or altered vascular response to flicker light with regard to the development of disease, it is necessary to try and understand the mechanisms by which flicker light provokes the retinal vessels.



Figure 3.5: Diagrammatic Representation of a Typical Retinal Vessel Response to Flicker Light on Dynamic Retinal Vessel Analysis

3.3.5.8. Reaction Mechanism

Flicker light stimulation increases the neural demand of the retina, which, under normal circumstances, should trigger a neurovascular coupling response of the retinal microvasculature, resulting in vasodilation.⁵⁴⁷ In general terms, an increase in the metabolic rate of the photoreceptors, following stimulation by flickering light, is thought to trigger the release of NO from the retinal vascular endothelium, bringing about an increase in vessel diameter followed by an increase in blood supply to meet the increased demand.⁵⁵⁴ An altered vascular response to flicker light, therefore, could be indicative of impaired autoregulatory 91

H.A. Shokr, PhD Thesis, Aston University 2021

mechanisms and/or endothelial dysfunction in the form of reduced bioavailability of NO. However, due to the complexity of the neurovascular coupling response, it is highly likely that other causative factors such as altered endothelial dependant vasoactive substances levels, altered astrocyte activity or changes in the basal tonus of the vessels could also play a role in altering the retinal vascular response to flicker. The role of these alternative factors may be particularly relevant when considering the vascular constriction response following flicker.⁵⁵⁰

3.3.5.9. Data Analysis

Of primary interest when analysing the retinal vascular response profile to flicker light is the percentage dilation of the vessel in response to the stimulus and the time scale across which this happens, along with the percentage constriction or overshoot of the vessel following cessation of flicker and again the time scale across which this happens.

In this thesis, such analysis of the retinal vascular response was conducted using elements of a newly defined method of DVA analysis, termed 'Sequential and Diameter Response Analysis' (SDRA) in conjunction with our novel vascular profile imaging methods. The inbuilt software of the DVA device itself does have the capability to provide an analysis of the retinal vascular response to flicker light; however, this has been identified to have a number of shortfalls and the need to move away from this traditional inbuilt software analysis and to evaluate the retinal vascular response profile in more detail has been increasingly realised by numerous authors in recent years.^{534,548,550}

As described in section 3.3.5.3., the spatially and temporally defined segments of the selected vessels translate into 25 captured video frames per second and for the duration of the recording (350 seconds). These data include 25 diameter recordings per second (25 x 350) as well as a spatial recording along the length of segment drawn (usually 1 mm), which is then compressed by the device software to provide an averaged spatial-temporal reading or local temporal course for each vessel (Figure 3.6).



Figure 3.6: Local Temporal Course Representation of Retinal Vessel Analysis. For each examination, a vessel segment approximately 1 mm in length and approximately 1 to 2 disc diameters away from the optic disc was selected. The vessel longitudinal section within the region of interest was scanned 25 times per second and continuous diameter data was obtained along the vessel over time, thus creating a three-dimensional matrix of values as depicted.

Indeed SDRA, first introduced by Heitmar et al. ⁵⁴⁸, was primarily designed to overcome the shortfalls of the inbuilt software analysis programme. An overview of this inbuilt DVA analysis and its limitations, along with a summary of SDRA and the novel imaging analysis methods used in this thesis, is given in the following sections.

3.3.5.10. Previous Methods of Analysis

The inbuilt DVA software calculates the vascular dilation response to flicker light by averaging all three of the stimulation cycles and then taking the average diameter from the last +/- 3 seconds of flicker stimulation as the maximum diameter response to flicker (i.e., average diameter reached between 17-23 seconds from the start of flicker taken as maximum diameter response). The shortfalls of this method largely arise due to its incorporation of both time and diameter responses and its inaccurate assumptions about the nature of the vascular response.

Indeed, subjects who reach their maximum dilation outside the 17-23 second window would have their maximum dilatory response underestimated by this technique. Furthermore, averaging the results from all three flicker cycles, differences in the reaction pattern or time course of each individual cycle cannot be determined.⁵⁴⁸ Additionally, this method of analysis does not take in to consideration baseline fluctuation in vessel diameter (BDF), a parameter first highlighted as important in DVA analysis by Nagel et al.⁵²⁰ BDF refers to the spontaneous variations in vessel diameter, which occur under normal resting conditions, as a result of vascular tone and arterial pulsation and are superimposed on the vascular response profile.⁵²⁰ In order to account for the influence of BDF, Nagel et al. introduced the concept of baseline-corrected flicker response (BCFR), where BDF is accounted for by subtracting it from the dilation amplitude (DA) of the vascular response, as shown in Equations 3.2 and 3.3.⁵²⁰

BCFR = DA - BDF

BCFR = Baseline corrected flicker response DA = Dilation amplitude

BDF = Baseline diameter fluctuation

Equation 3.2: Baseline Corrected Flicker Response

$$DA = MD - MC$$

DA = dilation amplitude

MD = maximum dilation

MC = maximum constriction

Equation 3.3: Dilation Amplitude

3.3.5.11. Sequential and Diameter Response Analysis (SDRA)

SDRA has the advantage of utilising the raw data set generated by the DVA device and allowing each individual flicker cycle to be considered separately. This enables a more accurate assessment of dynamic vessel response to be obtained and enables the inclusion of the parameters BDF, BCFR and DA on top of the standard percentage dilation and constriction response parameters. Additionally, it allows the time taken to reach maximum dilation and the time taken to reach maximum constriction to be determined for both the artery and vein for each individual flicker cycle.

It is important to note that analysis of individual flicker cycles is reliant on a full data set having been obtained from the participant on each subsequent cycle. If this is not achieved, for

example, due to poor patient fixation, loss of concentration or excessive blinking, calculating and analysing the average data set using SDRA is considered more reliable.

Overall the SDRA method has been validated and shown to be a sensitive measure of the vascular response to flicker light with good coefficients of variation.⁵⁴⁸ Furthermore, it is able to overcome a number of the shortfalls of the inbuilt RVA analysis software. It has therefore been the analysis method of choice for this thesis; however, we have taken it one step further and using the principles of SDRA have developed an additional way of imaging the retinal vascular profile to allow further aspects of the vascular response to be explored.

3.3.5.12. Novel Analysis

Whilst SDRA overcomes many of the limitations of the inbuilt RVA software analysis, it still does not allow visualisation of the entire dynamic retinal vessel response profile. Furthermore, it has been suggested that, in addition to the parameters illustrated in sections 3.3.5.10 and 3.3.5.11, evaluation of the slope of both the dilation and constriction responses to flicker light could give additional important information about the state of the retinal microvasculature in health and disease. In order to address this we have developed a new method of analysing and interpreting the retinal vascular response to flickering light using Matlab (MATLAB R2010a; MathWorks Inc., Natick,MA). Our method expands on the SDRA methodology by extracting the raw response data and applying a statistical polynomial regression algorithm, implemented using the polyfit and polyval functions of the Matlab high level programming language (MATLAB R2010a; MathWorks Inc., Natick, MA).

Given the measurements y_i at times t_i , i=1 ...T, we approximated y = f(t) by a polynomial of degree b as:

$$p(t) = p_1 t_1^n + p_2 t^{n-1} + \dots + p_n t + p_{n+1}$$

The polyfit function locates the coefficients P_1 , P_2 ,... P_n , P_{n+1} such that the error $\sum_{i=1}^{T} (y_i - p(t_i))^2$ is minimized. This involves solving the system of equations:

$$p_{1}t_{1}^{n} + \dots + p_{n}t_{1} + p_{n+1} = y_{1}$$

$$p_{1}t_{T}^{n} + \dots + p_{n}t_{T} + p_{n+1} = y_{T}$$

If we denote $t_i^{n-j+1} = v_{ij}$ then $V = (v_{ij})$ is the Vandermonde matrix and the least squares problem to be solved can be written as $V_P = y$



The polyval function was then used to calculate the fitted polynomials that ultimately provided us with curves representative of the dynamic vascular response profile, which could then be used for analysis. The degree of the polynomial, n, is an adjustable parameter. In this case, n = 20 was applied for consistency as this provided the closest fit polynomials on the data points. Also based on the principles of SDRA in a more recent study our lab introduced slope⁵⁴¹ as an additional parameter to more accurately describe vessel behaviour as slope characterizes the interaction between the change in vessel diameter and the rate at which this change occurs. Slope can be determined independently for both the dilation (Slope_D) and constriction (Slope_C) components and in both arteries and veins as per (Equation 3.4) and (Equation 3.5).

$$Slope_D = \frac{MD - Baseline}{tMD}$$

Equation 3.4: Dilation Slope

$$Slope_{C} = \frac{MC - MD}{tMC}$$

Equation 3.5: Constriction Slope

Visualization plots can be created using the algorithm for each individual flicker cycle as well as an averaged or composite response plot for all flicker cycles, with the artery and vein regarded separately. An example of a visualization plot is illustrated in Figure 3.7. A summary of the main parameters of interest to this thesis is provided in Table 3.4.



Figure 3.7: Diagrammatic Representation of the SDRA Parameters Abbreviations: SDRA, sequential diameter response analysis; BDF, baseline diameter fluctuation; MD, maximum diameter; MD%, percent dilation relative to average baseline diameter; MC, minimum constriction diameter; MC%, percent constriction relative to average baseline diameter.
 Table 3.4: Summary of DVA Parameters Calculated and Used for Analysis

Acronym	Parameter	Description/Calculation
Baseline	Baseline diameter	Average diameter during baseline recording
BDF	Baseline diameter	Maximal range of diameter measurements during
	fluctuation	baseline
MD	Maximum diameter	Point of maximum dilation following onset of flicker
MD%	Percentage dilation	Percentage change in vessel diameter relative to
		baseline ((MD – Baseline)/Baseline) *100
MC	Minimum diameter	Point of maximum constriction after MD (post-flicker)
MC%	Percentage	Percentage change in vessel diameter relative to
	constriction	baseline ((MC-Baseline)) /Baseline) *100
tMD*	Reaction Time	Time (seconds) taken to reach MD following the onset of flicker
tMC*	Constriction Time	Time (seconds) taken to reach MC from MD post-flicker
DA	Dilation amplitude	MD-MC
BCFR	Baseline corrected	Change in vessel diameter taking into consideration
	flicker response	baseline diameter fluctuation (DA–BDF)
Slope _D	Dilation slope	(MD–baseline)/tMD
Slope _C	Constriction slope	(MC–MD)/tMC

*Since the tMD and tMC values derived using the matlab algorithm reflect the absolute time at MD or MC, these values were expressed so as to only account for the exact number of seconds taken to reach MD since the onset of flicker (i.e., time at MD - 30 seconds) and the time in seconds taken to reach MC from MD (i.e., time at MC - time at MD).

3.3.6. Systemic Vascular Assessment (Digital thermal monitoring (DTM))

Peripheral microvascular reactivity at the level of the fingertips was assessed by way of DTM using VENDYS 5000BC Digital Thermal Monitoring (DTM) system (Endothelix, Inc., Houston, TX, USA). This FDA approved device consists of a computer-based thermometry system (0.006°C thermal resolution), with two special thermocouple fingertip probes designed to minimize the area of skin-probe contact and fingertip pressure. A standard sphygmomanometer cuff and a compressor unit to control cuff inflation and deflation is included to facilitate the occlusion-hyperaemia protocol

3.3.6.1. Measurement Protocol

The test is conducted with the patient at rest for 30 minutes in the supine position, in a quiet, dimmed room with an ambient temperature of 22°C to 26°C. VENDYS DTM probes are affixed to the index finger of each hand, and after a period of stabilization of basal skin temperature (defined as stabilization within a 0.05°C threshold), the temperature is measured in the index fingers of both hands (of which the right arm only is subjected to occlusion-hyperaemia) with an automated, operator-independent protocol. The right upper arm cuff is rapidly inflated to \geq 50 mmHg above systolic pressure for 5 minutes and then rapidly deflated to invoke reactive hyperaemia distally.⁵⁵⁵

Thermal tracings are measured continuously and digitized automatically using a computerbased thermometry system with 0.006°C thermal resolution. Dual-channel temperature data is simultaneously acquired at a 1 Hz sample rate.

(Figure 3.8) shows a representative example of a temperature-time trace and the primary DTM-derived measures related to thermal debt and recovery that were recorded and calculated. Temperature rebound (TR): maximum temperature- start temperature (just before cuff inflation); adjusted temperature rebound (aTR): temperature rebound/ start temperature; area under the curve temperature rebound (AUC_{TR}): area under the curve between maximum temperature and minimum temperature. The post-occlusive adjusted temperature rebound aTR determined by the software algorithm is directly associated with the extent of the subjects vascular reactivity.⁵⁵⁶ An aTR below 1 is considered to show poor cardiovascular reactivity, whereas an aTR of between 1 and 2 is considered intermediate, and an aTR of more than 2 is considered healthy.



Figure 3.8: Graphical Representation of The Digital Thermal Monitor Software Analysis. T MAX, maximum temperature; TMIN, minimum temperature; AUCTR, Area under the curve temperature rebound.

3.3.7. Assessment of Anterior Ocular Structure and Function These tests are a collaborative work with Dr Sonia Trave Huarte 3.3.7.1. Keratograph

The Keratogrpah (K5M, Optikgerate GmbH, Wetzlar, Germany), combines keratometry measuring processes with topographic mapping. It is an illumination system which has a special reflector illuminating a placido bowl which contains a series of 22 white concentric rings, and thus images obtained are reflected from this placido bowl from the patients' eye. Besides being an advanced corneal topographer it also contains additional imaging modalities to measure the anterior ocular surface. These include infra-red measurements of the Meibomian glands at 840nm, evaluation of the tear break-up time non-invasively, measuring the amount of bulbar and limbal hyperaemia, tear meniscus height, and a dynamic evaluation of the lipid layer. It also has a built-in video recorder and software to analyse the data. It can be used for future reference when evaluating the ocular surface after treatment or post-surgical interventions.

3.3.7.2. Procedure

The test is conducted in a dim room to eliminate reflections from the surface of the keratograph. The patient is in a seated position with their chin placed on the chin rest and the outer canthus is aligned with the chin-forehead reference bar. The patient focuses on the red light located directly in the centre of the concentric rings. All subsequent measurements are taken from this reference point. The following procedures described below are in the order of the protocol used in this thesis.

3.3.7.3. Tear Meniscus Height

Images obtained of the tear meniscus height (TMH) from the keratograph are taken from the lower lid. The tear meniscus can be evaluated with different options to magnify the view. The height of the tear meniscus can give an indication of the tear volume, i.e., Quantity of tear reserve. TMH should be directly measured under the pupil centre at the lower lid. The classification of normal tear reserve is as follows: good >0.2mm; normal = 0.2mm; poor < 0.2mm.⁵⁵⁷ The software allows the user to quantify the height of the tear meniscus with an integrated ruler. An example of the assessment is presented below in Figure 3.9.



Figure 3.9. Example of TMH Measurement

3.3.7.4. Non-Invasive Keratograph Break up Time (NIKBUT)

The non-invasive break up time is evaluated and displayed in a colour coded map of the cornea Figure 3.10. Once the patient is aligned, the clinician will require the patient to blink twice consecutively. This second blink triggers the program to initiate the recording. The recording of break time up time ceases when one of two events occur; the patient blinks or there is a significant amount of distortion from the reflected image of the placido rings. The information is then encoded by the software and displayed to the clinician. The time it takes for the tears to break up can give an indication of the quality of the tear film. Normal tear break up time is > 10 seconds. Areas highlighted in red indicate an unstable tear film, and areas in green indicate a stable tear film.



Figure 3.10. Example of NIKBUT Measurement

3.3.7.5. Dynamic Evaluation of the Lipid Layer

The evaluation of the lipid layer is performed with the patient looking directly at the centre of the placido disc Figure 3.11. Once aligned, the measurement commences, and the clinician should advise the patient to blink normally as to reveal the spread and formation of the lipid layer across the corneal surface. This measurement is usually recorded to assess the dynamic behaviour of the tear film, as well as giving an indication of the thickness of the lipid layer. If the interference pattern displays colours and structures, it is regarded as normal; however, if no colours or structures are visible it could be an indication of early tear film evaporation.

Recordings can range from 5 sec to 10 seconds or longer if the clinician deems so. There is also the ability to capture images within the video recording for analysing images later on.



Figure 3.11. Example Measurement of Dynamic Evaluation of Lipid Layer

3.3.7.6. Redness

The Keratograph is also able to evaluate the amount of limbal and bulbar hyperaemia. This feature can record and grade redness automatically and objectively with the inbuilt software that compares the amount of redness and analyses it according to the Efron grading scale. This area analysed is based on the area percentage between the vessels and the rest of the analysed area represented in figure 3.12.



Figure 3.12. Measurement Example of Bulbar Redness

3.3.7.7. Meibography

The Meibomian glands structure and can be imaged with infra-red light. The superior lid as well as the lower lid must be everted before capturing the image. The morphological changes in the glandular tissue can be made visible using a 3D meibo scan represented in Figure 3.13. Once the image is captured the software makes a contrast enhanced image alongside a normal image which can be viewed by the clinician and is usually subjectively graded.



Figure 3.13. Example of Infrared Superior and Inferior Meibography

3.3.7.8. Data Analysis

The inbuilt software is capable of analysing data for a number of images taken. These are limited to the amount of Hyperaemia, which is graded according to the EFRON grading scale and the NIKBUT with the use of the placido discs. These values are generated by the software algorithm itself and have been previously used in studies that show good agreement with previous techniques and is highly repeatable and reproducible.^{558,559}

The quantification of tear meniscus height is made on the actual image with the use of an inbuilt ruler which measures to 1/10th of a mm and can be graded accordingly. The classification of lipid layer thickness can be made subjectively on the interference pattern observed during the video recording. An open/closed meshwork correlates to a thickness of 13-50nm, wave or flow pattern is between 50-70nm, an amourphous pattern is between 80-90nm, Colour fringes represent a thickness between from 90-180nm, and finally a globular network represents a thickness of >200nm.

The Meibomian glands can be analysed according to a number of different protocols. Originally scales were developed on a 4 point scale to assess the level of meibomian gland drop out or a change in its morphology.^{560,561} More recently, a new scale was introduced by Pult et al ⁵⁶², which incorporated a 5 point scale based on the level of Meibomian gland dropout. The infrared image is captured and then analysed with the imajeJ software V1.49 (Wayne Rasband, NIH, USA). The software allows the user to trace the whole area which is going under analysis and then to trace the area where the glands drop out represented in Figure 3.14. The percentage of gland dropouts can be calculated by dividing the (Area of dropout/whole area) *100. In a further study Pult et al. ⁵⁶³ analysed the repeatability of objective grading and subjective grading using the Meibomian gland loss ratio and determined that intra-observer and inter-observer agreement was better in computerized grading following the subjective 5 point scale and 4 point scale.



Figure 3.14. Measurement Example with Image J representing area of loss of the superior tarsus.

3.3.8. Blood Sampling

After ensuring that the patient had fasted for at least 10-12 hours, blood samples were drawn from the antecubital fossa vein for subsequent analyses. The author was trained and certified on all aspects of the blood draw procedure prior to obtaining blood samples. Briefly, the procedure involved placing a tourniquet around the patient's forearm approximately 2 cm above the proposed collection site, and the desired vein was inspected and located via palpation. The skin over the selected area was then sterilized, and the vein was anchored below the collection site. The needle of the syringe assembly, held parallel to and alongside the vein, was quickly inserted into the vein, and blood was aspirated into EDTA Vacutainer® tubes. The tourniquet was released just as the last collection tube was filled, the needle was deftly withdrawn, and pressure was simultaneously applied to the site. The patient was instructed to continue applying light pressure on the site until bleeding, if any, had ceased.

3.3.9. Blood Analysis

The fasting EDTA blood samples obtained were immediately assessed for glucose (GLUC) and TG, and plasma total cholesterol (TChol) and HDL-c using a Reflotron Desktop Analyser (Roche Diagnostics, UK). In addition, LDL-C levels were estimated as per the Friedewald equation (Equation 3.6). Aliquots of the remaining EDTA blood and plasma were then processed and stored at approximately -80°C for further analyses.⁵⁶⁴

 $LDL = \frac{Total \ cholesterol - (HDL + TG)}{2.17}$ LDL= Low-density lipoprotein HDL= High-density Lipoprotein TG= Triglycerides

Equation 3.6: Friedewald Equation

3.3.10. Systemic Circulatory Markers – Biochemical Assays

3.3.10.1. Glutathione/Glutathione Disulphide Redox System

a. Glutathione

As discussed in section 1.4.4 glutathione/glutathione disulphide (GSH/GSSG) is the most abundant thiol redox system and is essential for the maintenance of redox balance in cells.⁵⁶⁵ Cellular glutathione mainly exists in reduced form (GSH), which affords an elegant mechanism for redox control of metabolic processes, the failure of which has important implications for optimal endothelial function.⁵⁶⁶ In the present thesis, glutathione levels were based on the principles of an enzymatic-recycling assay.^{567,568} This technique, developed in-house and validated previously ⁵⁶⁹, relies on the reaction of free thiol groups with a sulfhydryl reagent DTNB (5,5'-dithiobis-2-nitrobenzoic acid) to form the yellow derivative (TNB, 5'-thio-2-nitrobenzoic acid), which can then be measured via absorbance spectrophotometry at 410 nm. The blood GSH and GSSG concentrations measured in this thesis were in good agreement with literature data in control patients in the ranges of 150 – 1500 μ M and 1 to 500 μ M, respectively ⁵⁷⁰, and suggests that the experimental conditions reported in this thesis are suitable for the analysis of total glutathione and glutathione disulphide concentrations in whole blood. The validity and reliability of the spectrophotometric method of detection has also previously been established.⁵⁷¹

b. Glutathione Recycling Assay Principle

An evaluation of systemic oxidative stress status was performed in this thesis through the determination of circulating reduced glutathione (GSH) in fasting venous blood samples. GSH, a tripeptide (γ -glutamylcysteinylglycine) with a free thiol group, is a major antioxidant in human tissues. During the reduction of hydrogen peroxide (H2O2) to water (H2O) and the respective alcohol – a reaction that is catalysed by glutathione peroxidase (GPx) – GSH becomes oxidized glutathione (GSSG) (Equation 3.7). GSSG can, in turn, be recycled to back to GSH in the presence of glutathione reductase (GSR) and β -nicotinamide adenine dinucleotide phosphate (NADPH) (Equation 3.8).

$$H_2O_2 + 2GSH \xrightarrow{GPX} GSSG + 2H_2O$$



$$GSSG + NADPH + H^+ \xrightarrow{GSR} 2GSH + NADP^+$$

Equation 3.8: Oxidized Glutathione Reduction

During increased oxidative stress in cells, the ratio of GSH/GSSG decreases as a consequence of GSSG accumulation. The measurement of total GSH, GSSG and the GSH/GSSG ratio is therefore considered a useful indicator of oxidative stress status. The assays and analysis described below were all conducted by the author and optimized in-house according to previously reported and validated methods.⁵⁷² The method used is described below.

c. GSH Assay

Protocol: To minimize auto-oxidation of thiols or the enzymatic reduction of disulphides, sample processing was carried out immediately whereby a 30 µL aliquot of the blood sample was pre-treated with 33.3 µL of 100 mg/mL SSA (5-sulfosalicylic acid) and 936.7 µL of sodium phosphate buffer (pH 7.5) to release GSH via cellular disruption and protein precipitation. The sample was centrifuged at 13,000 rpm for 5 minutes, and aliquots of the supernatant were stored at -80oC for further analyses. Based on previous reports of sample stability, assays were conducted within 2 months of collection.⁵⁷³ GSH standards were prepared from 0 to 80 μ M in increments of 20 μ M with the same final concentrations of SSA (1%) as in the samples. To each well of a 96-well plate 150 µL of daily buffer (125 mM sodium phosphate, 6.3 mM disodium EDTA and 0.3 mg/mL NADPH), 50 µL of 6mM DTNB solution, and 25 µL of standards and samples were added in triplicate. The plate was incubated for 3 minutes at 37°C following which 25 µL of GSR was added to each well. Any GSSG formed was thereby recycled to GSH by GSR in the presence of NADPH and the plate was read at 410 nm at 0, 1, 2, 3, and 5 minutes. Standard curves of the GSH concentration were generated using a linear regression program (Microsoft Excel, Microsoft Corporation, USA). An example of a standard curve is provided in Figure 3.15.


Figure 3.15: Standard Curve for the GSH Assay

d. GSSG Assay

The accurate measurement of GSSG has in the past been more challenging due to lower concentrations in tissues and the lack of effective methods to prevent the oxidation of GSH during sample preparation. For the measurement of GSSG levels, therefore, the reagents used were the same as those described above for the GSH assay. In addition, the standards and samples were pre-treated with 2-vinylpyridine (2-VP) in order to derivative GSH without interfering with GSR reaction. A summary of the GSH/GSSG recycling assay is depicted in Figure 3.16.



Protocol: GSSG standards were prepared from 0 to 10 μ M in 1 μ M increments. In addition, the 100 μ L aliquots of the standards and samples pre-treated with 2-VP were adjusted to a pH of 7.5 with triethanolamine (TEA). The assay was then carried out as described above for GSH where 25 μ L of standards and samples were added in triplicate to a 96-well plate containing 150 μ L of daily buffer and 50 μ L of DTNB in each well, incubated at 37 °C for 3 minutes, treated with 25 μ L GSR and read at 0, 1, 2, 3, and 5 minutes. Standard curves of the GSSG concentrations were similarly generated using the linear regression program described above (Figure 3.17).



e. Analyte Concentration Calculations

The GSH and GSSG concentrations were based on the net reaction rate, construction of the standard curves, and calculation of the analyte concentrations in the samples from which total GSH levels and the redox index were then calculated as follows.

The change in absorbance (A) at 410 nm is a linear function of the analyte concentration in the reaction mixture (Equation 3.9):

 $A_{410} = Slope \times minutes \times intercept$

Equation 3.9: Absorbance Change

The net rate is the difference between the rate at each concentration of the GSH or GSSG standard and the blank rate (Equation 3.10). The general form of the equation describing the calibration curve is:

 $Net rate = Slope \times GSH \text{ or } GSSG + intercept$



Therefore, to calculate the GSH or GSSG concentration (Equation 3.11):



Equation 3.11: GSH and GSSG Concentration

Finally, the total GSH and redox index were calculated as per (Equation 3.12) and (Equation 3.13).

 $tGSH = GSH + (2 \times GSSG)$

Equation 3.12: Total GSH Calculation

Redox index = GSH/GSSG



3.3.10.2. Griess Assay for Nitrite

Disturbances in NO bioavailability or production are thought to be responsible for functional vascular alterations associated with endothelial dysfunction and atherosclerosis.⁵⁷⁴ Nevertheless, the transient and volatile nature of NO makes it unsuitable for most analytical assessment in complex matrices such as blood and plasma. Thus endothelial NOS activity is typically assessed as the plasma concentration of nitrite (NO₂) and nitrate (NO₃)⁵⁷⁵, based on observations that NO is converted to NO₂ and NO₃ when inhaled or added to blood since NO₂ is oxidized to NO₃ by haemoglobin.⁵⁷⁶ Previous reports propose that NO₂ more specifically represents a delivery source for intravascular NO and reflects acute changes in regional eNOS activity.⁵⁷⁷ The Griess assay is based on a diazotization reaction and detects the presence of nitrite. In order to measure both NO₂ and NO₃ the NO₃ must be enzymatically converted to NO₂. The Griess reagent system includes a 1% sulphanilamide in 5% phosphoric acid solution, 0.1% N-(1-napthyl) ethylenediamine dihydrochloride (NED) in distilled water and a nitrite standard stock solution of 0.1 mmol/L sodium nitrite in distilled water.

a. Protocol

Reagent preparation: the 1% sulphanilamide solution was prepared by first adding 0.5 mL phosphoric acid (H2PO4) to 10 mL of distilled water and dissolving 0.1 g sulphanilamide. The 0.1% NED solution was prepared by dissolving 0.01 g in 10 mL. For 1 mL of 100 μ M (0.1 M) nitrite standard solution, 6.9 mg of sodium nitrite was dissolved in 1 mL of distilled water based on formula weight (FW) calculations i.e., mg = (mM x FW x mL) /1000).

Standard preparation: twenty-four wells of the 96-well plate were designated for the nitrite standard reference curve (Figure 3.18) 100 μ l of the 100 μ M nitrite solution was added to three wells in row A, and 50 μ l of the appropriate buffer (phosphate buffered saline, PBS) was pipetted into 3 wells in rows B-H. A serial dilution was then carried out (50 μ l/well) in triplicate down the plate to generate the nitrite standard reference curve (100, 50, 25, 12.5,6.25, 3.13, 1.56 μ M), discarding the 50 μ l from the 1.56 μ M set of wells. The final volume in each well was 50 μ l, and the nitrite concentration range was 0-100 μ M.

Procedure: 50 μ l of each experimental sample was then added to the remaining wells in triplicate, followed by 50 μ l of sulphanilamide solution to all wells (standards and samples). The plate was incubated for 5-10 minutes at room temperature protected from light before adding 50 μ l of the NED solution to all wells. The plate was incubated for a further 5-10 minutes at room temperature protected from light and the absorbance was read using plate reader (absorbance spectrophotometry) with a filter of 550 nm. A standard curve was generated to determine the NO₂ concentration in each sample.



3.3.10.3. Relative Telomere Length (RTL)

DNA extractions were performed using the QIAamp DNA Mini Kit (Qiagen) according to the manufacturer's protocol.⁵⁷⁸

DNA purity was detected by NanoDrop ™ 1000/c (Spectrophotometers, Thermo Fisher Scientific, USA) and samples were stored at -80°C until further analysis.

Relative telomere length was measured using real-time polymerase chain reaction (RT-PCR) according to a previously published method ⁵⁷⁹ using LightCycler® 480 Instrument (Roche Diagnostics GmbH, Germany). Briefly, primers for telomere repeats and a normalising genomic sequence (Table 3.5) was prepared in a 25 μ I PCR reaction, consisting of Precision 2× qPCR Mastermix (0.025 U/ μ I Taq polymerase, 5 mM MgCl₂, dNTP mix 200 μ M each dNTP) and 15 ng of template DNA. Samples for both the telomere and single-copy gene amplifications were performed in triplicate with non-template control. The ratio of telomere to the normalising genomic control sequence (T/S ratio) was calculated as previously described ^{579,580} to provide an indication of relative telomere length (RTL).

Table 3.5: Telomere Primers Genomic Sequence

CGG TTT GTT TGG GTT TGG GTT TGG GTT TGG GTT	hTelo Forward
TGG GTT	
GGC TTG CCT TAC CCT TAC CCT TAC CCT TAC CCT	hTelo Reverse
TAC CCT	
TCCAGGCTTTGGGCATCA	36B4 Forward
CTTTATCAGCTGCACATCACTCAGA	36B4 Reverse

3.3.10.4. Extraction and Liquid Chromatography-Tandem Mass Spectrometry (LC-MS/MS) of Plasma Oxysterols.

This analysis is a collaborative work with Dr Irundika Dias

Oxysterols were extracted from 70 μ l of human plasma, spiked with 1 ng of internal standards following the protocol described by Dias et al.³²³ Plasma were mixed with 430 μ l methanol, vortexed and incubated on ice for 10 min in the presence of 4 mg/ml BHT before centrifugation at 14,000 × g for 10 min. The methanolic supernatant was diluted with acidified water up to 10 % of methanol for loading onto an SPE cartridge. Oxysterols were eluted with 1.8 ml of butyl acetate and dried under a vacuum.

The lipid extracts were dissolved in 40 μ L of 40:60 Methanol:H2O and analysed with a liquid chromatography UltiMate 3000 HPLC system (Dionex, Thermo Scientific Ltd.) coupled on-line with an electrospray tandem triple quadrupole-linear ion trap mass spectrometer (QTrap 5500, ABSciex) as described previously (Dias et al., 2018) with some modifications.³²³ Briefly, the lipid extracts were separated using a reverse phase C18 column (100 × 3.2 mm, 5.0 μ m particle size; Macherey-Nagel) with mobile phases A (70:30 Methanol:H2O with 0.1% formic acid) and B (90:10 Isopropanol: Methanol with 0.1% formic acid). The flow rate was 200 μ L/min, and the column was maintained at 45 °C.

3.3.11. Symbolic Regression

This analysis is a collaborative work with Dr Aniko Ekart and Dr Victoria Lush

To model and subsequently predict the cardiovascular risk factors of age and BP a symbolic regression-based analysis was used.^{581,582} For this, we applied the GPLEARN genetic programming with symbolic regression library in python.⁵⁸³

As all these measurements are numerical values, and the relationships between these are unknown and, possibly, of nonlinear nature, symbolic regression represents an excellent framework for modelling CVD risk factors. While classical regression methods rely on a priori definition of the model structure and only provide parameters for the pre-selected models, symbolic regression derives both the model structure and its parameters automatically. The mechanism powering symbolic regression is genetic programming, a form of evolutionary computation for automatically generating computer programs or functions that best describe the data provided to it.

The practitioner only has to provide the set of functions and variables to base the model on and genetic programming will learn both the model structure and parameters automatically from the data, only including those functions and variables that are necessary. If not controlled, the models evolved by genetic programming quickly become very large and hence both time consuming to evaluate and difficult to interpret, without necessarily improving in quality. For this reason, we are applying so-called parsimony pressure to control model size. Symbolic regression has already been applied successfully in a variety of domains, recently including medical, for example an automatic machine learning approach for brain age prediction.⁵⁸⁴ Differently from other so-called black-box artificial intelligence (AI) approaches, symbolic regression solutions will be interpretable and will enable healthcare professionals to draw conclusions about the model and the relationships it uncovers.

Models were generated for each age group separately. With this division we ensure equitable consideration of all ages, without the need to apply data augmentation methods to generate synthetic data. As symbolic regression automatically determines the most relevant features as well as the structure of the regression model, there is no need to separately apply any dimensionality reduction method.

For each age group, we compared predicted age based on the models generated on the combined telomere relative expression and artery measurements, artery measurements only and telomere relative expression only. We used k-fold cross-validation and mean absolute error (MAE) for assessing model quality.

An example age prediction tree generated by Symbolic Regression and its corresponding expression formula are shown in Figure 3.19. The tree represents best model produced for 115

age group 1 arterial measurements and telomere experiment (fold 5, MAE 1.157639). Table 3.6 presents Symbolic Regression feature name encodings used in prediction trees and expression formulas and their corresponding feature names.

Table 3.6: Artery measurements and telomere encodings for the Symbolic Regression formulas.

Feature Representation	Corresponding Measurements
X0	Telomere Relative expression
X1	Artery baseline
X2	Artery Baseline Diameter Fluctuation
Х3	Artery Maximum Dilation
X4	Artery Time to Maximum Dilation
X5	Artery Maximum Dilation Percentage
X6	Artery Maximum Constriction
Х7	Artery Time to Maximum Constriction
X8	Artery Maximum Constriction Percentage
X9	Artery Dilation Amplitude
X10	Artery Baseline Corrected Flicker Response
X11	Artery Dilation Slope
X12	Artery Constriction Slope



Figure 3.19: Symbolic Regression tree for age group 1, telomere + arteries experiment (MAE 1.157639). Corresponding expression formula:

àdd(add(add(add(add(add(add(add(add(div(X1, X4), add(X4, X12)), div(X10, X2)), div(X10, X2)), X12), div(X10, add(X3, mul(add(div(X1, X4), add(X4, X12)), X8)))), X12), X12), X11), div(X10, add(sub(add(sub(add(add(add(add(add(add(mul(div(X6, mul(X6, X12)), X4), mul(0.467, X6)), X12), X12), X11), div(X10, add(div(X2, add(sub(div(div(X0, add(X1, X3)), X2), sub(X6, add(X9, X7))), add(add(add(mul(div(X6, mul(X6, X12)), X4), mul(0.467, X6)), X12), X12), div(X5, X10)), add(X3, mul(add(div(X1, X4), add(X4, X12)), X8))))), 0.927), div(X10, X2)), 0.927), div(div(X10, X2), X2)))).

3.3.12. Statistical Analysis and Sample Size Calculation

Sample size was calculated using G-power software. The first step in power calculation is seeking out published papers in the area of empirical interest that answer theoretically, conceptually, or physiologically similar research questions and use the reported values associated with the statistical results to define the population of interest. Next the researcher defined the inclusion and exclusion criteria for each study in order to define the effect size that will be taken. Probability sampling is used in experimental designs and a P-values of <0.05 were considered significant with a β error of 80%.

All data are reported as mean (SD) unless otherwise indicated. The Shapiro–Wilk test was used to determine the distribution of the data. Univariate associations were determined using Pearson's (normally distributed data) or Spearman's method (non-normally distributed data), and forward stepwise regression analyses were performed to test the influences of systemic and circulating markers on the measured variables. In multivariate regression models the β coefficient value was considered to answer the question of which of the independent variables has a greater effect on the dependent variable as β refers to the SD change in the dependent variable per SD increase in the predictor variable and is particularly useful when variables are measured in different units. Differences between groups were subsequently assessed using one-way analysis of variance (anova) or analysis of covariance (ancova). All analyses were performed using Statistica® (version 13.3, StatSoft Inc., Tusla, OK, USA) software.

4. Study 1: Microvascular Function and Oxidative Stress in Adult Individuals with Early Onset of Cardiovascular Disease

This study has been published in Scientific Reports, Nature (Appendix 13.4.1) ⁵¹⁴ 4.1. Abstract

Aims: The current study aims to investigate retinal vascular function and its relationship with systemic anti-oxidative defence capacity in normal individuals versus those with early hypertensive changes according to the current ESC/ESH guidelines.

Methods: Retinal microvascular function was assessed in 201 participants by means of dynamic retinal vessel analysis. Blood pressure, lipid panel, oxidized (GSH) & reduced glutathione (GSSG) were also evaluated for each participant.

Results: Individuals classed as grade 1 hypertension demonstrated higher retinal arterial baseline diameter fluctuation (p = 0.0012), maximum dilation percentage (p = 0.0007), time to maximum constriction (p = 0.0003) and lower arterial constriction slope (p = 0.0131). Individuals classed as high normal, and grade 1 hypertension also demonstrated higher time to maximum dilation than individuals classed as optimal or normal. GSH levels correlated negatively with SBP, DBP and MBP values in all participants (p = 0.0010; p = 0.0350 and p = 0.0050) as well as with MBP values in high normal and grade 1 hypertension (p = 0.0290). The levels of GSSG correlated positively with SBP, DBP and MBP values in high normal and grade 1 hypertension (p = 0.0290).

Conclusion: Our results point to the fact that microvascular alterations can be identifiable at BP values still considered within normal values and go in parallel with the changes observed in the level of oxidative stress.

4.2. Introduction

The assessment of the microvascular function represents an important part in establishing the pathophysiology but also the risk stratification of cardiovascular disease (CVD).585 Indeed, endothelial dysfunction, one of the main culprits for the development of atherosclerosis, occurs much earlier at the microvascular than at the macrovascular level.^{586,587} Dynamic retinal vessel analysis (DVA) was identified as a useful measure of early changes that signal endothelial dysfunction at the microvascular level. This method can also be used to identify risk for future cardiovascular pathologies in individuals at risk for ^{521,539,540,548,588,589} or already suffering from CVD.^{550,590} This is generally possible due to the fact that the retinal microvascular response to flicker provocation is, in part, dependent on nitric oxide (NO) release⁵⁹¹, and compromised NO homoeostasis is known to be a key factor in endothelial dysfunction at all vascular levels. The assessment of retinal microvessels is used in research but also in clinical practice for diagnosis and follow-up of various CVD, including hypertension.478,592 Nevertheless, the overwhelming majority of protocols that look at retinal microvessels in the course of various CVD, use static imaging to detect abnormalities associated with various degrees of pathology. This is, however, useful only when CVD already established itself and not at earlier, pre-clinical stages. In order to determine the risk at earlier stages to allow preventive measures to be adopted, using the assessment of the retinal microvascular function represents a better alternative to structural imaging because it provides integrated and dynamic data to help establishing possible CVD risk.

In certain conditions that accelerate degradation of NO, such as high oxidative stress, microvascular dilation can be severely impaired.⁵⁹³ In order to counteract such effects, the human body uses various anti-oxidative mechanisms including glutathione. Therefore, any condition associated with low levels of circulating glutathione result in a higher rate of oxidative reactions that contribute towards low NO bioavailability and, consequently, to an impaired microvascular function.^{539,540,594,595} We have already shown that retinal microvascular dilation and constriction responses to stress levels are influenced by systemic antioxidant capacity, and circulating markers for CVD risk in healthy individuals with low to moderate cardiovascular risk.⁵⁸⁹ Moreover, we have suggested that, by providing an integrated and dynamic analysis of vascular function that is, indeed, specific for each individual, retinal vessel reactivity could also be used for profiling a so-called individualized vascular risk for CVD.^{540,589}

Nevertheless, we have not tested this hypothesis in individuals with early stages of CVD. Therefore, the present study aims to study the retinal vascular function and its relationship with systemic anti-oxidative markers in individuals with various levels of early BP abnormalities as defined according to the 2018 European Society of Cardiology/European Society of Hypertension Guidelines.⁵⁹⁶

4.3. Methods

4.3.1. Study Participants

Study participants were recruited through advertisements at the Vascular Research Laboratory and Health Clinics at Aston University (Birmingham, UK). The inclusion / exclusion criteria as defined in section 3.2.1. The main study-specific inclusion criteria for this study were defined as those individuals aged above 30 years, with no current or prior history of cardiovascular, cerebrovascular diseases, peripheral vascular disease, severe dyslipidaemia, diabetes and metabolic disorders. Individuals treated for systemic hypertension as well as those using other vasoactive medications such as dietary supplements containing vitamins or antioxidants and bronchodilators were also excluded from the study.

4.3.2. General Assessment

Standard anthropometric measures of height and weight were recorded to determine body mass index (BMI = weight/height). General clinical assessments for all participants as detailed in sections 3.3.2, 3.3.4 including general health history questionnaires and IOP profiles.

4.3.3. Blood Pressure Assessment and Patients Grouping

Measurements of BP and heart activity were first performed in-clinic as described in section 3.3.3.a. They were then confirmed using a computer-operated ambulatory BP and electrocardiogram (ECG) monitor (Cardiotens-01, Meditech Ltd, Budapest, Hungary) according to the described protocol in section 3.3.3.b. Using the 24-h SBP and DBP values, study participants were then stratified into four subcategories, "optimal " "normal", "high normal" and "Grade I" as recommended by the 2018 European Society of Cardiology/European Society of Hypertension arterial hypertension Guidelines.⁵⁷

4.3.4. Dynamic Retinal Vessel Analysis

Vascular assessments of interest to this study, detailed in the methods section 3.3.5. For each participant, one unselected eye was evaluated by DVA machine. For all of the selected participants, the raw data was re-analysed using the mathematical approach for the analysis of retinal response parameters described in this thesis (section 3.3.5.12). The following parameters, averaged across three flicker cycles and with the arteries and veins regarded separately, were evaluated in this study: baseline diameter, BDF, DA, BCFR, MD%, MC%, tMD, tMC as well as dilation (Slope_D) and constriction slopes (Slope_C) (section 3.3.5.11).

4.3.5. Biomarkers Assays

Bloods samples drawn in the morning of the appointment. Fasting TG, T-CHOL and HDL-C were analysed according to the detailed protocol in section 3.3.9. LDL-C values were calculated as per the Friedewald equation.

4.3.6. Measurement of GSH and oxidized glutathione (GSSG)

Initial processing of blood GSH and GSSG levels were assessed by the GSH recycling assay as detailed previously in section 3.3.10.1.

4.3.7. Sample Size and Statistical Analysis

Based on previous studies, a change of 30% with a SD of 2.5% in retinal vessels reactivity was shown to be significant.^{540,597} As the study design was multifactorial in nature, it was calculated that a sample size of n = 201 in each group was sufficient to provide 95% power at an alpha level of 0.05.

All statistical analyses were performed using the methods explained 3.3.12.

4.4. Results

A total number of 218 participants were initially screened for the study inclusion, of which 17 individuals were excluded based on the quality of retinal vascular image analysis. The remaining 201 participants (93 men and 108 women) were included in the final analysis and classified in one of 4 study groups using the current ESC/ESH Guidelines: group 1 (optimal BP: 56 individuals 22 men and 34women), group 2 (normal BP: 54 individuals, 35 men and 19 women), group 3 (high normal BP: 44 individuals, 28 men and 16 women) and group 4 (grade 1 hypertension: 47 individuals, 28 men and 19 women).

Table 4.1 shows the clinical characteristics of the study population. Although the first group were significantly younger than group 4, there were no statistically significant differences between the study groups with regards to gender, BMI and circulating levels of glucose, t-Chol and HDL-Chol between the study groups (all p > 0.05).

Nevertheless, as expected, there were statistically significant differences between the 4 study groups with regards to SBP, DBP and MBP (all p < 0.0001). In addition, individuals classed as high normal and grade 1 hypertension demonstrated higher levels of circulating TG (p = 0.0021, ANCOVA) and GSSG (p = 0.0111, ANCOVA) and lower levels of GSH (p = 0.0060, ANCOVA). Moreover, individuals classed as grade 1 hypertension demonstrated higher levels of LDL-C (p = 00265, ANCOVA) than the rest of the study groups.

With regards to arterial and venous retinal vascular reactivity parameters as characterized in Tables 4.2 and 4.3, all values reported are based on averaged data across three flicker cycles with the artery and vein regarded separately.

The measured retinal arteriolar parameters, after controlling for all influential covariates identified by multivariate regression analysis, and independent of age, individuals classed as grade 1 hypertension demonstrated significantly higher BDF (p = 0.0012, Figure 4. 1), MD% (p = 0.0007, Figure 4.1), tMC (p = 0.0003, Figure 4.1) and lower Slope_{AC} (p = 0.0131, Figure 4.1) than the rest of the study groups. In addition, individuals classed as high normal and grade 1 hypertension also demonstrated higher tMD than individuals classed as optimal or normal (p < 0.0001, Figure 4.1).

There were no significant differences between participants with regard to measured venous parameters (all p > 0.05, Table 4.3).

4.4.1. Correlations between Vascular and Systemic Circulatory Parameters

Univariate analysis revealed that the whole blood GSH levels correlated significantly and negatively with SBP, DBP and MBP values in all participants (r = -0.25, p = 0.0010; r = -0.17, p = 0.0350 and r = -0.22, p = 0.0050, as well as with MBP values in those classed as high normal and with grade 1 hypertension (r = -0.38, p = 0.0290) but not in those classed as having optimal or normal BP (p > 0.05). In addition, the levels of whole blood GSSG correlated significantly and positively with SBP, DBP and MBP values in all participants (r = 0.16, p = 0.0410; r = 0.17, p = 0.0330 and r = 0.18, p = 0.0220, but not in any groups separately (all p > 0.05). There were no correlations between the levels of GSH and retinal vascular parameters in any of the study groups (all p > 0.05). However, GSSG levels correlated significantly and positively with artery BDF (r = 0.30, p = 0.0370) and negatively with Slope_{AC} only in individuals classed as having an optimal BP level (r = -0.37, p = 0.0200).

Table 4.1. Summary of the Systemic Characteristics of the Study Participants

Variable	Optimal (1) (Mean +SD)	Normal (2) (Mean +SD)	High-Normal (3) (Mean +SD)	Grade 1 (4) (Mean +SD)	<i>p</i> -value	Post – Hoc
Number	56	54	44	47	-	-
Age (years)	40.25(1.80)	42.67(1.87)	46.50(2.06)	47.60(2.03)	0.0260*	-
SBP (mmHg)	109.11(0.79)	124.59(0.82)	131.95(0.90)	145.31(0.90)	0.0000*	1<2<3<4
DBP (mmHg)	67.38(1.07)	74.71(1.11)	81.17(1.22)	87.59(1.22)	0.0000*	1<2<3<4
HR (bpm)	66.69(1.27)	67.61(1.32)	68.48(1.46)	70.24(1.46)	0.3149	-
IOP (mmHg)	14.31(0.82)	15.15(0.85)	16.76(0.93)	16.30(0.93)	0.1870	-
MAP (mmHg)	81.29(1.30)	91.33(1.35)	98.09(1.48)	104.35(1.47)	0.0000*	1<2<3<4
BMI (kg/m ²)	25.32(0.90)	26.88(1.027)	27.89(0.99)	28.21(0.88)	0.0808	-
Glucose	4.77(0.12)	4.80(0.12)	4.90(0.14)	5.01(0.13)	0.4875	-
TG (mmol/L)	1.01(0.06)	1.15(0.06)	1.29(0.07)	1.43(0.07)	0.0021*	1=2<3,4 3=4
CHOL	4.53(0.13)	4.38(0.13)	4.62(0.15)	4.80(0.14)	0.1875	-
HDL-C (mmol/L)	1.21(0.07)	1.13(0.07)	1.19(0.08)	1.16(0.07)	0.6838	-
LDL-C (mmol/L)	2.92(0.15)	3.06(0.16)	3.27(0.18)	3.59(0.17)	0.0265*	1=2=3<4
GSH (umol)	712.37(68.79)	601.47(74.18)	412.52(77.17)	428.63(83.43)	0.0060*	1=2>3,4 3=4
GSSG (umol)	36.15(7.92)	55.53(8.54)	62.49(9.75)	76.67(9.47)	0.0111*	1=2<3,4 3=4

Table 4.1: Abbreviations: SBP: systolic blood pressure; DBP: diastolic blood pressure; HR: heart rate; IOP:intraocular pressure; BMI: body mass index; TG: triglycerides; CHOL: total cholesterol; HDL-C: high-densitylipoprotein cholesterol; LDL-C: low-density lipoprotein; GSSG: oxidized glutathione; GSH: reducedglutathione. *Significant p-values are indicated where p < 0.05 was considered significant.</td>

 Table 4.2. Summary of Retinal Arterial Vascular Function Parameters

Variable	Optimal BP (1) (Mean +SD)	Normal BP (2) (Mean +SD)	High-Normal BP (3) (Mean +SD)	Grade 1 BP (4) (Mean +SD)	p-value Anova/ Ancova	Post- Hoc
Baseline	109.00(2.23)	110.42(2.48)	113.81(2.97)	111.77(4.45)	0.08319	-
BDF	5.36(0.40)	5.75 (0.42)	6.62(0.46)	8.12(0.46)	0.0012*	1=2=3< 4
BCFR	4.62(0.40)	3.85(0.48)	4.08(0.53)	5.74(0.81)	0.2646	-
MD	114.22(2.44)	115.54(2.72)	119.86(3.25)	118.81(4.87)	0.1194	-
tMD	12.76(0.698)	14.77(0.735)	18.04(0.842)	23.10(0.79)	0.0000*	1=2<3< 4
MD %	5.46(0.53)	4.31(0.58)	5.17(0.57)	7.31(0.513)	0.0007*	1=2=3< 4
MC	106.14(2.39)	107.39(2.39)	110.35(3.18)	108.60(4.76)	0.2053	
tMC	23.85(1.07)	25.40(1.109)	26.77(1.22)	30.84(1.22)	0.0003*	1=2=3< 4
MC%	-4.31(0.282)	-3.63(0.293)	-4.042(0.32)	-3.23(0.32)	0.1696	-
DA	9.11(0.66)	9.04(0.735)	9.46(0.88)	11.32(1.35)	0.4951	-
Slope _{AD}	0.69(0.066)	0.57(0.069)	0.37(0.08)	0.51(0.075)	0.1539	-
Slope _{AC}	-0.45(0.07)	-0.44(0.07)	-0.48(0.08)	-0.74(0.08)	0.0131*	1=2=3> 4

Table 4.2: Abbreviations: ANOVA, analysis of variance; ANCOVA, analysis of covariance; Baseline, baseline diameter; BDF, baseline diameter fluctuation; BCFR, Baseline corrected flicker response; tMD, time to reach MD; MD (%), percent dilation; tMC, time to reach MC; MC (%), percent constriction; DA, dilation amplitude (difference between MD and MC during flicker) Slope_{AD}, slope of arterial dilation; Slope_{AC}, slope of arterial constriction. * Significant *p*-values are indicated where *p* < 0.05 was considered significant.

Table 4.3. Summary of Retinal Venous Vascular Function Parameters

Variable	Optimal BP	Normal BP	High-Normal	Grade 1 BP	p-value
	(1)	(2)	BP (3)	(4)	Anova/Ancova
	(Mean +SD)	(Mean +SD)	(Mean +SD)	(Mean +SD)	
Baseline	146.41(3.93)	144.11(4.08)	151.22(4.50)	143.30(4.44)	0.5542
BDF	6.08(0.50)	4.769(0.60)	5.21(0.75)	6.51(0.93)	0.1913
BCFR	4.42(0.41)	4.21(0.43)	4.86(0.47)	3.88(0.47)	0.4718
MD	153.77(4.07)	152.77(4.18)	161.22(4.63)	152.10(4.59)	0.4623
tMD	19.97(0.79)	20.56(0.94)	21.18(1.198)	22.05(1.46)	0.0525
MD%	5.28(0.35)	5.44(0.37)	5.42(0.40)	6.10(0.41)	0.2231
MC	143.04(3.89)	141.74(4.038)	148.13(4.45)	140.56(4.40)	0.6051
tMC	34.30(1.29)	31.10(1.43)	32.90(1.70)	32.50(2.53)	0.4114
MC%	-1.76(0.20)	-1.57(0.22)	-1.48(0.31)	-1.90(0.38)	0.7218
DA	9.79(0.72)	8.61(0.73)	11.22(0.84)	11.16(0.83)	0.1069
Slope _{VD}	0.39(0.03)	0.38(0.04)	0.36(0.05)	0.41(0.056)	0.8740
Slope _{vC}	-0.48(0.10)	-0.39(0.11)	-0.498(0.12)	-0.83(0.12)	0.0663

Table 4.3: Abbreviations ANOVA, analysis of variance; ANCOVA, analysis of covariance; Baseline, baseline diameter; BDF, baseline diameter fluctuation; BCFR, Baseline corrected flicker response; tMD, time to reach MD; MD (%), percent dilation; tMC, time to reach MC; MC (%), percent constriction; DA, dilation amplitude (difference between MD and MC during flicker) Slopev_D, slope of venous dilation; Slopev_C, slope of venous constriction. * Significant *p*-values are indicated where *p* < 0.05 was considered significant.



Figure 4.1: Comparison of Retinal Arterial Response Profile across Groups.

AU, arbitrary units; BDF, baseline diameter fluctuation calculated as the maximum range in vessel diameter during first 30 seconds of baseline readings; MD%, calculated as the percentage change in vessel diameter from baseline to maximum following onset of flicker; tMD, time to reach maximum diameter during flicker; tMC, time to reach maximum constriction post flicker; Slope_{AC},calculated as (MC-MD)/(tMC).

4.5. Discussion

In the present study, we have examined the differences in retinal vascular function between individuals with normal and early stages of abnormal BP according to the current ESC/ESH guidelines. We have also looked at the relationship between systemic BP, retinal vessel reactivity and systemic antioxidant defence capacity in each of the study groups. For the first time, our results showed that individuals graded as having high normal BP values or grade 1 hypertension displayed abnormal dilatory and constrictor responses to flickering light stimulation in retinal arteries but not in the retinal veins when compared to those classed as having optimal or normal BP values.

Hypertension is a polygenic disease; however, endothelial dysfunction and enhanced oxidative stress are among the established modifiable risk factors for this disease.⁵⁹⁸ Alteration of vascular reactivity characterized by augmented contractility and impaired relaxation is a prominent feature in hypertension. Moreover, in essential hypertension, NO availability is reduced earlier, resulting in early vascular endothelial dysfunction.⁵⁹⁹ Indeed, our relatively young patients with high normal BP and grade 1 hypertension, according to the current ESC/EHS guidelines, demonstrated abnormal retinal microvascular dilation and constriction after stimulation with flickering light.

It has already been demonstrated that pre-hypertension is associated with retinal microvascular alterations early in life.⁶⁰⁰ Nevertheless, it seems that these alterations can also be identifiable at BP values previously considered within normal values. Indeed, our study participants classed as having high normal BP values and without additional risks for CVD, have also exhibited signs of microvascular dysfunction at the retinal level. At these BP values, treatments for hypertension are not yet indicated; however, in the light of the above observations, additional measures to prevent further damage may be considered. Moreover, ESC/ESH also recommends treating at lower BP levels in very high-risk patients with a highnormal BP and established CVD.⁵⁷ Nevertheless, the potential impact of pushing for an earlier diagnosis is multiple, including pressures on the health system and economy.⁶⁰¹ Therefore, careful consideration should be given to the actual benefits of treatment in each individual separately, rather than in a given risk category. At present, all the new biomarkers for CVD risk drive the shift towards personalized medicine. Therefore, due to its personalised approach, retinal vessel reactivity can be used for profiling individualized vascular risk by providing an integrated and dynamic analysis of vascular function specific for each individual, therefore, aiding a personalised management of abnormal BP values.

As already mentioned above, increased oxidative stress is an important risk factor for the development of hypertension. Among other defence mechanisms, glutathione represents a major redox buffer.²⁹⁷ It has been already shown that GSH significantly lower, and the GSSG is higher in the red blood cells in patients with hypertension but not in normal individuals.⁶⁰²

Similarly, we have also shown that whole-blood GSH levels were lower and GSSG levels were higher in individuals classed as high normal and grade 1 hypertension. These results show for the first time that such changes may actually occur at a much earlier stage than previously noticed and go in parallel in the changes observed in the retinal microvasculature, confirming that, in the presence of abnormal BP levels, the pathological inhibition of vascular relaxation involves not only NO production by endothelial cells but also a glutathione-dependent bioavailability of NO.³⁰⁰ In addition, we also demonstrated that GSSG correlated with retinal arterial baseline and constriction characterisation parameters only in individuals with optimal BP levels and without additional CVD risks. It is very interesting that, although the levels of antioxidative products were abnormal in individuals with higher BP levels, the link between redox status and retinal vascular reactivity was lost in these groups. This observation links, however, to our previous results in in individuals with low to moderate CVD risk, indicating that in addition to other processes, antioxidant mechanisms support normal NO levels during flicker provocation at the retinal microvascular level.⁵⁸⁹ However, in the present study we could not identify such relationship in individuals with higher levels of BP. As the effects of radical oxygen species (ROS) on vascular tone is dependent upon the concentration and type of species, as well as on the type of vascular beds and various experimental conditions⁶⁰³, it is possible that these effects were not visible in our specific setup. Indeed, it has been proposed that there is a lack of sensitive methods to accurately evaluate the oxidative stress in the human cardiovascular system, especially when it comes to establishing the role redox stress in the small artery vasculopathy of human hypertension.⁶⁰⁴

4.6. Conclusion

Abnormalities of microvascular dysfunction may be an important risk factor for systemic hypertension. In addition, excessive oxidative stress is associated results in cardiovascular diseases by impairing endothelial function and, consequently, microvascular dysfunction. This complex interlink cannot be ignored, and measures to address both dysfunctions should be applied. This is especially important since, despite significant advancement in understanding the pathophysiology of human hypertension and the large number of antihypertensive drugs, strict BP control is still proving insufficient for the prevention of future vascular complications, the so-called "Hypertension Paradox".⁶⁰⁵ Therefore, more research is necessary to develop specific mechanism-based personalised therapies that will address the vascular redox pathobiology.⁶⁰⁴ These therapies will possibly need to be applied at a much earlier stage when abnormalities are only functional and can still be reversed.

5. European Society of Cardiology/European Society of Hypertension versus the American College of Cardiology/American Heart Association guidelines on the cut-off values for early hypertension: a microvascular perspective

This study has been published in Scientific Reports, Nature (Appendix 13.4.2).⁶⁰⁶ 5.1 Abstract

Aims: To investigate retinal and peripheral microvascular function in asymptomatic individuals who fall into different BP groups using either the ESC/ESH or the ACC/AHA guidelines.

Methods: Retinal and peripheral microvascular function was assessed in 358 participants by means of dynamic retinal vessel analysis and digital thermal monitoring, respectively. Blood pressure and lipid panel were also evaluated.

Retinal vascular function measured in all groups belonging to the ACC/ASH classifications were within the normal values for age-matched normal population.

Results: Individuals classed as grade 1 hypertension according to the ESC/ESH guidelines, however, exhibited a significantly decreased artery baseline (p=0.0004) and MC (p=0.040), higher slope_{AD} (p=0.0018) and decreased vein MC (p=0.0446) compared to age-matched normal individuals. In addition, they also had significant lower artery baseline, artery BDF, MD and MC than individuals classed as stage 1 hypertension based on the ACC/ASH guidelines (p=0.00022, p=0.0179, p=0.0409 and p=0.0329 respectively).

Peripheral vascular reactivity (aTR) was lower in ESC /ESH grade I compared to those graded ACC/ASH stage I hypertension (p=0.0122).

Conclusion: Microvascular dysfunctions are present at multiple levels only in individuals with ESC/ESH grade 1 hypertension.

This observation could be important when deciding personalised care in individuals with early hypertensive changes.

5.2. Introduction:

One of the main differences between the latest guidelines published in 2018 by the European Society Of Cardiology/European Society of Hypertension (ESC/ESH) and those published in 2017 by The American College of Cardiology/American Heart Association (ACC/AHA) is the cut-off for what is considered elevated blood pressure (BP) and the first stage of hypertension (HTN) diagnosis.⁶⁰¹ As it can be expected, such difference is bound to have a high impact on clinical diagnosis and management of HTN. By using the ACC/AHA guidelines, the number of patients diagnosed with having this disease increased significantly and so did the pressure on the health system and economy. In addition, such change in practice has also a significant impact on the patients' physical and psychological wellbeing.⁶⁰¹ Nevertheless, it would seem sensible that, in individuals with higher risk for cardiovascular disease (CVD), early diagnosis and interventions are applied at lower BP values. Indeed, both ECC/ESH guidelines and ACC/AHA have recognised this fact and recommended considering treating patients with high CVD risk at a BP threshold lower than their current cut-off for grade/stage 1 HTN.⁵⁷ In this way, positive progress is made towards considering individual over population factors when considering someone of being at risk for CVD.

But what is the real measure of a high CVD risk? Today, the gold standard for absolute CVD risk is based on the Framingham 10-year CVD risk score (FRS).⁶⁰⁷ Other risk scores, such as the Prospective Cardiovascular Münster (PROCAM) and the European Society of Cardiology Systematic Coronary Risk Evaluation (SCORE)^{608,609} are also being used for the same purpose. These calculations are based, however, only on few non-modifiable as well as some modifiable risk factors, thus limiting their ability to accurately predict the risk for CVD in all individuals. Indeed, it has been shown that they could either over- or underestimate actual risk for CVD in a large number cases.^{608,610,611} Consequently, it has been suggested that other factors, such as obesity, psychosocial stress and lifestyle, to name just a few, should also be included for more precise estimations.⁶¹² In addition, vascular endothelial function, a parameter often overlooked, is essential not only for investigations into the pathophysiology of CVD but also for better CVD risk stratification. ^{400,613,614} Assessing vascular endothelial function is usually performed using techniques such as ultrasound flow-mediated dilation (FMD), pulse wave analysis (PWA), plethysmography and iontophoresis.⁶¹⁵ These tests can, however, be complex and time-consuming, and are performed only in highly specialized services, this contributing to the lack of inclusion of this parameter in the more largely circulated risk scores. Nevertheless, parameters such as retinal microvascular function has been found to show a good association not only with various circulatory markers for CVD ^{514,539} , but also with other modifiable and non-modifiable risk factors for this disease such as obesity ⁶¹⁶, family history ⁵¹³ and age.^{588,589} Retinal microvascular function assessments provide instant integrated and dynamic data analysis that is specific to each individual. In addition, it also correlates with peripheral markers for endothelial dysfunction. Indeed, we have recently shown that signs of abnormal vascular function are similarly present and detectable in various microvascular beds, despite existing differences in their anatomical and physiological properties.⁵¹⁵ This observation is important and helps clinician to detect signs of microvascular dysfunction regardless of the method they can access to. By having a choice, practitioners can make better decisions whether to treat or not selected patients with lower BP values but with additional CVD risk, thus making decisions based on individual, rather that population risk factors, an important step towards personalised management of HTN. Nevertheless, in individuals without higher risk, the decision to treat might not be always justified. In order to shed a possible light on the validity of lowering the cut-off for the diagnosis of early hypertension in individuals without additional CVD risk, this study aimed to analyse the retinal and peripheral microvascular function in subjects that fall into different BP groups when using either the ESC/ESH or the ACC/AHA guidelines.^{57,536}

5.3. Methods

5.3.1. Study Participants

Study participants were recruited through advertisements at the Vascular Research Laboratory and Health Clinics at Aston University (Birmingham, UK). The inclusion / exclusion criteria as defined in section 3.2.1. The main study-specific inclusion criteria for this study were defined as those individuals aged above 30 years, with no current or prior history of cardiovascular, cerebrovascular diseases, peripheral vascular disease, severe dyslipidaemia, diabetes and metabolic disorders. Individuals treated for systemic hypertension and those using other vasoactive medications such as dietary supplements containing vitamins or antioxidants and bronchodilators were also excluded from the study.

5.3.2. General Assessment

Standard anthropometric measures of height and weight were recorded to determine body mass index (BMI = weight/height). General clinical assessments were conducted for all the participants as detailed in sections 3.3.2, 3.3.4 including general health history questionnaires and IOP profiles.

5.3.3. Biomarkers Assays:

Bloods samples were drawn on the morning of the appointment. Fasting TG, T-CHOL and HDL-C were analysed according to the detailed protocol in section 3.3.9. LDL-C values were calculated as per the Friedewald equation.

5.3.4. Blood Pressure Assessment and Patients Grouping

Measurements of BP were performed on two separate occasions, one in-clinic and one outof-clinic.^{535,536} During the in-clinic visit, BP was measured in a quiet environment, with the patient seated for 5 min, 3 times at 1-2 min intervals using the same protocol described in section 3.3.3.a. The BP values were confirmed by a second measurement using a computeroperated ambulatory BP and electrocardiogram (ECG) monitor (Cardiotens-01, Meditech Ltd, Budapest, Hungary) as explained in section 3.3.3.b. Using the daytime SBP and DBP values, study participants were stratified into three subcategories: "normal", "high normal", and "Grade I" as recommended by the 2018 ESC/ESH Guidelines. They were then also further classified into three other subcategories: "normal", "elevated", and "stage I", as recommended by the 2017 ACC/AHA guidelines. Subjects classed as "optimal" according to the 2018 ESC/ESH guidelines were excluded, and their values were included in our pool of "normal" data.⁵¹⁴

5.3.5. Dynamic Retinal Vessel Analysis

Vascular assessments of interest to this study, detailed in the methods section 3.3.5. For each participant, one unselected eye was evaluated by DVA. All raw data were re-analysed using the mathematical approach for the analysis of retinal response parameters described in section 3.3.5.12. The following parameters averaged across three flicker cycles and with the arteries and veins regarded separately, were evaluated in this study: baseline diameter, BDF, DA, BCFR, MD%, MC%, tMD, tMC as well as dilation (Slope_D) and constriction slopes (Slope_c) (section 3.3.5.11).

5.3.6. Digital Thermal Monitoring (DTM)

The peripheral microvascular reactivity at the level of the fingertips was assessed using VENDYS 5000 BCE DTM system (Endothelix, Inc, Houston, TX, USA) according to an established protocol as mentioned in section 3.3.6.⁶¹⁷ The post-occlusive adjusted temperature rebound aTR determined by the software algorithm is directly associated with the extent of the subjects vascular reactivity.⁶¹⁸

5.3.7. Statistical analysis

Data were analysed according to methods explained in 3.3.12.

5.4. Results

A total number of 372 participants were initially screened for study inclusion of which 14 individuals were excluded based on the quality of retinal vascular image analysis or to the fact that they did not tolerate the ambulatory BP measurements. The remaining 358 participants were included in the final analysis and classified into 3 study groups using the current

ESC/ESH Guidelines (normal BP:66 individuals; high normal BP: 40 individuals; and grade 1 hypertension: 53 individuals). A further three groups were decided based on the ACC/AHA guidelines: (normal BP: 83 individuals; Elevated BP: 46 individuals; and stage 1 hypertension: 70 individuals).

General characteristics of the study population are presented in Table 5.1. There were no significant differences in age, gender, BMI, HR, IOP, glucose, TG, cholesterol, LDL-c, HDL-c and body composition between all the study groups (all p > 0.05). There was a significant difference between groups in SBP and DBP (p < 0.001). In addition, there were also significant differences between the peripheral vascular reactivity (aTR) between ESC /ESH grade I compared to ACC/ASH stage I hypertension, with individuals graded as ESC /ESH grade I having lower aTR than those graded ACC/ASH stage I hypertension (p=0.0122).

Comparisons of group differences in measured retinal arterial and venous DVA parameters after flicker stimulation are presented in Table 5.2& 5.3. All values reported are based on data averaged across the three flicker cycles with arteries and veins considered separately. Values were further compared to the average normal values for the groups' corresponding age range.

5.4.1. Retinal Vascular Response:

After controlling all the influential covariates identified in multivariate analysis, there were no statistically significant differences in baseline diameter, BDF, BCFR, tMD and tMC, slopeAD and $slope_{AC}$ between the ACC/AHA and ESC/ESH normal BP groups (p > 0.05, Table 5.2). Similarly, there were no significant group differences found between ACC/AHA elevated and ESC/ESH high normal BP groups (p > 0.05, Table 5.2). However, individuals classed as grade 1 hypertension according to the ESC/ESH guidelines had significant lower artery baseline, artery BDF, MD and MC than individuals classed as stage 1 hypertension based on the ACC/ASH guidelines (p=0.0002, p=0.0179, p=0.0409 and p=0.0329, respectively). In addition, while the retinal vascular reactivity parameters measured in individuals classed as stage 1 hypertension according to ACC/ASH guidelines were still within the normal limits for their age, the ESC/ESH grade I hypertensive group exhibited a significantly decreased artery baseline (p=0.0004) and MC (p=0.040) as well as higher slope_{AD} (p=0.0018) compared to age matched population average (Table 5.2, Figure 5.1). Similarly, vein MC was significantly decreased in ESC /ESH grade I compared to ACC/ASH stage I hypertension (p=0.0383). In addition, vein MC was significantly decreased only in the ESC/ESH grade I hypertension compared to age matched population average (p=0.0446) (Table 5.3, Figure 5.1).

No significant group differences in any of the other measured arterial DVA parameters were identified (all p > 0.05).

Table 5.1: General Characteristics of the Study Population

	ACC/AHA	ESC/ESH	р-	ACC/AHA	ESC/ESH	p-	ACC/AHA	ESC/ESH	р-
	<u>Normal</u>	<u>Normal</u>	value	Elevated	<u>High</u>	value	<u>Stage I</u>	<u>Grade I</u>	value
	<120&<80	SBP:120-		SBP: 120-	Normal		SBP:130-	SBP:140-	
		129		129	SBP:130-		139	159	
		DBP:80-84		DBP: <80	139		DBP:80-89	DBP:90-99	
					DBP:85-89				
Number	83	66	-	46	40	-	70	53	-
Gender	30M:53F	34M:32F	-	29M:17F	22:18F	-	44M:37F	32M:21F	-
Age (years)	40.1.8 (1.35)	43.91 (1.70)	0.0859	42.83 (2.31)	45.80 (2.53)	0.3897	44.88(3.4)	48.77(2.7)	0.1063
SBP (mmHg)	109.43 7.08)	122.52(4.98)	0.0000*	124.17(6.88)	131.03(6.74)	0.0002*	134.61(8.72)	142.00(7.52)	0.0000*
DBP (mmHg)	68.33 (6.15)	76.15 (5.92)	0.0000*	72.76 (4.98)	83.93 (5.37)	0.0000*	83.95(4.71)	89.47(6.54)	0.0000*
HR (bpm)	68.055(9.28)	65.06(10.15)	0.0659	63.31 (9.38)	67.37(10.85)	0.1085	68.37(10.87)	69.59(10.93)	0.5467
IOP	13.30 (2.34)	13.41 (2.64)	0.8139	13.25 (2.59)	13.43 (2.25)	0.7123	14.33(2.63)	15.29(3.01)	0.0970
BMI (kg/m2)	24.69 (0.36)	25.66 (0.45)	0.0934	26.73 (0.71)	25.70 (0.78)	0.3175	26.49(4.20)	28.09(4.40)	0.0680
Body fat %	28.00(1.68)	23.76(1.76)	0.0864	23.02(1.90)	28.31(2.06)	0.0670	27.65(2.43)	31.93(3.97)	0.3651
Body water %	52.51(1.65)	51.56(1.73)	0.6907	54.25(2.30)	48.84(2.48)	0.1180	46.80(2.96)	49.44(4.84)	0.6445
Muscle mass %	47.45(2.59)	50.89(2.72)	0.1673	60.94(3.13)	55.68(3.38)	0.2599	52.75(4.29)	43.30(6.10)	0.2583
Glucose(mmol/L)	4.87 (0.55)	4.93 (0.86)	0.6178	5.07 (0.82)	5.03 (0.60)	0.8433	4.93(0.92)	5.15(1.31)	0.3356
TG (mmol/L)	1.139(0.56)	1.19 (0.61)	0.6345	1.15 (0.48)	1.25 (0.46)	0.4397	1.263(0.62)	1.44(0.80)	0.2210
T.Chol (mmol/L)	4.40 (0.78)	4.70 (1.02)	0.5642	4.88 (1.07)	4.55 (0.91)	0.2040	4.50(0.89)	4.72(0.86)	0.2516
HDL-C (mmol/L)	1.33 (0.41)	1.37 (0.39)	0.5212	1.35 (0.38)	1.23 (0.58)	0.3288	1.33(0.56)	1.21(0.44)	0.2848
LDL-C (mmol/L)	2.57 (0.91)	2.78 (0.12)	0.1639	2.10 (1.09)	2.82 (1.05)	0.5199	2.59(0.94)	2.90(0.93)	0.1131
TG/HDL	1.00 (0.81)	1.01 (0.81)	0.9561	0.98 (0.63)	1.35 (0.91)	0.0683	1.21(0.92)	1.36(0.94)	0.4460
aTR	1.60(0.13)	1.93(0.11)	0.0365*	1.70(0.11)	1.63(0.11)	0.7790	1.80(0.12)	1.20(0.20)	0.0122*

Table 5.1: Abbreviations: SBP: systolic blood pressure; DBP: diastolic blood pressure; HR: heart rate; IOP: intraocular pressure; BMI: body mass index; TG: triglycerides; T. Chol: total cholesterol; HDL-C: high-density lipoprotein cholesterol; LDL-C: low-density lipoprotein; aTR: adjusted temperature rebound *Significant p-values are indicated where P<0.05 was

Table 5.2. Summary of Retinal Arterial Vascular Function Parameters

AverageNormalNormalNormalNormalAAverage<	lormal	ACC/AHA	ESC/ESH	ANOVA/ANCOA	ACC/AHA	ESC/ESH	ANOVA/ANOV	ACC/AHA	ESC/ESH	ANOVA/ANCOV
DBP:80-84 DBP:30-84 DBP:30-84 DBP:35-89 DBP:35-89 Artery baseline 99.38 (0.12) 99.38 (0.77) 0.1979 99.33 (0.95) 99.97 (0.92) 0.43 99.99(0.002) 5.64 (2.650) 6.14 (3.46) 0.3461 6.28 (2.87) 5.83 (2.74) 0.52 Artery-BDF 5.64 (2.650) 6.14 (3.46) 0.3461 6.28 (2.87) 5.83 (2.74) 0.52 Artery-BDF 5.64 (2.650) 6.14 (3.46) 0.3461 6.28 (2.87) 5.83 (2.74) 0.52 Artery-MD 103.57 (2.20) 104.07 (2.24) 0.2000 104.44 (2.2) 104.19 (2.74) 0.57 Artery-MD 21.88 (9.63) 20.87 (6.27) 0.5047 20.34 (5.83) 20.27 (6.44) 0.77 22.14(1009) 21.88 (9.63) 20.87 (6.27) 0.1400 4.64 (2.63) 4.48 (2.75) 0.79 23.67 (2.21) 3.59 (2.20) 4.20 (2.53) 0.1400 4.64 (2.63) 4.48 (2.75) 0.79 Artery-MD% 3.59 (2.20) 97.53 (1.82) 97.53 (2.15) 0.76 0.75	Werage	<u>Normal</u> <120&<80	<u>Normal</u> SBP:120-129	<i>p</i> -value	Elevated SBP:120-129	High Normal SBP:130-139	A <i>p</i> -value	<u>Stage I</u> SBP:130-139	Grade I SBP:140-159	A <i>p</i> -value
Artery baseline99.98 (0.12)99.88 (0.77)0.197999.83(0.95)99.97(0.92)0.4399.99(0.002) $5.64 (2.650)$ $6.14 (3.46)$ 0.3461 $6.28(2.87)$ $5.83(2.74)$ 0.57 $5.66(2.63)$ $5.64 (2.650)$ $6.14 (3.46)$ 0.3461 $6.28(2.87)$ $5.83(2.74)$ 0.57 $5.66(2.63)$ $103.57(2.20)$ $104.07(2.24)$ 0.2000 $104.44(2.2)$ $104.19(2.74)$ 0.57 $Artery-MD$ $103.57(2.20)$ $104.07(2.24)$ 0.2000 $104.44(2.2)$ $104.19(2.74)$ 0.71 $Artery-MD$ $21.88 (9.63)$ $20.87 (6.27)$ 0.5047 $20.84(5.83)$ $20.27(6.44)$ 0.71 $Artery-MD$ $21.88 (9.63)$ $20.87 (6.27)$ 0.5047 $20.84(5.83)$ $20.27(6.44)$ 0.71 $Artery-MD$ $3.59 (2.20)$ $4.20 (2.53)$ 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 $Artery-MD$ $3.59 (2.20)$ $4.20 (2.53)$ 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 $3.67(2.21)$ $97.53 (0.19)$ $97.64 (0.25)$ 0.7455 $20.27(6.44)$ 0.74 $3.67(2.21)$ $97.73 (0.19)$ $97.64 (0.25)$ 0.7455 $22.09(6.60)$ $28.05(6.44)$ 0.54 $Artery-MC$ $27.47 (7.36)$ $27.88 (6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.95 $Artery-MC$ $27.47 (7.36)$ $27.88 (6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.97 $Artery-MC$ $27.47 (7.36)$ $27.78 (6.98)$ 0.7455 $28.09(6.60)$ <t< th=""><th></th><th></th><th>DBP:80-84</th><th></th><th>DBP: <80</th><th>DBP:85-89</th><th></th><th>DBP:80-89</th><th>DBP:90-99</th><th>-</th></t<>			DBP:80-84		DBP: <80	DBP:85-89		DBP:80-89	DBP:90-99	-
Artery-BDF 5.64 (2.650) 6.14 (3.46) 0.3461 $6.28(2.87)$ $5.83(2.74)$ 0.52 $5.66(2.63)$ $103.57(2.20)$ $104.07(2.24)$ 0.2000 $104.44(2.2)$ $104.19(2.74)$ 0.57 Artery-MD $103.57(2.20)$ $104.07(2.24)$ 0.2000 $104.44(2.2)$ $104.19(2.74)$ 0.57 Artery-MD 21.88 (9.63) 20.87 (6.27) 0.5047 $20.84(5.83)$ $20.27(6.44)$ 0.71 Artery-MD % 3.59 (2.20) 4.20 (2.53) 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 Artery-MD % 3.59 (2.20) 4.20 (2.53) 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 Artery-MD % 3.59 (2.20) 4.20 (2.53) 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 Artery-MC 97.73 (0.19) 97.64 (0.25) 0.7691 $97.83(1.82)$ $97.53(2.15)$ 0.54 Artery-MC 27.47 (7.36) 27.88 (6.98) 0.7455 28.09 (6.60) 28.05 (6.44) 0.98 Artery-MC % -2.22 (1.84) -2.056 (1.87) 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 Artery-MC % -2.22 (1.84) 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 Artery-MC % -2.22 (1.87) 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 Artery-MC % -2.22 (1.87) 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.36 Artery-DA 5.77 (2.87) 0.26 (2.97) 0.5191 $0.43(2.74)$	Artery baseline 9.99(0.002)	99.98 (0.12)	99.88 (0.77)	0.1979	99.83(0.95)	99.97(0.92)	0.4346	99.99(0.62)	97.93(0.27)	0.0002*
Artery-MD $103.57(2.20)$ $104.07(2.24)$ 0.2000 $104.44(2.2)$ $104.19(2.74)$ 0.571 $103.67(2.20)$ $21.88(9.63)$ $20.87(6.27)$ 0.5047 $20.84(5.83)$ $20.27(6.44)$ 0.71 $Artery-tMD$ $21.88(9.63)$ $20.87(6.27)$ 0.5047 $20.84(5.83)$ $20.27(6.44)$ 0.71 $22.14(10.09)$ $3.59(2.20)$ $4.20(2.53)$ 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 $Artery-MD$ $3.59(2.20)$ $4.20(2.53)$ 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 $3.67(2.21)$ $97.73(0.19)$ $97.64(0.25)$ 0.7691 $97.83(1.82)$ $97.53(2.15)$ 0.54 $3.67(2.21)$ $97.73(0.19)$ $97.64(0.25)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.98 $97.51(1.89)$ $27.47(7.36)$ $27.88(6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.38 $Artery-MC$ $27.47(7.36)$ $27.88(6.187)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.38 $Artery-MC$ $-2.22(1.84)$ $-2.056(1.87)$ 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 $Artery-MC$ $-2.22(1.84)$ $-2.056(1.87)$ 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 $Artery-MC$ $5.77(2.87)$ $6.17(2.96)$ 0.5191 $6.61(2.95)$ $7.30(3.63)$ 0.43 $6.16(2.95)$ $0.30(2.57)$ $0.26(2.97)$ 0.9244 $0.43(2.74)$ $0.44(3.02)$ $0.44(3.02)$ $Artery-BCFR0.30(2.57)0.26(2.97)0.92440.43$	\rtery-BDF .66(2.63)	5.64 (2.650)	6.14 (3.46)	0.3461	6.28(2.87)	5.83(2.74)	0.5250	5.98(3.42)	5.02(2.41)	0.0179*
\mathbf{v} constrained v	httery-MD	103.57(2.20)	104.07(2.24)	0.2000	104.44(2.2)	104.19(2.74)	0.5780	104.14(2.43)	102.75(4.26)	0.0409*
Artery-MD% $3.59(2.20)$ $4.20(2.53)$ 0.1400 $4.64(2.63)$ $4.48(2.75)$ 0.79 $3.67(2.21)$ $97.73(0.19)$ $97.64(0.25)$ 0.7691 $97.83(1.82)$ $97.53(2.15)$ 0.54 $\mathbf{Artery-MC}$ $97.73(0.19)$ $97.64(0.25)$ 0.7691 $97.83(1.82)$ $97.53(2.15)$ 0.54 $\mathbf{Artery-MC}$ $27.47(7.36)$ $27.88(6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.98 $\mathbf{Artery-MC}$ $27.47(7.36)$ $27.88(6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.98 $\mathbf{Artery-MC}$ $27.22(1.84)$ $-2.056(1.87)$ 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 $\mathbf{Artery-MC}$ $-2.22(1.84)$ $-2.056(1.87)$ 0.6124 $-1.99(1.95)$ $7.30(3.63)$ 0.43 $\mathbf{Artery-MC}$ $5.77(2.87)$ $6.17(2.96)$ 0.5191 $6.61(2.95)$ $7.30(3.63)$ 0.43 $\mathbf{Artery-BCFR}$ $0.30(2.57)$ $0.26(2.97)$ 0.9244 $0.43(2.74)$ $0.44(3.02)$ 0.16	vrtery-tMD 2.14(10.09)	21.88 (9.63)	20.87 (6.27)	0.5047	20.84(5.83)	20.27(6.44)	0.7120	19.86(6.74)	20.10(7.11)	0.9493
Artery-MC $97.53(0.19)$ $97.64(0.25)$ 0.7691 $97.83(1.82)$ $97.53(2.15)$ 0.54 $97.51(1.89)$ $27.47(7.36)$ $27.88(6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.98 $27.05(8.31)$ $27.47(7.36)$ $27.88(6.98)$ 0.7455 $28.09(6.60)$ $28.05(6.44)$ 0.98 $27.05(8.31)$ $27.22(1.84)$ $-2.056(1.87)$ 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 $Artery-MC%$ $-2.22(1.84)$ $-2.056(1.87)$ 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 $Artery-MC%$ $5.77(2.87)$ $6.17(2.96)$ 0.5191 $6.61(2.95)$ $7.30(3.63)$ 0.43 $Artery-BCFR$ $0.30(2.57)$ $0.26(2.97)$ 0.9244 $0.43(2.74)$ $0.44(3.02)$ 0.16 $0.56(2.73)$ $0.56(2.73)$ $0.612(2.91)$ $0.43(2.74)$ $0.44(3.02)$ 0.16	Nrtery-MD% .67(2.21)	3.59 (2.20)	4.20 (2.53)	0.1400	4.64(2.63)	4.48(2.75)	0.7901	4.15(2.44)	4.01(2.55)	0.1020
Artery -tMC 27.47 (7.36) 27.88 (6.98) 0.7455 28.09 (6.60) 28.05 (6.44) 0.98 27.05 (8.31) 27.05 (8.31) 28.05 (6.44) 0.98 27.05 (8.31) -2.22 (1.84) -2.056 (1.87) 0.6124 -1.99 (1.95) -2.14 (2.14) 0.38 Artery-MC% -2.22 (1.84) -2.056 (1.87) 0.6124 -1.99 (1.95) -2.14 (2.14) 0.38 Artery-MC% 5.77 (2.87) 6.17 (2.96) 0.5191 6.61 (2.95) 7.30 (3.63) 0.43 Artery-DA 5.77 (2.87) 0.26 (2.97) 0.9244 0.43 (2.74) 0.44 (3.02) 0.16 Artery-BCFR 0.30 (2.57) 0.26 (2.97) 0.9244 0.43 (2.74) 0.44 (3.02) 0.16	\rtery-MC 7.51(1.89)	97.73 (0.19)	97.64 (0.25)	0.7691	97.83(1.82)	97.53(2.15)	0.5410	97.58(1.80)	96.08(4.97)	0.0329*
Artery-MC%-2.22 (1.84)-2.056 (1.87) 0.6124 $-1.99(1.95)$ $-2.14(2.14)$ 0.38 $-2.49(1.89)$ $-2.77 (2.87)$ $6.17 (2.96)$ 0.5191 $6.61(2.95)$ $7.30(3.63)$ 0.43 Artery-DA $5.77 (2.87)$ $6.17 (2.96)$ 0.5191 $6.61(2.95)$ $7.30(3.63)$ 0.43 Artery-BCFR $0.30 (2.57)$ $0.26 (2.97)$ 0.9244 $0.43(2.74)$ $0.44(3.02)$ 0.16 $0.56(2.73)$ $0.56(2.73)$ $0.56(2.73)$ $0.44(3.02)$ 0.16	rttery -tMC 7.05(8.31)	27.47 (7.36)	27.88 (6.98)	0.7455	28.09(6.60)	28.05(6.44)	0.9810	27.49(6.70)	28.85(7.96)	0.3641
Artery-DA 5.77 (2.87) 6.17 (2.96) 0.5191 6.61(2.95) 7.30(3.63) 0.43 6.16(2.95) 6.17 (2.96) 0.5191 6.61(2.95) 7.30(3.63) 0.43 Artery-BCFR 0.30 (2.57) 0.26 (2.97) 0.9244 0.43(2.74) 0.44(3.02) 0.16 0.56(2.73) 0.56(2.73) 0.526 (2.97) 0.9244 0.43(2.74) 0.44(3.02) 0.16	\rtery-MC% 2.49(1.89)	-2.22 (1.84)	-2.056 (1.87)	0.6124	-1.99(1.95)	-2.14(2.14)	0.3810	-2.41(1.80)	-1.94(1.72)	0.2002
Artery-BCFR 0.30 (2.57) 0.26 (2.97) 0.9244 0.43 (2.74) 0.44 (3.02) 0.16 0.56(2.73) 0.56(2.73) 0.44 (3.02) 0.16 0.16	Ittery-DA .16(2.95)	5.77 (2.87)	6.17 (2.96)	0.5191	6.61(2.95)	7.30(3.63)	0.4310	6.57(3.02)	6.71(2.91)	0.8096
	Nttery-BCFR 56(2.73)	0.30 (2.57)	0.26 (2.97)	0.9244	0.43(2.74)	0.44(3.02)	0.1681	0.67(3.07)	0.89(3.12)	0.7326
Artery-Slope _{AD} 0.21 (0.16) 0.25 (0.15) 0.1347 0.27 (0.16) 0.26 (0.32) 0.19 0.21 (0.13) 0.21 (0.13) 0.21 (0.16) 0.26 (0.32) 0.19	Artery-Slope _{AD} 1.21(0.13)	0.21 (0.16)	0.25 (0.15)	0.1347	0.27(0.16)	0.26(0.32)	0.1902	0.28(0.20)	0.35(0.30)	0.2335
Artery-Slope _{AC} -0.36 (0.45) -0.30 (0.22) 0.3600 -0.31(0.22) -0.32(0.18) 0.96 -0.36(0.30) -0.36(0.30) -0.31(0.22) -0.32(0.18) 0.96	Nrtery-Slope _{AC} 0.36(0.30)	-0.36 (0.45)	-0.30 (0.22)	0.3600	-0.31(0.22)	-0.32(0.18)	0.9603	-0.29(0.20)	-0.27(0.51)	0.2826

flicker response; tMD, time to reach MD; MD(%), percent dilation; tMC, time to reach MC; MC (%), percent constriction; DA, dilation amplitude (difference between MD and MC Table 5.2: Abbreviations: ANOVA, analysis of variance; ANCOVA, analysis of covariance; Baseline, baseline diameter; BDF, baseline diameter fluctuation; BCFR, Baseline corrected during flicker) SlopeAD, slope of arterial dilation; SlopeAC, slope of arterial constriction. * Significant p-values are indicated where p < 0.05 was considered significant.

Table 5.3. Summary of Retinal Venous Vascular Function Parameters

Normal	ACC Normal	ESC Normal	ANOVA/ANCOVA	ACC	ESC High	ANOVA/ANCOVA	ACC Stage I	ESC Grade I	ANOVA/ANCOVA
Average	<120&<80	SBP:120-129	<i>p</i> -value	Elevated	Normal	<i>p</i> -value	SBP:130-139	SBP:140-159	<i>p</i> -value
		DBP:80-84		SBP:120-129	SBP:130-139		DBP:80-89	DBP:90-99	
				DBP: <80	DBP:85-89				
Vein-Baseline	99.96(0.20)	99.90 (0.78)	0.3996	99.83(0.97)	99.99(0.002)	0.3733	99.99(0.0025)	99.91(0.0022)	0.4251
99.98(0.15)									
Vein-BDF	4.427(2.22)	3.94(2.02)	0.2059	4.09(2.22)	5.149(3.48)	0.1583	4.70(2.74)	4.92(2.48)	0.7667
4.39(2.04)									
Vein-MD	104.19(2.16)	104.06(2.24)	0.7239	104.01(2.19)	105.31(2.83)	0.5790	105.10(2.73)	104.61(2.245)	0.6224
104.34(2.26)									
Vein-tMD	22.37(6.90)	21.53 (7.40)	0.5058	22.68(8.62)	21.25(6.46)	0.5343	19.82(5.45)	20.43(5.81)	0.4213
22.55(6.93)									
Vein-MD%	4.24(2.15)	4.19 (2.61)	0.9114	4.21(2.75)	5.310(2.93)	0.1326	5.10(2.73)	4.612(2.24)	0.5127
4.36(2.26)									
Vein MC	98.87(1.14)	99.20 (1.12)	0.0957	98.36(1.02)	98.70(1.51)	0.2408	98.46(1.123)	97.13(12.5)	0.0383*
98.71(1.27)									
Vein- tMC	31.23(5.62)	30.15 (6.38)	0.2990	31.26(6.61)	30.54(6.34)	0.6757	30.37(5.76)	31.0384(5.20)	0.6642
31.68(5.51)									
Vein-MC%	-1.09(1.18)	-0.70 (1.55)	0.0744	-0.85(0.70)	-1.29(0.81)	0.1973	-1.17(1.25)	-1.06925902	0.5289
-1.27(1.28)									
Vein-DA	5.325(2.69)	4.85 (2.81)	0.3302	4.86(1.69)	6.60(3.20)	0.0869	6.27(3.16)	5.6811(2.60)	0.4637
5.63(2.88)									
Vein-BCFR	0.94(2.37)	0.98 (2.86)	0.9383	0.66(2.79)	1.56(2.84)	0.2202	1.52(2.49)	0.76220(1.97)	0.2591
1.23(2.19)									
Vein -Slope _{AD}	0.23(0.14)	0.24 (0.16)	0.5595	0.24(0.175)	0.50(1.109)	0.1881	0.39(0.74)	0.2697(0.13)	0.4007
0.23(0.13)									
Vein -Slope _{vc}	-0.23(0.14)	-0.20 (0.13)	0.2551	-0.20(0.13)	-0.27(0.16)	0.1062	-0.25(0.17)	-0.2318(0.16)	0.7306
-0.24(0.15)									

Table 5.3 Abbreviations ANOVA, analysis of variance; ANCOVA, analysis of covariance; Baseline, baseline diameter; BDF, baseline diameter fluctuation; BCFR, Baseline corrected flicker response; tMD, time to reach MD; MD%, percent dilation; tMC, time to reach MC; MC%, percent constriction; DA, dilation amplitude (difference between MD and MC during flicker) Slope_{AD}, slope of venous dilation; Slope_{VC}, slope of venous constriction. * Significant p-values are indicated where p < 0.05 was considered significant.





Figure 5.1: Comparison of Retinal Arterial and Venous Response Profile across Groups.

AU, arbitrary units; BDF, baseline diameter fluctuation calculated as the maximum range in vessel diameter during first 30 seconds of baseline readings; MD%, calculated as the percentage change in vessel diameter from baseline to maximum following onset of flicker; tMD, time to reach maximum diameter during flicker; tMC, time to reach maximum constriction post flicker; slope_{AC} & Slope_{VC}, calculated as (MC-MD)/(tMC).

5.5. Discussion:

The present study looked at the peripheral and retinal microvascular function in various BP groups divided according to either the ESC/ESH or the ACC/AHA guidelines. Our report shows, for the first time, that independent of all systemic influences, retinal microvascular dysfunction firstly manifest itself only in individuals classed as ESC/ESH grade 1 hypertension. In contrast, subjects classed as ACC/AHA stage 1 hypertension exhibited normal retinal microvascular behaviour after provocation using flickering light. In addition, ESC/ESH grade 1 hypertension individuals also exhibited lower peripheral vascular reactivity than those graded as ACC/AHA stage 1 hypertension.

It is well known that microvascular dysfunction is a parameter highly predictive of cardiovascular events. ⁴⁷¹ Due to this quality, its early identification is extremely important, especially in individuals at risk for CVD, including those with borderline BP values or classed as grade/stage 1 hypertension. Indeed, similar to what is known for so long in the literature about the effects of systemic hypertension on microvasculature ⁶¹⁹ in the present study we have shown that individuals classed as ESC/ESH grade 1 hypertension demonstrated abnormal peripheral and retinal microvascular function. Indeed, at the retinal vascular level, both vasodilation and vasoconstriction of the retinal arteries and veins were affected. Nevertheless, as all patients with essential hypertension are known to demonstrate microvascular endothelial dysfunction from early stages ⁶²⁰, it is puzzling how individuals classed as ACC/AHA stage 1 hypertension failed to show such abnormalities. This is an important observation that requires further careful consideration since changes in diagnostic cut-off for systemic hypertension towards lower BP values in individuals without additional risk factors has a very high impact not only on the clinical practice and economy but also on each individual well-being.⁶⁰¹

Beside its direct effect on the vascular functionality, hypertension is also known to cause an acceleration of the ageing process of the vascular function.⁵⁹⁹ We have previously demonstrated that, in ageing, normal individuals there is a decline in retinal vasoregulative capacity in both dilatory and constrictory phases that was independent of any systemic influences ⁵⁸⁸ and we have linked this abnormal response to the possibility of an age-related increase in oxidative stress.⁵¹⁵ Nevertheless, the subjects included in the current paper were much younger that those included in our ageing studies. It is possible that, the abnormal microvascular function observed in individuals classed as ESC/ESH grade 1 hypertension was indeed due to a reduced nitric oxide (NO) availability through excess inhibition by reactive oxygen species (ROS) ⁶²¹; however, in this case, the main culprit for ROS excess was not ageing but, possibly, an increased BP. Indeed, it has been shown that, in hypertensive

patients, NO availability is reduced early in the course of the disease .⁵⁹⁹ Although we did not measure either the level of NO or antioxidative markers in our current cohort of patients, in the light of our previous discovery showing that retinal microvascular dysfunction occurs in parallel with changes observed in the level of oxidative stress both in normal ⁵¹⁵ and individuals with abnormal BP levels ⁶¹⁶, it is logic to presume that similar abnormalities occurred in our current cohort of individuals with ESC/ESH grade 1 hypertension, thus contributing to the abnormal retinal microvascular function measured in these individuals.

Our ESC/ESH grade 1 hypertension also exhibited an abnormal retinal venous response to flicker. In previous studies, we have already documented that individuals with other categories of cardiovascular risk also exhibit abnormal retinal venous responses to flicker-induced provocation.^{513,514,622,623} Structural retinal venular dilation has previously been implicated as a strong prediction of adverse CVD events.^{624,625} Nevertheless, the abnormal functional retinal venous responses, as measured by DVA, are still to be confirmed as signs of endothelial dysfunction in their own right or as epiphenomena of retinal arterial dysfunction.^{514,588} In this particular context, our observed retinal venular dysfunctionality could reflect a compensatory adaptation following sustained arterial dilation during flicker. Further investigation is required to understand the relevance of this kind of response; however, it could be hypothesized that a change in venous calibre associated with either structural or endothelial irregularities could also be used as a marker for cardiovascular risk in individuals with early hypertension. ^{513,514,588}

In addition to a general retinal microvascular dysfunction, our cohort of individuals with ESC/ESH grade 1 hypertension also exhibited abnormal peripheral microvascular reactivity as assessed by DTM, showing that abnormal microvascular function is a general process in these individuals and can be detected at multiple levels despite differences in vascular physiology and methodologies.

5.6. Conclusion

This study found microvascular dysfunctions to be present at multiple levels only in individuals with ESC/ESH grade 1 hypertension and not in those classed as stage 1 hypertension according to the latest ACC/AHA guidelines. The implications of these observations could be important, especially when clinicians decide their intervention based only on controversial, borderline BP values measured in individuals without other CVD risk.

6. Dry Eye Disease is Associated with Retinal Microvascular Dysfunction and Possible Risk for Cardiovascular Disease

This study was published in Acta Ophthalmologica, Wiley (Appendix 13.4.3).⁶²⁶

6.1. Abstract

Purpose: To explore the presence of microvascular endothelial dysfunction as a measure for early cardiovascular disease in individuals diagnosed with Dry Eye Disease (DED) as compared to age-matched normal controls.

Methods: Blood pressure profiles, Body Mass Index, intraocular pressure, blood analyses for glucose and lipid metabolism markers (TG, CHOL, HDL-c, LDL-c), as well as retinal and peripheral microvascular function were assessed in twenty-five 35-50 years old with DED as diagnosed by the TFOS DEWS II criteria and twenty-five age and sex-matched controls.

Results: After controlling all the influential covariates, individuals diagnosed with DED exhibited significant lower retinal artery baseline (p=0.027), artery maximum diameter (p=0.027), minimum constriction (p=0.039) and dilation amplitude (p=0.029) than individuals in the non-dry eye group. In addition, the time to reach the vein maximum diameter was significantly longer in the DED patients than in normal controls (p=0.0052). Only in individuals diagnosed with DED, artery maximum constriction correlated statistically significantly and positively with HDL-C blood levels (p=0.006). Similarly, artery slope_{AD} correlated positively with T-CHOL and LDL-C (p=0.006 & 0.011 respectively). Additionally, artery baseline diameter and maximum constriction were significantly and negatively correlated to T-CHOL/HDL-C ratio (p=0.032 and p=0.013 respectively) in DED individuals only.

Conclusions: Individuals with a positive diagnosis of Dry Eye Disease exhibit a higher risk for cardiovascular disease than age- and sex-matched normal individuals. Closer monitoring and prompt personalised intervention will possibly contribute to reduced cardiovascular disease morbidity in these individuals.

6.2. Introduction

Dry eye disease (DED) represents a multifactorial, chronic and debilitating pathology of the ocular surface characterized by loss of homeostasis of the tear film and accompanied by ocular symptoms. Tear film instability and hyperosmolarity, ocular surface inflammation and damage, as well as neurosensory abnormalities, play etiological roles in this disease.³²⁵ In addition to other risk factors, DED has previously been associated with dyslipidaemia, a group of metabolic abnormalities characterized by any or a combination of the following: raised low-density lipoprotein cholesterol (LDL-C), raised total cholesterol (TC), raised triglycerides (TG) and low high-density lipoprotein cholesterol (HDL-C).⁶²⁷ Indeed, as lipid homeostasis is important for the stability of the tear film, the association between dyslipidaemia and DED is entirely justified. Moreover, disruptions of cholesterol biosynthesis are also associated with sebaceous/ Meibomian gland (MG) dysfunctions ⁶²⁸, another cause of tear film instability and dry eye. Dyslipidaemia also represents a significant risk factor for cardiovascular disease (CVD), especially due to its contribution to the pathogenesis of atherosclerosis in mediumsized and large arteries but also at the microvascular level.^{629,630} This affects not only the anatomy of these vessels but, most importantly, their function. Indeed, at the functional level, it impairs endothelium-dependent vasodilation because of defects in nitric oxide (NO) bioavailability.⁶³⁰ This has catastrophic effects on the balance between the physiological vascular dilatory and constrictory states, which, in turn, will also affect other important circulatory functions and, most importantly, vascular protection against oxidation, inflammation and thrombosis.⁶³¹ This imbalance characterises the so-called endothelial dysfunction (ED), an initial reversible step in the development of atherogenesis; nevertheless, it is also one of the most important stages in the development of CVD. Therefore, its identification, as early as possible, represents a key factor in CVD prevention.⁶³¹

Besides having common risk factors, other relationships between DED and CVD are not clearly understood. However, as both CVD and DED disease are common and important health problems encountered frequently in the general population, a closer look at their other possible links is warranted. The present study explores the presence of microvascular ED (as a measure for early CVD) in individuals diagnosed with DED as compared to age-matched normal controls.

6.3. Subjects and Methods

6.3.1. Study Participants

Healthy individuals aged between 35 and 50-year-old were recruited for this case-control study through advertisements at the Vascular Research Laboratory and the Dry Eye Clinic, Aston University (Birmingham, UK). Ethical approval was sought from the relevant local ethics committees, and written informed consent was received from all participants prior to study

enrolment. The study was designed and conducted in accordance with the tenets of the Declaration of Helsinki, and all study-related procedures adhered to institutional guidelines. The inclusion/exclusion criteria as defined in section 3.2.1.

6.3.2. General Assessment

Standard anthropometric measures of height and weight were recorded to determine body mass index (BMI = weight/height). General clinical assessments were conducted for all the participants as detailed in sections 3.3.2, 3.3.3.a 3.3.4 included general health history questionnaires, BP and IOP profiles.

6.3.3. Biomarkers Assays

Bloods samples were drawn on the morning of the appointment. Fasting TG, T-CHOL and HDL-C were analysed according to the detailed protocol in section 3.3.9. LDL-C values were calculated as per the Friedewald equation.

6.3.4. DED Diagnosis

Following the protocols stated in section 3.3.7. All subjects underwent dry eye assessment using a digital slit lamp and the Keratograph K5m (Oculus, Wetzlar, Germany), including objective non-invasive breakup time, taking the average of three readings and ocular surface staining with fluorescein (i-DEW Flo, Mainline, Derby, UK) and lissamine green (GreenGlo, HUB Pharmaceuticals, Plymouth, MI, USA) using the TFOS DEWS II recommended methodology. ³³⁸ Osmolarity was assessed in each eye using the Tearlab (Dallas-Fort Worth, TX, USA).

Diagnosis of dry eye was based on the latest TFOS DEWS II criteria ³³⁸, which involves positive symptoms screening with the Ocular Surface Disease Index (cut-off >+13) and one or more of non-invasive tear breakup time (<10s), hyperosmolarity (>+308mOsm/L in the higher eye or an intereye difference .8mIsm/L) and ocular surface staining (>5 corneal, >9 conjunctival punctate spots or lid margin (>=2mm length and >+25% width).

6.3.5. Dynamic Retinal Vessel Analysis

Vascular assessments of interest to this study, detailed in the methods section 3.3.5. For each participant, one unselected eye was evaluated by DVA. All raw data were re-analysed using the mathematical approach for the analysis of retinal response parameters described in section 3.3.5.12. The following parameters averaged across three flicker cycles and with the arteries and veins regarded separately, were evaluated in this study: baseline diameter, BDF, DA, BCFR, MD%, MC%, tMD, tMC as well as dilation (Slope_D) and constriction slopes (Slope_C) (section 3.3.5.11).

6.3.6. Digital Thermal Monitoring (DTM)

The peripheral microvascular reactivity at the level of the fingertips was assessed using VENDYS 5000 BCE DTM system (Endothelix, Inc, Houston, TX, USA) according to an established protocol as mentioned in section 3.3.6.⁶¹⁷

6.3.7. Sample size

Based on previous studies, a change of 30% with a SD of 2.5% in retinal vessel reactivity was shown to be significant. As the study design was multi-factorial in nature it was calculated that n=25 in each group was sufficient to provide 90% power with an alpha of 0.05.

6.3.8. Statical Analysis

All data were analysed according to methods explained in 3.3.12.

6.4. Results

A total number of 66 participants were initially screened for study inclusion, of which 16 individuals were excluded based on the quality of retinal vascular image analysis. The remaining 50 participants were included in the final analysis and classified into 2 study groups based on a positive diagnosis for DED: 25 individuals (12 men and 13 women) and negative diagnosis: 25 individuals (15 men and 10 women).

General characteristics of the study population are presented in Table 6.1. There were no significant differences in age, sex, BMI, HR, IOP, glucose, TG, and LDL-C, between the study groups (all p > 0.05). Although still within the normal range, the individuals with DED had a statistically significant higher T-CHOL (5.0 vs 4.44 mmol/L) and lower HDL-C (1.21 vs 1.58 mmol/L) values than those without DED (p = 0.0356 and 0.0137 respectively), albeit close to the upper limit for these parameters (Table 6.1).

There were no significant differences between the study groups with regard to the peripheral microvascular function parameters as measured using the DTM method (all p>0.05, Table 6.1). In addition, after controlling all the influential covariates identified using multivariate analysis, there were no significant differences between the study groups with regards to the retinal microvascular parameters BDF, BCFR, MD%, MC%, tMD and tMC, Slope_{AD} and Slope_{AC} (all p > 0.05, Table 6.2). However, individuals diagnosed with DED exhibited significant lower artery baseline diameter, artery MD, MC and DA than individuals in the non-dry eye group (p=0.0269, p=0.0273, p=0.0386 and p=0.0291, respectively, Figure 6.1). In addition, vein tMD was significantly longer in the DED patients than in normal controls (p=0.0052) (Table 6.3).

Only in individuals diagnosed with DED, artery MC correlated statistically significantly and positively with HDL-C blood levels (p= 0.006). Similarly, artery slope_{AD} correlated positively 144
with T-CHOL and LDL-C (p= 0.006 & 0.011 respectively). Additionally, artery baseline diameter and MC were significantly and negatively correlated to T-CHOL/HDL-C ratio (p=0.032 and p=0.013 respectively) in DED individuals only (Figure 6.2).

	Dry Eye diagnosis	Non-Dry Eye	P-value
Number	25	25	-
Sex	12 Male:13 Female	15 Male:10 Female	-
Age (years)	44.1(2.5)	37.6(2.5)	0.073
BMI (Kg/m²) ^b	26.03(1.03)	26.71(1.03)	0.645
SBP(mmHg)	116.90(4.96)	120.5(4.96)	0.611
DBP (mmHg)	68.94(3.43)	71.3(3.43)	0.629
HR (bpm)	62.14(2.10)	66.00(2.98)	0.366
IOP	13.48(0.42)	13.31(0.44)	0.776
GLUC (mmol/L)	4.81(0.15)	4.54(0.16)	0.228
TG (mmol/L)	1.01(0.07)	0.94(0.07)	0.502
T-CHOL (mmol/L)	5.0(0.172)	4.44(0.19)	0.036*
HDL-C (mmol/L)	1.21(0.109)	1.58(0.09)	0.014*
LDL-C (mmol/L)	3.15(0.23)	2.81(0.23)	0.301
T-CHOL/HDL-C	4.36(0.377)	3.086(2.34)	0.020*
aTR	1.52(0.14)	1.76(0.14)	0.237

Table 6.1: General Characteristics of the Study Population

Table 6.1: BMI, body mass index; SBP, systolic blood pressure; DBP, diastolic blood pressure; HR, heart rate (in beats per minute); IOP, intraocular pressure; GLUC, glucose; TG, triglycerides; T-CHOL, total cholesterol; HDL-C, high-density lipoprotein cholesterol; LDL-C, low-density lipoprotein cholesterol; aTR, adjusted temperature rebound.

*Significant p-values are indicated where p < 0.05 was considered significant.

^a Data are presented as mean (SD) unless otherwise indicated.

^b Calculated as weight in kilograms divided by height in metres squared.

	Mean (SD)					
Parameter	Dry Eye	None-Dry Eye	P-value			
Artery baseline	110.59(2.52)	112.72(2.51)	0.027*			
Artery-BDF	5.47(0.47)	5.434(0.47)	0.954			
Artery-DA ^a	8.29(0.77)	10.73(0.77)	0.029*			
Artery-BCFR ^b	3.84(0.51)	4.33(0.51)	0.502			
Artery-MD	118.75(2.64)	123.28(2.85)	0.027*			
Artery-tMD	16.73(0.92)	15.67(0.92)	0.414			
Artery-MD%	4.95(0.54)	5.56(0.53)	0.434			
Artery-MC	109.50(2.38)	111.67(2.38)	0.039*			
Artery-tMC	23.95(1.59)	25.71(1.58)	0.436			
Artery-MC%	-3.05(0.38)	-3.63(0.38)	0.287			
Artery-Slope _{AD} ^c	0.41(0.04)	0.39(0.043)	0.764			
Artery-Slope _{AC} ^d	-0.52(0.08)	-0.41(0.085)	0.361			

Table 6.2: Summary of Retinal Arterial Vascular Functional Parameters

Table 6.2. Baseline, baseline diameter; BDF, baseline diameter fluctuation; DA, dilation amplitude; BCFR, baseline-corrected flicker response; MD, artery maximum dilation; tMD, reaction time to maximum dilation diameter; MD%, percentage change in diameter from baseline to maximum dilation; MC, Maximum constriction; tMC, reaction time to maximum constriction diameter from maximum dilation diameter; MC%, percentage constriction below baseline; Slope_{AD}, slope of arterial dilation; Slope_{AC}, slope of arterial constriction.

Unless otherwise indicated, all values are expressed in arbitrary units, which approximately correspond to micrometres (μ m) in a normal Gullstrand eye.

*Significant p-values are indicated where p < 0.05 was considered significant.

^a Calculated as MD – MC.

^b Calculated as DA – BDF ⁵²³

^c Calculated as (MD – baseline)/tMD ⁵⁴¹ ^d Calculated as (MC – MD)/tMC ⁵⁴¹

	Mean (SD)					
Parameter	Dry Eye	None- Dry Eye	P-Value			
Vein-Baseline	133.62(3.92)	139.94(3.92)	0.260			
Vein-BDF	5.37(0.43)	5.23(0.43)	0.822			
Vein-DA ^a	9.11(0.61)	8.96(0.62)	0.867			
Vein-BCFR [♭]	3.73(0.43)	3.72(0.43)	0.988			
Vein-MD	140.04(4.58)	149.92(4.58)	0.134			
Vein-tMD	19.46(0.70)	16.40(0.70)	0.005*			
Vein-MD%	4.94(0.38)	4.50(0.38)	0.415			
Vein MC	130.94(4.35)	140.96(4.35)	0.110			
Vein- tMC	33.49(1.61)	33.29(1.61)	0.930			
Vein-MC%	-2.0(0.19)	-1.76(0.19)	0.371			
Vein -Slope _{AD} ^c	0.49(0.07)	0.48(0.07)	0.908			
Vein -Slope _{VC} ^d	-0.52(0.08)	-0.41(0.086)	0.361			

Table 6.3. Summary of Retinal Venous Vascular Function Parameters

Table 6.3. Baseline, baseline diameter; BDF,baseline diameter fluctuation; DA, dilation amplitude; BCFR, baseline-corrected flicker response; MD, vein maximum dilation; tMD, reaction time to maximum dilation diameter; MD%,percentage change in diameter from baseline to maximum dilation; MC, Maximum constriction; tMC, reaction time to maximum constriction diameter from maximum dilation diameter; MC%, percentage constriction below baseline; Slope_{VD}, slope of venous dilation; Slope_{VC}, slope of venous constriction.

Unless otherwise indicated, all values are expressed in arbitrary units, which approximately correspond to micrometres (μ m) in a normal Gullstrand eye.

*Significant p-values are indicated where p < 0.05 was considered significant.

^a Calculated as MD – MC.

^bCalculated as DA – BDF ⁵²³

^cCalculated as (MD – baseline)/tMD ⁵⁴¹

^dCalculated as (MC – MD)/tMC ⁵⁴¹



Figure 6.1: Comparison of Retinal Arterial Response Profile across Groups.

Abbreviations: AU, arbitrary units; BDF, baseline diameter fluctuation calculated as the maximum range in vessel diameter during first 30 seconds of baseline readings; MD%, calculated as the percentage change in vessel diameter from baseline to maximum following onset of flicker; tMD, time to reach maximum diameter during flicker; tMC, time to reach maximum constriction post flicker; slope_{AC}, calculated as (MC-MD)/ (tMC).



Figure 6.2: Correlation between Retinal Arterial Responses and Systemic Blood Lipids Abbreviations: Artery- MC, Arterial maximum constriction; Artery- Slope_{AD}, slope of arterial dilation; Artery-BL, artery baseline diameter; Artery- MC; Arterial maximum constriction; HDL-C, high density lipoprotein cholesterol; T-CHOL, total cholesterol; LDL-C, low density lipoprotein cholesterol; T-CHOL/HDL, total cholesterol/ high density lipoprotein ratio. Note: normalized unit values used on y-axes.

6.5. Discussion

This study examined the link between DED and CVD risk, as assessed using known circulatory markers as well as measurements of microvascular function at the retinal and peripheral level. Through this approach, it identified that individuals diagnosed with DED exhibit abnormal retinal, but not peripheral microvascular function and these abnormalities correlate with plasma levels of circulating cholesterols. To our knowledge, this is the first study to reveal that middle-aged individuals with DED have increased retinal microvascular dysfunction compared to normal, sex and age-matched controls.

Similar to other research ^{632–634}, this study also demonstrated that individuals with DED exhibit higher, albeit at the upper normal levels of circulating T-CHOL when compared to those without DED. It has been previously proposed that the relationship between hypercholesterolemia and DED can be explained as increased levels of cholesterol in the meibomian lipid would increase its melting point, thus leading to increased viscosity and plugging of the meibomian orifice.⁶³⁵ It is also important to note that individuals with DED included in our study also exhibited statistically significant lower, albeit still normal levels of HDL-C and higher T-CHOL/HDL-C ratio than the normal controls group. It is well known that HDL-C transports cholesterol from the tissues to the liver to be disposed, making it beneficial in the prevention of CVD.

Moreover, decreases in HDL-C have also been linked with endothelial dysfunction and a reduction in the bioavailability of NO.⁶³⁶ In addition, T-CHOL/HDL-C ratio has previously been specified as a better indicator of premature CVD risk in than T-CHOL levels ⁶³⁷, making these observations very important.

The present study revealed a novel finding that, independent of other characteristic, individuals with DED exhibit abnormal retinal arterial and venous microvascular dysfunction. To date, little research looked at microvessels in individuals with DED and only at the level of conjunctival vessels. It is interesting to mention that patients with DED have been previously shown to exhibit abnormal microvascular response and reduced blood flow at the conjunctival vessels level after trigeminal stimulation, suggesting that these patients suffer of an imbalance in the autonomic nervous system (ANS).⁶³⁸ Indeed, conjunctival vessels have a dual autonomic innervation.⁶³⁹

Moreover, DED has been proposed to be associated with various ANS dysfunctions as autonomic nerves are abundant in meibomian gland tissue and play an important role in regulating the secretory activities of meibomian gland in animals.^{640,641} Endothelial dysfunction and ANS imbalance often co-exist in the development of various cardiovascular disease processes.⁶⁴² At the retinal microvascular level, in the absence of autonomic innervation, 150

metabolic and myogenic stimuli are more involved in retinal autoregulation of the microvascular calibre.⁶⁴ Although the function of retinal microvessels are not under the influence of ANS; this study suggests that both the ANS and endothelial dysregulation co-exist in individuals with DED, and the results of this imbalance are evident and can be measurable at different vascular levels. In addition, abnormalities in the retinal venous functionality have also been found in the DED group. Since retinal veins typically incite a more passive regulatory contribution to increases in blood flow whether this observation may reflect some kind of reconciliation of alterations in arterial outflow to the venous side via downstream autoregulatory mechanisms ⁵³⁸ is unclear at present.

The positive correlation between the retinal microvascular function parameters and the levels of circulating HDL-C in DED individuals reinforces the fact that the HDL-C has a vital role in the prevention of CVD. Indeed, retinal vascular calibres have been shown to be independently associated with risk factor variables such as age, blood pressure, HDL-C, and LDL-C. Our DED group, however, also exhibited a negative relationship between T- CHOL/HDL-C ratio and retinal artery MC. As this circulatory parameter is a strong indicator of risk for CVD ^{643,644}, this observation is very important and, in addition to the above mentioned microcirculatory abnormalities, points to the fact that DED individuals could exhibit higher risk for CVD than age- and sex-matched normal individuals. Further follow-up studies to confirm the actual development of CVD in these individuals are warranted.

6.6. Conclusion

Significant attenuations in retinal vascular function exist and can be detected in persons diagnosed with DED. Moreover, these abnormalities correlate with known circulatory markers for CVD. Functional retinal assessments could therefore be useful for early vascular screening, possibly contributing to a reduced risk for CVD morbidity in these individuals.

7. Novel Composite Early Risk Markers for Vascular Ageing and Risk for Cardiovascular Disease

Abstract of this study has been presented in the European Network for Oxysterol Research Conference and the study was published in Antioxidants (Appendix 13.4.4).⁶⁴⁵

7.1. Abstract:

Abnormal redox balance is one of the main determinants of ageing, abnormal vascular function and risk for cardiovascular diseases. The aim of the present paper is to assess the relationship between various oxidative stress markers and microvascular function in individuals of various age groups, free of overt disease.

Forty-two apparently healthy individuals were included in the study (group 1: 19–30 years, group 2: 31-50 years, and group 3: 51-70 years). Parameters of microvascular function (dilation and constriction) have been assessed at the retinal microvascular level using the dynamic retinal vessel analyzer (DVA). Fasting plasma has been obtained from all subjects, and quantification of monohydroxy and dihydroxy oxysterols was performed by LC-MS/MS following reverse phase chromatography. Griess assay was used to evaluate the Nitric Oxide concentration. Glutathione redox ratio was also analyzed by means of a whole blood glutathione recycling assay. In all participants, the levels of 7-Ketocholesterol, 25hydroxycholesterol and 7 β -hydroxycholesterol correlated significantly and positively with the time to maximum arteriolar dilation. In addition, 25- hydroxycholesterol and 7βhydroxycholesterol negatively correlated to the percentage of maximum arteriolar dilation. A negative correlation was observed for 27- hydroxycholesterol and 7β- hydroxycholesterol with microvascular arteriolar constriction. In addition, the glutathione redox ratio decreased significantly with age and correlated significantly and negatively with 7β -hydroxycholesterol and 25- hydroxycholesterol. These results suggest that oxysterol profile correlates with changes in microvascular beds function with age, and it's possible to be used as a biomarker of vascular ageing and CVD risk.

7.2. Introduction:

While we all age at the same pace chronologically, it was observed that, physiologically, individuals do not age at a similar rate.¹⁶⁸ This observation has led to the concept of biological ageing, which is a measure of premature progressive decline in organs' function.⁶⁴⁶ At the vascular level, as blood vessels constantly withstand against mechanical forces, any changes to the blood components, including cells, macromolecules, ions or external pathogenic insults, could affect vessels' integrity and their performance.⁶⁴⁷ As a result, modifiable cardiovascular risk factors such as obesity, smoking, lifestyle, as well as non-modifiable risk factors such as genetics, ethnicity and age, could affect vascular function.⁶⁴⁸

In the past few years, functional evaluation of the microvessels gained a lot of clinical interest since endothelial dysfunction, the main culprit for cardiovascular diseases (CVDs), was found to occur earlier at the microvessel level than the macrovessel level.⁵⁸⁶ Accordingly, many methods were developed to evaluate functional microvascular responses to biological stimuli in different vascular beds.⁶⁴⁹ One of these methods, the dynamic retinal vessel analysis (DVA), was highlighted as one of the most sensitive and reliable methods to measure early vascular changes that point out early endothelial dysfunction and future risk of cardiovascular pathologies in individuals with and without overt clinical symptoms.^{515,548,606} It has also been shown that retinal microvascular function is modified by age.⁵³¹ However, although this is a consistently demonstrable effect, little is known about what factors in particular directly determine this outcome? We have previously demonstrated that high levels of circulating reactive oxygen species (ROS) have a direct effect on the retinal microvascular function.^{514,589} Indeed, evidences suggest that oxidative stress induces cellular senescence is a key influencer of age-induced mechanical and structural vascular dysfunction 650-653 as well as of the development and progression of cardiovascular diseases (CVD). 621,654-657 Additionally, high levels of ROS also result in low levels of Nitric Oxide (NO), a well-known known critical molecule in the regulation of the vascular function and tone, therefore further contributing to the above-mentioned abnormalities.⁶⁵⁷

High levels of cholesterol are also an important factor affecting microvascular function at all levels, including the retinal microvascular function.^{539,589,626} Cholesterol can be oxidized by non-enzymatic and enzymatic pathways.^{658,659} The non-enzymatic pathway can be free-radical mediated and results in the formation of 7-hydroxycholesterol, which is further oxidized to oxysterols (7-Ketocholesterol, 7 α -hydroxycholesterol and 7 β -hydroxycholesterol).³⁰⁹ The enzymatic oxidation by cholesterol-hydroxylase enzymes produces 4 β -hydroxycholesterol, 7 α -hydroxycholesterol and 27 hydroxycholesterol) and high levels of oxysterols in plasma were highlighted as significant contributors to macrovessels atherogenesis and endothelial dysfunction.^{660–663} Moreover, oxysterols have also been implicated in the pathogenesis of vascular ageing because of their marked pro-oxidant, 153

H.A. Shokr, PhD Thesis, Aston University 2021

proinflammatory and proapoptotic properties.⁶⁶⁴ There is an active cholesterol metabolism at the retinal level, where LDL uptake is the main source of lipids in the retina ^{665,666}, and preclinical and clinical studies have suggested a connection between oxysterols and retinal degeneration.⁶⁶⁷ Similarly, it can be proposed that their various effects can also be observed at the retinal microvascular level. Nevertheless, this effect has never been previously determined. Therefore, this study aims to investigate the correlation between oxysterols, glutathione redox ratio and retinal microvascular functional reactivity in apparently healthy individuals of various ages.

7.3. Materials and Methods

Healthy individuals aged between 35 and 50-year-old were recruited for this case-control study through advertisements at the Vascular Research Laboratory, Aston University (Birmingham, UK). Ethical approval was sought from the relevant local ethics committees, and written informed consent was received from all participants prior to study enrolment. The study was designed and conducted in accordance with the tenets of the Declaration of Helsinki, and all study-related procedures adhered to institutional guidelines.

The inclusion/exclusion criteria as defined in section 3.2.1.

7.3.1. Sample Size

Based on previous studies, a change of 30% with a SD of 2.5% in retinal vessels reactivity was shown to be significant.^{540,597} As the study design was multifactorial in nature, it was calculated that a total sample size of n = 42 was sufficient to provide 95% power at an alpha level of 0.05.

7.3.2. General Investigations

Standard anthropometric measures of height and weight were recorded to determine body mass index (BMI = weight/height). General clinical assessments were conducted for all participants as detailed in sections 3.3.2, 3.3.3.a 3.3.4 included general health history questionnaires, BP and IOP profiles.

7.3.3. Dynamic Retinal Vessel Analysis

Vascular assessments of interest to this study, detailed in the methods section 3.3.5. For each participant, one unselected eye was evaluated by DVA. All raw data were re-analysed using the mathematical approach for the analysis of retinal response parameters described in section 3.3.5.12. The following parameters averaged across three flicker cycles and with the arteries and veins regarded separately, were evaluated in this study: baseline diameter, BDF, DA, BCFR, MD%, MC%, tMD, tMC as well as dilation (Slope_D) and constriction slopes (Slope_c) (section 3.3.5.11).

7.3.4. Biomarkers Assays:

Bloods samples were drawn on the morning of the appointment. Fasting TG, T-CHOL and HDL-C were analysed according to the detailed protocol in section 3.3.9. LDL-C values were calculated as per the Friedewald equation.

7.3.5. Measurement of GSH and Oxidized Glutathione (GSSG)

Initial processing of blood GSH and GSSG levels were assessed by the GSH recycling assay as detailed previously in section 3.3.10.1.

7.3.6. Measurement of Nitric Oxide (NO)

Nitric oxide levels were measured according to the protocol explained in section 3.3.10.2.

7.3.7. Extraction and Liquid Chromatography-Tandem Mass Spectrometry (LC-MS/MS) of Plasma Oxysterols

Plasma oxidized lipids were extracted and measured according to the protocol previously detailed in section in section 3.3.10.4

7.3.8. Statistical Analysis

All data were analysed according to methods explained in 3.3.12.

7.4. Results:

7.4.1. Clinical characteristics

A total number of 56 participants were initially screened for study inclusion, of which 14 individuals were excluded based on the quality of retinal vascular image analysis. The remaining 42 participants were included in the final analysis and classified into 3 age groups. Group 1: 19–30 years (10 male: 6 female); Group 2: 31–50 years (8 male: 8 female); and Group 3: 51–70 years (4 male: 6 female). The general characteristics of the study population are presented in Table 7.1. There were no significant differences in sex, BMI, HR, IOP, glucose, TG, and HDL-C, between the study groups (all p > 0.05). Although still within the normal range, older participants had higher T-CHOL and LDL-C values than younger ones, albeit close to the upper limit for these parameters (p = 0.0071 and 0.0112 respectively) (Table 7.1).

7.4.2. Retinal microvascular function

After controlling all the influential covariates identified using multivariate analysis, there were no significant differences between the study groups with regard to arterial baseline, BCFR, MD, MC, tMC, Slope_{AD} and all venous microvascular parameters (all p > 0.05, Table 7.2&7.3).

However, the post hoc analysis revealed a significantly decreased BDF, MD%, MC%, DA and Slope_{AC} with age (p=0.0140, p=0.0360, p=0.0040, p=0.0230 and p=0.0055 respectively, Table 7.2, Figure 7.2). Moreover, artery tMD increased significantly with age (p=0.0254, Table 7.2). No statistically significant differences were found between retinal venous vascular function parameters among the study groups (Table 7.3).

Variable	19-30	31-50	51-70	p-value	Post -hoc
Number	16	16	10	-	-
	10M:6F	8M:8F	4M:6F		
Age (years)	24.54 (1.74)	37.81 (1.57)	60.75 (2.22)	0.0001*	1<2<3
SBP (mmHg)	115.27(3.80)	114.7 (3.95)	118.5 (5.10)	0.82946	-
DBP (mmHg)	64.64 (2.54)	70.9 (2.67)	72.5 (3.44)	0.13002	-
HR (bpm)	63.17 (3.29)	68.1 (2.55)	66.18(2.43)	0.50522	-
IOP (mmHg)	12.93 (0.57)	13.23 (0.60)	14.83 (0.80)	0.14723	-
BMI (kg/m²)	23.34 (1.13)	26.02 (1.19)	26.59 (1.53)	0.16039	-
Glucose (mmol/L)	4.41(0.21)	4.99 (0.28)	4.68 (0.28)	0.26740	-
TG (mmol/L)	0.83 (0.065)	1.0 (0.081)	0.96 (0.088)	0.22376	-
CHOL (mmol/L)	3.95 (0.17)	4.14 (0.21)	4.93 (0.24)	0.0071*	1<2<3
HDL-C (mmol/L)	1.33 (0.08)	1.44 (0.10)	1.25 (0.12)	0.50590	-
LDL-C (mmol/L)	2.28 (0.17)	2.60(0.21)	3.01 (0.24)	0.01116*	1<2<3

Table 7.1: Summary of the Systemic Characteristics of the Study Participants

Table 7.1: Abbreviations: SBP: systolic blood pressure; DBP: diastolic blood pressure; HR: heart rate; IOP: intraocular pressure; BMI: body mass index; TG: triglycerides; CHOL: total cholesterol; HDL-C: high-density lipoprotein cholesterol; LDL-C: low-density lipoprotein.

*Significant p-values are indicated where p < 0.05 was considered significant.

					—
Variable	19-30	31-50	51-70	P-value Anova/Ancova	Post-Hoc
Baseline	117.44 (4.24)	116.60 (4.45)	113.41(5.74)	0.61902	-
BDF	7.06 (0.65)	5.92 (0.70)	3.49 (0.88)	0.0140*	1>2>3
BCFR	4.90 (0.72)	4.81 (0.80)	2.54 (0.90)	0.1058	-
MD	124.94 (4.13)	121.76 (4.60)	117.02 (5.20)	0.4986	-
tMD	14.67 (0.77)	16.15 (0.73)	19.29 (0.83)	0.0254*	1<2<3
MD %	5.64 (0.63)	4.50 (0.70)	3.24 (0.79)	0.0360*	1>2>3
MC	112.99 (4.32)	113.24 (4.78)	112.80 (5.42)	0.5985	-
tMC	25.39 (2.34)	28.93 (2.60)	27.523 (2.93)	0.5974	-
MC%	-3.85 (0.36)	-2.43 (0.4)	-1.70 (0.46)	0.0040*	1>2>3
DA	11.95 (1.27)	9.74 (1.40)	6.04 (1.60)	0.0230*	1>2>3
Slope _{AD}	0.52 (0.11)	0.51 (0.12)	0.26 (0.14)	0.3030	-
Slope _{AC}	-0.62 (0.062)	-0.38 (0.07)	-0.30 (0.08)	0.0055*	1>2, 2=3, 1>3

Table 7.2. Summary of Retinal Arterial Vascular Function Parameters

Table 7.2: Abbreviations: ANOVA, analysis of variance; ANCOVA, analysis of covariance; Baseline, baseline diameter; BDF, baseline diameter fluctuation; BCFR, Baseline corrected flicker response; tMD, time to reach M.D.; M.D. (%), percent dilation; tMC, time to reach MC; MC (%), percent constriction; DA, dilation amplitude (difference between M.D. and MC during flicker) Slope_{AD}, slope of arterial dilation; Slope_{AC}, slope of arterial constriction. * Significant *p*-values are indicated where *p* < 0.05 was considered significant.



Figure 7.1: Comparison of Retinal Arterial Response Profile across Groups.

Abbreviations: AU, arbitrary units; BDF, baseline diameter fluctuation calculated as the maximum range in vessel diameter during first 30 seconds of baseline readings; MD%, calculated as the percentage change in vessel diameter from baseline to maximum following onset of flicker; tMD, time to reach maximum diameter during flicker; tMC, time to reach maximum constriction post flicker; slope_{AC}, calculated as (MC-MD)/ (tMC).

Variable	19-30	31-50	51-70	P-value Anova/Ancova	Post-Hoc
Baseline	142.47(4.25)	143.55 (4.70)	131.39 (5.33)	0.19389	-
BDF	5.45 (0.75)	5.02 (0.84)	5.61 (0.95)	0.88379	-
BCFR	3.00 (0.85)	4.87 (0.95)	3.14 (1.07)	0.30859	-
MD	149.09 (4.44)	150.26 (4.91)	138.19 (5.60)	0.22687	-
tMD	23.76 (1.47)	19.37 (1.63)	21.95 (1.85)	0.15744	-
MD %	4.66 (0.66)	4.75 (0.74)	5.15 (0.83)	0.89666	-
MC	140.64 (4.24)	140.37 (4.69)	129.44 (5.32)	0.22113	-
tMC	32.12 (2.16)	34.41 (2.40)	33.71 (2.71)	0.76678	-
MC%	-1.30 (0.45)	-2.20 (0.49)	-1.50 (0.56)	0.38622	-
DA	8.45 (1.36)	9.90 (1.50)	8.75 (1.70)	0.76728	-
Slope _{AD}	0.33 (0.046)	0.36 (0.051)	0.35 (0.06)	0.86531	-
Slope _{AC}	-0.60 (0.12)	-0.31 (0.13)	-0.39 (0.14)	0.22394	-

Table 7.3: Abbreviations ANOVA, analysis of variance; ANCOVA, analysis of covariance; Baseline, baseline diameter; BDF, baseline diameter fluctuation; BCFR, Baseline corrected flicker response; tMD, time to reach MD; MD (%), percent dilation; tMC, time to reach MC; MC (%), percent constriction; DA, dilation amplitude (difference between MD and MC during flicker) SlopevD, slope of venous dilation; SlopevC, slope of venous constriction. * Significant *p*-values are indicated where *p* < 0.05 was considered significant.

7.4.3. Oxidative Stress markers

After controlling all the influential covariates identified using multivariate analysis, post hoc analysis showed a significant increase in 7-Ketocholesterol (7-KC), 25 Hydroxycholesterol (25-OHC), 27 Hydroxycholesterol (27-OHC), 7 β Hydroxycholesterol (7 β -OHC) and 7,25 Hydroxycholesterol (7,27-OHC) with age (p=0.0051, p=0.0067, p=0.0192, p=0.0168, p= 0.0113 respectively).

The GSH/GSSG ratio decreased significantly with age among the study's populations (p=0.0279). No other significant differences were recorded (all p>0.05, table 4).

No statistically significant differences were found in the NO concentrations between the study groups (all p >0.05)

4. Correlations between vascular and systemic circulatory parameters:

Univariate analysis revealed that blood 7-KC, 25-OHC and 7 β -OHC levels correlated significantly and positively with tMD in all the study participants (r=0.4318, p=0.025; r=0.0301, p=.742; r=0.4029, p=0.037 respectively).

Additionally, 25-OHC and 7 β -OHC are negatively correlated to arterial MD% (r= -0.3879, p=0.046; r=-0.5324, p=0.004 respectively). Similarly, 27-OHC and 7 β -OHC significantly correlated negatively with artery Slope_{AC} (r= -0.2291, p=0.0250; r= -0.4576, p=0.016 respectively).

There were no correlations between the GSH redox ratio and retinal vascular parameters in any of the study groups (all p>0.05). However, GSH/GSSG levels correlated significantly and negatively with 7 β -OHC and 25-OHC (r=-0.5129, p= 0.021; r= -0.4799, p= 0.032 respectively).

Table 7.4. Summary of Normalized Oxidative Stress and Inflammatory Systemic	2
Circulatory markers	

Variable	19-30	31-50	51-70	P-value Anova/Ancova	Post- Hoc
7-KC (nmol/mmol)	10.49 (1.71)	13.28 (1.17)	17.26 (1.17)	0.0051*	1<2<3
25-OHC (nmol/mmol)	16.42 (3.86)	20.86 (2.65)	30.56 (2.65)	0.0067*	1<2<3
27-OHC (nmol/mmol)	16.61 (3.05)	20.55 (2.09)	26.79 (2.09)	0.0192*	1<2<3
4β-OHC (nmol/mmol)	1.02 (0.35)	0.81 (0.24)	1.14 (0.24)	0.6054	-
7β-OHC (nmol/mmol)	6.65 (1.70)	9.86 (1.17)	12.74 (1.17)	0.0168*	1<2<3
7,27-OHC (nmol/mmol)	8.45 (2.30)	8.64 (1.57)	10.65 (1.57)	0.5986	-
7,25-OHC (nmol/mmol)	5.60 (2.60)	8.89 (1.78)	11.41 (1.80)	0.0113*	1<2<3
NO (uM)	46.45 (4.37)	43.02 (3.39)	39.73 (3.10)	0.4495	-
GSH/GSSG	27.25(7.95)	20.84 (7.95)	17.05 (8.60)	0.0279*	1<2<3

Table 7.4:Abbreviations: ANOVA, analysis of variance; ANCOVA, analysis of covariance; 7-KC, 7-
Ketocholesterol; 25-OHC, 25-Hydroxycholesterol; 27-OHC, 27-Hydroxycholesterol; 4β-OHC, 4β-
Hydroxycholesterol; 7β-OHC, 7β-Hydroxycholesterol; 7,27-OHC, 7,27-Hydroxycholesterol; 7,25-OHC, 7,25-
Hydroxycholesterol; IL-6, interleukin-6; NO, nitric oxide; GSH, glutathione; GSSG, Glutathione disulphide;
GSH/GSSG, redox ratio.

7.5. Discussion:

In the present study, we have examined the relationship between circulatory markers of oxidative stress and endothelial function with dynamic retinal microvascular changes in various age groups. Our results show, for the first time that, with age oxysterol levels change in parallel with microvascular function in individuals without overt CVD.

In accordance with the existing literature, our results demonstrated a systematic increase in plasma circulatory lipids ⁶⁶⁸ and oxidative stress markers ^{669,670} with age. In addition and similar to previous research, circulating oxysterols concentrations were also increased with age.^{671–673} Another consistency with previously published work was the fact that our healthy and yet older individuals (above 50 years old) displayed abnormal dilatory and constrictor responses to flickering light retinal arteries stimulation in comparison to younger and middle-aged participants.^{674,675}

Indeed, it is well known that healthy and young endothelium usually retains antithrombotic, vasodilatory, anti-inflammatory and antioxidant properties, which regulate the vessels response to vasodilatory stimulants.⁶⁷⁶ One hallmark of endothelium dysfunction is the progressively impaired vasodilatory response to blood flow and vasodilating compounds.⁷⁶ It was believed that a reduced vasodilation, even at the retinal level is mainly due to a diminished bioavailability of NO. Nevertheless, in our study, the NO levels were not different between the three groups, and the impaired microvascular responses were only correlated to abnormal plasma oxysterols levels.

Although a surprise, it is possible that these circulatory biomarkers are, indeed, the initial elusive culprits for the consistently observed age-related retinal vascular dysfunction and not early abnormal NO levels.⁵³¹ This observation and hypothesis does not eliminate, however, the well-demonstrated involvement of NO early microvascular dysfunction. Indeed, oxysterols are involved in the impairment of endothelial-eNOS-dependent NO generation. Moreover, as well-known reactive oxygen species (ROS) inducers, oxysterols also contribute to a decline in NO availability ⁶⁷⁷. Therefore, the fact that NO levels were not different between the 3 groups of individuals tested could simply mean that the oxysterol actions were the first to appear and be measured. More research is necessary to confirm our hypothesis.

Oxidative stress is known to be a major driving force in ageing and in the development of CVD. In addition, oxysterols may contribute to endothelial senescence and CVD by making the endothelial cells more susceptible to various risk factors for CVD.⁶⁶⁴ Therefore, our observations are highly significant and emphasise the fact that the assessment of plasma oxysterols as well as the retinal microvascular dysfunction are among the earliest markers for CVD risk. As for the fact that to date the clinical tools used to assess risk for CVDs rely mainly structural vascular changes and, therefore, as a direct result, either overestimate or underestimate individualized risk.⁶⁰⁸

7.6. Conclusion

Our observations suggest that using DVA and plasma oxysterols levels instead of other time consuming and less sensitive methods could increase the sensitivity of CVDs risk prediction, especially in individuals free of overt CVD but with various risk factors. Additionally, as molecular and imaging biomarkers drive the shift towards personalized medicine, DVA and oxysterols levels can be used as a quick, minimally invasive method of profiling individualized vascular risk, therefore, to be used in prediction, prevention, and personalized intervention.

8. Prediction of Cardiovascular Risk in Asymptomatic Individuals: A Symbolic Regression-Based Analysis of Single and Composite Vascular-Omics

8.1. Abstract:

Aims: To assess the performance of using either leucocyte telomere length or retinal microvascular dysfunction or the combination of both for predicting age and systemic blood pressure, two of the most known risk factors for cardiovascular diseases.

Methods: The final analysis included one hundred twenty-three healthy participants with low global cardiovascular risk and similar dietary habits. Study participants were divided into three age groups (30 and below, between 31 and 50 and over 50). Parameters of microvascular function (dilation and constriction) have been assessed at the retinal microvascular level using the dynamic retinal vessel analyzer. Relative telomere length was measured using real-time polymerase chain reaction, and a symbolic regression-based analysis was used to predict the cardiovascular risk factors of age and blood pressure.

Results: The three groups were also similar regarding heart rate, body mass index, triglycerides, high-density lipoprotein, glutathione redox system (all p-value >0.05). Total cholesterol and low-density lipoprotein were significantly higher in middle age and elderly groups compared to the younger group (p=0.0004, 0.006 and p=0.0001, 0.001). Relative telomere length decreased significantly with age (p=0.010). There were significant group differences in arterial time to maximum dilation (p=0.005), maximum constriction (p=0.007) and maximum constriction percentage (p=0.010). Time to maximum dilation and constriction significantly decreased in the youngest and middle age groups compared with the oldest group (p = 0.012, 0.013 and p= 0.0024, 0.0004 respectively). Artery maximum constriction percentage increased in the oldest age group compared to the youngest and middle age group (p = 0.0085 and p = 0.0148, respectively). In the two younger groups, the error between predicted versus actual values for age are smallest in the case of using artery measurements only, while the largest error occurred when using telomere only. In the older group, using telomere measurements only led to the lowest error. Systolic blood pressure was better predicted by telomere measurements only; however, the smallest error in diastolic blood pressure prediction occurred when both telomere and arterial measurements were used.

Conclusion: The assessment of retinal vascular function is a better predictor of chronological ageing than Relative telomere length. Nevertheless, composite biomarkers are better when predicting modifiable risk factors for cardiovascular diseases, such as systemic blood pressure.

8.2. Introduction:

At present, the identification of at-risk individuals for cardiovascular disease (CVD) is performed using risk calculators that consider population-based markers such as lipid profiles, smoking and hypertension.^{678,679} Nevertheless, it has been shown the CVD predictive accuracy of such scores either over- or underestimates actual risk at the individual level.^{608–610} Indeed, for complex, multifactorial disorders such as CVD, there is high individual variability in the resistance and susceptibility to pathological onset and progression.¹⁶³ Therefore, the hunt for reliable and easy to measure individual risk predictors for CVD is still on. Nevertheless, this is not an easy task. Most biomarkers, especially those used for early screening, should be easily available and fit a tight budget while being performant and reliable. Ageing is a significant risk factor for many pathologies, including CVDs.^{680–682} It has been, however, admitted that individuals do not age at a similar pace.¹⁶⁸ This observation has led to the concept of biological ageing, which is a measure of bodily functional decline ¹⁶⁸ and represents a parameter that can be used for the selection of the individuals in need for early CVD prevention.⁶⁸³ Indeed, the assessment of biological ageing can be, possibly, the earliest available markers for CVD risk.

Both reduced telomeres' length and vascular dysfunction have been linked to ageing as well as early CVD risk detection. Indeed, reduced leukocyte telomere length (LTL), beside its agepredictive power, has also been associated with known CVD risk factors, such as positive family history, lower high-density lipoprotein cholesterol (HDL-C) levels ⁶⁸⁴, smoking and alcohol consumption.^{685,686} but also with high oxidative stress levels ¹⁹⁷ and increased endothelial cell turnover.¹⁸⁷ Moreover, recent studies also link telomeres length to the development of hypertension.⁶⁸⁷ Nevertheless, the assessment of telomeres is a complicated process and is not readily available.

The assessment of vascular endothelial function is essential not only for investigations into the pathophysiology of CVD but also for better CVD risk stratification.^{76,163,683} Parameters such as retinal microvascular function are easily measured, potentially in primary care settings and have been found to show a good association not only with various circulatory markers for

H.A. Shokr, PhD Thesis, Aston University 2021

164

CVD^{514,688}, but also with other modifiable and non-modifiable risk factors for this disease such as obesity ⁶¹⁶, family history ⁵¹³ and age. ^{531,589} A highly desirable quality of this assessment is the fact that it provides instant integrated and dynamic data analysis that is specific to each individual rather to a population.^{513–515,531,606,626} Therefore, its quantification in apparently healthy individuals could offer quick and reliable data on their systemic health and possible risk for CVD.

So far, both above-discussed parameters were demonstrated to be used, with certain degrees of success, as predictors for CVD risk.^{5,8,689} Nevertheless, the question remains if the proportion of actual cases correctly predicted can be increased by using a combination of these two biomarkers? Indeed, combining multiple biomarkers to develop a composite biomarker with enhanced performance is an appealing perspective. Although composite biomarkers have been successfully validated in other multifactorial disorders, such as cancer ^{690,691}, their utility in predicting CVD risk in asymptomatic individuals without obviously elevated conventional risk factors is unclear. Indeed, as previously discussed, already existing composite biomarkers such as the Framingham Risk Score (FRS) are based on population risk parameters and are not very successful in predicting early CVD.

This is largely because this score is an incremental to known risk factors.⁶⁹² The aim of the present study is to assess the performance of using either LTL or retinal microvascular dysfunction or the combination of both for predicting age and systemic BP, two of the most known risk factors for CVD.

8.3. Methods

8.3.1. Study Participants

Healthy individuals aged 18-year-old and above were recruited for this study through advertisements at the Vascular Research Laboratory, Aston University (Birmingham, UK). The study was designed and conducted in accordance with the tenets of the Declaration of Helsinki, and all study-related procedures adhered to institutional guidelines.

The inclusion/exclusion criteria as defined in section 3.2.1.

8.3.2. General Investigations

Standard anthropometric measures of height and weight were recorded to determine body mass index (BMI = weight/height). General clinical assessments were conducted for all participants as detailed in sections 3.3.2, 3.3.3.a, 3.3.4 included general health history questionnaires, BP and IOP profiles.

8.3.3. Blood Analyses

Bloods samples were drawn on the morning of the appointment. Fasting TG, T-CHOL and HDL-C were analysed according to the detailed protocol in section 3.3.9. Low-density lipoprotein cholesterol (LDL-C) values were calculated as per the Friedewald equation.

8.3.4. Measurement of GSH and Oxidized Glutathione (GSSG)

Initial processing of blood GSH and GSSG levels were assessed by the GSH recycling assay as detailed previously in section 3.3.10.1.

8.3.5. Relative Telomere Length

RTL was measured using real-time polymerase chain reaction (RT-PCR) according to a previously detailed method in section 3.3.10.3.

8.3.6. Dynamic Retinal Microvascular Function Vessel Analysis

Vascular assessments of interest to this study, detailed in the methods section 3.3.5. For each participant, one unselected eye was evaluated by DVA. All raw data were re-analysed using the mathematical approach for the analysis of retinal response parameters described in section 3.3.5.12. The following parameters averaged across three flicker cycles and with the arteries and veins regarded separately, were evaluated in this study: baseline diameter, BDF, DA, BCFR, MD%, MC%, tMD, tMC as well as dilation (Slope_D) and constriction slopes (Slope_c) (section 3.3.5.11).

8.3.7. Sample Size

Based on previous studies, a change of 30% with a SD of 2.5% in retinal vessels reactivity was shown to be significant.^{540,597} As the study design was multifactorial in nature, it was

calculated that a total sample size of n = 120 was sufficient to provide 95% power at an alpha level of 0.05.

One hundred twenty-three healthy participants with low global cardiovascular risk (< 10% at 10 years as assessed by the FRS and similar dietary habits were included in the final analysis. We divided our study participants into three age groups (30 and below, between 31 and 50 and over 50).

8.3.8. Statistical Analysis

Data were analysed according to the methods explained in 3.3.12.

8.3.9. Symbolic Regression

A symbolic regression-based analysis was used to model and subsequently predict the cardiovascular risk factors of age and BP, as described in section 3.3.11.

8.4. Results

Table 8.1 shows the general and circulatory markers characteristics of the study population, stratified by age groups.

There were no statistically significant differences between the number of participants in each group (p>0.05). In addition, the number of men and women in each group was similar (p>0.05). The three groups were also similar with regards to HR, BMI, TG, HDL, GSH and GSSG (all p-value >0.05). However, although within the normal range, T-CHOL and LDL plasma concentrations were significantly different between the 3 study groups (p<0.001 and p=0.010), with middle age and elderly groups showing higher concentrations compared to the younger group (p= 0.0004, 0.006 and p= 0.0001, 0.001 respectively). On the other hand, RTL decreased significantly with age (p=0.010) Table 8.1.

8.4.1. Differences in Retinal Vascular Function

Group differences in flicker-induced retinal arterial diameter changes (DVA) are summarized in Table 8.2. All reported values are based on data averaged across the three flicker cycles, with the artery and vein regarded separately. After controlling for influential covariates identified in multivariate analysis, there were no significant group differences in baseline diameter, BDF, DA, BCFR, MD, MD%, MC, Slope_{AD} and Slope_{Ac} (all p > 0.01, Table 8.2). There were, however, significant group differences in arterial tMD (p=0.005), MC (p=0.007) and MC% (p=0.010) (Table 8.2). *Post hoc* comparisons showed tMD, and tMC significantly decreased in the youngest and middle age groups compared with the oldest group (p = 0.012, 0.013 and p= 0.0024, 0.0004 respectively). Additionally, artery MC% was increased in the oldest age group compared to the youngest and middle age groups group p = 0.0085 and p =0.0148, respectively) (Figure 8.1). There were no statistically significant differences between all measured venous retinal parameters in all study groups (all p>0.01).

Table 8.1: General Characteristics of the Study Population

Variables	<30	31-50	>50	p-value	Post-hoc analysis
Number	40	47	36	>0.05	-
Gender	20M:20F	23M:24F	19M:17F	>0.05	-
Age (years)	24.95 (0.72)	38.28 (0.67)	58.53 (0.76)	0.0000*	1<2<3
SBP	114.9(1.84)	114.96(1.70)	121.86(1.94)	0.0129*	1=2<3
DBP	66 (1.35)	70.64(1.25)	72.83 (1.43)	0.0022*	1=2<3
HR (bpm)	66.05 (1.32)	64.94(1.22)	62.11 (1.39)	0.1133	-
BMI (kg/m²)	24.92 (0.71)	26.07 (0.65)	26.53 (0.76)	0.2727	-
Glucose	4.43 (0.11)	4.68 (0.12)	5.00 (0.12)	0.0034*	1=2<3
TG (mmol/L)	0.83 (0.050)	0.92 (0.047)	0.98 (0.05)	0.1269	-
T-CHOL	3.94 (0.12)	4.63 (0.12)	4.51 (0.14)	<0.001*	2=3>1
HDL-C (mmol/L)	1.33 (0.06)	1.26 (0.06)	1.16 (0.06)	0.051	-
LDL-C (mmol/L)	2.21 (0.13)	3.13 (0.12)	2.92 (0.14)	0.001*	2=3>1
GSSG	31.19 (3.43)	36.80 (2.93)	28.70 (3.33)	0.169	-
GSH	348.79 (47.91)	412.13 (41.03)	410.09 (46.52)	0.549	-
RTL	0.64 (0.22)	0.09 (0.21)	-0.36 (0.24)	0.010*	1>2>3

Table 8.1: BMI, body mass index; SBP, systolic blood pressure; DBP, diastolic blood pressure; HR, heart rate (in beats per minute); GLUC, glucose; TG, triglycerides; T-CHOL, total cholesterol; HDL-C, high-density lipoprotein cholesterol; LDL-C, low-density lipoprotein cholesterol; GSH, Glutathione; GSSG, Glutathione disulphide; RTL, leucocyte telomere relative lenght.

*Significant p-values are indicated where p < 0.05 was considered significant.

^a Data are presented as mean (SD) unless otherwise indicated.

^b Calculated as weight in kilograms divided by height in meters squared.

		Mean (SD)			
Parameter	19-30	31-50	>50	P-Value	Post-hoc analysis
Artery baseline	116(5.02)	114.86 (2.22)	108.46 (6.07)	0.107	
Artery-BDF	6.34 (0.41)	5.22 (0.38)	4.80 (0.44)	0.029*	
Artery-DA ^a	10.80 (0.70)	9.94 (0.66)	8.18 (0.76)	0.041*	
Artery-BCFR ^b	4.48 (0.40)	4.69 (0.37)	3.26 (0.43)	0.034*	
Artery-MD	124.11 (2.04)	118.36 (1.90)	116.11 (2.15)	0.021*	
Artery-tMD	17.44 (0.58)	17.55 (0.53)	19.89 (0.61)	0.005*	1=2<3
Artery-MD%	5.31 (0.32)	4.66 (0.29)	4.30 (0.34)	0.104	-
Artery-MC	113.17 (2.30)	109.92 (2.12)	112.80 (2.43)	0.521	-
Artery-tMC	24.65 (2.49)	23.94 (1.13)	30.75 (3.11)	0.007*	1=2<3
Artery-MC%	-3.40 (0.30)	-3.55 (0.27)	-2.70 (0.30)	0.010*	1=2>3
Artery-Slope _{AD} ^c	0.45 (0.04)	0.39 (0.04)	0.38 (0.05)	0.567	-
Artery-Slope _{AC} ^d	-0.56 (0.04)	-0.44 (0.04)	-0.36 (0.05)	0.126	-

Table 8.2. Summary of Retinal Arterial Vascular Function Parameters

Table 8.2. Baseline, baseline diameter; BDF, baseline diameter fluctuation; DA, dilation amplitude; BCFR, baseline-corrected flicker response; MD, artery maximum dilation; tMD, reaction time to maximum dilation diameter; MD%, percentage change in diameter from baseline to maximum dilation; MC, Maximum constriction; tMC, reaction time to maximum constriction diameter from maximum dilation diameter; MC%, percentage constriction below baseline; Slope_{AD}, slope of arterial dilation; Slope_{AC}, slope of arterial constriction. Unless otherwise indicated, all values are expressed in arbitrary units, which approximately correspond to micrometres (μ m) in a normal Gullstrand eye.

*Significant p-values are indicated where p < 0.05 was considered significant.

a Calculated as MD – MC, b Calculated as DA – BDF

c Calculated as (MD - baseline)/tMD

d Calculated as (MC-MD)/ (tMC)



Figure 8.1: Comparison of Retinal Arterial Response Profile Across Groups.

Abbreviations: AU, arbitrary units; BDF, baseline diameter fluctuation calculated as the maximum range in vessel diameter during first 30 seconds of baseline readings; MD%, calculated as the percentage change in vessel diameter from baseline to maximum following onset of flicker; tMD, time to reach maximum diameter during flicker; tMC, time to reach maximum constriction post flicker; slope_{AC}, calculated as (MC-MD)/ (tMC).

8.4.2. Symbolic Regression-based Analysis

As differences between groups were mainly perceived at the retinal arteriolar level, only these measurements have been used for our analysis. We have performed three sets of Symbolic Regression experiments for each age group, based on the combined biomarkers of arterial measurements and telomere, arterial measurements only, and telomeres only. 5-fold cross validation with 30 Symbolic Regression runs per fold was used for each set of experiments.

8.4.2.1. Age prediction

The mean absolute error (MAE) values for each age group are presented in tables 8.3a, 3b, 3c and figure 8.2. In the two younger groups, the error between predicted versus actual values for age are smallest in the case of using artery measurements only, while the largest error occurred when using telomere only. The situation was completely the opposite in the older group, where using telomere measurements only led to the lowest error.

The Friedman test for age prediction data indicated that only for the youngest age group are the differences significant (p-value 0.0295), and in this case, the Nemenyi post-hoc test indicated that at α =0.05, only the results for arteries only and telomere only are significantly different, with p-value 0.043.

The Kruskal-Wallis test indicated that the differences are significant for Telomere + Arteries and Arteries groups in our case, and in these cases, we performed the Dunn post-hoc test to determine which groups have different means. With p-values of 0.012 and 0.000022, we established that the differences are significant for telomere and arteries, between the youngest group and the middle group 2 and between the youngest and oldest groups. In the case of arteries, all pairwise differences were indicated by the Dunn test as significant.

Arteries + Telomere		Arteries Only		Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	1.488355	1	1.457137	1	2.150139
2	2.03976	2	1.628655	2	2.466321
3	1.634165	3	1.580925	3	3.263452
4	2.157625	4	2.038026	4	3.874383
5	1.157639	5	1.40857	5	1.776776
Average	1.695509	Average	1.622663	Average	2.706214

 Table 8.3a:
 Mean Absolute Errors for age predictions (Age group 1).

Table 8.3b: Mean Absolute Errors for age predictions (Age group 2).

Arteries + Telomere		Arteries Only		Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	2.812663	1	2.527317	1	3.204149
2	2.107845	2	2.245985	2	1.518323
3	2.790105	3	2.620622	3	3.189129
4	3.892265	4	3.516084	4	4.30067
5	2.632783	5	2.965419	5	3.701804
Average	2.847132	Average	2.775085	Average	3.182815

Table 8.3c: Mean Absolute Errors for age predictions (Age group 3).

Arteries + Telomere		Arteries Only		Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	4.902158	1	6.013113	1	2.392696
2	2.922922	2	5.230758	2	4.52436
3	4.003612	3	3.691483	3	3.489574
4	4.052891	4	5.85124	4	4.656914
5	6.184841	5	4.847482	5	3.400724
Average	4.413285	Average	5.126815	Average	3.692854





Figure 8.2. Predicted age vs actual age using models based on artery data, telomere relative expression and artery and telomere combined data. Colour coded for the different age groups (green: 18 – 30, orange: 31 – 50, blue: 50+).

8.4.2.2. Blood Pressure Prediction

As shown in Tables 8.4a,4b,4c and 8.5a,5b,5c and figures 8.3 and 8.4, composite biomarkers are better when predicting systemic diastolic BP and using the telomeres alone gave a better prediction of the systolic blood pressure.

Arteries and Telomere		Arteries Only		Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	7.60824	1	9.781193	1	9.214715
2	14.67967	2	15.0163	2	8.569096
3	8.132338	3	6.466947	3	9.16523
4	8.570601	4	14.36258	4	6.129705
5	13.69032	5	7.829224	5	9.043941
Average	10.53623	Average	10.69125	Average	8.424537

Table 8.4a: Mean Absolute Errors for Systolic BP predictions (Age group 1).

Table 8.4b: Mean Absolute Errors for Systolic BP predictions (Age group 2).

Arteries and Telomere		Arteries	only	Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	8.581326	1	8.418168	1	7.495257
2	7.755753	2	7.490878	2	7.130104
3	10.37444	3	7.814303	3	5.463688
4	10.96121	4	8.422463	4	10.27759
5	8.381083	5	6.844803	5	8.355181
Average	9.210761	Average	7.798123	Average	7.744364

 Table 8.4c:
 Mean Absolute Errors for Systolic BP predictions (Age group 3).

Arteries and Telomere		Arteries Only		Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	17.1292	1	16.13368	1	12.3724
2	6.913672	2	5.783934	2	5.320442
3	8.221233	3	8.26078	3	8.073649
4	8.246475	4	6.949804	4	5.237078
5	7.398514	5	5.531346	5	6.151203
Average	9.581819	Average	8.531909	Average	7.430954





Figure 8.3: Predicted systolic blood pressure vs actual systolic blood pressure using models based on artery, telomere, artery + telomeres combined data. Colour coded for the different age groups (green: 18 – 30, orange: 31 – 50, blue: 50+).

Arteries and Telomere		Arteries	Only	Telomere Only		
Fold	MAE	Fold	MAE	Fold	MAE	
27	5.110043	23	6.43493	1	6.644792	
26	6.032743	3	7.535935	2	5.476441	
4	4.814686	14	4.070671	3	5.058709	
4	4.77134	11	4.221734	4	5.449827	
22	4.172079	23	3.580786	5	4.426252	
Average	4.980178	Average	5.168811	Average	5.411204	

Table 8.5a: Mean Absolute Errors for Diastolic BP predictions (Age group 1).

Table 8.5b: Mean Absolute Errors for Diastolic BP predictions (Age group 2).

Arteries and Telomere		Arteries Only		Telomere Only	
Fold	MAE	Fold	MAE	Fold	MAE
1	4.733007	1	5.99453	1	4.215411
2	4.618492	2	6.014625	2	5.779917
3	7.849373	3	8.224371	3	5.561008
4	6.65106	4	8.225318	4	5.896184
5	2.782652	5	3.812176	5	8.340123
Average	5.326917	Average	6.454204	Average	5.958529

Table 8.5c: Mean Absolute Errors for Diastolic BP predictions (age group 3).

Arteries and Telomere		Arteries Only		Telomere Only		
Fold	MAE	Fold	MAE	Fold	MAE	
1	8.175021	1	7.292121	1	7.451285	
2	6.597758	2	5.685044	2	6.589668	
3	4.907752	3	5.379021	3	7.05904	
4	3.739657	4	5.28883	4	4.040507	
5	5.058974	5	6.01973	5	3.199624	
Average	5.695832	Average	5.932949	Average	5.668025	





Figure 8.4: Predicted diastolic blood pressure vs actual diastolic blood pressure using models based on artery, telomeres and artery + telomeres combined data. Colour coded for the different age groups (green: 18 – 30, orange: 31 – 50, blue: 50+).

8.5. Discussion

Our results show, for the first time, that the assessment of retinal vascular function is better predictor of chronological ageing that RTL. Nevertheless, using composite biomarkers are better when predicting modifiable risk factors for CVD, such as systemic BP.

In line with our previous reports ⁵³¹, we have again demonstrated that in healthy individuals, retinal vascular function parameters are affected by the chronological age, therefore being a good predictor for vascular dysfunction associated with this physiological process. In addition, and as already published ^{693–695}, RTL values has also declined with age in our groups. Nevertheless, as many studies imply, individuals do not age at a similar pace, a concept included into the definition of the so-called "biological age", which is considered a measure of each individual's health status and, possibly, their health span, things that cannot always be predicted by looking at the chronological age only.⁶⁹⁶ Indeed, assessing biological age can help identify those individuals at risk from preventable diseases, such as CVD, through early applications of healthy-ageing intervention programs. They can also be used for selecting suitable candidates for clinical trials designed to delay the onset and the progression of ageing-associated pathological conditions.⁶⁹⁷ Therefore; it is understandable why the need of measuring biological ageing is so important.

Many studies use various circulatory markers in association with chronological age to try to predict everyone's biological age; however, this approach is not ideal, as using chronological age into this type of calculation is already skewing the results in the wrong direction. The use of RTL only as a marker of biological ageing is also criticised as recent research show that RTL per se represent only a rough estimate of the ageing rate and can hardly be regarded as a clinically important risk marker for age-related pathologies and mortality.⁶⁹⁷ Indeed, the process of RTL shortening with age is very complex, and the exact mechanisms underlying this process are not yet established.^{697–699} Indeed, our study revealed that for younger individuals (groups 1 and 2), even when it came to predicting the actual chronological age, the largest error was given when using the RTL measurements only. A much better prediction was obtained when the RTL values were combined with the retinal vascular function. However, what was particularly noticeable was that when using retinal arterial function measures alone, the prediction error was the lowest. However, the situation was completely the opposite when we considered individuals over 50 years old when assessing the RTL had the best prediction success when it came to chronological ageing. There is no clear explanation for this finding, and more research needs to be done to understand the exact mechanism behind it. Nevertheless, we can hypothesise that, in older individual, vascular function is a less predictable measurement due to the heterogeneity of their vascular health status. Indeed, in this age group, the influence of genetic, environmental and socioeconomic factors that had

179

influenced the function of their bodies can be, probably, perceived more and, therefore, will result in a large variability when it comes to assessing their vascular health and using is as a predictor of their chronological ageing.

The level of blood pressure is another risk factor for CVD, and many attempts were made to possibly predict these, even using retinal vascular imaging.⁷⁰⁰ Nevertheless, these methods have limitations as they use static imaging, and the computer will perceive structural changes that are present when the disease is already present, even if at the subclinical level. By assessing retinal vascular function rather than structure, we believe that our method will be far superior in this respect. Nevertheless, for the SBP and for all age groups, the assessment of RTL was far superior to that of retinal vascular function or of the combination of RTL and retinal vascular function. A similar situation was also found for DBP values, apart from younger individuals were using the retinal arterial function in combination with RTL assessment and was a much better predictor of the actual values.

This is a very interesting discovery. Telomere length has been previously associated with the development of hypertension. However, this parameter has never been used to predict BP values in asymptomatic individuals. More research is necessary to determine the validity of our findings.
9. General Summary and Discussion

Assessment of systemic vasculature is a crucial step in the early diagnosis and management of CVDs. The currently used vascular investigating techniques provided an invaluable understanding of many cardiovascular pathologies; however, they are either expensive or technically complicated to be used in the widespread screening of individuals at risk. Likewise, the commonly used risk scores though beneficial from a public health perspective; however, they have more inferior predictive value for the individual patient. As a result, a simpler and more accessible technique is required.

With the close correlation between the macrovascular blood supply in the cardiovascular system and the microvasculature of the retina and as both networks share close regulatory processes, it was thought that retinal vascular analyser could be used to assess the systemic vascular function. Owing to the homology between the two networks, any changes in the retinal vessels may reflect similar changes in the cardiac vasculature.

DVA also has a unique advantage owing to its ability to simultaneously and directly visualize and assess both the microvascular function and structure. It can also provide detailed vascular profiling at the individual rather than population level, which could guide the shift towards personalised medicine, especially in cases at high risk to develop CVD despite having low predicted cardiovascular risk score.

This thesis focused on analysing the microvascular function as reflected by the retinal vessels to assess whether microcirculatory dysfunctions can be present and detected at an early stage in otherwise healthy individuals with different cardiovascular risk factors. As the evaluation of one vascular bed can be limiting, in this thesis, we sought to provide a broader picture of systemic vascular health using additional validated surrogate markers for vascular endothelial structure, function and plasma oxidative stress status. It was thus hypothesised that outcomes substantiating the relevance of retinal vascular function against these measures could foster new strategies in cardiovascular diseases risk screening, prevention, and management in asymptomatic individuals.

In summary, the findings of this work were:

9.1. Microvascular Function and Oxidative Stress in Adult individuals With Early Onset of Cardiovascular Disease

Vascular remodelling represented by alterations in the endothelial structure and function of resistance vessels is well identified and assessed in cardiovascular patients. However, questions remain around individuals who are classified as having early-onset CVDs, such as high normal blood pressure and grade 1 hypertension. These classes of hypertension are not subjected to medical treatment, and routine follow-up techniques are not suitable for assessing

their preclinical stages of the diseases. Additionally, to date, no studies evaluated the microvascular function in this population, and our understanding of when the first recognizable signs of vascular dysfunction occur in patients with overt CVDs, or those at risk is limited.

Therefore, this study (chapter 4) aimed to evaluate for the first time the microvascular function in different hypertensive patients classified according to the European Society of Cardiology guidelines. Additionally, as oxidative stress and its role in the progression to more severe stages of hypertension are highlighted as a well-established causative factor of macrovascular dysfunction, we assessed if the degree of endothelial dysfunction discovered could relate to GSH redox system imbalance in these patients.

Following DVA assessment, alteration in the retinal vessels' response was detected in the high-normal and grade 1 hypertension groups in the form of higher baseline diameter fluctuation, impaired dilation responses, and post-flicker under-constrictions compared to individuals in the optimal BP group. In parallel with these changes, alterations in the glutathione redox buffering system were found. Furthermore, GSH and GSSG levels were found to correlate with SBP, DBP and MBP values in all the study participants.

Although we can only hypothesize the exact cause of the early signs of microvascular dysfunction in these groups, our findings support the early onset of CVDs in the microvascular level before the macrovascular level. Furthermore, this study confirmed that functional changes in the microvasculature are an early sign of losing macrovascular function and structure later in life.

While structural and functional changes in the macrovascular level are easy to assess and identify clinically, it is crucial to identify these vascular anomalies when the disease is in the subclinical phase to allow superior disease management. Thus, this study suggests that retinal vascular assessment could be a valuable tool in identifying early cardiovascular risks in individuals with early stages of CVDs. Additionally, since dysfunctional retinal responses correlated with an established cardiovascular risk indicator (oxidative stress), it could act as a helpful screening marker to identify asymptomatic individuals who could benefit from early targeted and individualized therapy.

9.2. European Society of Cardiology/European Society of Hypertension versus the American College of Cardiology/American Heart Association guidelines on the cut-off values for early hypertension: A Microvascular Perspective

The most recent and widely used guidelines for the diagnosis and management of high blood pressure are the 2017 American College of Cardiology (ACC)/American Heart Association (AHA) guidelines and the 2018 European Society of Cardiology (ESC)/European Society of Hypertension (ESH) guidelines. Although both guidelines adopted the same technique in

measuring BP, both have different cut-off values to diagnose hypertension (130/80 vs 140/90) and different levels to which BP should be reduced in hypertensive patients. These differences in the level of HTN that require treatment added a lot of pressure on the healthcare systems worldwide, increased the health insurance costs and increased unreimbursed physician time for visits. The ACC/AHA argues that patients' early treatment can help slow the progress of the disease and the development of diseases end-organ complications. On the other hand, the ESC claim that the risk of extrapolation to lower levels recommended in the ACC guidelines could result in overtreatment.

To assess the clinical validity of lower BP cut-off, we conducted the study presented in chapter 5 to compare the microvascular function in two beds (the retinal vessels and the peripheral microvascular beds) in healthy individuals free from CVDs risk factors who fall into different BP groups classified using both the ESC/ESH or the ACC/AHA guidelines. The analysis of the two vascular beds confirmed the initiation of microvascular dysfunction with blood pressure 140/90 and above in the form of an altered dynamic response profile and a steeper arterial dilation slope and venous constriction slope following cessation of flicker while significant peripheral vascular changes were only noticeable in the grade 1 BP group.

The Correlation between these two different vascular beds in individuals classified as ESC/ESH grade 1 hypertension only supports the introduction of higher BP targets when managing hypertension. It also supports the solo introduction of non-pharmacological treatments in individuals with BP below 140/90 and the routine use of retinal vascular analysis as a variable specific for each individual to help monitoring therapeutic responses in hypertension patients.

9.3. Dry Eye Disease is Associated with Retinal Microvascular Dysfunction and Possible Risk for Cardiovascular Diseases

Dry eye disease is an ocular surface disease that involves multiple interacting predisposing mechanisms. Several systemic clinical conditions were reported to negatively impact the ocular surface and lead to DED. Of these medical conditions, metabolic conditions such as diabetes mellitus and dyslipidaemia and CVDs such as peripheral vascular disorders, stroke and ischemic cardiac diseases were found to be positively associated with DED.⁷⁰¹ However, the type and strength of this association were neither consistent nor clear in the published literature.^{633,702,703} Moreover, most of these researchers based their diagnosis on signs and symptoms with no clinically objective ocular examinations. For this reason, our study in chapter 6 aimed to assess functional retinal and peripheral microvascular responses to flickering light in healthy and DED patients. Due to the numerous systemic associations of

DED ⁷⁰⁴, this study also included assessments of systemic circulatory biomarkers of vascular function.

The study's main finding was that individuals with dry eye disease exhibited impaired microvascular responses on the retinal beds, which was represented by the inability to reach proper arterial baseline diameter before and after the flicker stimulation and signs of arterial stiffness demonstrated as impaired constriction and dilation responses after stimulation. All these observed vascular impairments were significantly and positively correlated to systemic plasma lipids concentrations.

Although the impairments in retinal vascular function were not associated with peripheral arterial stiffness, this lack of significance can be justified by the pilot nature of the study as well as autoregulatory mechanisms of the retinal vessels and the absence of systemic compensatory mechanisms in this level which allow the appearance of signs of microvascular dysfunction in the retinal bed earlier than the peripheral microvascular beds.⁷⁰⁵

In this regard, further validation research in larger cohorts is recommended, and studies to evaluate the impact of the treatment of DED risk factors on the retinal and peripheral vasculature are still needed.

9.4. Novel composite early risk markers for vascular ageing and risk for cardiovascular diseases

The close association between ageing and vascular morphological and molecular alterations such as increased wall thickness, loss of elasticity and reduction of NO haemostasis has been verified extensively.⁷⁰⁶ Consequently, the risk of developing cardiovascular diseases increases in the elderly.⁷⁰⁷ Currently published literature highlighted imbalances in the redox status and consequent impairment of NO signalling as main determents of age-dependent vascular endothelial dysfunction. However, to date, no study examined the relationship with regard to microcirculation. Using oxysterols as a marker of oxidative stress, the study in chapter 7 examined the potential influences of established increased oxidation markers, systemic antioxidation defence capacity on retinal microvascular function in healthy individuals classified into different age groups. Study findings confirmed the systemic association between plasma oxysterol levels and gradual microvascular endothelial dysfunction at the retinal arterial level after the age of 30. This association was also correlated to decreased systemic antioxidant capacity (redox index) without any significant change in NO concentrations. This direct relationship between oxysterols and functional retinal response to stress and based of the risk category of individuals included in this study suggest the high sensitivity of the combination between oxysterols plasma concentration and DVA as a composite biomarker of early vascular ageing. Nevertheless, beside the action of local factors,

a better understanding of the possible systemic influences on microvascular retinal function in individuals without any CVD risks is still needed.

9.5. Prediction of Cardiovascular Risk in Asymptomatic Individuals: A Symbolic Regression-Based Analysis of Single and Composite Vascular-Omics

As discussed before, current clinically used risk prediction models are typically based on limited predictors with sub-optimal performance across all patient groups. Having established the presence of retinal vascular dysfunction in patients without overt clinical CVDs, an investigation into whether it can predict CVDs future risk was the next logical step. Hence, the study in section 8 aimed to examine the significance of the assessment of retinal vascular function alone and in combination with relative telomere length in predicting CVDs major risk factors (age, BP). On investigation, older age groups exhibited increased systemic arterial stiffness and alterations in retinal arterial reactivity to flicker light stimulation in the form of an altered dynamic response profile. Additionally, our results also confirmed the benefits that can be reached by using a composite measurement of retinal vascular responses and relative telomere length as a predictive biomarker of CVDs major risk factors such as age and blood pressure. These observations are of great scientific importance as combining these markers to currently used risk scores could be a cornerstone of preventative cardiology, which is the new era of interest in the international cardiovascular guidelines.

10. General Conclusion & Limitations

Studies in this thesis confirmed the usefulness of functional retinal vascular analysis using DVA machine as a surrogate endpoint of endothelial dysfunction and a marker of future risk of development of CVDs in apparently healthy individuals. However, it is appreciated that the included research in this thesis is still subject to some potential limitations. For instance, all the included studies were cross-sectional in nature, so follow up periods couldn't be established. Further longitudinal studies are recommended to investigate the progressive association between microvascular dysfunction and the development of clinical vascular and cardiac diseases. Further evaluation of the predictive ability of the suggested composite CVDs risk stratification markers is also recommended to strengthen the clinical value of routine functional retinal vascular assessment and oxidative stress biomarkers analysis in primary clinical settings as an easy non-invasive CVDs risk stratification tools.

The validity and reliability of the research findings should be tested in special populations such as African Americans, Asian, children and cancer survivors, as it is known these individuals have higher susceptibility to CVDs and abnormal responses to commonly used medications and interventions. Larger sample size is needed to confirm the external validity of the findings. It is also acknowledged that much of the variability in specific retinal vascular function parameters in the general population remains to be explained as the role of conventional risk factors in determining retinal vascular reactivity across the range usually found in the general population and not just at extremes of the range remains to be quantified.

Nevertheless, as supported by findings in this thesis, dynamic retinal functional assessments provide an opportunity to explore the relationship between cardiovascular risk factors, microvascular endothelial dysfunction, and associative cardiovascular risk. The presented results open up several avenues that can be explored in future work.

11. Future Directions

11.1. Individuals-focussed Studies

Despite statistically significant results being demonstrated in this thesis, it would be interesting to determine individual's risk of developing CVDs using the retinal vascular function parameters. The generated data would help interpret the inter-relationships between risk factors and further evaluate risk severity in sub-categories of at-risk individuals such as those with genetic risk factors or ethnic susceptibility. Identifying these individuals will enable early management of the modifiable risk factors and accordingly, prediction, diagnosis, treatment and, eventually, prevention of CVDs can be matched to individuals based on genetics, environmental risk factors and lifestyle.

11.2. Improving the Predictive Power of Currently Used Risk Scores

Owing to the poor predictive power of the currently used risk scores, large cohorts should be established to assess whether functional retinal vascular parameters could improve the predicted results of the clinically used risk models. Early identification of at-risk individuals will facilitate early clinical intervention and slow down the progression of the disease.

11.3. Expansion of Preliminary Data

Results of the included studies confirmed the clinical significance of using retinal arteries functional parameters in combination with oxysterols as an early vascular ageing biomarker and relative telomeres length as a predictor of cardiovascular risk factors; however, conducting a larger prospective cohort study could enhance the validity of our findings with regard to exposures, confounders, and endpoints and could also minimize the recall error. Furthermore, expansion of CVDs risk predictive biomarkers could offer additional information of both aetiological and clinical relevance with regard to CDVs.

12. References:

- 1. Mensah GA, Roth GA, Fuster V. The Global Burden of Cardiovascular Diseases and Risk Factors: 2020 and Beyond. J Am Coll Cardiol [Internet]. 2019 Nov 19 [cited 2021 Jul 14];74(20):2529–32. Available from: https://doi.org/10.1016/j.jacc.2019.10.009
- 2. Cardiovascular diseases [Internet]. [cited 2021 Feb 16]. Available from: https://www.who.int/health-topics/cardiovascular-diseases#tab=tab_1
- 3. Hajar R. Risk factors for coronary artery disease: Historical perspectives. Hear Views. 2017;
- 4. World Health Organisation. Implementation tools Package of Essential Noncommunicable (PEN) disease interventions for primary health care in low-resource settings. MscforumOrg. 2013;
- 5. Hajifathalian K, Ueda P, Lu Y, Woodward M, Ahmadvand A, Aguilar-Salinas CA, et al. A novel risk score to predict cardiovascular disease risk in national populations (Globorisk): A pooled analysis of prospective cohorts and health examination surveys. Lancet Diabetes Endocrinol. 2015;
- 6. Lloyd-Jones DM. Cardiovascular risk prediction: Basic concepts, current status, and future directions. Circulation. 2010.
- 7. Joseph P, Yusuf S, Lee SF, Ibrahim Q, Teo K, Rangarajan S, et al. Prognostic validation of a non-laboratory and a laboratory based cardiovascular disease risk score in multiple regions of the world. Heart [Internet]. 2018 Apr 1 [cited 2021 Jul 14];104(7):581–7. Available from: https://heart.bmj.com/content/104/7/581
- 8. Selvarajah S, Kaur G, Haniff J, Cheong KC, Hiong TG, Van Der Graaf Y, et al. Comparison of the Framingham Risk Score, SCORE and WHO/ISH cardiovascular risk prediction models in an Asian population. Int J Cardiol. 2014;
- 9. Hippisley-Cox J, Coupland C, Vinogradova Y, Robson J, Minhas R, Sheikh A, et al. Predicting cardiovascular risk in England and Wales: Prospective derivation and validation of QRISK2. BMJ. 2008;
- 10. Knuuti J, Ballo H, Juarez-Orozco LE, Saraste A, Kolh P, Rutjes AWS, et al. The performance of non-invasive tests to rule-in and rule-out significant coronary artery stenosis in patients with stable angina: a meta-analysis focused on post-test disease probability. Eur Heart J [Internet]. 2018 Sep 14 [cited 2021 Jul 14];39(35):3322–30. Available from: https://academic.oup.com/eurheartj/article/39/35/3322/5020750
- 11. Sechtem U. Non-invasive testing in patients with suspected coronary artery disease: Some may be more equal than others. Eur Heart J. 2018 Sep 14;39(35):3331–3.
- 12. Agabiti-Rosei E, Rizzoni D. Microvascular structure as a prognostically relevant endpoint. J Hypertens. 2017;35(5):914–21.
- 13. Flammer J, Konieczka K, Bruno RM, Virdis A, Flammer AJ, Taddei S. The eye and the heart. Eur Heart J. 2013;
- 14. Guo S, Yin S, Tse G, Li G, Su L, Liu T. Association Between Caliber of Retinal Vessels and Cardiovascular Disease: a Systematic Review and Meta-Analysis. [cited 2021 Sep 16]; Available from: https://doi.org/10.1007/s11883-020-0834-2
- Theuerle JD, Al-Fiadh AH, Amirul Islam FM, Patel SK, Burrell LM, Wong TY, et al. Impaired retinal microvascular function predicts long-term adverse events in patients with cardiovascular disease. Cardiovasc Res [Internet]. 2021 Jul 7 [cited 2021 Jul 14];117(8):1949–57. Available from:

https://academic.oup.com/cardiovascres/article/117/8/1949/5880582

- 16. The Circulatory System Susan Whittemore, M D Texas Heart Institute Clinical Susan Whittemore Introduction by Denton a Cooley - Google Books [Internet]. [cited 2021 Jul 14]. Available from: https://books.google.co.uk/books?hl=en&lr=&id=TmPMCHpJTW4C&oi=fnd&pg=PA5& dq=human+circulatory+system+anatomy&ots=GTXlqIIyLx&sig=GsHLSpqaCpz5AP_Y DIrKJm7-aas&redir_esc=y#v=onepage&q=human circulatory system
- anatomy&f=false17. Margaris KN, Black RA. Modelling the lymphatic system: Challenges and

opportunities. Journal of the Royal Society Interface. 2012.

- 18. Aird WC. Discovery of the cardiovascular system: From Galen to William Harvey. Journal of Thrombosis and Haemostasis. 2011.
- 19. Bowman WC, Rand MJ. Textbook of Pharmacology. Textb Pharmacol. (2nd ed.).
- 20. BOOK. B. Guyton and Hall Textbook of Medical Physiology 13ed. Igarss 2014. 2014.
- 21. Komaru T, Kanatsuka H, Shirato K. Coronary microcirculation: Physiology and pharmacology. Pharmacology and Therapeutics. 2000.
- 22. Shah S, Gnanasegaran G, Sundberg-Cohon J, Buscombe JR. The heart: Anatomy, physiology and exercise physiology. In: Integrating Cardiology for Nuclear Medicine Physicians: A Guide to Nuclear Medicine Physicians. 2009.
- 23. American Heart Association. All About Heart Rate (Pulse). Heart.Org. 2018.
- 24. Mahadevan V. Anatomy of the heart. Surgery (United Kingdom). 2015.
- 25. cardiac muscle | Definition, Function, & Structure | Britannica [Internet]. [cited 2021 Dec 30]. Available from: https://www.britannica.com/science/cardiac-muscle
- 26. Atlas of Human Anatomy, Professional Edition E-Book: including ... Frank H. Netter -Google Books [Internet]. [cited 2021 Sep 15]. Available from: https://books.google.co.uk/books?hl=en&Ir=&id=grIBCgAAQBAJ&oi=fnd&pg=PA1&dq =Atlas+of+Human+Anatomy+by+Frank+H.+Netter&ots=NvdUPbNGxt&sig=Xk5XPX2 bE-cXTw_u7hiVaTlOyIg&redir_esc=y#v=onepage&q=Atlas of Human Anatomy by Frank H. Netter&f=false
- Anderson RH, Ho SY. The architecture of the sinus node, the atrioventricular conduction axis, and the internodal atrial myocardium. J Cardiovasc Electrophysiol. 1998;
- Anderson RH, Yanni J, Boyett MR, Chandler NJ, Dobrzynski H. The anatomy of the cardiac conduction system. Clin Anat [Internet]. 2009 Jan 1 [cited 2021 Jul 14];22(1):99–113. Available from: https://onlinelibrary.wiley.com/doi/full/10.1002/ca.20700
- Anderson RH, Ho SY, Becker AE. Anatomy of the human atrioventricular junctions revisited. Anat Rec. 2000;
- 30. Anderson R, Mori S, Spicer D, Sanchez-Quintana D, Jensen B. The Anatomy, Development, and Evolution of the Atrioventricular Conduction Axis. J Cardiovasc Dev Dis. 2018;
- The Heart's Electrical Conduction System EMS Education [Internet]. [cited 2021 Dec 30]. Available from: http://emspjeducation.weebly.com/the-hearts-electricalconduction-system.html
- 32. Marieb EN, Hoehn K. Human anatomy & amp; physiology. Http://Www.Pharmatips.in/. 2007.
- 33. Vaziri SM, Larson MG, Lauer MS, Benjamin EJ, Levy D. Influence of Blood Pressure on Left Atrial Size. Hypertension [Internet]. 1995 [cited 2021 Jul 14];25(6):1155–60. Available from: https://www.ahajournals.org/doi/abs/10.1161/01.HYP.25.6.1155
- 34. Scott GBD. Functional Histology a Text and Colour Atlas. Postgrad Med J. 1988;
- 35. DiFiore's. Atlas of Histology with functional correlations. Vasa. 2008.
- 36. Circulatory System Dynamics Abraham Noordergraaf Google Books [Internet]. [cited 2021 Sep 15]. Available from: https://books.google.co.uk/books?hl=en&lr=&id=1iMHTv4MGyAC&oi=fnd&pg=PP1&d q=the+circulatory+system+&ots=kMqN6Czaya&sig=WhwgThm45tc7ZVHNEWzUqhvn Kcc&redir esc=y#v=onepage&q=the circulatory system&f=false
- 37. Simionescu M, Simionescu N, Palade GE. Differentiated microdomains on the luminal surface of capillary endothelium: Distribution of lectin receptors. J Cell Biol. 1982;
- 38. Tucker WD, Mahajan K. Anatomy, Blood Vessels. StatPearls. 2018.
- 39. Heistad DD, Marcus ML. Role of vaso vasorum in nourishment of the aorta. Blood Vessels. 1979;
- 40. Vasa vasorum [Internet]. [cited 2021 Dec 30]. Available from: https://clinicalanatomy.com/100-mtd/358-vasa-vasorum
- 41. Mazzoni A, Hansen CC. Surgical Anatomy of the Arteries. Arch Otolaryngol [Internet].

188

1970 Feb 1 [cited 2021 May 16];91(2):128–35. Available from: https://jamanetwork.com/journals/jamaotolaryngology/fullarticle/602783

- 42. Rhodin JAG. The ultrastructure of mammalian arterioles and precapillary sphincters. J Ultrasructure Res. 1967;
- Moore DH, Ruska H. THE FINE STRUCTURE OF CAPILLARIES AND SMALL ARTERIES. J Biophys Biochem Cytol [Internet]. 1957 May 25 [cited 2021 Jul 14];3(3):457–62. Available from: http://rupress.org/jcb/articlepdf/3/3/457/1068789/457.pdf
- 44. Gross RJ, Osterle JF. Membrane Transport Characteristics of Ultrafine Capillaries. J Chem Phys [Internet]. 2003 Sep 18 [cited 2021 Jul 14];49(1):228. Available from: https://aip.scitation.org/doi/abs/10.1063/1.1669814
- 45. Abularrage CJ, Sidawy AN, Aidinian G, Singh N, Weiswasser JM, Arora S. Evaluation of the microcirculation in vascular disease. J Vasc Surg. 2005;
- 46. Sarin H. Physiologic upper limits of pore size of different blood capillary types and another perspective on the dual pore theory of microvascular permeability. J Angiogenes Res. 2010;
- Torres-Vázquez J, Kamei M, Weinstein BM. Molecular distinction between arteries and veins. Cell Tissue Res 2003 3141 [Internet]. 2003 Sep 23 [cited 2021 Jul 14];314(1):43–59. Available from: https://link.springer.com/article/10.1007/s00441-003-0771-8
- 48. Tucker WD, Arora Y, Mahajan K. Anatomy, Blood Vessels. StatPearls [Internet]. 2021 Aug 11 [cited 2021 Dec 30]; Available from: https://www.ncbi.nlm.nih.gov/books/NBK470401/
- 49. Hicks JW, Badeer HS. Gravity and the circulation: "Open" vs. "closed" systems. American Journal of Physiology - Regulatory Integrative and Comparative Physiology. 1992.
- 50. Levy MN. The cardiac and vascular factors that determine systemic blood flow. Circ Res. 1979;
- Arthur Guyton BC, Coleman TG, Cowley AW, Davis Manning Jr R, Norman RA, Ferguson JD. Circulation Research A Systems Analysis Approach to Understanding Long-Range Arterial Blood Pressure Control and Hypertension. Circ Res [Internet]. 1974 [cited 2021 Jul 14];35. Available from: http://ahajournals.org
- 52. Spencer MP. SYSTEMIC CIRCULATION 1. [cited 2021 Jul 14]; Available from: www.annualreviews.org
- 53. Physical Characteristics of the Circulation [Internet]. [cited 2021 Dec 30]. Available from: https://www.brainkart.com/article/Physical-Characteristics-of-the-Circulation_19262/
- 54. Mayet J, Hughes A. Cardiac and vascular pathophysiology in hypertension. Heart. 2003.
- 55. Stamler J. Blood pressure and high blood pressure. Aspects of risk. Hypertension [Internet]. 1991 [cited 2021 Jul 14];18(3):I-95-I–107. Available from: https://www.ahajournals.org/doi/abs/10.1161/01.HYP.18.3 Suppl.I95
- 56. Burnier M, Wuerzner G. Pathophysiology of hypertension. In: Pathophysiology and Pharmacotherapy of Cardiovascular Disease. 2015.
- 57. Williams B, Mancia G, Spiering W, Rosei EA, Azizi M, Burnier M, et al. 2018 practice guidelines for the management of arterial hypertension of the European society of cardiology and the European society of hypertension ESC/ESH task force for the management of arterial hypertension. J Hypertens. 2018;36(12):2284–309.
- 58. Whelton PK, Carey RM, Aronow WS, Casey DE, Collins KJ, Dennison Himmelfarb C, et al. 2017 ACC/AHA/AAPA/ABC/ACPM/AGS/APhA/ASH/ASPC/NMA/PCNA Guideline for the Prevention, Detection, Evaluation, and Management of High Blood Pressure in Adults: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Pr. J Am Coll Cardiol. 2018;71(19):e127–248.
- 59. Ackermann U. Regulation of arterial blood pressure. Surg. 2004 May 1;22(5):120a-120f.

- 60. Greene AS, Tonellato PJ, Lui J, Lombard JH, Cowley AW. Microvascular rarefaction and tissue vascular resistance in hypertension. Am J Physiol Hear Circ Physiol. 1989;
- 61. Sandoo A, Zanten JJC. V van, Metsios GS, Carroll D, Kitas GD. The Endothelium and Its Role in Regulating Vascular Tone. Open Cardiovasc Med J [Internet]. 2010 Feb 26 [cited 2021 Sep 16];4(1):302. Available from: /pmc/articles/PMC3040999/
- Dampney R, Horiuchi J, Killinger S, Sheriff M, Tan P, McDowall L. LONG-TERM REGULATION OF ARTERIAL BLOOD PRESSURE BY HYPOTHALAMIC NUCLEI: SOME CRITICAL QUESTIONS. Clin Exp Pharmacol Physiol [Internet]. 2005 May 1 [cited 2021 Jul 14];32(5–6):419–25. Available from: https://onlinelibrary.wiley.com/doi/full/10.1111/j.1440-1681.2005.04205.x
- Kiel JW. The Ocular Circulation. Colloq Ser Integr Syst Physiol From Mol to Funct. 2011;
- 64. J. K, E.A. N, T. C-L. Cellular and physiological mechanisms underlying blood flow regulation in the retina and choroid in health and disease. Prog Retin Eye Res. 2012;
- 65. Gallar J, Acosta MC, Belmonte C. Activation of scleral cold thermoreceptors by temperature and blood flow changes. Investig Ophthalmol Vis Sci. 2003;
- 66. Iwase T, Yamamoto K, Kobayashi M, Ra E, Murotani K, Terasaki H. What ocular and systemic variables affect choroidal circulation in healthy eyes. Med (United States). 2016;
- 67. Uyama M. Ocular circulation. J Japanese Ophthalmol Soc. 1999;
- Siesky B, Harris A, Ehrlich R, Kheradiya N, Lopez CR. Glaucoma Risk Factors: Ocular Blood Flow. Glaucoma B A Pract Evidence-Based Approach to Patient Care [Internet]. 2010 [cited 2021 Dec 30];111–34. Available from: https://link.springer.com/chapter/10.1007/978-0-387-76700-0 11
- 69. Ocular Blood Flow Google Books [Internet]. [cited 2021 Jul 15]. Available from: https://books.google.co.uk/books?hl=en&Ir=&id=p9pjQYzPXcC&oi=fnd&pg=PR3&dq=ocular+blood+flow&ots=zzJusKFt6p&sig=z5RAaUv 06b4TN95nWc1jaoDuVEs&redir_esc=y#v=onepage&g=ocular blood flow&f=false
- 70. Murray CD. The Physiological Principle of Minimum Work: I. The Vascular System and the Cost of Blood Volume. Proc Natl Acad Sci. 1926;
- 71. Glucksberg MR, Dunn R. Direct measurement of retinal microvascular pressures in the live, anesthetized cat. Microvasc Res. 1993;
- 72. Tham YC, Lim SH, Gupta P, Aung T, Wong TY, Cheng CY. Inter-relationship between ocular perfusion pressure, blood pressure, intraocular pressure profiles and primary open-Angle glaucoma: The Singapore Epidemiology of Eye Diseases study. Br J Ophthalmol. 2018;
- Ngo S, Harris A, Siesky BA, Schroeder A, Eckert G, Holland S. Blood pressure, ocular perfusion pressure, and body mass index in glaucoma patients. Eur J Ophthalmol. 2013;
- 74. Hall ME, Hamid A, Oshunbade AA. The Eyes: A Window to the Heart (and Mortality). Circ Res. 2019;124(12):1692–4.
- 75. Vita JA. Endothelial function. Circulation. 2011.
- 76. Deanfield JE, Halcox JP, Rabelink TJ. Endothelial function and dysfunction: Testing and clinical relevance. Circulation. 2007.
- 77. Widlansky ME, Gokce N, Keaney JF, Vita JA. The clinical implications of endothelial dysfunction. Journal of the American College of Cardiology. 2003.
- 78. Shimokawa H. 2014 williams harvey lecture: Importance of coronary vasomotion abnormalities From bench to bedside. European Heart Journal. 2014.
- 79. Félétou M, Vanhoutte PM. Endothelial dysfunction: A multifaceted disorder. In: American Journal of Physiology - Heart and Circulatory Physiology. 2006.
- 80. Schiffrin EL. A critical review of the role of endothelial factors in the pathogenesis of hypertension. J Cardiovasc Pharmacol. 2002;38(SUPPL. 2):7–10.
- 81. Verma S, Anderson TJ. Fundamentals of endothelial function for the clinical cardiologist. Circulation. 2002.

- 82. Vita JA. Endothelial Function. Circulation [Internet]. 2011 Dec 20 [cited 2021 Jul 14];124(25). Available from:
- https://www.ahajournals.org/doi/abs/10.1161/CIRCULATIONAHA.111.078824
 83. Levine GN, Keaney JF, Vita JA. Cholesterol reduction in cardiovascular disease _____
- Clinical Benefits and Possible Mechanisms. New England Journal of Medicine. 1995.
 84. Balligand JL, Cannon PJ. Nitric oxide synthases and cardiac muscle: Autocrine and paracrine influences. Arterioscler Thromb Vasc Biol. 1997;
- Singh S, Evans TW. Nitric oxide, the biological mediator of the decade: Fact or fiction? Eur Respir J. 1997;10(3):699–707.
- 86. Brüne B, Lapetina EG. Protein thiol modification of glyceraldehyde-3-phosphate dehydrogenase as a target for nitric oxide signaling. Genetic engineering. 1995.
- 87. Yermilov V, Rubio J, Becchi M, Friesen MD, Pignatelli B, Ohshima H. Formation of 8nitroguanine by the reaction of guanine with peroxynitrite in vitro. Carcinogenesis. 1995;
- 88. Bredt DS. Nitric oxide signaling specificity The heart of the problem. Journal of Cell Science. 2003.
- 89. Münzel T, Daiber A, Ullrich V, Mülsch A. Vascular consequences of endothelial nitric oxide synthase uncoupling for the activity and expression of the soluble guanylyl cyclase and the CGMP-dependent protein kinase. Arteriosclerosis, Thrombosis, and Vascular Biology. 2005.
- 90. Lundberg JO, Weitzberg E, Gladwin MT. The nitrate-nitrite-nitric oxide pathway in physiology and therapeutics. Nature Reviews Drug Discovery. 2008.
- 91. Hermann M, Flammer A, Lüscher TF. Nitric oxide in hypertension. Journal of clinical hypertension (Greenwich, Conn.). 2006.
- 92. Thomas DD, Heinecke JL, Ridnour LA, Cheng RY, Kesarwala AH, Switzer CH, et al. Signaling and stress: The redox landscape in NOS2 biology. Free Radical Biology and Medicine. 2015.
- 93. Hausladen A, Gow AJ, Stamler JS. Nitrosative stress: Metabolic pathway involving the flavohemoglobin. Proc Natl Acad Sci U S A. 1998;
- 94. Thomas DD, Ridnour LA, Isenberg JS, Flores-Santana W, Switzer CH, Donzelli S, et al. The chemical biology of nitric oxide: Implications in cellular signaling. Free Radical Biology and Medicine. 2008.
- 95. Farah C, Michel LYM, Balligand JL. Nitric oxide signalling in cardiovascular health and disease. Nature Reviews Cardiology. 2018.
- 96. Förstermann U, Sessa WC. Nitric oxide synthases: Regulation and function. European Heart Journal. 2012.
- 97. Lei J, Vodovotz Y, Tzeng E, Billiar TR. Nitric oxide, a protective molecule in the cardiovascular system. Nitric Oxide Biology and Chemistry. 2013.
- 98. Ritchie RH, Drummond GR, Sobey CG, De Silva TM, Kemp-Harper BK. The opposing roles of NO and oxidative stress in cardiovascular disease. Pharmacol Res. 2017;
- 99. Furchgott RF, Zawadzki J V. The obligatory role of endothelial cells in the relaxation of arterial smooth muscle by acetylcholine. Nature. 1980;
- 100. Buchwalow IB, Cacanyiova S, Neumann J, Samoilova VE, Boecker W, Kristek F. The role of arterial smooth muscle in vasorelaxation. Biochem Biophys Res Commun. 2008;
- 101. Ignarro LJ, Buga GM, Wood KS, Byrns RE, Chaudhuri G. Endothelium-derived relaxing factor produced and released from artery and vein is nitric oxide. Proc Natl Acad Sci U S A. 1987;
- 102. Palmer RMJ, Ferrige AG, Moncada S. Nitric oxide release accounts for the biological activity of endothelium-derived relaxing factor. Nature. 1987;
- 103. Stuehr DJ, Santolini J, Wang ZQ, Wei CC, Adak S. Update on mechanism and catalytic regulation in the NO synthases. Journal of Biological Chemistry. 2004.
- 104. Daff S. NO synthase: Structures and mechanisms. Nitric Oxide Biology and Chemistry. 2010.
- 105. Gorren ACF, Mayer B. Nitric-oxide synthase: A cytochrome P450 family foster child.

Biochimica et Biophysica Acta - General Subjects. 2007.

- 106. Drapier JC, Hibbs JB. Differentiation of murine macrophages to express nonspecific cytotoxicity for tumor cells results in L-arginine-dependent inhibition of mitochrondrial iron-sulfur enzymes in the macrophage effector cells. J Immunol. 1988;
- 107. Ding AH, Nathan CF, Stuehr DJ. Release of reactive nitrogen intermediates and reactive oxygen intermediates from mouse peritoneal macrophages. Comparison of activating cytokines and evidence for independent production. J Immunol. 1988;
- 108. Marietta MA, Yoon PS, Iyengar R, Leaf CD, Wishnok JS. Macrophage Oxidation of L-Arginine to Nitrite and Nitrate: Nitric Oxide Is an Intermediate. Biochemistry. 1988;
- 109. Bredt DS, Snyder SH. Isolation of nitric oxide synthetase, a calmodulin-requiring enzyme. Proc Natl Acad Sci U S A. 1990;
- Mayer B, Schmidt K, Humbert P, Böhme E. Biosynthesis of endothelium-derived relaxing factor: A cytosolic enzyme in porcine aortic endothelial cells Ca2+dependently converts L-arginine into an activator of soluble guanylyl cyclase. Biochem Biophys Res Commun. 1989;
- 111. Derbyshire ER, Marletta MA. Structure and regulation of soluble guanylate cyclase. Annu Rev Biochem. 2012;
- 112. Montfort WR, Wales JA, Weichsel A. Structure and Activation of Soluble Guanylyl Cyclase, the Nitric Oxide Sensor. Antioxidants and Redox Signaling. 2017.
- 113. Koesling D, Russwurm M, Mergia E, Mullershausen F, Friebe A. Nitric oxide-sensitive guanylyl cyclase: Structure and regulation. In: Neurochemistry International. 2004.
- 114. Budworth J, Meillerais S, Charles I, Powell K. Tissue distribution of the human soluble guanylate cyclases. Biochem Biophys Res Commun. 1999;
- 115. Winger JA, Marletta MA. Expression and characterization of the catalytic domains of soluble guanylate cyclase: Interaction with the heme domain. Biochemistry. 2005;
- 116. Münzel T, Feil R, Mülsch A, Lohmann SM, Hofmann F, Walter U. Physiology and Pathophysiology of Vascular Signaling Controlled by Cyclic Guanosine 3',5'-Cyclic Monophosphate-Dependent Protein Kinase. Circulation. 2003.
- 117. Murad F. Nitric oxide and cyclic GMP in cell signaling and drug development. N Engl J Med. 2006;
- 118. Warner TD, Mitchell JA, Sheng H, Murad F. Effects of Cyclic GMP on Smooth Muscle Relaxation. Adv Pharmacol. 1994;
- 119. Childers KC, Garcin ED. Structure/function of the soluble guanylyl cyclase catalytic domain. Nitric Oxide. 2018 Jul 1;77:53–64.
- 120. Brady AJB, Warren JB, Poole-Wilson PA, Williams TJ, Harding SE. Nitric oxide attenuates cardiac myocyte contraction. Am J Physiol Hear Circ Physiol. 1993;
- 121. Shah AM, MacCarthy PA. Paracrine and autocrine effects of nitric oxide on myocardial function. Pharmacology and Therapeutics. 2000.
- 122. Rastaldo R, Pagliaro P, Cappello S, Penna C, Mancardi D, Westerhof N, et al. Nitric oxide and cardiac function. Life Sciences. 2007.
- 123. Liu VWT, Huang PL. Cardiovascular roles of nitric oxide: A review of insights from nitric oxide synthase gene disrupted mice. Cardiovascular Research. 2008.
- 124. Ziolo MT, Bers DM. The real estate of NOS signaling location, location, location. Circulation Research. 2003.
- 125. Mohan P, Sys SU, Brutsaert DL. Positive inotropic effect of nitric oxide in myocardium. Int J Cardiol. 1995;
- 126. Bendall JK, Damy T, Ratajczak P, Loyer X, Monceau V, Marty I, et al. Role of myocardial neuronal nitric oxide synthase-derived nitric oxide in β-adrenergic hyporesponsiveness after myocardial infarction-induced heart failure in rat. Circulation. 2004;
- 127. Damy T, Ratajczak P, Robidel E, Bendall JK, Oliviéro P, Boczkowski J, et al. Upregulation of cardiac nitric oxide synthase 1-derived nitric oxide after myocardial infarction in senescent rats. FASEB J. 2003;
- 128. Bolli R. Cardioprotective Function of Inducible Nitric Oxide Synthase and Role of Nitric

Oxide in Myocardial Ischemia and Preconditioning: An Overview of a Decade of Research. Journal of Molecular and Cellular Cardiology. 2001.

- 129. Burkard N, Williams T, Czolbe M, Blömer N, Panther F, Link M, et al. Conditional overexpression of neuronal nitric oxide synthase is cardioprotective in ischemia/reperfusion. Circulation. 2010;
- 130. Loyer X, Gómez AM, Milliez P, Fernandez-Velasco M, Vangheluwe P, Vinet L, et al. Cardiomyocyte overexpression of neuronal nitric oxide synthase delays transition toward heart failure in response to pressure overload by preserving calcium cycling. Circulation. 2008;
- 131. Jones SP, Bolli R. The ubiquitous role of nitric oxide in cardioprotection. Journal of Molecular and Cellular Cardiology. 2006.
- 132. Tziros C, Freedman J. The Many Antithrombotic Actions of Nitric Oxide. Curr Drug Targets. 2006 Oct 2;7(10):1243–51.
- 133. Vanhoutte PM. Endothelial dysfunction The first step toward coronary arteriosclerosis. Circulation Journal. 2009.
- 134. Mitchell JA, Ali F, Bailey L, Moreno L, Harrington LS. Role of nitric oxide and prostacyclin as vasoactive hormones released by the endothelium. In: Experimental Physiology. 2008.
- 135. Radomski MW, Palmer RMJ, Moncada S. The role of nitric oxide and cGMP in platelet adhesion to vascular endothelium. Biochem Biophys Res Commun. 1987;
- 136. Vanhoutte PM, Shimokawa H, Tang EHC, Feletou M. Endothelial dysfunction and vascular disease. Acta Physiologica. 2009.
- 137. Motley ED, Eguchi K, Patterson MM, Palmer PD, Suzuki H, Eguchi S. Mechanism of endothelial nitric oxide synthase phosphorylation and activation by thrombin. Hypertension. 2007;
- 138. Margetic S. Inflammation and hemostasis. Biochem Medica. 2012;
- 139. Becker BF, Heindl B, Kupatt C, Zahler S. Endothelial function and hemostasis. In: Zeitschrift fur Kardiologie. 2000.
- 140. Bogdan C. Nitric oxide and the immune response. Nature Immunology. 2001.
- 141. Tripathi P, Tripathi P, Kashyap L, Singh V. The role of nitric oxide in inflammatory reactions. FEMS Immunology and Medical Microbiology. 2007.
- 142. Mulligan MS, Moncada S, Ward PA. Protective effects of inhibitors of nitric oxide synthase in immune complex-induced vasculitis. Br J Pharmacol. 1992;
- 143. Lanas A. Role of nitric oxide in the gastrointestinal tract. Arthritis Research and Therapy. 2008.
- 144. Davignon J, Ganz P. Role of endothelial dysfunction in atherosclerosis. Circulation. 2004.
- 145. Reinoso EMS. Endothelial dysfunction. In: Oxidative Stress and the Critically III Patient. 2012.
- 146. Banez MJ, Geluz MI, Chandra A, Hamdan T, Biswas OS, Bryan NS, et al. A systemic review on the antioxidant and anti-inflammatory effects of resveratrol, curcumin, and dietary nitric oxide supplementation on human cardiovascular health. Nutr Res. 2020 Jun 1;78:11–26.
- 147. Attur MG, Patel R, Thakker G, Vyas P, Levartovsky D, Patel P, et al. Differential antiinflammatory effects of immunosuppressive drugs: Cyclosporin, rapamycin and FK-506 on inducible nitric oxide synthase, nitric oxide, cyclooxygenase-2 and PGE 2 production. Inflamm Res 2000 491 [Internet]. 2000 [cited 2021 Jul 14];49(1):20–6. Available from: https://link.springer.com/article/10.1007/PL00000199
- 148. Xu W, Liu LZ, Loizidou M, Ahmed M, Charles IG. The role of nitric oxide in cancer. Cell Research. 2002.
- 149. Brüne B, Von Knethen A, Sandau KB. Nitric oxide and its role in apoptosis. European Journal of Pharmacology. 1998.
- 150. Rapozzi V, Della Pietra E, Bonavida B. Dual roles of nitric oxide in the regulation of tumor cell response and resistance to photodynamic therapy. Redox Biol. 2015;
- 151. Kröncke KD, Fehsel K, Kolb-Bachofen V. Nitric oxide: Cytotoxicity versus

cytoprotection - How, why, when, and where? Nitric Oxide - Biol Chem. 1997;

- 152. Taylor EL, Megson IL, Haslett C, Rossi AG. Nitric oxide: a key regulator of myeloid inflammatory cell apoptosis. Cell Death Differ 2003 104 [Internet]. 2003 Apr 28 [cited 2021 Jul 14];10(4):418–30. Available from: https://www.nature.com/articles/4401152
- 153. Fukumura D, Gohongi T, Kadambi A, Izumi Y, Ang J, Yun C-O, et al. Predominant role of endothelial nitric oxide synthase in vascular endothelial growth factor-induced angiogenesis and vascular permeability. Proc Natl Acad Sci [Internet]. 2001 Feb 27 [cited 2021 Jul 14];98(5):2604–9. Available from: https://www.pnas.org/content/98/5/2604
- 154. Kubes P, Suzuki M, Granger DN. Nitric oxide: An endogenous modulator of leukocyte adhesion. Proc Natl Acad Sci U S A. 1991;
- 155. Kunieda T, Minamino T, Miura K, Katsuno T, Tateno K, Miyauchi H, et al. Reduced nitric oxide causes age-associated impairment of circadian rhythmicity. Circ Res. 2008;
- 156. Drexler H, Hornig B. Endothelial Dysfunction in Human Disease. J Mol Cell Cardiol. 1999 Jan 1;31(1):51–60.
- 157. Siti HN, Kamisah Y, Kamsiah J. The role of oxidative stress, antioxidants and vascular inflammation in cardiovascular disease (a review). Vascular Pharmacology. 2015.
- 158. Cai H, Harrison DG. Endothelial dysfunction in cardiovascular diseases: The role of oxidant stress. Circulation Research. 2000.
- 159. Bonetti PO, Lerman LO, Lerman A. Endothelial dysfunction: A marker of atherosclerotic risk. Arteriosclerosis, Thrombosis, and Vascular Biology. 2003.
- 160. Chistiakov DA, Orekhov AN, Bobryshev Y V. Endothelial barrier and its abnormalities in cardiovascular disease. Frontiers in Physiology. 2015.
- 161. Hamczyk MR, del Campo L, Andrés V. Aging in the Cardiovascular System: Lessons from Hutchinson-Gilford Progeria Syndrome. Annu Rev Physiol [Internet]. 2018 Feb 10 [cited 2021 Feb 28];80(1):27–48. Available from:
- http://www.annualreviews.org/doi/10.1146/annurev-physiol-021317-121454
 162. Dantas AP, Jiménez-Altayó F, Vila E. Vascular Aging: Facts and Factors. Front Physiol [Internet]. 2012 Aug 10 [cited 2021 Feb 28];3(1):325. Available from: http://journal.frontiersin.org/article/10.3389/fphys.2012.00325/abstract
- 163. Savji N, Rockman CB, Skolnick AH, Guo Y, Adelman MA, Riles T, et al. Association between advanced age and vascular disease in different arterial territories: A population database of over 3.6 million subjects. J Am Coll Cardiol. 2013;
- 164. Santos-Parker JR, LaRocca TJ, Seals DR. Aerobic exercise and other healthy lifestyle factors that influence vascular aging. Adv Physiol Educ [Internet]. 2014 Dec [cited 2021 Feb 28];38(4):296–307. Available from: https://www.physiology.org/doi/10.1152/advan.00088.2014
- 165. Burke GL, Arnold AM, Bild DE, Cushman M, Fried LP, Newman A, et al. Factors Associated with Healthy Aging: The Cardiovascular Health Study. J Am Geriatr Soc [Internet]. 2001 Mar 1 [cited 2021 Feb 28];49(3):254–62. Available from: http://doi.wiley.com/10.1046/j.1532-5415.2001.4930254.x
- 166. Shakeri H, Lemmens K, Gevaert AB, De Meyer GRY, Segers VFM. Cellular senescence links aging and diabetes in cardiovascular disease [Internet]. Vol. 315, American Journal of Physiology - Heart and Circulatory Physiology. American Physiological Society; 2018 [cited 2021 Feb 28]. p. H448–62. Available from: www.ajpheart.org
- 167. Corella D, Ordovás JM. Aging and cardiovascular diseases: The role of gene-diet interactions. Vol. 18, Ageing Research Reviews. Elsevier Ireland Ltd; 2014. p. 53–73.
- 168. Hamczyk MR, Nevado RM, Barettino A, Fuster V, Andrés V. Biological Versus Chronological Aging: JACC Focus Seminar. Journal of the American College of Cardiology. 2020.
- 169. Chennupati R, Lamers WH, Koehler SE, De Mey JGR. Endothelium-dependent hyperpolarization-related relaxations diminish with age in murine saphenous arteries of both sexes. Br J Pharmacol. 2013;

- 170. Herrera MD, Mingorance C, Rodríguez-Rodríguez R, Alvarez de Sotomayor M. Endothelial dysfunction and aging: An update. Ageing Research Reviews. 2010.
- 171. Higashi Y, Kihara Y, Noma K. Endothelial dysfunction and hypertension in aging. Hypertension Research. 2012.
- 172. Katusic ZS. Mechanisms of endothelial dysfunction induced by aging: Role of arginase I. Circulation Research. 2007.
- Mahdi A, Kövamees O, Pernow J. Improvement in endothelial function in cardiovascular disease - Is arginase the target? International Journal of Cardiology. 2020.
- 174. Santhanam L, Christianson DW, Nyhan D, Berkowitz DE. Arginase and vascular aging. Journal of Applied Physiology. 2008.
- 175. Santos-Parker JR, Strahler TR, Bassett CJ, Bispham NZ, Chonchol MB, Seals DR. Curcumin supplementation improves vascular endothelial function in healthy middleaged and older adults by increasing nitric oxide bioavailability and reducing oxidative stress. Aging (Albany NY). 2017;
- 176. Yang C, DeMars KM, Candelario-Jalil E. Age-dependent decrease in adropin is associated with reduced levels of endothelial nitric oxide synthase and increased oxidative stress in the rat brain. Aging Dis. 2018;
- 177. Csiszar A, Ungvari Z, Edwards JG, Kaminski P, Wolin MS, Koller A, et al. Aginginduced phenotypic changes and oxidative stress impair coronary arteriolar function. Circ Res. 2002;
- 178. Lesniewski LA, Durrant JR, Connell ML, Folian BJ, Donate AJ, Seals DR. Salicylate treatment improves age-associated vascular endothelial dysfunction: Potential role of nuclear factor κb and forkhead box o phosphorylation. Journals Gerontol Ser A Biol Sci Med Sci. 2011;
- 179. Man AWC, Li H, Xia N. The role of sirtuin1 in regulating endothelial function, arterial remodeling and vascular aging. Frontiers in Physiology. 2019.
- 180. Yildiz O. Vascular smooth muscle and endothelial functions in aging. In: Annals of the New York Academy of Sciences. 2007.
- 181. Koga T, Takata Y, Kobayashi K, Takishita S, Yamashita Y, Fujishima M. Age and hypertension promote endothelium-dependent contractions to acetylcholine in the aorta of the rat. Hypertension. 1989;
- 182. Shi Y, Man RYK, Vanhoutte PM. Two isoforms of cyclooxygenase contribute to augmented endothelium- dependent contractions in femoral arteries of 1-year-old rats. Acta Pharmacol Sin. 2008;
- 183. Kim JW, Baek BS, Kim YK, Herlihy JT, Ikeno Y, Yu BP, et al. Gene expression of cyclooxygenase in the aging heart. Journals Gerontol Ser A Biol Sci Med Sci. 2001;
- 184. Tang EHC, Vanhoutte PM. Gene expression changes of prostanoid synthases in endothelial cells and prostanoid receptors in vascular smooth muscle cells caused by aging and hypertension. Physiol Genomics. 2008;
- 185. Numaguchi Y, Harada M, Osanai H, Hayashi K, Toki Y, Okumura K, et al. Altered gene expression of prostacyclin synthase and prostacyclin receptor in the thoracic aorta of spontaneously hypertensive rats. Cardiovasc Res. 1999;
- 186. Gluais P, Lonchampt M, Morrow JD, Vanhoutte PM, Feletou M. Acetylcholine-induced endothelium-dependent contractions in the SHR aorta: The Janus face of prostacyclin. Br J Pharmacol. 2005;
- 187. Chang E, Harley CB. Telomere length and replicative aging in human vascular tissues. Proc Natl Acad Sci U S A. 1995;
- Aubert G, Lansdorp PM. Telomeres and aging [Internet]. Vol. 88, Physiological Reviews. American Physiological Society; 2008 [cited 2021 Feb 28]. p. 557–79. Available from: www.prv.org
- 189. de Lange T. What I got wrong about shelterin. J Biol Chem. 2018 Jul 6;293(27):10453–6.
- 190. Zhang J, Rane G, Dai X, Shanmugam MK, Arfuso F, Samy RP, et al. Ageing and the telomere connection: An intimate relationship with inflammation. Vol. 25, Ageing

Research Reviews. Elsevier Ireland Ltd; 2016. p. 55–69.

- 191. Blackburn EH, Epel ES, Lin J. Human telomere biology: A contributory and interactive factor in aging, disease risks, and protection [Internet]. Vol. 350, Science. American Association for the Advancement of Science; 2015 [cited 2021 Feb 28]. p. 1193–8. Available from: http://science.sciencemag.org/
- 192. Blackburn EH. Telomere states and cell fates. Nature. 2000;408(6808):53-6.
- 193. Kim NW, Piatyszek MA, Prowse KR, Harley CB, West MD, Ho PLC, et al. Specific association of human telomerase activity with immortal cells and cancer. Science (80-) [Internet]. 1994 Dec 23 [cited 2021 Feb 28];266(5193):2011–5. Available from: http://science.sciencemag.org/
- 194. Liu Y, Bloom SI, Donato AJ. The role of senescence, telomere dysfunction and shelterin in vascular aging. Microcirculation [Internet]. 2019 Feb 1 [cited 2021 Feb 28];26(2):e12487. Available from: http://doi.wiley.com/10.1111/micc.12487
- 195. Lindsey J, McGill NI, Lindsey LA, Green DK, Cooke HJ. In vivo loss of telomeric repeats with age in humans. Mutat Res DNAging [Internet]. 1991 [cited 2021 Feb 28];256(1):45–8. Available from: https://pubmed.ncbi.nlm.nih.gov/1944386/
- Blackburn EH. Telomere states and cell fates [Internet]. Vol. 408, Nature. Nature Publishing Group; 2000 [cited 2021 Feb 28]. p. 53–6. Available from: www.nature.com53
- 197. Reichert S, Stier A. Does oxidative stress shorten telomeres in vivo? A review. Biology Letters. 2017.
- 198. Demissie S, Levy D, Benjamin EJ, Cupples LA, Gardner JP, Herbert A, et al. Insulin resistance, oxidative stress, hypertension, and leukocyte telomere length in men from the Framingham Heart Study. Aging Cell. 2006;5(4):325–30.
- 199. Fotiadou P, Henegariu O, Sweasy JB. DNA polymerase β interacts with TRF2 and induces telomere dysfunction in a murine mammary cell line. Cancer Res [Internet]. 2004 Jun 1 [cited 2021 Feb 28];64(11):3830–7. Available from: https://pubmed.ncbi.nlm.nih.gov/15172990/
- 200. Petersen S, Saretzki G, Von Zglinicki T. Preferential accumulation of single-stranded regions in telomeres of human fibroblasts. Exp Cell Res [Internet]. 1998 Feb 25 [cited 2021 Feb 28];239(1):152–60. Available from: https://pubmed.ncbi.nlm.nih.gov/9511733/
- 201. Kawanishi S, Oikawa S. Mechanism of telomere shortening by oxidative stress. In: Annals of the New York Academy of Sciences [Internet]. New York Academy of Sciences; 2004 [cited 2021 Feb 28]. p. 278–84. Available from: https://pubmed.ncbi.nlm.nih.gov/15247029/
- 202. Oikawa S, Kawanishi S. Site-specific DNA damage at GGG sequence by oxidative stress may accelerate telomere shortening. FEBS Lett [Internet]. 1999 Jun 25 [cited 2021 Feb 28];453(3):365–8. Available from: https://pubmed.ncbi.nlm.nih.gov/10405177/
- 203. Oikawa S, Tada-Oikawa S, Kawanishi S. Site-specific DNA damage at the GGG sequence by UVA involves acceleration of telomere shortening. Biochemistry [Internet]. 2001 Apr 17 [cited 2021 Mar 1];40(15):4763–8. Available from: https://pubmed.ncbi.nlm.nih.gov/11294644/
- 204. Von Zglinicki T. Oxidative stress shortens telomeres [Internet]. Vol. 27, Trends in Biochemical Sciences. Trends Biochem Sci; 2002 [cited 2021 Mar 1]. p. 339–44. Available from: https://pubmed.ncbi.nlm.nih.gov/12114022/
- 205. Mittal M, Siddiqui MR, Tran K, Reddy SP, Malik AB. Reactive oxygen species in inflammation and tissue injury [Internet]. Vol. 20, Antioxidants and Redox Signaling. Antioxid Redox Signal; 2014 [cited 2021 Mar 1]. p. 1126–67. Available from: https://pubmed.ncbi.nlm.nih.gov/23991888/
- 206. Wadley AJ, Veldhuijzen Van Zanten JJCS, Aldred S. The interactions of oxidative stress and inflammation with vascular dysfunction in ageing: The vascular health triad. Age (Omaha). 2013 Jun;35(3):705–18.
- 207. Beyne-Rauzy O, Prade-Houdellier N, Demur C, Recher C, Ayel J, Laurent G, et al.

Tumor necrosis factor-α inhibits hTERT gene expression in human myeloid normal and leukemic cells. Blood [Internet]. 2005 Nov [cited 2021 Mar 1];106(9):3200–5. Available from: https://pubmed.ncbi.nlm.nih.gov/16020509/

- 208. Rahman I. Oxidative stress, transcription factors and chromatin remodelling in lung inflammation. Biochem Pharmacol. 2002 Sep 1;64(5–6):935–42.
- 209. Lustig A, Liu HB, Metter EJ, An Y, Swaby MA, Elango P, et al. Telomere shortening, inflammatory cytokines, and anti-cytomegalovirus antibody Follow Distinct ageassociated trajectories in humans. Front Immunol [Internet]. 2017 Aug 24 [cited 2021 Mar 1];8(AUG):1027. Available from: /pmc/articles/PMC5609584/
- 210. Brazvan B, Farahzadi R, Mohammadi SM, Saheb SM, Shanehbandi D, Schmied L, et al. Key immune cell cytokines affects the telomere activity of cord blood cells in vitro. Adv Pharm Bull [Internet]. 2016 [cited 2021 Mar 1];6(2):153–61. Available from: /pmc/articles/PMC4961972/
- 211. Brouilette S, Singh RK, Thompson JR, Goodall AH, Samani NJ. White cell telomere length and risk of premature myocardial infarction. Arterioscler Thromb Vasc Biol [Internet]. 2003 May 1 [cited 2021 Mar 1];23(5):842–6. Available from: https://pubmed.ncbi.nlm.nih.gov/12649083/
- 212. Samani NJ, Boultby R, Butler R, Thompson JR, Goodall AH. Telomere shortening in atherosclerosis. Lancet [Internet]. 2001 Aug 11 [cited 2021 Mar 1];358(9280):472–3. Available from: https://pubmed.ncbi.nlm.nih.gov/11513915/
- Brouilette SW, Moore JS, McMahon AD, Thompson JR, Ford I, Shepherd J, et al. Telomere length, risk of coronary heart disease, and statin treatment in the West of Scotland Primary Prevention Study: a nested case-control study. Lancet [Internet]. 2007 Jan 13 [cited 2021 Mar 1];369(9556):107–14. Available from: https://pubmed.ncbi.nlm.nih.gov/17223473/
- 214. Wilson WRW, Herbert KE, Mistry Y, Stevens SE, Patel HR, Hastings RA, et al. Blood leucocyte telomere DNA content predicts vascular telomere DNA content in humans with and without vascular disease. Eur Heart J [Internet]. 2008 Nov [cited 2021 Mar 1];29(21):2689–94. Available from: https://pubmed.ncbi.nlm.nih.gov/18762552/
- 215. Ogami M, Ikura Y, Ohsawa M, Matsuo T, Kayo S, Yoshimi N, et al. Telomere Shortening in Human Coronary Artery Diseases. Arterioscler Thromb Vasc Biol [Internet]. 2004 Mar [cited 2021 Mar 1];24(3):546–50. Available from: https://pubmed.ncbi.nlm.nih.gov/14726417/
- 216. Matthews C, Gorenne I, Scott S, Figg N, Kirkpatrick P, Ritchie A, et al. Vascular smooth muscle cells undergo telomere-based senescence in human atherosclerosis: Effects of telomerase and oxidative stress. Circ Res [Internet]. 2006 Jul [cited 2021 Mar 1];99(2):156–64. Available from: https://pubmed.ncbi.nlm.nih.gov/16794190/
- 217. Brouilette SW, Whittaker A, Stevens SE, Van Der Harst P, Goodall AH, Samani NJ. Telomere length is shorter in healthy offspring of subjects with coronary artery disease: Support for the telomere hypothesis. Heart [Internet]. 2008 Apr [cited 2021 Mar 1];94(4):422–5. Available from: https://pubmed.ncbi.nlm.nih.gov/18347373/
- 218. Farzaneh-Far R, Cawthon RM, Na B, Browner WS, Schiller NB, Whooley MA. Prognostic value of leukocyte telomere length in patients with stable coronary artery disease: Data from the heart and soul study. Arterioscler Thromb Vasc Biol [Internet]. 2008 Jul 1 [cited 2021 Mar 1];28(7):1379–84. Available from: https://pubmed.ncbi.nlm.nih.gov/18467646/
- 219. Liu P, Zhang Y, Ma L. Telomere length and associated factors in older adults with hypertension [Internet]. Vol. 47, Journal of International Medical Research. SAGE Publications Ltd; 2019 [cited 2021 Mar 1]. p. 5465–74. Available from: https://pubmed.ncbi.nlm.nih.gov/31662013/
- 220. Lung FW, Ku CS, Kao WT. Telomere length may be associated with hypertension. J Hum Hypertens [Internet]. 2008 Mar [cited 2021 Mar 1];22(3):230–2. Available from: https://pubmed.ncbi.nlm.nih.gov/18046431/
- 221. Valdes AM, Andrew T, Gardner JP, Kimura M, Oelsner E, Cherkas LF, et al. Obesity, cigarette smoking, and telomere length in women. Lancet [Internet]. 2005 Aug 20

197

[cited 2021 Mar 1];366(9486):662–4. Available from: https://pubmed.ncbi.nlm.nih.gov/16112303/

- 222. Morlá M, Busquets X, Pons J, Sauleda J, MacNee W, Agustí AGN. Telomere shortening in smokers with and without COPD. Eur Respir J [Internet]. 2006 Mar [cited 2021 Mar 1];27(3):525–8. Available from: https://pubmed.ncbi.nlm.nih.gov/16507852/
- 223. Sadr M, Noori Mugahi SMH, Hassanzadeh G, Nadji SA, Kiani A, Abedini A, et al. Telomere Shortening in Blood Leukocytes of Patients with Chronic Obstructive Pulmonary Disease. Tanaffos [Internet]. 2015 [cited 2021 Mar 1];14(1):10–6. Available from: http://www.ncbi.nlm.nih.gov/pubmed/26221147
- 224. Córdoba-Lanús E, Cazorla-Rivero S, Espinoza-Jiménez A, de-Torres JP, Pajares MJ, Aguirre-Jaime A, et al. Telomere shortening and accelerated aging in COPD: findings from the BODE cohort. Respir Res. 2017 Apr 13;18(1):59.
- 225. Babizhayev MA, Savel'yeva EL, Moskvina SN, Yegorov YE. Telomere length is a biomarker of cumulative oxidative stress, biologic age, and an independent predictor of survival and therapeutic treatment requirement associated with smoking behavior. Am J Ther. 2011 Nov;18(6).
- 226. Córdoba-Lanús E, Cabrera-López C, Cazorla-Rivero S, Rodríguez-Pérez MC, Aguirre-Jaime A, Celli B, et al. Shorter telomeres in non-smoking patients with airflow limitation. Respir Med. 2018 May 1;138:123–8.
- 227. Nordfjäll K, Eliasson M, Stegmayr B, Melander O, Nilsson P, Roos G. Telomere length is associated with obesity parameters but with a gender difference. Obesity [Internet]. 2008 Dec [cited 2021 Mar 1];16(12):2682–9. Available from: https://pubmed.ncbi.nlm.nih.gov/18820651/
- 228. Lu D, Kassab GS. Role of shear stress and stretch in vascular mechanobiology. J R Soc Interface [Internet]. 2011 Oct 7 [cited 2021 Jul 14];8(63):1379. Available from: /pmc/articles/PMC3163429/
- 229. Bellien J, Iacob M, Gutierrez L, Isabelle M, Lahary A, Thuillez C, et al. Crucial role of NO and endothelium-derived hyperpolarizing factor in human sustained conduit artery flow-mediated dilatation. Hypertension. 2006;
- 230. Dumont O, Kauffenstein G, Guihot AL, Guérineau NC, Abraham P, Loufrani L, et al. Time-related alteration in flow-(shear stress-) mediated remodeling in resistance arteries from spontaneously hypertensive rats. Int J Hypertens. 2014;
- 231. Davies PF. Hemodynamic shear stress and the endothelium in cardiovascular pathophysiology. Nature Clinical Practice Cardiovascular Medicine. 2009.
- 232. Chistiakov DA, Orekhov AN, Bobryshev Y V. Effects of shear stress on endothelial cells: go with the flow. Acta Physiologica. 2017.
- 233. Stone PH, Coskun AU, Prati F. Ongoing Methodological Approaches to Improve the In Vivo Assessment of Local Coronary Blood Flow and Endothelial Shear Stress: The Devil Is in the Details. Journal of the American College of Cardiology. 2015.
- 234. Wu C, Huang RT, Kuo CH, Kumar S, Kim CW, Lin YC, et al. Mechanosensitive PPAP2B regulates endothelial responses to atherorelevant hemodynamic forces. Circ Res. 2015;
- 235. Dunn J, Qiu H, Kim S, Jjingo D, Hoffman R, Kim CW, et al. Flow-dependent epigenetic DNA methylation regulates endothelial gene expression and atherosclerosis. J Clin Invest. 2014;
- 236. Dunn J, Simmons R, Thabet S, Jo H. The role of epigenetics in the endothelial cell shear stress response and atherosclerosis. International Journal of Biochemistry and Cell Biology. 2015.
- 237. Matouk CC, Marsden PA. Epigenetic regulation of vascular endothelial gene expression. Circulation Research. 2008.
- 238. Zhao Y, Flavahan S, Leung SW, Xu A, Vanhoutte PM, Flavahan NA. Elevated pressure causes endothelial dysfunction in mouse carotid arteries by increasing local angiotensin signaling. Am J Physiol Hear Circ Physiol. 2015;
- 239. Schulz E, Gori T, Münzel T. Oxidative stress and endothelial dysfunction in

hypertension. Hypertension Research. 2011.

- 240. Brandes RP. Endothelial dysfunction and hypertension. Hypertension. 2014;
- 241. Wu BJ, Chen K, Shrestha S, Ong KL, Barter PJ, Rye KA. High-density lipoproteins inhibit vascular endothelial inflammation by increasing 3β-hydroxysteroid-[INCREMENT]24 reductase expression and inducing heme oxygenase-1. Circ Res. 2013;
- 242. Kuvin JT, Rämet ME, Patel AR, Pandian NG, Mendelsohn ME, Karas RH. A novel mechanism for the beneficial vascular effects of high-density lipoprotein cholesterol: Enhanced vasorelaxation and increased endothelial nitric oxide synthase expression. Am Heart J. 2002;
- 243. Noor R, Shuaib U, Wang CX, Todd K, Ghani U, Schwindt B, et al. High-density lipoprotein cholesterol regulates endothelial progenitor cells by increasing eNOS and preventing apoptosis. Atherosclerosis. 2007;
- 244. Bandeali S, Farmer J. High-density lipoprotein and atherosclerosis: The role of antioxidant activity. Curr Atheroscler Rep. 2012;
- 245. Tran-Dinh A, Diallo D, Delbosc S, Varela-Perez LM, Dang QB, Lapergue B, et al. HDL and endothelial protection. British Journal of Pharmacology. 2013.
- 246. Lüscher TF, Landmesser U, Von Eckardstein A, Fogelman AM. High-density lipoprotein: Vascular protective effects, dysfunction, and potential as therapeutic target. Circulation Research. 2014.
- 247. Kaufmann PA, Gnecchi-Ruscone T, Schäfers KP, Lüscher TF, Camici PG. Low density lipoprotein cholesterol and coronary microvascular dysfunction in hypercholesterolemia. J Am Coll Cardiol. 2000;
- 248. Hogg N, Kalyanaraman B, Joseph J, Struck A, Parthasarathy S. Inhibition of lowdensity lipoprotein oxidation by nitric oxide Potential role in atherogenesis. FEBS Lett. 1993;
- 249. August M, Wingerter O, Oelze M, Wenzel P, Kleschyov AL, Daiber A, et al. Mechanisms underlying dysfunction of carotid arteries in genetically hyperlipidemic rabbits. Nitric Oxide - Biol Chem. 2006;
- 250. Mihaylova B, Emberson J, Blackwell L, Keech A, Simes J, Barnes EH, et al. The effects of lowering LDL cholesterol with statin therapy in people at low risk of vascular disease: Meta-analysis of individual data from 27 randomised trials. Lancet. 2012;
- 251. Speer T, Owala FO, Holy EW, Zewinger S, Frenzel FL, Stähli BE, et al. Carbamylated low-density lipoprotein induces endothelial dysfunction. Eur Heart J. 2014;
- 252. Carracedo J, Merino A, Briceño C, Soriano S, Buendía P, CaUeros L, et al. Carbamylated low-density lipoprotein induces oxidative stress and accelerated senescence in human endothelial progenitor cells. FASEB J. 2011;
- 253. Badar A, Arif Z, Alam K. Role of Carbamylated Biomolecules in Human Diseases. IUBMB Life. 2018.
- 254. Trpkovic A, Resanovic I, Stanimirovic J, Radak D, Mousa SA, Cenic-Milosevic D, et al. Oxidized low-density lipoprotein as a biomarker of cardiovascular diseases. Critical Reviews in Clinical Laboratory Sciences. 2015.
- 255. Blair A, Shaul PW, Yuhanna IS, Conrad PA, Smart EJ. Oxidized low density lipoprotein displaces endothelial nitric-oxide synthase (eNOS) from plasmalemmal caveolae and impairs eNOS activation. J Biol Chem. 1999;
- 256. Ahsan A, Han G, Pan J, Liu S, Padhiar AA, Chu P, et al. Phosphocreatine protects endothelial cells from oxidized low-density lipoprotein-induced apoptosis by modulating the PI3K/Akt/eNOS pathway. Apoptosis. 2015;
- 257. Sydow K, Münzel T. ADMA and oxidative stress. Atheroscler Suppl. 2003;
- 258. Bouras G, Deftereos S, Tousoulis D, Giannopoulos G, Chatzis G, Tsounis D, et al. Asymmetric Dimethylarginine (ADMA): A Promising Biomarker for Cardiovascular Disease? Curr Top Med Chem. 2013;
- 259. Wu F, Szczepaniak WS, Shiva S, Liu H, Wang Y, Wang L, et al. Nox2-dependent glutathionylation of endothelial NOS leads to uncoupled superoxide production and endothelial barrier dysfunction in acute lung injury. Am J Physiol Lung Cell Mol 199

Physiol. 2014;

- 260. Daiber A, Xia N, Steven S, Oelze M, Hanf A, Kröller-Schön S, et al. New therapeutic implications of endothelial nitric oxide synthase (eNOS) function/dysfunction in cardiovascular disease. International Journal of Molecular Sciences. 2019.
- 261. Montezano AC, Touyz RM. Reactive oxygen species and endothelial function Role of nitric oxide synthase uncoupling and nox family nicotinamide adenine dinucleotide phosphate oxidases. Basic and Clinical Pharmacology and Toxicology. 2012.
- 262. Virdis A, Bacca A, Colucci R, Duranti E, Fornai M, Materazzi G, et al. Endothelial dysfunction in small arteries of essential hypertensive patients: Role of cyclooxygenase-2 in oxidative stress generation. Hypertension. 2013;
- 263. Masi S, Uliana M, Virdis A. Angiotensin II and vascular damage in hypertension: Role of oxidative stress and sympathetic activation. Vascular Pharmacology. 2019.
- 264. Wong MSK, Vanhoutte PM. COX-mediated endothelium-dependent contractions: From the past to recent discoveries. Acta Pharmacologica Sinica. 2010.
- 265. Tang EHC, Leung FP, Huang Y, Feletou M, So KF, Man RYK, et al. Calcium and reactive oxygen species increase in endothelial cells in response to releasers of endothelium-derived contracting factor. Br J Pharmacol. 2007;
- 266. Fukai T, Ushio-Fukai M. Superoxide dismutases: Role in redox signaling, vascular function, and diseases. Antioxidants and Redox Signaling. 2011.
- 267. Schulz E, Wenzel P, Münzel T, Daiber A. Mitochondrial redox signaling: Interaction of mitochondrial reactive oxygen species with other sources of oxidative stress. Antioxidants and Redox Signaling. 2014.
- 268. Freed JK, Gutterman DD. Mitochondrial reactive oxygen species and vascular function: Less is more. Arteriosclerosis, Thrombosis, and Vascular Biology. 2013.
- 269. Kadlec AO, Beyer AM, Ait-Aissa K, Gutterman DD. Mitochondrial signaling in the vascular endothelium: beyond reactive oxygen species. Basic Research in Cardiology. 2016.
- 270. Rowlands DJ, Chapple S, Siow RCM, Mann GE. Equol-stimulated mitochondrial reactive oxygen species activate endothelial nitric oxide synthase and redox signaling in endothelial cells: Roles for F-actin and GPR30. Hypertension. 2011;
- 271. Macarthur H, Westfall TC, Wilken GH. Oxidative stress attenuates NO-induced modulation of sympathetic neurotransmission in the mesenteric arterial bed of spontaneously hypertensive rats. Am J Physiol Hear Circ Physiol. 2008;
- 272. Förstermann U, Xia N, Li H. Roles of vascular oxidative stress and nitric oxide in the pathogenesis of atherosclerosis. Circulation Research. 2017.
- 273. Priviero FBM, Zemse SM, Teixeira CE, Webb RC. Oxidative stress impairs vasorelaxation induced by the soluble guanylyl cyclase activator BAY 41-2272 in spontaneously hypertensive rats. Am J Hypertens. 2009;
- 274. Panth N, Paudel KR, Parajuli K. Reactive Oxygen Species: A Key Hallmark of Cardiovascular Disease. Adv Med. 2016;
- 275. Zhang Y, Janssens SP, Wingler K, Schmidt HHHW, Moens AL. Modulating endothelial nitric oxide synthase: A new cardiovascular therapeutic strategy. American Journal of Physiology - Heart and Circulatory Physiology. 2011.
- 276. Dulce RA, Schulman IH, Hare JM. S-glutathionylation: A redox-sensitive switch participating in nitroso-redox balance. Circulation Research. 2011.
- 277. Pham-Huy LA, He H, Pham-Huy C. Free radicals, antioxidants in disease and health. International Journal of Biomedical Science. 2008.
- 278. Sies H, Berndt C, Jones DP. Oxidative Stress. https://doi.org/101146/annurevbiochem-061516-045037 [Internet]. 2017 Jun 27 [cited 2021 Jul 14];86:715–48. Available from: https://www.annualreviews.org/doi/abs/10.1146/annurev-biochem-061516-045037
- 279. Marrocco I, Altieri F, Peluso I. Measurement and Clinical Significance of Biomarkers of Oxidative Stress in Humans. Oxid Med Cell Longev. 2017 Jun 18;2017.
- 280. Wu G, Fang YZ, Yang S, Lupton JR, Turner ND. Glutathione Metabolism and Its Implications for Health. Journal of Nutrition. 2004.

- 281. Lu SC. Glutathione synthesis. Biochimica et Biophysica Acta General Subjects. 2013.
- 282. Meister A, Anderson ME. Glutathione. Annu Rev Biochem. 1983;
- 283. Pompella A, Corti A. Editorial: The changing faces of glutathione, a cellular protagonist. Frontiers in Pharmacology. 2015.
- 284. Rossi R, Dalle-Donne I, Milzani A, Giustarini D. Oxidized forms of glutathione in peripheral blood as biomarkers of oxidative stress. Clin Chem. 2006;
- 285. Margaritelis N V., Veskoukis AS, Paschalis V, Vrabas IS, Dipla K, Zafeiridis A, et al. Blood reflects tissue oxidative stress: A systematic review. Biomarkers. 2015.
- 286. Giustarini D, Colombo G, Garavaglia ML, Astori E, Portinaro NM, Reggiani F, et al. Assessment of glutathione/glutathione disulphide ratio and S-glutathionylated proteins in human blood, solid tissues, and cultured cells. Free Radic Biol Med. 2017 Nov 1;112:360–75.
- Asensi M, Sastre J, Pallardo F V., Lloret A, Lehner M, Garcia-De-La Asuncion J, et al.
 [23] Ratio of reduced to oxidized glutathione as indicator of oxidative stress status and DNA damage. Methods Enzymol. 1999 Jan 1;299:267–76.
- 288. Niacin and Oxidative Stress: A Mini-Review | ClinMed International Library | Journal of Nutritional Medicine and Diet Care | [Internet]. [cited 2021 Dec 30]. Available from: https://clinmedjournals.org/articles/jnmdc/journal-of-nutritional-medicine-and-diet-care-jnmdc-2-014.php?jid=jnmdc
- 289. Shimizu H, Kiyohara Y, Kato I, Kitazono T, Tanizaki Y, Kubo M, et al. Relationship Between Plasma Glutathione Levels and Cardiovascular Disease in a Defined Population. Stroke [Internet]. 2004 Sep 1 [cited 2021 Aug 22];35(9):2072–7. Available from: https://www.ahajournals.org/doi/abs/10.1161/01.STR.0000138022.86509.2d
- 290. Chaves FJ, Mansego ML, Blesa S, Gonzalez-Albert V, Jiménez J, Tormos MC, et al. Inadequate Cytoplasmic Antioxidant Enzymes Response Contributes to the Oxidative Stress in Human Hypertension. Am J Hypertens [Internet]. 2007 Jan 1 [cited 2021 Aug 22];20(1):62–9. Available from: https://academia.oup.com/cit/c2/121442
 - https://academic.oup.com/ajh/article/20/1/62/131443
- Redón J, Oliva MR, Tormos C, Giner V, Chaves J, Iradi A, et al. Antioxidant Activities and Oxidative Stress Byproducts in Human Hypertension. Hypertension [Internet].
 2003 May 1 [cited 2021 Aug 22];41(5):1096–101. Available from: https://www.ahajournals.org/doi/abs/10.1161/01.HYP.0000068370.21009.38
- Damy T, Kirsch M, Khouzami L, Caramelle P, Corvoisier P Le, Roudot-Thoraval F, et al. Glutathione Deficiency in Cardiac Patients Is Related to the Functional Status and Structural Cardiac Abnormalities. PLoS One [Internet]. 2009 Mar 25 [cited 2021 Aug 22];4(3):e4871. Available from:
 - https://journals.plos.org/plosone/article?id=10.1371/journal.pone.0004871
- 293. Kovacs V, Gasz B, Balatonyi B, Jaromi L, Kisfali P, Borsiczky B, et al. Polymorphisms in glutathione S-transferase are risk factors for perioperative acute myocardial infarction after cardiac surgery: a preliminary study. Mol Cell Biochem 2014 3891 [Internet]. 2014 Jan 17 [cited 2021 Aug 23];389(1):79–84. Available from: https://link.springer.com/article/10.1007/s11010-013-1929-7
- 294. Lapenna D, Gioia S de, Ciofani G, Mezzetti A, Ucchino S, Calafiore AM, et al. Glutathione-Related Antioxidant Defenses in Human Atherosclerotic Plaques. Circulation [Internet]. 1998 May 19 [cited 2021 Aug 23];97(19):1930–4. Available from: https://www.ahajournals.org/doi/abs/10.1161/01.CIR.97.19.1930
- 295. Blankenberg S, Rupprecht HJ, Bickel C, Torzewski M, Hafner G, Tiret L, et al. Glutathione Peroxidase 1 Activity and Cardiovascular Events in Patients with Coronary Artery Disease. http://dx.doi.org/101056/NEJMoa0305355 [Internet]. 2009 Oct 7 [cited 2021 Aug 23];349(17):1605–13. Available from: https://www.nejm.org/doi/full/10.1056/NEJMoa030535
- 296. Kugiyama K, Ohgushi M, Motoyama T, Hirashima O, Soejima H, Misumi K, et al. Intracoronary Infusion of Reduced Glutathione Improves Endothelial Vasomotor Response to Acetylcholine in Human Coronary Circulation. Circulation [Internet]. 1998 201

Jun 16 [cited 2021 Aug 23];97(23):2299–301. Available from: https://www.ahajournals.org/doi/abs/10.1161/01.CIR.97.23.2299

- 297. Robaczewska J, Kedziora-Kornatowska K, Kozakiewicz M, Zary-Sikorska E, Pawluk H, Pawliszak W, et al. ROLE OF GLUTATHIONE METABOLISM AND GLUTATHIONE-RELATED ANTIOXIDANT DEFENSE SYSTEMS IN HYPERTENSION. J Physiol Pharmacol [Internet]. 2016 [cited 2021 Jul 19];67:331–7. Available from: www.jpp.krakow.pl
- 298. Shahid SU, Shabana, Humphries S. The SNP rs10911021 is associated with oxidative stress in coronary heart disease patients from Pakistan. Lipids Heal Dis 2018 171 [Internet]. 2018 Jan 5 [cited 2021 Aug 23];17(1):1–6. Available from: https://link.springer.com/articles/10.1186/s12944-017-0654-8
- 299. Hamanishi T, Furuta H, Kato H, Doi A, Tamai M, Shimomura H, et al. Functional Variants in the Glutathione Peroxidase-1 (GPx-1) Gene Are Associated With Increased Intima-Media Thickness of Carotid Arteries and Risk of Macrovascular Diseases in Japanese Type 2 Diabetic Patients. Diabetes [Internet]. 2004 Sep 1 [cited 2021 Aug 23];53(9):2455–60. Available from: https://diabetes.diabetesiournals.org/content/53/9/2455
- Rybka J, Kupczyk D, Ke dziora-Kornatowska K, Motyl J, Czuczejko J,
 Szewczyk-Golec K, et al. Glutathione-Related Antioxidant Defense System in Elderly
 Patients Treated for Hypertension. 2010;
- 301. Dalle-Donne I, Rossi R, Giustarini D, Milzani A, Colombo R. Protein carbonyl groups as biomarkers of oxidative stress. Clin Chim Acta. 2003 Mar 1;329(1–2):23–38.
- 302. Berlett BS, Stadtman ER. Protein Oxidation in Aging, Disease, and Oxidative Stress *. J Biol Chem [Internet]. 1997 Aug 15 [cited 2021 Aug 18];272(33):20313–6. Available from: http://www.jbc.org/article/S0021925818387283/fulltext
- 303. M C, E B, ER S. Human studies related to protein oxidation: protein carbonyl content as a marker of damage. Free Radic Res [Internet]. 2000 Nov 1 [cited 2021 Aug 18];33 Suppl(SUPPL.):S99-108. Available from: https://europepmc.org/article/med/11191280
- 304. Beal MF. Oxidatively modified proteins in aging and disease,. Free Radic Biol Med. 2002 May 1;32(9):797–803.
- 305. Descamps-Latscha B, Witko-Sarsat V, Nguyen-Khoa T, Nguyen AT, Gausson V, Mothu N, et al. Advanced oxidation protein products as risk factors for atherosclerotic cardiovascular events in nondiabetic predialysis patients. Am J Kidney Dis. 2005 Jan 1;45(1):39–47.
- 306. Mocatta TJ, Pilbrow AP, Cameron VA, Senthilmohan R, Frampton CM, Richards AM, et al. Plasma Concentrations of Myeloperoxidase Predict Mortality After Myocardial Infarction. J Am Coll Cardiol. 2007 May 22;49(20):1993–2000.
- 307. Semba RD, Ferrucci L, Sun K, Walston J, Varadhan R, Guralnik JM, et al. Oxidative Stress Is Associated with Greater Mortality in Older Women Living in the Community. J Am Geriatr Soc [Internet]. 2007 Sep 1 [cited 2021 Aug 24];55(9):1421–5. Available from: https://agsjournals.onlinelibrary.wiley.com/doi/full/10.1111/j.1532-5415.2007.01308.x
- 308. Poli G, Biasi F, Leonarduzzi G. Oxysterols in the pathogenesis of major chronic diseases. Vol. 1, Redox Biology. Elsevier B.V.; 2013. p. 125–30.
- 309. Brown AJ, Jessup W. Oxysterols and atherosclerosis. Vol. 142, Atherosclerosis. Elsevier; 1999. p. 1–28.
- 310. Miyoshi N, Iuliano L, Tomono S, Ohshima H. Implications of cholesterol autoxidation products in the pathogenesis of inflammatory diseases. Biochem Biophys Res Commun. 2014 Apr 11;446(3):702–8.
- 311. Johnson CB. Isolation of cholesterol oxidation products from animal fat using aminopropyl solid-phase extraction. J Chromatogr A. 1996 Jun 7;736(1–2):205–10.
- 312. Leonarduzzi G, Sottero B, Poli G. Oxidized products of cholesterol: dietary and metabolic origin, and proatherosclerotic effects (review). J Nutr Biochem. 2002 Dec 1;13(12):700–10.
- 313. Testa G, Rossin D, Poli G, Biasi F, Leonarduzzi G. Implication of oxysterols in chronic 202

H.A. Shokr, PhD Thesis, Aston University 2021

inflammatory human diseases. Biochimie. 2018 Oct 1;153:220-31.

- 314. Helmschrodt C, Becker S, Schröter J, Hecht M, Aust G, Thiery J, et al. Fast LC– MS/MS analysis of free oxysterols derived from reactive oxygen species in human plasma and carotid plaque. Clin Chim Acta. 2013 Oct 21;425:3–8.
- 315. Virginio VWM, Nunes VS, Moura FA, Menezes FH, Andreollo NA, Rogerio F, et al. Arterial tissue and plasma concentration of enzymatic-driven oxysterols are associated with severe peripheral atherosclerotic disease and systemic inflammatory activity. http://dx.doi.org/103109/107157622014992894 [Internet]. 2014 Feb 1 [cited 2021 Aug 22];49(2):199–203. Available from: https://www.tandfonline.com/doi/abs/10.3109/10715762.2014.992894
- Baila-Rueda L, Cenarro A, Lamiquiz-Moneo I, Mateo-Gallego R, Bea AM, Perez-Calahorra S, et al. Bile acid synthesis precursors in subjects with genetic hypercholesterolemia negative for LDLR/APOB/PCSK9/APOE mutations. Association with lipids and carotid atherosclerosis. J Steroid Biochem Mol Biol. 2017 May 1;169:226–33.
- 317. Barrett HL, Nitert MD, McIntyre HD, Callaway LK. Maternal lipids in pre-eclampsia: innocent bystander or culprit? https://doi.org/103109/106419552014946614 [Internet]. 2014 Nov 1 [cited 2021 Aug 22];33(4):508–23. Available from: https://www.tandfonline.com/doi/abs/10.3109/10641955.2014.946614
- 318. Williamson RD, McCarthy C, McCarthy FP, Kenny LC. Oxidative stress in preeclampsia; have we been looking in the wrong place? Pregnancy Hypertens An Int J Women's Cardiovasc Heal [Internet]. 2017 Jan 18 [cited 2021 Aug 22];8:1–5. Available from: https://cora.ucc.ie/handle/10468/5202
- 319. Moon JY, Moon MH, Kim KT, Jeong DH, Kim YN, Chung BC, et al. Cytochrome P450-mediated metabolic alterations in preeclampsia evaluated by quantitative steroid signatures. J Steroid Biochem Mol Biol. 2014 Jan 1;139:182–91.
- 320. Ferreira GS, Pinto PR, Iborra RT, Del Bianco V, Santana MFM, Nakandakare ER, et al. Aerobic Exercise Training Selectively Changes Oxysterol Levels and Metabolism Reducing Cholesterol Accumulation in the Aorta of Dyslipidemic Mice. Front Physiol. 2017 Sep 5;0(SEP):644.
- 321. Poirier J, Cockell KA, Ratnayake WMN, Scoggan KA, Hidiroglou N, Gagnon C, et al. Antioxidant Supplements Improve Profiles of Hepatic Oxysterols and Plasma Lipids in Butter-fed Hamsters: https://doi.org/104137/NMIS3911 [Internet]. 2010 Feb 11 [cited 2021 Aug 22];3:NMI.S3911. Available from: https://journals.sagepub.com/doi/full/10.4137/NMI.S3911
- 322. Dumolt JH, Radhakrishnan SK, Moghadasian MH, Le K, Patel MS, Browne RW, et al. Maternal hypercholesterolemia enhances oxysterol concentration in mothers and newly weaned offspring but is attenuated by maternal phytosterol supplementation. J Nutr Biochem. 2018 Feb 1;52:10–7.
- 323. Dias IHK, Milic I, Lip GYH, Devitt A, Polidori MC, Griffiths HR. Simvastatin reduces circulating oxysterol levels in men with hypercholesterolaemia. Redox Biol. 2018 Jun 1;16:139–45.
- 324. Gianazza E, Brioschi M, Fernandez AM, Banfi C. Lipoxidation in cardiovascular diseases. Redox Biol. 2019 May 1;23:101119.
- 325. Craig JP, Nichols KK, Akpek EK, Caffery B, Dua HS, Joo CK, et al. TFOS DEWS II Definition and Classification Report. Ocular Surface. 2017.
- 326. Farrand KF, Fridman M, Stillman IÖ, Schaumberg DA. Prevalence of Diagnosed Dry Eye Disease in the United States Among Adults Aged 18 Years and Older. Am J Ophthalmol. 2017;
- 327. Paulsen AJ, Cruickshanks KJ, Fischer ME, Huang GH, Klein BEK, Klein R, et al. Dry eye in the beaver dam offspring study: Prevalence, risk factors, and health-related quality of life. Am J Ophthalmol. 2014;
- 328. Santaella RM, Fraunfelder FW. Ocular adverse effects associated with systemic medications: Recognition and management. Drugs. 2007.
- 329. Jaenen N, Baudouin C, Pouliquen P, Manni G, Figueiredo A, Zeyen T. Ocular

symptoms and signs with preserved and preservative-free glaucoma medications. Eur J Ophthalmol. 2007;

- 330. Engler D, Abraham V. Dry eyes. SA Pharmaceutical Journal. 2019.
- 331. Pflugfelder SC. Prevalence, burden, and pharmacoeconomics of dry eye disease. American Journal of Managed Care. 2008.
- 332. Yu J, Asche C V., Fairchild CJ. The economic burden of dry eye disease in the United States: A decision tree analysis. Cornea. 2011;
- 333. F. S, M. A, V.Y. B, I. J, K. L, F. M, et al. TFOS DEWS II Epidemiology Report. Ocul Surf. 2017;
- 334. Stern ME, Beuerman RW, Fox RI, Gao J, Mircheff AK, Pflugfelder SC. The pathology of dry eye: The interaction between the ocular surface and lacrimal glands. Cornea. 1998.
- 335. Belmonte C, Acosta MC, Gallar J. Neural basis of sensation in intact and injured corneas. Exp Eye Res. 2004;
- 336. Belmonte C, Nichols JJ, Cox SM, Brock JA, Begley CG, Bereiter DA, et al. TFOS DEWS II pain and sensation report. Ocular Surface. 2017.
- 337. Smith JA, Albenz J, Begley C, Caffery B, Nichols K, Schaumberg D, et al. The epidemiology of dry eye disease: Report of the epidemiology subcommittee of the international Dry Eye WorkShop (2007). In: Ocular Surface. 2007.
- 338. Wolffsohn JS, Arita R, Chalmers R, Djalilian A, Dogru M, Dumbleton K, et al. TFOS DEWS II Diagnostic Methodology report. Ocular Surface. 2017.
- 339. Hesse L, Johnson MW. Intravitreal injection of tissue plasminogen activitator: Four considerations [2]. Archives of Ophthalmology. 2001.
- 340. Chalmers RL, Begley CG, Caffery B. Validation of the 5-Item Dry Eye Questionnaire (DEQ-5): Discrimination across self-assessed severity and aqueous tear deficient dry eye diagnoses. Contact Lens Anterior Eye. 2010;
- 341. Rajagopalan K, Abetz L, Mertzanis P, Espindle D, Begley C, Chalmers R, et al. Comparing the discriminative validity of two generic and one disease-specific healthrelated quality of life measures in a sample of patients with dry eye. Value Heal. 2005;
- 342. Rubin GS, West SK, Muñoz B, Bandeen-Roche K, Zeger S, Schein O, et al. A comprehensive assessment of visual impairment in a population of older Americans: The SEE Study. Investig Ophthalmol Vis Sci. 1997;
- 343. Basak S. Physiology of the Eye. Essentials Ophthalmol. 2016;25–25.
- 344. Kawashima M. Systemic Health and Dry Eye. Invest Ophthalmol Vis Sci [Internet].
 2018 Nov 1 [cited 2021 Aug 27];59(14):DES138–42. Available from: https://doi.org/10.1167/iovs.17-23765
- 345. Choi HR, Lee JH, Lee HK, Song JS, Kim HC. Association Between Dyslipidemia and Dry Eye Syndrome Among the Korean Middle-Aged Population. Cornea [Internet]. 2020 Feb 1 [cited 2021 Aug 27];39(2):161–7. Available from: https://journals.lww.com/corneajrnl/Fulltext/2020/02000/Association_Between_Dyslipi demia_and_Dry_Eye.5.aspx
- 346. Achtsidis V, Eleftheriadou I, Kozanidou E, Voumvourakis KI, Stamboulis E, Theodosiadis PG, et al. Dry Eye Syndrome in Subjects With Diabetes and Association With Neuropathy. Diabetes Care [Internet]. 2014 Oct 1 [cited 2021 Aug 27];37(10):e210–1. Available from: https://care.diabetesjournals.org/content/37/10/e210
- 347. Methods to assess vascular endothelial function. 2000 Nov 3;
- 348. Ludmer PL, Selwyn AP, Shook TL, Wayne RR, Mudge GH, Alexander RW, et al. Paradoxical Vasoconstriction Induced by Acetylcholine in Atherosclerotic Coronary Arteries. N Engl J Med. 1986;
- 349. Suwaidi J Al, Hamasaki S, Higano ST, Nishimura RA, Holmes DR, Lerman A. Longterm follow-up of patients with mild coronary artery disease and endothelial dysfunction. Circulation. 2000;
- 350. Michelsen MM, Pena A, Mygind ND, Frestad D, Gustafsson I, Hansen HS, et al. Coronary Flow Velocity Reserve Assessed by Transthoracic Doppler: The iPOWER

Study: Factors Influencing Feasibility and Quality. J Am Soc Echocardiogr. 2016;

- 351. Fearon WF, Balsam LB, Farouque HMO, Robbins RC, Fitzgerald PJ, Yock PG, et al. Novel index for invasively assessing the coronary microcirculation. Circulation. 2003;
- 352. Martínez GJ, Yong ASC, Fearon WF, Ng MKC. The index of microcirculatory resistance in the physiologic assessment of the coronary microcirculation. Coronary Artery Disease. 2015.
- 353. Anderson TJ, Uehata A, Gerhard MD, Meredith IT, Knab S, Delagrange D, et al. Close relation of endothelial function in the human coronary and peripheral circulations. J Am Coll Cardiol. 1995;
- 354. Broxterman RM, Witman MA, Trinity JD, Groot HJ, Rossman MJ, Park SY, et al. Strong Relationship between Vascular Function in the Coronary and Brachial Arteries: A Clinical Coming of Age for the Updated Flow-Mediated Dilation Test? Hypertension. 2019;
- 355. Gori T, Dragoni S, Lisi M, Di Stolfo G, Sonnati S, Fineschi M, et al. Conduit Artery Constriction Mediated by Low Flow. A Novel Noninvasive Method for the Assessment of Vascular Function. J Am Coll Cardiol. 2008;
- 356. Bellien J, Joannides R, Iacob M, Arnaud P, Thuillez C. Evidence for a basal release of a cytochrome-related endothelium-derived hyperpolarizing factor in the radial artery in humans. Am J Physiol - Hear Circ Physiol. 2006;
- 357. Mather KJ, Verma S, Anderson TJ. Improved endothelial function with metformin in type 2 diabetes mellitus. J Am Coll Cardiol. 2001;
- 358. Bonetti PO, Pumper GM, Higano ST, Holmes DR, Kuvin JT, Lerman A. Noninvasive identification of patients with early coronary atherosclerosis by assessment of digital reactive hyperemia. J Am Coll Cardiol. 2004;
- 359. Budoff M, Ahmadi N, Kleis S, Akhtar W, McQuilkin G, Gul K, et al. Digital (Fingertip) Thermal Monitoring of Vascular Function: A Novel, Noninvasive, Nonimaging Test to Improve Traditional Cardiovascular Risk Assessment and Monitoring of Response to Treatments. In: Asymptomatic Atherosclerosis. 2011.
- 360. Goor DA, Sheffy J, Schnall RP, Arditti A, Caspi A, Bragdon EE, et al. Peripheral Arterial Tonometry: A Diagnostic Method for Detection of Myocardial Ischemia Induced during Mental Stress Tests: A Pilot Study. Clin Cardiol. 2004;
- 361. Lerman A, Zeiher AM. Endothelial function: Cardiac events. Circulation. 2005.
- Kjaer A, Meyer C, Nielsen FS, Parving HH, Hesse B. Dipyridamole, cold pressor test, and demonstration of endothelial dysfunction: A PET study of myocardial perfusion in diabetes. J Nucl Med. 2003;
- Johnson NP, Gould KL. Clinical evaluation of a new concept: Resting myocardial perfusion heterogeneity quantified by Markovian analysis of PET identifies coronary microvascular dysfunction and early atherosclerosis in 1,034 subjects. J Nucl Med. 2005;
- 364. Yilmaz A, Athanasiadis A, Mahrholdt H, Voehringer M, Ong P, Hill S, et al. Diagnostic value of perfusion cardiovascular magnetic resonance in patients with angina pectoris but normal coronary angiograms assessed by intracoronary acetylcholine testing. Heart. 2010;
- Blum A, Tamir S, Mualem K, Ben-Shushan RS, Keinan-Boker L, Paritsky M. Endothelial dysfunction is reversible in helicobacter pylori-positive subjects. Am J Med. 2011;
- 366. Joensen JB, Juul S, Henneberg E, Thomsen G, Ostergaard L, Lindholt JS. Can longterm antibiotic treatment prevent progression of peripheral arterial occlusive disease? A large, randomized, double-blinded, placebo-controlled trial. Atherosclerosis. 2008;
- 367. Beckman JA, Thakore A, Kalinowski BH, Harris JR, Creager MA. Radiation therapy impairs endothelium-dependent vasodilation in humans. J Am Coll Cardiol. 2001;
- 368. Chow AY, Chin C, Dahl G, Rosenthal DN. Anthracyclines cause endothelial injury in pediatric cancer patients: A pilot study. J Clin Oncol. 2006;
- 369. Hollenberg SM, Klein LW, Parrillo JE, Scherer M, Burns D, Tamburro P, et al. Changes in coronary endothelial function predict progression of allograft vasculopathy 205

after heart transplantation. J Hear Lung Transplant. 2004;

- 370. Thijssen DHJ, Black MA, Pyke KE, Padilla J, Atkinson G, Harris RA, et al. Assessment of flow-mediated dilation in humans: A methodological and physiological guideline. American Journal of Physiology - Heart and Circulatory Physiology. 2011.
- 371. Wilkinson IB, Webb DJ. Venous occlusion plethysmography in cardiovascular research: Methodology and clinical applications. British Journal of Clinical Pharmacology. 2001.
- 372. Patvardhan EA, Heffernan KS, Ruan JM, Soffler MI, Karas RH, Kuvin JT. Assessment of vascular endothelial function with peripheral arterial tonometry: Information at your fingertips? Cardiology in Review. 2010.
- 373. Kuvin JT, Patel AR, Sliney KA, Pandian NG, Rand WM, Udelson JE, et al. Peripheral vascular endothelial function testing as a noninvasive indicator of coronary artery disease. J Am Coll Cardiol. 2001;
- 374. Kharalkar N, Valvano JW, Naghavi M, Wei CL. Novel temperature based technique for measurement of endothelial dysfunction. In: Annual International Conference of the IEEE Engineering in Medicine and Biology Proceedings. 2003.
- 375. Dhindsa M, Sommerlad SM, DeVan AE, Barnes JN, Sugawara J, Ley O, et al. Interrelationships among noninvasive measures of postischemic macro- and microvascular reactivity. J Appl Physiol. 2008;
- 376. Rossi M, Carpi A, Di Maria C, Franzoni F, Galetta F, Santoro G. Post-ischaemic peak flow and myogenic flowmotion component are independent variables for skin postischaemic reactive hyperaemia in healthy subjects. Microvasc Res. 2007;
- 377. Ahmadi N, Nabavi V, Nuguri V, Hajsadeghi F, Flores F, Akhtar M, et al. Low fingertip temperature rebound measured by digital thermal monitoring strongly correlates with the presence and extent of coronary artery disease diagnosed by 64-slice multi-detector computed tomography. Int J Cardiovasc Imaging. 2009;
- 378. Gul KM, Ahmadi N, Wang Z, Jamieson C, Nasir K, Metcalfe R, et al. Digital thermal monitoring of vascular function: A novel tool to improve cardiovascular risk assessment. Vasc Med. 2009;
- 379. Ahmadi N, Usman N, Shim J, Nuguri V, Vasinrapee P, Hajsadeghi F, et al. Vascular dysfunction measured by fingertip thermal monitoring is associated with the extent of myocardial perfusion defect. J Nucl Cardiol. 2009;
- 380. Frolow M, Drozdz A, Kowalewska A, Nizankowski R, Chlopicki S. Comprehensive assessment of vascular health in patients; Towards endothelium-guided therapy. Pharmacological Reports. 2015.
- 381. Vanhoutte PM, Shimokawa H, Feletou M, Tang EHC. Endothelial dysfunction and vascular disease a 30th anniversary update. Acta Physiologica. 2017.
- 382. Panza JA, Garcia CE, Kilcoyne CM, Quyyumi AA, Cannon RO. Impaired endotheliumdependent vasodilation in patients with essential hypertension: Evidence that nitric oxide abnormality is not localized to a single signal transduction pathway. Circulation. 1995;
- 383. Versari D, Daghini E, Virdis A, Ghiadoni L, Taddei S. Endothelium-dependent contractions and endothelial dysfunction in human hypertension. British Journal of Pharmacology. 2009.
- Casino PR, Kilcoyne CM, Quyyumi AA, Hoeg JM, Panza JA. The role of nitric oxide in endothelium-dependent vasodilation of hypercholesterolemic patients. Circulation. 1993;
- 385. John S, Schneider MP, Delles C, Jacobi J, Schmieder RE. Lipid-independent effects of statins on endothelial function and bioavailability of nitric oxide in hypercholesterolemic patients. Am Heart J. 2005;
- Williams SB, Cusco JA, Roddy MA, Johnstone MT, Creager MA. Impaired nitric oxidemediated vasodilation in patients with non-insulin-dependent diabetes mellitus. J Am Coll Cardiol. 1996;
- 387. Mäkimattila S, Virkamäki A, Groop PH, Cockcroft J, Utriainen T, Fagerudd J, et al. Chronic hyperglycemia impairs endothelial function and insulin sensitivity via different 206

mechanisms insulin-dependent diabetes mellitus. Circulation. 1996;

- 388. Hadi HAR, Al Suwaidi JA. Endothelial dysfunction in diabetes mellitus. Vascular Health and Risk Management. 2007.
- 389. Baradaran A. Concepts towards endothelial dysfunction in diabetes mellitus. 2016;1(1):4–5.
- 390. Dikalov Ś, Itani H, Richmond B, Vergeade A, Jamshedur Rahman SM, Boutaud O, et al. Tobacco smoking induces cardiovascular mitochondrial oxidative stress, promotes endothelial dysfunction, and enhances hypertension. Am J Physiol Hear Circ Physiol. 2019;
- 391. Lynch J, Jin L, Richardson A, Conklin DJ. Tobacco Smoke and Endothelial Dysfunction: Role of Aldehydes? Curr Hypertens Rep. 2020;22(9).
- 392. Engin A. Endothelial dysfunction in obesity. In: Advances in Experimental Medicine and Biology. 2017.
- 393. Mauricio M, Aldasoro M, Ortega J, Vila J. Endothelial Dysfunction in Morbid Obesity. Curr Pharm Des. 2013;
- 394. Virdis A, Ghiadoni L, Cardinal H, Favilla S, Duranti P, Birindelli R, et al. Mechanisms responsible for endothelial dysfunction induced by fasting hyperhomocystinemia in normotensive subjects and patients with essential hypertension. J Am Coll Cardiol. 2001;
- 395. Obradovic M, Zaric BL, Haidara MA, Isenovic ER. Link between Homocysteine and Cardiovascular Diseases. Curr Pharmacol Reports. 2018;4(1):1–9.
- 396. Ungvari Z, Tarantini S, Kiss T, Wren JD, Giles CB, Griffin CT, et al. Endothelial dysfunction and angiogenesis impairment in the ageing vasculature. Nature Reviews Cardiology. 2018.
- 397. Tesauro M, Mauriello A, Rovella V, Annicchiarico-Petruzzelli M, Cardillo C, Melino G, et al. Arterial ageing: from endothelial dysfunction to vascular calcification. In: Journal of Internal Medicine. 2017.
- 398. Corban MT, Lerman LO, Lerman A. Endothelial dysfunction cardiovascular disease pathophysiology hidden in plain sight. Arteriosclerosis, Thrombosis, and Vascular Biology. 2019.
- 399. Halcox JPJ, Donald AE, Ellins E, Witte DR, Shipley MJ, Brunner EJ, et al. Endothelial function predicts progression of carotid intima-media thickness. Circulation. 2009;
- 400. Anderson TJ, Charbonneau F, Title LM, Buithieu J, Rose MS, Conradson H, et al. Microvascular function predicts cardiovascular events in primary prevention: Longterm results from the firefighters and their endothelium (FATE) study. Circulation. 2011;123(2):163–9.
- 401. Rossi R, Nuzzo A, Olaru AI, Origliani G, Modena MG. Endothelial function affects early carotid atherosclerosis progression in hypertensive postmenopausal women. J Hypertens. 2011;
- 402. Halcox JPJ, Schenke WH, Zalos G, Mincemoyer R, Prasad A, Waclawiw MA, et al. Prognostic value of coronary vascular endothelial dysfunction. Circulation. 2002;
- 403. Schindler TH, Hornig B, Buser PT, Olschewski M, Magosaki N, Pfisterer M, et al. Prognostic value of abnormal vasoreactivity of epicardial coronary arteries to sympathetic stimulation in patients with normal coronary angiograms. Arterioscler Thromb Vasc Biol. 2003;
- 404. DeFilippis AP, Young R, Carrubba CJ, McEvoy JW, Budoff MJ, Blumenthal RS, et al. An analysis of calibration and discrimination among multiple cardiovascular risk scores in a modern multiethnic cohort. Ann Intern Med. 2015;
- 405. J. W, C.A. L, P. A, Y. H, D.M. D, S.E. R. Predictive validity of common cardiovascular disease risk scores overall and in relation to race and gender. Circulation. 2016;
- 406. Gokce N, Keaney JF, Hunter LM, Watkins MT, Menzoian JO, Vita JA. Risk stratification for postoperative cardiovascular events via noninvasive assessment of endothelial function: A prospective study. Circulation. 2002;
- 407. Rossi R, Chiurlia E, Nuzzo A, Cioni E, Origliani G, Modena MG. Flow-mediated vasodilation and the risk of developing hypertension in healthy postmenopausal

207

women. J Am Coll Cardiol. 2004;

- 408. Neunteufl T, Heher S, Katzenschlager R, Wölfl G, Kostner K, Maurer G, et al. Late prognostic value of flow-mediated dilation in the brachial artery of patients with chest pain. Am J Cardiol. 2000;
- 409. Frick M, Suessenbacher A, Alber HF, Dichtl W, Ulmer H, Pachinger O, et al. Prognostic value of brachial artery endothelial function and wall thickness. J Am Coll Cardiol. 2005;
- 410. Shechter M, Issachar A, Marai I, Koren-Morag N, Freinark D, Shahar Y, et al. Longterm association of brachial artery flow-mediated vasodilation and cardiovascular events in middle-aged subjects with no apparent heart disease. Int J Cardiol. 2009;
- 411. Hirsch L, Shechter A, Feinberg MS, Koren-Morag N, Shechter M. The impact of early compared to late morning hours on brachial endothelial function and long-term cardiovascular events in healthy subjects with no apparent coronary heart disease. Int J Cardiol. 2011;
- 412. Yeboah J, Crouse JR, Hsu FC, Burke GL, Herrington DM. Brachial flow-mediated dilation predicts incident cardiovascular events in older adults: The cardiovascular health study. Circulation. 2007;
- 413. Yeboah J, Folsom AR, Burke GL, Johnson C, Polak JF, Post W, et al. Predictive value of brachial flow-mediated dilation for incident cardiovascular events in a population-based study: The multi-ethnic study of atherosclerosis. Circulation. 2009;
- 414. Matsuzawa Y, Kwon TG, Lennon RJ, Lerman LO, Lerman A. Prognostic value of flowmediated vasodilation in brachial artery and fingertip artery for cardiovascular events: A systematic review and meta-analysis. J Am Heart Assoc. 2015;
- 415. Shechter M, Shechter A, Koren-Morag N, Feinberg MS, Hiersch L. Usefulness of brachial artery flow-mediated dilation to predict long-term cardiovascular events in subjects without heart disease. Am J Cardiol. 2014;
- 416. Green DJ. Exercise training as vascular medicine: Direct impacts on the vasculature in humans. Exercise and Sport Sciences Reviews. 2009.
- 417. Wierzbicki AS. New directions in cardiovascular risk assessment: The role of secondary risk stratification markers. International Journal of Clinical Practice. 2012.
- 418. Lind L, Berglund L, Larsson A, Sundström J. Endothelial function in resistance and conduit arteries and 5-year risk of cardiovascular disease. Circulation. 2011;
- 419. Matsuzawa Y, Sugiyama S, Sugamura K, Nozaki T, Ohba K, Konishi M, et al. Digital Assessment of Endothelial Function and Ischemic Heart Disease in Women. J Am Coll Cardiol. 2010;
- 420. Rubinshtein R, Kuvin JT, Soffler M, Lennon RJ, Lavi S, Nelson RE, et al. Assessment of endothelial function by non-invasive peripheral arterial tonometry predicts late cardiovascular adverse events. Eur Heart J. 2010;
- 421. Piepoli MF, Hoes AW, Agewall S, Albus C, Brotons C, Catapano AL, et al. 2016 European guidelines on cardiovascular disease prevention in clinical practice. The Sixth Joint Task Force of the European Society of Cardiology and Other Societies on Cardiovascular Disease Prevention in Clinical Practice (constituted by representati. G Ital Cardiol (Rome). 2017;
- 422. Piepoli MF, Abreu A, Albus C, Ambrosetti M, Brotons C, Catapano AL, et al. Update on cardiovascular prevention in clinical practice: A position paper of the European Association of Preventive Cardiology of the European Society of Cardiology*. Eur J Prev Cardiol. 2020;
- 423. Greenland P, Alpert JS, Beller GA, Benjamin EJ, Budoff MJ, Fayad ZA, et al. 2010 ACCF/AHA guideline for assessment of cardiovascular risk in asymptomatic adults: Executive summary: A report of the American College of cardiology foundation/American Heart association task force on practice guidelines. Circulation. 2010;
- 424. Charakida M, Masi S, Lüscher TF, Kastelein JJP, Deanfield JE. Assessment of atherosclerosis: The role of flow-mediated dilatation. European Heart Journal. 2010.
- 425. Hamburg NM, Palmisano J, Larson MG, Sullivan LM, Lehman BT, Vasan RS, et al.

Relation of brachial and digital measures of vascular function in the community: The framingham heart study. Hypertension. 2011;

- 426. Schnabel RB, Schulz A, Wild PS, Sinning CR, Wilde S, Eleftheriadis M, et al. Noninvasive vascular function measurement in the community cross-sectional relations and comparison of methods. Circ Cardiovasc Imaging. 2011;
- 427. Kuvin JT, Patel AR, Sliney KA, Pandian NG, Sheffy J, Schnall RP, et al. Assessment of peripheral vascular endothelial function with finger arterial pulse wave amplitude. Am Heart J. 2003;
- 428. Araki T, Banchhor SK, Londhe ND, Ikeda N, Radeva P, Shukla D, et al. Reliable and Accurate Calcium Volume Measurement in Coronary Artery Using Intravascular Ultrasound Videos. J Med Syst. 2016;
- 429. Lorenz MW, Polak JF, Kavousi M, Mathiesen EB, Völzke H, Tuomainen TP, et al. Carotid intima-media thickness progression to predict cardiovascular events in the general population (the PROG-IMT collaborative project): A meta-analysis of individual participant data. Lancet. 2012;
- 430. Bittner M, Faes L, Boehni SC, Bachmann LM, Schlingemann RO, Schmid MK. Colour Doppler analysis of ophthalmic vessels in the diagnosis of carotic artery and retinal vein occlusion, diabetic retinopathy and glaucoma: systematic review of test accuracy studies. BMC Ophthalmol. 2016;
- 431. Jorzik JJ, Bindewald A, Dithmar S, Holz FG. Digital simultaneous fluorescein and indocyanine green angiography, autofluorescence, and red-free imaging with a solid-state laser-based confocal scanning laser ophthalmoscope. Retina. 2005;
- 432. Pournaras CJ, Riva CE. Retinal blood flow evaluation. Ophthalmologica. 2013.
- 433. Marmor MF, Ravin JG. Fluorescein angiography: Insight and serendipity a half century ago. Arch Ophthalmol. 2011;
- 434. Wolf S, Jung F, Kiesewetter H, Körber N, Reim M. Video fluorescein angiography: Method and clinical application. Graefe's Arch Clin Exp Ophthalmol. 1989;
- 435. Lira RPC, Oliveira CL de A, Marques MVRB, Silva AR, Pessoa C de C. Adverse reactions of fluorescein angiography: A prospective study. Arq Bras Oftalmol. 2007;
- 436. Yang Y, Kim S, Kim J. Fluorescent dots in fluorescein angiography and fluorescein leukocyte angiography using a scanning laser ophthalmoscope in humans. Ophthalmology. 1997;
- 437. Tanaka T, Muraoka K, Shimizu K. Fluorescein Fundus Angiography with Scanning Laser Ophthalmoscope: Visibility of Leukocytes and Platelets in Perifoveal Capillaries. Ophthalmology. 1991;
- 438. Almasi R, Vafaei A, Ghasemi Z, Ommani MR, Dehghani AR, Rabbani H. Registration of fluorescein angiography and optical coherence tomography images of curved retina via scanning laser ophthalmoscopy photographs. Biomed Opt Express. 2020;
- 439. Truffer F, Geiser M, Chappelet M-A, Strese H, Maître G, Amoos S, et al. Absolute retinal blood flowmeter using a laser Doppler velocimeter combined with adaptive optics. J Biomed Opt. 2020;
- 440. Petrig BL, Riva CE, Hayreh SS. Laser Doppler flowmetry and optic nerve head blood flow. Am J Ophthalmol. 1999;
- 441. Riva CE, Geiser M, Petrig BL. Ocular blood flow assessment using continuous laser Doppler flowmetry: Review Article. Acta Ophthalmologica. 2010.
- 442. Michelson G. Principle, validity, and reliability of scanning laser Doppler flowmetry. J Glaucoma. 1996;
- 443. Yaoeda K, Shirakashi M, Funaki S, Funaki H, Nakatsue T, Abe H. Measurement of microcirculation in the optic nerve head by laser speckle flowgraphy and scanning laser doppler flowmetry. Am J Ophthalmol. 2000;
- 444. Stalmans I, Vandewalle E, Anderson DR, Costa VP, Frenkel REP, Garhofer G, et al. Use of colour Doppler imaging in ocular blood flow research. Acta Ophthalmologica. 2011.
- 445. Dimitrova G, Kato S. Color Doppler Imaging of Retinal Diseases. Survey of Ophthalmology. 2010.

- 446. Fallon TJ, Maxwell DL, Kohner EM. Autoregulation of Retinal Blood Flow in Diabetic Retinopathy Measured by the Blue-light Entoptic Technique. Ophthalmology. 1987;
- 447. Petrig BL, Lorenz B, Riva CE. Effect of pupil aperture size on blood flow parameters measured by the blue field simulation technique. Investig Ophthalmol Vis Sci. 1996;
- 448. Yap MKH, Brown B. The repeatability of the noninvasive blue field entoptic phenomenon method for measuring macular capillary blood flow. Optom Vis Sci. 1994;
- 449. Maram J, Srinivas S, Sadda SR. Evaluating ocular blood flow. Indian Journal of Ophthalmology. 2017.
- 450. Bhan A, Bhargava J, Vernon SA, Armstrong S, Bhan K, Tong L, et al. Repeatability of ocular blood flow pneumotonometry. Ophthalmology. 2003;
- 451. Danielewska ME, Messner A, Werkmeister RM, Placek MM, Dos Santos VA, Rękas M, et al. Relationship between the parameters of corneal and fundus pulse signals acquired with a combined ultrasound and laser interferometry technique. Transl Vis Sci Technol. 2019;
- 452. Lim M, Sasongko MB, Ikram MK, Lamoureux E, Wang JJ, Wong TY, et al. Systemic Associations of Dynamic Retinal Vessel Analysis: A Review of Current Literature. Microcirculation. 2013.
- 453. Hardarson SH. Retinal oximetry. Acta Ophthalmol. 2013;
- 454. Hardarson SH, Harris A, Karlsson RA, Halldorsson GH, Kagemann L, Rechtman E, et al. Automatic retinal oximetry. Investig Ophthalmol Vis Sci. 2006;
- 455. Beach J. Pathway to Retinal Oximetry. Transl Vis Sci Technol. 2014;
- 456. Small artery remodeling is the most prevalent (earliest?) fo... : Journal of Hypertension [Internet]. [cited 2021 Jul 15]. Available from: https://journals.lww.com/jhypertension/Abstract/2001/05000/Small_artery_remodeling is the most prevalent.13.aspx
- 457. Laurent S, Boutouyrie P. The Structural Factor of Hypertension: Large and Small Artery Alterations. Circulation Research. 2015.
- 458. Sorop O, Olver TD, Van DeWouw J, Heinonen I, Van Duin RW, Duncker DJ, et al. Themicrocirculation: A key player in obesity-associated cardiovascular disease. Cardiovascular Research. 2017.
- 459. Wiernsperger N, Nivoit P, Bouskela E. Microcirculation in obesity: An unexplored domain. An Acad Bras Cienc. 2007;
- 460. Bagi Z, Broskova Z, Feher A. Obesity and Coronary Microvascular Disease Implications for Adipose Tissue-Mediated Remote Inflammatory Response. Curr Vasc Pharmacol. 2014;
- 461. van de Wouw J, Broekhuizen M, Sorop O, Joles JA, Verhaar MC, Duncker DJ, et al. Chronic Kidney Disease as a Risk Factor for Heart Failure With Preserved Ejection Fraction: A Focus on Microcirculatory Factors and Therapeutic Targets. Frontiers in Physiology. 2019.
- 462. Leone A, Landini L. Vascular Pathology from Smoking: Look at the Microcirculation! Curr Vasc Pharmacol. 2013;
- 463. Messner B, Bernhard D. Smoking and cardiovascular disease: Mechanisms of endothelial dysfunction and early atherogenesis. Arteriosclerosis, Thrombosis, and Vascular Biology. 2014.
- 464. Rizzoni D, Porteri E, Boari GEM, De Ciuceis C, Sleiman I, Muiesan ML, et al. Prognostic Significance of Small-Artery Structure in Hypertension. Circulation. 2003;
- 465. Mathiassen ON, Buus NH, Sihm I, Thybo NK, Mørn B, Schroeder AP, et al. Small artery structure is an independent predictor of cardiovascular events in essential hypertension. J Hypertens. 2007;
- 466. Muiesan ML, Rizzoni D, Salvetti M, Porteri E, Monteduro C, Guelfi D, et al. Structural changes in small resistance arteries and left ventricular geometry in patients with primary and secondary hypertension. J Hypertens. 2002;
- 467. Rizzoni D, Porteri E, Duse S, De Ciuceis C, Rosei CA, La Boria E, et al. Relationship between media-to-lumen ratio of subcutaneous small arteries and wall-to-lumen ratio

of retinal arterioles evaluated noninvasively by scanning laser Doppler flowmetry. J Hypertens [Internet]. 2012 [cited 2021 Jul 15];30(6):1169–75. Available from: https://journals.lww.com/jhypertension/Fulltext/2012/06000/Relationship_between_me dia_to_lumen_ratio_of.19.aspx

- 468. Liew G, Wang JJ. Retinal Vascular Signs: A Window to the Heart? Rev Española Cardiol (English Ed. 2011;
- 469. Liew G, Wang JJ, Mitchell P, Wong TY. Retinal vascular imaging: a new tool in microvascular disease research. Circulation. Cardiovascular imaging. 2008.
- 470. Akay F, Gündoğan FC, Yolcu U, Toyran S, Tunç E, Uzun S. Retinal structural changes in systemic arterial hypertension: An OCT study. Eur J Ophthalmol. 2016;
- 471. Touyz RM. Vascular remodeling, retinal arteries, and hypertension. Hypertension. 2007;50(4):603–4.
- 472. Al-Fiadh AH, Farouque O, Kawasaki R, Nguyen TT, Uddin N, Freeman M, et al. Retinal microvascular structure and function in patients with risk factors of atherosclerosis and coronary artery disease. Atherosclerosis. 2014;
- Zhang Y, Shi C, Chen Y, Wang W, Huang S, Han Z, et al. Retinal Structural and Microvascular Alterations in Different Acute Ischemic Stroke Subtypes. J Ophthalmol. 2020;
- 474. Zhao Y, Yang B, Xu A-D, Ruan Y-W, Xu Y, Hu H-L, et al. Retinal Microvascular Changes in Subtypes of Ischemic Stroke. Front Neurol. 2021;
- 475. Kurdoğlu Z. Retinal findings in cases of preeclampsia. Perinat J. 2011;
- 476. Gonul S, Yilmaz H, Gedik S, Ozturk BT, Oflaz AB, Sahin M. Evaluation of the choroidal thickness and retinal nerve fiber layer and visual fields in morbid obesity: Does bariatric surgery affect retinal structure and function? Indian J Ophthalmol. 2021;
- 477. Walsh JB. Hypertensive Retinopathy: Description, Classification, and Prognosis. Ophthalmology. 1982;
- 478. Wong TY, Klein R, Klein BEK, Tielsch JM, Hubbard L, Nieto FJ. Retinal microvascular abnormalities and their relationship with hypertension, cardiovascular disease, and mortality. Surv Ophthalmol. 2001;
- 479. Keith NM, Wagener HP, Barker NW. Some different types of essential hypertension: their course and prognosis. Am J Med Sci. 1974;
- 480. McGeechan K, Liew G, Macaskill P, Irwig L, Klein R, Klein BEK, et al. Meta-analysis: Retinal vessel caliber and risk for coronary heart disease. Annals of Internal Medicine. 2009.
- 481. Seidelmann SB, Claggett B, Bravo PE, Gupta A, Farhad H, Klein BE, et al. Retinal Vessel Calibers in Predicting Long-Term Cardiovascular Outcomes: The Atherosclerosis Risk in Communities Study. Circulation. 2016;
- 482. Cuspidi C, Sala C, Grassi G. Updated classification of hypertensive retinopathy: Which role for cardiovascular risk stratification? Journal of Hypertension. 2015.
- 483. Hubbard LD, Brothers RJ, King WN, Clegg LX, Klein R, Cooper LS, et al. Methods for evaluation of retinal microvascular abnormalities associated with hypertension/sclerosis in the Atherosclerosis Risk in Communities Study. Ophthalmology. 1999;
- 484. Knudtson MD, Lee KE, Hubbard LD, Wong TY, Klein R, Klein BEK. Revised formulas for summarizing retinal vessel diameters. Curr Eye Res. 2003;
- 485. Ahn SJ, Woo SJ, Park KH. Retinal and choroidal changes with severe hypertension and their association with visual outcome. Investig Ophthalmol Vis Sci. 2014;
- 486. Ruiz-Medrano J, Arias L, Ruiz-Moreno JM. Optical coherence tomography. In: Central Serous Chorioretinopathy. 2019.
- 487. Kawasaki R, Cheung N, Islam FMA, Klein R, Klein BEK, Cotch MF, et al. Is diabetic retinopathy related to subclinical cardiovascular disease? Ophthalmology. 2011;
- 488. Mimoun L, Massin P, Steg G. Retinal microvascularisation abnormalities and cardiovascular risk. Archives of Cardiovascular Diseases. 2009.
- 489. Wong TY, Kamineni A, Klein R, Sharrett AR, Klein BE, Siscovick DS, et al.

Quantitative retinal venular caliber and risk of cardiovascular disease in older persons: The cardiovascular health study. Arch Intern Med. 2006;

- 490. Takayama K, Kaneko H, Ito Y, Kataoka K, Iwase T, Yasuma T, et al. Novel Classification of Early-stage Systemic Hypertensive Changes in Human Retina Based on OCTA Measurement of Choriocapillaris. Sci Rep. 2018;
- 491. Fraser-Bell S, Symes R, Vaze A. Hypertensive eye disease: a review. Clinical and Experimental Ophthalmology. 2017.
- 492. DellaCroce JT, Vitale AT. Hypertension and the eye. Current Opinion in Ophthalmology. 2008.
- 493. Katsi V, Marketou M, Vlachopoulos C, Tousoulis D, Souretis G, Papageorgiou N, et al. Impact of arterial hypertension on the eye. Curr Hypertens Rep. 2012;
- 494. Konstantinidis L, Guex-Crosier Y. Hypertension and the eye. Current Opinion in Ophthalmology. 2016.
- 495. Ikram MK, Witteman JCM, Vingerling JR, Breteler MMB, Hofman A, De Jong PTVM. Retinal vessel diameters and risk of hypertension: The Rotterdam Study. Hypertension. 2006;
- 496. Wong TY, Shankar A, Klein R, Klein BEK, Hubbard LD. Prospective cohort study of retinal vessel diameters and risk of hypertension. Br Med J. 2004;
- 497. Smith W, Wang JJ, Wong TY, Rochtchina E, Klein R, Leeder SR, et al. Retinal arteriolar narrowing is associated with 5-year incident severe hypertension: The Blue Mountains Eye Study. Hypertension. 2004;
- 498. Kawasaki R, Cheung N, Wang JJ, Klein R, Klein BE, Cotch MF, et al. Retinal vessel diameters and risk of hypertension: The Multiethnic Study of Atherosclerosis. J Hypertens. 2009;
- 499. Lona G, Endes K, Köchli S, Infanger D, Zahner L, Hanssen H. Retinal vessel diameters and blood pressure progression in children. Hypertension. 2020;
- 500. He Y, Li SM, Kang MT, Liu LR, Li H, Wei SF, et al. Association between blood pressure and retinal arteriolar and venular diameters in Chinese early adolescent children, and whether the association has gender difference: A cross-sectional study. BMC Ophthalmol. 2018;
- 501. Köchli S, Endes K, Infanger D, Zahner L, Hanssen H. Obesity, blood pressure, and retinal vessels: A meta-analysis. Pediatrics. 2018.
- 502. Ritchie Mackenzie LD, Campbell NC, Murchie P. New NICE guidelines for hypertension. BMJ (Online). 2011.
- 503. Van Den Born BJH, Hulsman CAA, Hoekstra JBL, Schlingemann RO, Van Montfrans GA. Value of routine funduscopy in patients with hypertension: Systematic review. Br Med J. 2005;
- 504. Elliott WJ. Value of Routine Funduscopy in Patients With Hypertension: Systematic Review. Yearb Cardiol. 2006;
- 505. Mancia G, Fagard R, Narkiewicz K, Redon J, Zanchetti A, Böhm M, et al. 2013 ESH/ESC guidelines for the management of arterial hypertension: The Task Force for the management of arterial hypertension of the European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC). Eur Heart J. 2013;
- 506. Parr JC, Spears GFS. Mathematic relationships between the width of a retinal artery and the widths of its branches. Am J Ophthalmol. 1974;
- 507. Parr JC, Spears GFS. General caliber of the retinal arteries expressed as the equivalent width of the central retinal artery. Am J Ophthalmol. 1974;
- 508. Ikram MK, De Jong FJ, Vingerling JR, Witteman JCM, Hofman A, Breteler MMB, et al. Are retinal arteriolar or venular diameters associated with markers for cardiovascular disorders? The Rotterdam study. Investig Ophthalmol Vis Sci. 2004;
- 509. Knudtson MD, Klein BEK, Klein R, Wong TY, Hubbard LD, Lee KE, et al. Variation associated with measurement of retinal vessel diameters at different points in the pulse cycle. Br J Ophthalmol. 2004;
- 510. Aiello LP, Northrup JM, Keyt BA, Takagi H, Iwamoto MA. Hypoxic Regulation of Vascular Endothelial Growth Factor in Retinal Cells. Arch Ophthalmol. 1995;

- 511. Manalo DJ, Rowan A, Lavoie T, Natarajan L, Kelly BD, Ye SQ, et al. Transcriptional regulation of vascular endothelial cell responses to hypoxia by HIF-1. Blood. 2005;
- 512. Polak K, Schmetterer L, Riva CE. Influence of flicker frequency on flicker-induced changes of retinal vessel diameter. Investig Ophthalmol Vis Sci. 2002;
- 513. Seshadri S, Karimzad SE, Shokr H, Gherghel D. Retinal vascular function in asymptomatic individuals with a positive family history of cardiovascular disease. Acta Ophthalmol. 2018 Dec 1;96(8):e956–62.
- 514. Shokr H, Dias IHK, Gherghel D. Microvascular function and oxidative stress in adult individuals with early onset of cardiovascular disease. Sci Rep [Internet]. 2020;10(1):1–8. Available from: http://dx.doi.org/10.1038/s41598-020-60766-0
- 515. Karimzad SE, Shokr H, Gherghel D. Retinal and peripheral vascular function in healthy individuals with low cardiovascular risk. Microvasc Res. 2019 Nov 1;126.
- 516. Flammer AJ, Anderson T, Celermajer DS, Creager MA, Deanfield J, Ganz P, et al. The assessment of endothelial function: From research into clinical practice. Circulation. 2012;126(6):753–67.
- 517. Pemp B, Garhofer G, Weigert G, Karl K, Resch H, Wolzt M, et al. Reduced retinal vessel response to flicker stimulation but not to exogenous nitric oxide in type 1 diabetes. Investig Ophthalmol Vis Sci. 2009;
- 518. Les tendances en matière de décentralisation. 2007;30(12):1.
- 519. Reimann M, Weiss N, Ziemssen T. Different responses of the retinal and cutaneous microcirculation to transient dysmetabolic conditions. Atheroscler Suppl. 2015;
- 520. Nagel E, Vilser W, Lanzl I. Age, blood pressure, and vessel diameter as factors influencing the arterial retinal flicker response. Investig Ophthalmol Vis Sci. 2004;
- 521. Nagel E, Vilser W, Fink A, Riemer T, Lanzl I. Blood pressure effects on retinal vessel diameter and flicker response: A 1.5-year follow-up. Eur J Ophthalmol. 2006;
- 522. Anna M, Ewa PS, Katarzyna B, Anna S, Krzysztof S, Miłosz Piotr K, et al. Correlation between Flicker-Induced Retinal Vessel Vasodilatation and Plasma Biomarkers of Endothelial Dysfunction in Hypertensive Patients. Curr Eye Res. 2018;
- 523. Nagel E, Vilser W, Lanzl I. Age, blood pressure, and vessel diameter as factors influencing the arterial retinal flicker response. Investig Ophthalmol Vis Sci. 2004;45(5):1486–92.
- 524. Nguyen TT, Wong TY. Retinal vascular manifestations of metabolic disorders. Vol. 17, Trends in Endocrinology and Metabolism. Elsevier Current Trends; 2006. p. 262–8.
- 525. Pulido JS, Anderson WB, Flynn JT, Lichter PR, Sanders D. Multicenter prospective, randomized, double-masked, placebo-controlled study of rheopheresis to treat nonexudative age-related macular degeneration: Interim analysis. In: Transactions of the American Ophthalmological Society [Internet]. American Ophthalmological Society; 2002 [cited 2021 May 23]. p. 85–107. Available from: /pmc/articles/PMC1358951/?report=abstract
- 526. Reimann M, Prieur S, Lippold B, Bornstein SR, Reichmann H, Julius U, et al. Retinal vessel analysis in hypercholesterolemic patients before and after LDL apheresis. Atheroscler Suppl. 2009 Dec 29;10(5):39–43.
- 527. Celermajer DS, Sorensen KE, Spiegelhalter DJ, Georgakopoulos D, Robinson J, Deanfield JE. Aging is associated with endothelial dysfunction in healthy men years before the age-related decline in women. J Am Coll Cardiol. 1994;24(2):471–6.
- 528. Kneser M, Kohlmann T, Pokorny J, Tost F. Age related decline of microvascular regulation measured in healthy individuals by retinal dynamic vessel analysis. Med Sci Monit [Internet]. 2009 Aug 1 [cited 2021 May 23];15(8):CR436–41. Available from: https://www.medscimonit.com/abstract/index/idArt/878144
- 529. Nagel E, Vilser W. Flicker observation light induces diameter response in retinal arterioles: A clinical methodological study. Br J Ophthalmol [Internet]. 2004 Jan 1 [cited 2021 May 23];88(1):54–6. Available from: http://bjo.bmj.com/
- 530. Influence of Flicker Frequency on Flicker-Induced Changes of Retinal Vessel Diameter | IOVS | ARVO Journals [Internet]. [cited 2021 May 23]. Available from: https://iovs.arvojournals.org/article.aspx?articleid=2123956

- 531. Seshadri S, Ekart A, Gherghel D. Ageing effect on flicker-induced diameter changes in retinal microvessels of healthy individuals. Acta Ophthalmol. 2016;94(1):e35–42.
- 532. Upadhyay RK. Emerging Risk Biomarkers in Cardiovascular Diseases and Disorders. J Lipids. 2015;2015:1–50.
- 533. Butt M, Dwivedi G, Blann A, Khair O, Y.H. Lip G. Endothelial Dysfunction: Methods of Assessment & Implications for Cardiovascular Diseases. Curr Pharm Des. 2010 Dec 1;16(31):3442–54.
- 534. Garhofer G, Bek T, Boehm AG, Gherghel D, Grunwald J, Jeppesen P, et al. Use of the retinal vessel analyzer in ocular blood flow research [Internet]. Vol. 88, Acta Ophthalmologica. John Wiley & Sons, Ltd; 2010 [cited 2021 May 25]. p. 717–22. Available from: www.imedos.de
- 535. Williams B, Mancia G, Spiering W, Rosei EA, Azizi M, Burnier M, et al. 2018 practice guidelines for the management of arterial hypertension of the European society of cardiology and the European society of hypertension ESC/ESH task force for the management of arterial hypertension. J Hypertens. 2018;36(12):2284–309.
- 536. Whelton PK, Carey RM, Aronow WS, Casey DE, Collins KJ, Himmelfarb CD, et al. 2017 ACC/AHA/AAPA/ABC/ACPM/AGS/APhA/ ASH/ASPC/NMA/PCNA guideline for the prevention, detection, evaluation, and management of high blood pressure in adults a report of the American College of Cardiology/American Heart Association Task Force on Clinical pr. Vol. 71, Hypertension. 2018. 13–115 p.
- 537. Flammer J, Konieczka K, Flammer AJ. The primary vascular dysregulation syndrome: implications for eye diseases. EPMA J [Internet]. 2013 Dec 7 [cited 2021 May 24];4(1):1–33. Available from: http://www.epmajournal.com/content/4/1/14
- 538. Kotliar KE, Vilser W, Nagel E, Lanzl IM. Retinal vessel reaction in response to chromatic flickering light. Graefe's Arch Clin Exp Ophthalmol. 2004;
- 539. Patel SR, Bellary S, Qin L, Gill PS, Taheri S, Heitmar R, et al. Abnormal retinal vascular function and lipid levels in a sample of healthy UK South Asians. Br J Ophthalmol. 2011;95(11):1573–6.
- 540. Patel SR, Bellary S, Qin L, Balanos GM, McIntyre D, Gherghel D. Abnormal retinal vascular reactivity in individuals with impaired glucose tolerance: A preliminary study. Investig Ophthalmol Vis Sci. 2012;
- 541. Mroczkowska S, Ekart A, Sung V, Negi A, Qin L, Patel SR, et al. Coexistence of macro- and micro-vascular abnormalities in newly diagnosed normal tension glaucoma patients. Acta Ophthalmol. 2012 Nov;90(7).
- 542. Mroczkowska S, Benavente-Perez A, Patel S, Qin L, Bentham P, Gherghel D. Retinal vascular dysfunction relates to cognitive impairment in alzheimer disease. Alzheimer Dis Assoc Disord [Internet]. 2014 Dec 4 [cited 2021 May 24];28(4):366–7. Available from: https://journals.lww.com/alzheimerjournal/Fulltext/2014/10000/Retinal_Vascular_Dvsf

https://journals.lww.com/alzheimerjournal/Fulltext/2014/10000/Retinal_Vascular_Dysf unction_Relates_to_Cognitive.11.aspx

- 543. Seifert U, Vilser W. Retinal Vessel Analyzer (RVA)-Design and Function. Vol. 47, Biomed Tech. 2002.
- 544. Pache M, Nagel E, Flammer J. Reproducibility of measurements with the retinal vessel analyzer under optimal conditions. Klin Monbl Augenheilkd. 2002;
- 545. [Comparison of diameter response of retinal arteries and veins to flickering light. A clinical study with healthy people]. Abstract Europe PMC [Internet]. [cited 2021 May 25]. Available from: https://europepmc.org/article/med/15759121
- 546. Faraci FM, Heistad DD. Regulation of the cerebral circulation: Role of endothelium and potassium channels [Internet]. Vol. 78, Physiological Reviews. American Physiological Society; 1998 [cited 2021 May 25]. p. 53–97. Available from: https://journals.physiology.org/doi/abs/10.1152/physrev.1998.78.1.53
- 547. Riva CE, Logean E, Falsini B. Visually evoked hemodynamical response and assessment of neurovascular coupling in the optic nerve and retina. Vol. 24, Progress in Retinal and Eye Research. Pergamon; 2005. p. 183–215.
- 548. Heitmar R, Blann AD, Cubbidge RP, Lip GYH, Gherghel D. Continuous retinal vessel

H.A. Shokr, PhD Thesis, Aston University 2021

diameter measurements: The future in retinal vessel assessment? Investig Ophthalmol Vis Sci. 2010 Nov 1;51(11):5833–9.

- 549. Response of Retinal Vessel Diameters to Flicker Stimulation...: Journal of Glaucoma [Internet]. [cited 2021 May 25]. Available from: https://journals.lww.com/glaucomajournal/Fulltext/2004/08000/Response_of_Retinal_ Vessel_Diameters_to_Flicker.13.aspx?casa_token=SKEs1Ktt_3AAAAAA:Vp7bcSc6B i8IC4_Ksir0b0gyUs1NQ5gOxnki43o73vogt2m3lq_YxwioxOYyrqgw0Xv73yX7NqWSsf oVFD0m_ehaFAE
- 550. Lanzl IM, Seidova SF, Maier M, Lohmann C, Schmidt-Trucksäss A, Halle M, et al. Dynamic retinal vessel response to flicker in age-related macular degeneration patients before and after vascular endothelial growth factor inhibitor injection. Acta Ophthalmol [Internet]. 2011 Aug 1 [cited 2021 May 25];89(5):472–9. Available from: https://onlinelibrary.wiley.com/doi/full/10.1111/j.1755-3768.2009.01718.x
- 551. Nguyen TT, Kawasaki R, Jie JW, Kreis AJ, Shaw J, Vilser W, et al. Flicker lightinduced retinal vasodilation in diabetes and diabetic retinopathy. Diabetes Care. 2009;
- 552. Bek T, Hajari J, Jeppesen P. Interaction between flicker-induced vasodilatation and pressure autoregulation in early retinopathy of Type 2 diabetes. Graefe's Arch Clin Exp Ophthalmol [Internet]. 2008 May 12 [cited 2021 May 25];246(5):763–9. Available from: https://link.springer.com/article/10.1007/s00417-008-0766-y
- 553. Mandecka A, Dawczynski J, Blum M, Müller N, Kloos C, Wolf G, et al. Influence of flickering light on the retinal vessels in diabetic patients. Diabetes Care [Internet]. 2007 Dec 1 [cited 2021 May 25];30(12):3048–52. Available from: http://care.diabetesjournals.org
- 554. Dorner GT, Garhofer G, Kiss B, Polska E, Polak K, Riva CE, et al. Nitric oxide regulates retinal vascular tone in humans. Am J Physiol Hear Circ Physiol. 2003;
- 555. Schier R, Hinkelbein J, Marcus H, Smallwood A, Correa AM, Mehran R, et al. A novel technique for the assessment of preoperative cardiovascular risk: Reactive hyperemic response to short-term exercise. Biomed Res Int. 2013;2013.
- 556. Akhtar MW, Kleis SJ, Metcalfe RW, Naghavi M. Sensitivity of digital thermal monitoring parameters to reactive hyperemia. J Biomech Eng. 2010 May 1;132(5).
- 557. Kawai M, Yamada M, Kawashima M, Inoue M, Goto E, Mashima Y, et al. Quantitative evaluation of tear meniscus height from fluorescein photographs. Cornea [Internet]. 2007 May [cited 2021 Jul 16];26(4):403–6. Available from: https://journals.lww.com/corneajrnl/Fulltext/2007/05000/Quantitative_Evaluation_of_T ear Meniscus Height.4.aspx
- 558. Wu S, Hong J, Tian L, Cui X, Sun X, Xu J. Assessment of Bulbar Redness with a Newly Developed Keratograph. Optom Vis Sci [Internet]. 2015 Aug 5 [cited 2021 Mar 15];92(8):892–9. Available from: https://journals.lww.com/00006324-201508000-00009
- 559. Tian L, Qu JH, Zhang XY, Sun XG. Repeatability and reproducibility of noninvasive keratograph 5m measurements in patients with dry eye disease. J Ophthalmol. 2016;2016.
- 560. Arita R, Itoh K, Inoue K, Amano S. Noncontact Infrared Meibography to Document Age-Related Changes of the Meibomian Glands in a Normal Population. Ophthalmology. 2008 May 1;115(5):911–5.
- 561. Nichols JJ, Berntsen DA, Mitchell GL, Nichols KK. An assessment of grading scales for meibography images. Cornea [Internet]. 2005 May [cited 2021 Jul 16];24(4):382–8. Available from: https://journals.lww.com/corneajrnl/Fulltext/2005/05000/An_Assessment_of_Grading_
- Scales_for_Meibography.3.aspx
 562. Pult H, Nichols JJ. A Review of Meibography. Optom Vis Sci [Internet]. 2012 May [cited 2021 Mar 15];89(5):E760–9. Available from: https://journals.lww.com/00006324-201205000-00027
- 563. Pult H, Riede-Pult B. Comparison of subjective grading and objective assessment in meibography. Contact Lens Anterior Eye. 2013 Feb 1;36(1):22–7.

- 564. Friedewald WT, Levy RI, Fredrickson DS. Estimation of the concentration of lowdensity lipoprotein cholesterol in plasma, without use of the preparative ultracentrifuge. Clin Chem. 1972;
- 565. Schafer FQ, Buettner GR. Redox environment of the cell as viewed through the redox state of the glutathione disulfide/glutathione couple. Vol. 30, Free Radical Biology and Medicine. Pergamon; 2001. p. 1191–212.
- 566. Kemp M, Go YM, Jones DP. Nonequilibrium thermodynamics of thiol/disulfide redox systems: A perspective on redox systems biology. Vol. 44, Free Radical Biology and Medicine. Pergamon; 2008. p. 921–37.
- 567. Tietze F. Enzymic method for quantitative determination of nanogram amounts of total and oxidized glutathione: Applications to mammalian blood and other tissues. Anal Biochem. 1969;
- 568. Anderson ME. Glutathione: An overview of biosynthesis and modulation. Chem Biol Interact. 1998;
- 569. Gherghel D, Mroczkowska S, Qin L. Reduction in blood glutathione levels occurs similarly in patients with primary-open angle or normal tension glaucoma. Investig Ophthalmol Vis Sci. 2013;
- 570. Rossi R, Milzani A, Dalle-Donne I, Giustarini D, Lusini L, Colombo R, et al. Blood glutathione disulfide: In vivo factor or in vitro artifact? Clin Chem. 2002;
- 571. Rahman I, Kode A, Biswas SK. Assay for quantitative determination of glutathione and glutathione disulfide levels using enzymatic recycling method. Nat Protoc. 2007;
- 572. Gherghel D, Griffiths HR, Hilton EJ, Cunliffe IA, Hosking SL. Systemic reduction in glutathione levels occurs in patients with primary open-angle glaucoma. Investig Ophthalmol Vis Sci. 2005;
- 573. Jones DP, Carlson JL, Samiec PS, Sternberg P, Mody VC, Reed RL, et al. Glutathione measurement in human plasma. Evaluation of sample collection, storage and derivatization conditions for analysis of dansyl derivatives by HPLC. Clin Chim Acta. 1998;
- 574. Liuzzo G. Atherosclerosis: an inflammatory disease. [Internet]. Vol. 26, Rays. 2001 [cited 2021 May 31]. p. 221–30. Available from: https://europepmc.org/article/med/12696277
- 575. MONCADA, S. Nitric oxide : Physiology, pathophysiology and pharmacology. Pharmacol Rev [Internet]. 1991 [cited 2021 May 31];43:109–42. Available from: https://ci.nii.ac.jp/naid/10009922604
- 576. Kelm M. Nitric oxide metabolism and breakdown. Vol. 1411, Biochimica et Biophysica Acta - Bioenergetics. Elsevier; 1999. p. 273–89.
- 577. Lauer T, Preik M, Rassaf T, Strauer BE, Deussen A, Feelisch M, et al. Plasma nitrite rather than nitrate reflects regional endothelial nitric oxide synthase activity but lacks intrinsic vasodilator action. Proc Natl Acad Sci U S A [Internet]. 2001 Oct 23 [cited 2021 May 31];98(22):12814–9. Available from: www.pnas.orgcgidoi10.1073pnas.221381098
- 578. Resources QIAGEN [Internet]. [cited 2021 Feb 9]. Available from: https://www.qiagen.com/us/resources/
- 579. Cawthon RM. Telomere measurement by quantitative PCR. Nucleic Acids Res. 2002;
- 580. Cawthon RM. Telomere length measurement by a novel monochrome multiplex quantitative PCR method. Nucleic Acids Res. 2009;
- 581. Quade M, Gout J, Abel M. Glyph: Symbolic Regression Tools. J Open Res Softw [Internet]. 2019 Jun 17 [cited 2021 Sep 5];7(1):1–8. Available from: http://openresearchsoftware.metajnl.com/articles/10.5334/jors.192/
- 582. Orzechowski P, Cava W La, Moore JH. Where are we now? A large benchmark study of recent symbolic regression methods. 2018 [cited 2021 Sep 5]; Available from: https://doi.org/10.1145/3205455.3205539
- 583. Introduction to GP gplearn 0.4.1 documentation [Internet]. [cited 2021 Sep 20]. Available from: https://gplearn.readthedocs.io/en/stable/intro.html
- 584. Dafflon J, Pinaya WHL, Turkheimer F, Cole JH, Leech R, Harris MA, et al. An

216
automated machine learning approach to predict brain age from cortical anatomical measures. Hum Brain Mapp [Internet]. 2020 Sep 1 [cited 2021 Sep 5];41(13):3555–66. Available from: https://onlinelibrary.wiley.com/doi/full/10.1002/hbm.25028

- 585. Rizzoni D, Aalkjaer C, Ciuceis C De, Porteri E, Rossini C, Rosei CA, et al. How to Assess Microvascular Structure in Humans. High Blood Press Cardiovasc Prev 2011 184 [Internet]. 2012 Nov 4 [cited 2021 Jul 18];18(4):169–77. Available from: https://link.springer.com/article/10.2165/11593640-000000000-00000
- 586. Gariano RF, Gardner TW. Retinal angiogenesis in development and disease. Nature. 2005.
- 587. Brown MD. Endothelial ageing: molecular mechanisms and functional significance. Exp Physiol [Internet]. 2009 Mar 1 [cited 2021 Jul 18];94(3):297–8. Available from: https://physoc.onlinelibrary.wiley.com/doi/full/10.1113/expphysiol.2008.043125
- 588. Seshadri S, Ekart A, Gherghel D. Ageing effect on flicker-induced diameter changes in retinal microvessels of healthy individuals. Acta Ophthalmol [Internet]. 2016 Feb 1 [cited 2021 May 23];94(1):e35–42. Available from: https://onlinelibrary.wiley.com/doi/full/10.1111/aos.12786
- 589. Seshadri S, Mroczkowska S, Qin L, Patel S, Ekart A, Gherghel D. Systemic circulatory influences on retinal microvascular function in middle-age individuals with low to moderate cardiovascular risk. Acta Ophthalmol. 2015;93(4):e266–74.
- 590. Pemp B, Weigert G, Karl K, Petzl U, Wolzt M, Schmetterer L, et al. Correlation of Flicker-Induced and Flow-Mediated Vasodilatation in Patients With Endothelial Dysfunction and Healthy Volunteers. Diabetes Care [Internet]. 2009 Aug 1 [cited 2021 Jul 18];32(8):1536–41. Available from: https://care.diabetesjournals.org/content/32/8/1536
- 591. Dorner GT, Garhöfer G, Selenko N, Fasching P, Bayerle-Eder M, Schmetterer L, et al. The ocular hemodynamic response to nitric oxide synthase inhibition is unaltered in patients with early type I diabetes. Graefe's Arch Clin Exp Ophthalmol 2003 2418 [Internet]. 2003 Jun 26 [cited 2021 Jul 18];241(8):619–24. Available from: https://link.springer.com/article/10.1007/s00417-003-0708-7
- 592. McGeechan K, Liew G, Macaskill P, Irwig L, Klein R, Klein BEK, et al. Meta-analysis: Retinal vessel caliber and risk for coronary heart disease. Ann Intern Med [Internet]. 2009 Sep 15 [cited 2021 Jul 16];151(6):404–13. Available from: www.annals.org
- 593. Cines DB, Pollak ES, Buck CA, Loscalzo J, Zimmerman GA, McEver RP, et al. Endothelial Cells in Physiology and in the Pathophysiology of Vascular Disorders. Blood [Internet]. 1998 May 15 [cited 2021 Jul 18];91(10):3527–61. Available from: http://ashpublications.org/blood/article-pdf/91/10/3527/1416608/3527.pdf
- 594. Frisard MI, Study for the LHA, Broussard A, Study for the LHA, Davies SS, Study for the LHA, et al. Aging, Resting Metabolic Rate, and Oxidative Damage: Results From the Louisiana Healthy Aging Study. Journals Gerontol Ser A [Internet]. 2007 Jul 1 [cited 2021 Jul 18];62(7):752–9. Available from: https://academic.oup.com/biomedgerontology/article/62/7/752/581912
- 595. Salmon AB, Richardson A, Pérez VI. Update on the oxidative stress theory of aging: Does oxidative stress play a role in aging or healthy aging? Free Radic Biol Med. 2010 Mar 1;48(5):642–55.
- 596. Williams B, Mancia G, Spiering W, Agabiti Rosei E, Azizi M, Burnier M, et al. 2018 ESC/ESH Guidelines for the management of arterial hypertensionThe Task Force for the management of arterial hypertension of the European Society of Cardiology (ESC) and the European Society of Hypertension (ESH). Eur Heart J [Internet]. 2018 Sep 1 [cited 2021 Jul 18];39(33):3021–104. Available from: https://academic.oup.com/eurhearti/article/39/33/3021/5079119
- 597. Garhöfer G, Zawinka C, Resch H, Kothy P, Schmetterer L, Dorner GT. Reduced response of retinal vessel diameters to flicker stimulation in patients with diabetes. Br J Ophthalmol. 2004;
- 598. Abdilla N, Tormo MC, Fabia MJ, Chaves FJ, Saez G, Redon J. Impact of the components of metabolic syndrome on oxidative stress and enzymatic antioxidant

217

activity in essential hypertension. J Hum Hypertens 2007 211 [Internet]. 2006 Oct 26 [cited 2021 Jul 19];21(1):68–75. Available from: https://www.nature.com/articles/1002105

- 599. Bruno RM, Masi S, Taddei M, Taddei S, Virdis A. Essential Hypertension and Functional Microvascular Ageing. High Blood Press Cardiovasc Prev [Internet]. 2018;25(1):35–40. Available from: https://doi.org/10.1007/s40292-017-0245-9
- 600. Imhof K, Zahner L, Schmidt-Trucksäss A, Hanssen H. Association of body composition and blood pressure categories with retinal vessel diameters in primary school children. Hypertens Res 2016 396 [Internet]. 2016 Jan 14 [cited 2021 Jul 19];39(6):423–9. Available from: https://www.nature.com/articles/hr2015159
- 601. Goel H, Tayel H, Nadar SK. Aiming higher in hopes to achieve lower: the European Society of Cardiology/European Society of Hypertension versus the American College of Cardiology/American Heart Association guidelines for diagnosis and management of hypertension. J Hum Hypertens [Internet]. 2019;33(9):635–8. Available from: http://dx.doi.org/10.1038/s41371-019-0227-6
- 602. Homocysteine and red blood cell glutathione as indices for m...: Journal of Hypertension [Internet]. [cited 2021 Jul 19]. Available from: https://journals.lww.com/jhypertension/Abstract/2003/12000/Homocysteine_and_red_ blood_cell_glutathione_as.22.aspx
- 603. Lee MY, Griendling KK. Redox Signaling, Vascular Function, and Hypertension. https://home.liebertpub.com/ars [Internet]. 2010 Sep 30 [cited 2021 Jul 19];10(6):1045–59. Available from: https://www.liebertpub.com/doi/abs/10.1089/ars.2007.1986
- 604. Touyz RM, Montezano AC, Rios F, Widlansky ME, Liang M. Redox Stress Defines the Small Artery Vasculopathy of Hypertension. Circ Res [Internet]. 2017 May 26 [cited 2021 Jul 19];120(11):1721–3. Available from: https://www.ahajournals.org/doi/abs/10.1161/CIRCRESAHA.117.310672
- 605. The Hypertension Paradox. N Engl J Med. 2009 Nov 26;361(22):2195–7.
- 606. Shokr H, Gherghel D. European Society of Cardiology/European Society of Hypertension versus the American College of Cardiology/American Heart Association guidelines on the cut-off values for early hypertension: a microvascular perspective. Sci Rep [Internet]. 2021 Dec 1 [cited 2021 Jun 13];11(1):3473. Available from: https://doi.org/10.1038/s41598-021-83096-1
- 607. Of R, In I. JBS 2: Joint British Societies' guidelines on prevention of cardiovascular disease in clinical practice. Heart. 2005;91 Suppl 5(December).
- 608. Cohn JN. Identifying the Risk and Preventing the Consequences of Cardiovascular Disease. Hear Lung Circ [Internet]. 2013;22(7):512–6. Available from: http://dx.doi.org/10.1016/j.hlc.2013.03.083
- 609. Koenig W. Cardiovascular biomarkers: Added value with an integrated approach? Circulation. 2007;116(1):3–5.
- 610. Vasan RS. Biomarkers of cardiovascular disease: Molecular basis and practical considerations. Circulation. 2006;113(19):2335–62.
- 611. Koenig W, Khuseyinova N. Biomarkers of atherosclerotic plaque instability and rupture. Arterioscler Thromb Vasc Biol. 2007;27(1):15–26.
- 612. Cochrane T, Davey R, Gidlow C, Iqbal Z, Kumar J, Mawby Y, et al. Contribution of individual risk factor changes to reductions in population absolute cardiovascular risk. Biomed Res Int. 2014;2014.
- 613. Mitchell GF, Vita JA, Larson MG, Parise H, Keyes MJ, Warner E, et al. Crosssectional relations of peripheral microvascular function, cardiovascular disease risk factors, and aortic stiffness: The Framingham Heart Study. Circulation. 2005;112(24):3722–8.
- 614. Cooper LL, Palmisano JN, Benjamin EJ, Larson MG, Vasan RS, Mitchell GF, et al. Microvascular function contributes to the relation between aortic stiffness and cardiovascular events. Circ Cardiovasc Imaging. 2016;9(12):1–8.
- 615. Blanch N, Clifton PM, Keogh JB. A systematic review of vascular and endothelial

function: Effects of fruit, vegetable and potassium intake. Nutr Metab Cardiovasc Dis [Internet]. 2015;25(3):253–66. Available from:

- http://dx.doi.org/10.1016/j.numecd.2014.10.001
- 616. Patel SR, Bellary S, Karimzad S, Gherghel D. Overweight status is associated with extensive signs of microvascular dysfunction and cardiovascular risk. Sci Rep [Internet]. 2016;6(August):1–8. Available from: http://dx.doi.org/10.1038/srep32282
- 617. Schier R, Marcus HE, Mansur E, Lei X, El-Zein R, Mehran R, et al. Evaluation of digital thermal monitoring as a tool to assess perioperative vascular reactivity. J Atheroscler Thromb. 2013;20(3):277–86.
- 618. Lin HF, Dhindsa MS, Tarumi T, Miles SC, Umpierre D, Tanaka H. Impact of blood pressure cuff inflation rates on flow-mediated dilatation and contralateral arm response. J Hum Hypertens. 2012;26(1):35–40.
- 619. Levy BI, Ambrosio G, Pries AR, Struijker-Boudier HAJ. Microcirculation in hypertension: A new target for treatment? Circulation. 2001;104(6):735–40.
- 620. Virdis A, Taddei S. Endothelial Dysfunction in Resistance Arteries of Hypertensive Humans: Old and New Conspirators. Journal of Cardiovascular Pharmacology. 2016.
- 621. Nasri H, Baradaran A, Rafieian-Kopaei M. Oxidative stress and hypertension: Possibility of hypertension therapy with antioxidants. J Res Med Sci. 2014;
- 622. Mroczkowska S, Benavente-Perez A, Negi A, Sung V, Patel SR, Gherghel D. Primary open-angle glaucoma vs normal-tension glaucoma: The vascular perspective. Arch Ophthalmol. 2013;131(1):36–43.
- 623. Qin L, Mroczkowska SA, Ekart A, Patel SR, Gibson JM, Gherghel D. Patients with early age-related macular degeneration exhibit signs of macro- and micro-vascular disease and abnormal blood glutathione levels. Graefe's Arch Clin Exp Ophthalmol. 2014;252(1):23–30.
- 624. Wong T, Mitchell P. The eye in hypertension. Lancet. 2007.
- 625. Tien YW, Islam FMA, Klein R, Klein BEK, Cotch MF, Castro C, et al. Retinal vascular caliber, cardiovascular risk factors, and inflammation: The Multi-Ethnic Study of Atherosclerosis (MESA). Investig Ophthalmol Vis Sci. 2006;47(6):2341–50.
- 626. Shokr H, Wolffsohn JS, Trave Huarte S, Scarpello E, Gherghel D. Dry eye disease is associated with retinal microvascular dysfunction and possible risk for cardiovascular disease. Acta Ophthalmol [Internet]. 2021 Feb 11 [cited 2021 Mar 15];aos.14782. Available from: https://onlinelibrary.wiley.com/doi/10.1111/aos.14782
- 627. Musunuru K. Atherogenic dyslipidemia: Cardiovascular risk and dietary intervention. Lipids. 2010.
- 628. Bu J, Wu Y, Cai X, Jiang N, Jeyalatha MV, Yu J, et al. Hyperlipidemia induces meibomian gland dysfunction. Ocul Surf. 2019;
- 629. Pereira T. Dyslipidemia and Cardiovascular Risk: Lipid Ratios as Risk Factors for Cardiovascular Disease. In: Dyslipidemia From Prevention to Treatment. 2012.
- 630. Padró T, Vilahur G, Badimon L. Dyslipidemias and Microcirculation. Curr Pharm Des. 2018;
- 631. Mudau M, Genis A, Lochner A, Strijdom H. Endothelial dysfunction: The early predictor of atherosclerosis. Cardiovascular Journal of Africa. 2012.
- Dao AH, Spindle JD, Harp BA, Jacob A, Chuang AZ, Yee RW. Association of dyslipidemia in moderate to severe meibomian gland dysfunction. Am J Ophthalmol. 2010;
- 633. Wang TJ, Wang IJ, Hu CC, Lin HC. Comorbidities of dry eye disease: A nationwide population-based study. Acta Ophthalmol. 2012;
- 634. Chun YH, Kim HR, Han K, Park YG, Song HJ, Na KS. Total cholesterol and lipoprotein composition are associated with dry eye disease in Korean women. Lipids Health Dis. 2013;
- 635. Butovich IA, Millar TJ, Ham BM. Understanding and analyzing meibomian lipids A review. Current Eye Research. 2008.
- 636. Campbell S, Genest J. HDL-C: Clinical equipoise and vascular endothelial function. Expert Review of Cardiovascular Therapy. 2013.

- 637. Tewari S, Kumar S, Kapoor A, Singh U, Agarwal A, Bharti BB, et al. Premature coronary artery disease in North India: An angiography study of 1971 patients. Indian Heart J. 2005;
- 638. Chen W, Batawi HIM, Alava JR, Galor A, Yuan J, Sarantopoulos CD, et al. Bulbar conjunctival microvascular responses in dry eye. Ocul Surf. 2017;
- 639. Ruskell GL. Innervation of the conjunctiva. Trans Ophthalmol Soc U K. 1985;
- 640. Li L xiang, Jin DL, Gao JS, Wang LG, Liu XJ, Wang JZ, et al. Activities of autonomic neurotransmitters in meibomian gland tissues are associated with menopausal dry eye. Neural Regen Res. 2012;
- 641. LeDoux MS, Zhou Q, Murphy RB, Greene ML, Ryan P. Parasympathetic innervation of the meibomian glands in rats. Investig Ophthalmol Vis Sci. 2001;
- 642. Amiya E, Watanabe M, Komuro I. The Relationship between Vascular Function and the Autonomic Nervous System. Ann Vasc Dis. 2014;
- 643. Davidson MH, Abate N, Ballantyne CM, Catapano AL, Xu X, Lin J, et al. Ezetimibe/simvastatin compared with atorvastatin or rosuvastatin in lowering to specified levels both LDL-C and each of five other emerging risk factors for coronary heart disease: Non-HDL-cholesterol, TC/HDL-C, apolipoprotein B, apo-B/apo-A-I, or C-react. J Clin Lipidol [Internet]. 2008;2(6):436–46. Available from: http://dx.doi.org/10.1016/j.jacl.2008.10.003
- 644. Agirbasli M, Tanrikulu A, Acar Sevim B, Azizy M, Bekiroglu N. Total cholesterol-tohigh-density lipoprotein cholesterol ratio predicts high-sensitivity C-reactive protein levels in Turkish children. J Clin Lipidol. 2015;
- 645. Shokr H, Dias IHK, Gherghel D. Oxysterols and Retinal Microvascular Dysfunction as Early Risk Markers for Cardiovascular Disease in Normal, Ageing Individuals. Antioxidants 2021, Vol 10, Page 1756 [Internet]. 2021 Nov 3 [cited 2021 Dec 25];10(11):1756. Available from: https://www.mdpi.com/2076-3921/10/11/1756/htm
- 646. Rattan SIS. Theories of biological aging: Genes, proteins, and free radicals. https://doi.org/101080/10715760600911303 [Internet]. 2009 Dec [cited 2021 Jul 7];40(12):1230–8. Available from: https://www.tandfonline.com/doi/abs/10.1080/10715760600911303
- 647. Endemann DH, Schiffrin EL. Endothelial Dysfunction. J Am Soc Nephrol [Internet]. 2004 Aug 1 [cited 2021 Jul 6];15(8):1983–92. Available from: https://jasn.asnjournals.org/content/15/8/1983
- 648. Thanassoulis G, Lyass A, Benjamin EJ, Larson MG, Vita JA, Levy D, et al. Relations of Exercise Blood Pressure Response to Cardiovascular Risk Factors and Vascular Function in the Framingham Heart Study. Circulation [Internet]. 2012 Jun 12 [cited 2021 Jul 29];125(23):2836–43. Available from: https://www.ahajournals.org/doi/abs/10.1161/CIRCULATIONAHA.111.063933
- 649. Alam TA, Seifalian AM, Baker D. A Review of Methods Currently Used for Assessment of In vivo Endothelial Function. Eur J Vasc Endovasc Surg. 2005 Mar 1;29(3):269–76.
- 650. Tyrrell DJ, Goldstein DR. Ageing and atherosclerosis: vascular intrinsic and extrinsic factors and potential role of IL-6 [Internet]. Vol. 18, Nature Reviews Cardiology. Nature Research; 2021 [cited 2021 Jun 25]. p. 58–68. Available from: www.nature.com/nrcardio
- 651. Nilsson PM, Boutouyrie P, Cunha P, Kotsis V, Narkiewicz K, Parati G, et al. Early vascular ageing in translation: From laboratory investigations to clinical applications in cardiovascular prevention [Internet]. Vol. 31, Journal of Hypertension. Lippincott Williams and Wilkins; 2013 [cited 2021 Jun 16]. p. 1517–26. Available from: https://journals.lww.com/jhypertension/Fulltext/2013/08000/Early_vascular_ageing_in _translation_from.3.aspx
- 652. Glavic MM, Blagus L, Bosnjak V, Frkanec S, Katic L, Domislovic V, et al. CHARACTERISTICS OF HEALTHY VASCULAR AGEING (HVA) AND EARLY VASCULAR AGEING (EVA) IN GENERAL POPULATION. EH-UH STUDY (CROATIAN SCIENTIFIC FOUNDATION). J Hypertens. 2021 Apr;39(Supplement

1):e67.

- 653. O'Rourke MF, Safar ME, Dzau V. The Cardiovascular Continuum extended: Aging effects on the aorta and microvasculature. Vasc Med [Internet]. 2010 Dec 5 [cited 2021 Jun 16];15(6):461–8. Available from: http://vmj.sagepub.com
- 654. Tsutsui H, Kinugawa S, Matsushima S. Oxidative stress and heart failure. https://doi.org/101152/ajpheart005542011 [Internet]. 2011 Dec [cited 2021 Aug 28];301(6):2181–90. Available from: https://journals.physiology.org/doi/abs/10.1152/ajpheart.00554.2011
- 655. Samman Tahhan A, Sandesara PB, Hayek SS, Alkhoder A, Chivukula K, Hammadah M, et al. Association between oxidative stress and atrial fibrillation. [cited 2021 Aug 28]; Available from: http://dx.doi.org/
- 656. Franco R, Schoneveld OJ, Pappa A, Panayiotidis MI. The central role of glutathione in the pathophysiology of human diseases. https://doi.org/101080/13813450701661198 [Internet]. 2008 Oct [cited 2021 Aug 29];113(4–5):234–58. Available from: https://www.tandfonline.com/doi/abs/10.1080/13813450701661198
- 657. Chen C-A, Wang T-Y, Varadharaj S, Reyes LA, Hemann C, Talukder MAH, et al. Sglutathionylation uncouples eNOS and regulates its cellular and vascular function. Nat 2010 4687327 [Internet]. 2010 Dec 22 [cited 2021 Aug 29];468(7327):1115–8. Available from: https://www.nature.com/articles/nature09599
- 658. DeLuca HF. Overview of general physiologic features and functions of vitamin D. [Internet]. Vol. 80, The American journal of clinical nutrition. Oxford Academic; 2004 [cited 2021 Jun 22]. p. 1689S-1696S. Available from: https://academic.oup.com/ajcn/article/80/6/1689S/4690513
- 659. Payne AH, Hales DB. Overview of steroidogenic enzymes in the pathway from cholesterol to active steroid hormones [Internet]. Vol. 25, Endocrine Reviews. Oxford Academic; 2004 [cited 2021 Jun 22]. p. 947–70. Available from: https://academic.oup.com/edrv/article/25/6/947/2195014
- 660. Kummerow FA. Interaction between sphingomyelin and oxysterols contributes to atherosclerosis and sudden death. Am J Cardiovasc Dis [Internet]. 2013 [cited 2021 Jun 17];3(1):17–26. Available from: http://www.ncbi.nlm.nih.gov/pubmed/23459228
- 661. Griffiths WJ, Wang Y. Oxysterol research: A brief review [Internet]. Vol. 47, Biochemical Society Transactions. Portland Press Ltd; 2019 [cited 2021 Jun 17]. p. 517–26. Available from: /pmc/articles/PMC6490702/
- 662. Terao J. Cholesterol hydroperoxides and their degradation mechanism. Subcell Biochem [Internet]. 2014 [cited 2021 Jun 17];77:83–91. Available from: https://link.springer.com/chapter/10.1007/978-94-007-7920-4_7
- 663. Khatib S, Vaya J. Oxysterols and symptomatic versus asymptomatic human atherosclerotic plaque. Vol. 446, Biochemical and Biophysical Research Communications. Academic Press Inc.; 2014. p. 709–13.
- 664. Gargiulo S, Gamba P, Testa G, Leonarduzzi G, Poli G. The role of oxysterols in vascular ageing. J Physiol [Internet]. 2016 Apr 15 [cited 2021 Sep 6];594(8):2095–113. Available from: https://physoc.onlinelibrary.wiley.com/doi/full/10.1113/JP271168
- 665. Gordiyenko N, Campos M, Lee JW, Fariss RN, Sztein J, Rodriguez IR. RPE Cells Internalize Low-Density Lipoprotein (LDL) and Oxidized LDL (oxLDL) in Large Quantities In Vitro and In Vivo. Invest Ophthalmol Vis Sci. 2004 Aug 1;45(8):2822–9.
- 666. Uptake of cholesterol by the retina occurs primarily via a low density lipoprotein receptor-mediated process PubMed [Internet]. [cited 2021 Aug 31]. Available from: https://pubmed.ncbi.nlm.nih.gov/17110914/#:~:text=Conclusions%3A The retina is capable,and uptake by the retina.
- 667. Zhang X, Alhasani RH, Zhou X, Reilly J, Zeng Z, Strang N, et al. Oxysterols and retinal degeneration. Br J Pharmacol [Internet]. 2021 Aug 1 [cited 2021 Sep 6];178(16):3205–19. Available from: https://bpspubs.onlinelibrary.wiley.com/doi/full/10.1111/bph.15391
- 668. Miles EA, Rees D, Banerjee T, Cazzola R, Lewis S, Wood R, et al. Age-related increases in circulating inflammatory markers in men are independent of BMI, blood

221

pressure and blood lipid concentrations. Atherosclerosis. 2008 Jan 1;196(1):298–305.

- 669. Liguori I, Russo G, Curcio F, Bulli G, Aran L, Della-Morte D, et al. Oxidative stress, aging, and diseases. Clin Interv Aging [Internet]. 2018 Jan 1 [cited 2021 Sep 6];13:757. Available from: /pmc/articles/PMC5927356/
- 670. Schleicher E, Friess U. Oxidative stress, AGE, and atherosclerosis. Kidney Int. 2007 Aug 1;72(SUPPL. 106):S17–26.
- 671. Anderson A, Campo A, Fulton E, Corwin A, Jerome WG, O'Connor MS. 7-Ketocholesterol in disease and aging. Redox Biol. 2020 Jan 1;29:101380.
- 672. van den Kommer TN, Dik MG, Comijs HC, Fassbender K, Lütjohann D, Jonker C. Total cholesterol and oxysterols: Early markers for cognitive decline in elderly? Neurobiol Aging. 2009 Apr 1;30(4):534–45.
- 673. Zarrouk A, Vejux A, Mackrill J, O'Callaghan Y, Hammami M, O'Brien N, et al. Involvement of oxysterols in age-related diseases and ageing processes. Ageing Res Rev. 2014 Nov 1;18:148–62.
- 674. Lakatta EG, Levy D. Arterial and cardiac aging: Major shareholders in cardiovascular disease enterprises: Part I: Aging arteries: A "set up" for vascular disease. Vol. 107, Circulation. Lippincott Williams and Wilkins; 2003. p. 139–46.
- 675. Sun Z. Aging, arterial stiffness, and hypertension. Hypertension [Internet]. 2015 Feb 21 [cited 2021 Jun 22];65(2):252–6. Available from: http://ahajournals.org
- 676. Félétou M. Endothelium-Dependent Regulation of Vascular Tone. 2011 [cited 2021 Sep 5]; Available from: https://www.ncbi.nlm.nih.gov/books/NBK57147/
- 677. Bosch-Marce M, Okuyama H, Wesley JB, Sarkar K, Kimura H, Liu Y V., et al. Effects of aging and hypoxia-inducible factor-1 activity on angiogenic cell mobilization and recovery of perfusion after limb ischemia. Circ Res [Internet]. 2007 Dec [cited 2021 Sep 6];101(12):1310–8. Available from: https://jhu.pure.elsevier.com/en/publications/effects-of-aging-and-hypoxia-inducible-factor-1-activity-on-angio-3
- 678. Grundy SM, Pasternak R, Greenland P, Smith S, Fuster V. Assessment of cardiovascular risk by use of multiple-risk-factor assessment equations: A statement for healthcare professionals from the American Heart Association and the American College of Cardiology. Circulation. 1999.
- 679. He X, Matam BR, Bellary S, Ghosh G, Chattopadhyay AK. CHD Risk Minimization through Lifestyle Control: Machine Learning Gateway. Sci Rep. 2020;
- 680. McDermott MMG. The international pandemic of chronic cardiovascular disease. Journal of the American Medical Association. 2007.
- 681. Rosamond W, Flegal K, Furie K, Go A, Greenlund K, Haase N, et al. Heart disease and stroke statistics-2008 Update: A report from the American heart association statistics committee and stroke statistics subcommittee. Circulation. 2008.
- 682. Nichols M, Townsend N, Scarborough P, Rayner M. European Cardiovascular Disease Statistics 4th edition 2012: EuroHeart II. Eur Heart J. 2013;
- 683. Hamczyk MR, Nevado RM, Barettino A, Fuster V, Andrés V. Biological Versus Chronological Aging: JACC Focus Seminar. J Am Coll Cardiol [Internet]. 2020 Mar 3 [cited 2021 Jul 7];75(8):919–30. Available from: https://doi.org/10.1016/j.jacc.2019.11.062
- 684. Dei Cas A, Spigoni V, Franzini L, Preti M, Ardigò D, Derlindati E, et al. Lower endothelial progenitor cell number, family history of cardiovascular disease and reduced HDL-cholesterol levels are associated with shorter leukocyte telomere length in healthy young adults. Nutr Metab Cardiovasc Dis. 2013;
- 685. Voghel G, Thorin-Trescases N, Farhat N, Nguyen A, Villeneuve L, Mamarbachi AM, et al. Cellular senescence in endothelial cells from atherosclerotic patients is accelerated by oxidative stress associated with cardiovascular risk factors. Mech Ageing Dev. 2007;
- 686. Fyhrquist F, Saijonmaa O, Strandberg T. The roles of senescence and telomere shortening in cardiovascular disease. Nature Reviews Cardiology. 2013.
- 687. Huang Y-Q, Liu L, Lo K, Huang J-Y, Zhang B, Feng Y-Q. The relationship between

222

mean telomere length and blood pressure: results from the National Health and Nutrition Examination Surveys. Ann Transl Med [Internet]. 2020 Apr [cited 2021 Sep 5];8(8):535–535. Available from: /pmc/articles/PMC7214907/

- 688. Gates PE, Strain WD, Shore AC. Human endothelial function and microvascular ageing. In: Experimental Physiology. 2009.
- 689. D'Agostino RB, Grundy S, Sullivan LM, Wilson P. Validation of the Framingham coronary heart disease prediction scores: Results of a multiple ethnic groups investigation. J Am Med Assoc [Internet]. 2001 Jul 11 [cited 2021 May 25];286(2):180–7. Available from: https://jamanetwork.com/
- 690. Kim DS, Choi YD, Moon M, Kang S, Lim J-B, Kim KM, et al. Composite Three-Marker Assay for Early Detection of Kidney Cancer. Cancer Epidemiol Prev Biomarkers [Internet]. 2013 Mar 1 [cited 2021 Sep 5];22(3):390–8. Available from: https://cebp.aacrjournals.org/content/22/3/390
- 691. Bramsen JB, Rasmussen MH, Ongen H, Mattesen TB, Ørntoft MBW, Árnadóttir SS, et al. Molecular-Subtype-Specific Biomarkers Improve Prediction of Prognosis in Colorectal Cancer. Cell Rep. 2017 May 9;19(6):1268–80.
- 692. Vasan RS. Biomarkers of Cardiovascular Disease. Circulation [Internet]. 2006 May 16 [cited 2021 Sep 5];113(19):2335–62. Available from: https://www.ahajournals.org/doi/abs/10.1161/circulationaha.104.482570
- 693. Müezzinler A, Zaineddin AK, Brenner H. A systematic review of leukocyte telomere length and age in adults. Ageing Res Rev. 2013 Mar 1;12(2):509–19.
- 694. Chen W, Kimura M, Kim S, Cao X, Srinivasan SR, Berenson GS, et al. Longitudinal versus Cross-sectional Evaluations of Leukocyte Telomere Length Dynamics: Age-Dependent Telomere Shortening is the Rule. Journals Gerontol Ser A [Internet]. 2011 Mar 1 [cited 2021 Sep 21];66A(3):312–9. Available from: https://academic.oup.com/biomedgerontology/article/66A/3/312/600510
- 695. Rizvi S, Raza ST, Mahdi F. Telomere Length Variations in Aging and Age-Related Diseases.
- 696. Wu JW, Yaqub A, Ma Y, Koudstaal W, Hofman A, Ikram MA, et al. Biological age in healthy elderly predicts aging-related diseases including dementia. Sci Reports 2021 111 [Internet]. 2021 Aug 5 [cited 2021 Sep 21];11(1):1–10. Available from: https://www.nature.com/articles/s41598-021-95425-5
- 697. Vaiserman A, Krasnienkov D. Telomere Length as a Marker of Biological Age: Stateof-the-Art, Open Issues, and Future Perspectives. Front Genet. 2021 Jan 21;0:1816.
- 698. Koliada AK, Krasnenkov DS, Vaiserman AM. Telomeric aging: mitotic clock or stress indicator? Front Genet. 2015;0(FEB):82.
- 699. Notterman DA, Schneper L. Telomere Time—Why We Should Treat Biological Age Cautiously. JAMA Netw Open [Internet]. 2020 May 1 [cited 2021 Sep 21];3(5):e204352–e204352. Available from: https://jamanetwork.com/journals/jamanetworkopen/fullarticle/2765370
- 700. Ding J, Wai KL, McGeechan K, Ikram MK, Kawasaki R, Xie J, et al. Review: Retinal vascular caliber and the development of hypertension: A meta-analysis of individual participant data. Journal of Hypertension. 2014.
- 701. Stapleton F, Alves M, Bunya VY, Jalbert I, Lekhanont K, Malet F, et al. TFOS DEWS II Epidemiology Report. Ocular Surface. 2017.
- 702. Uchino M, Nishiwaki Y, Michikawa T, Shirakawa K, Kuwahara E, Yamada M, et al. Prevalence and risk factors of dry eye disease in Japan: Koumi study. Ophthalmology. 2011 Dec 1;118(12):2361–7.
- 703. Tan LL, Morgan P, Cai ZQ, Straughan RA. Prevalence of and risk factors for symptomatic dry eye disease in Singapore. https://doi.org/101111/cxo12210 [Internet]. 2021 Jan 1 [cited 2021 Aug 28];98(1):45–53. Available from: https://www.tandfonline.com/doi/abs/10.1111/cxo.12210
- 704. Kobia-Acquah E, Ankamah-Lomotey S, Owusu E, Forfoe S, Bannor J, Koomson JA, et al. Prevalence and associated risk factors of symptomatic dry eye in Ghana: A cross-sectional population-based study. Contact Lens Anterior Eye. 2021 Jan

²²³

16;101404.

- 705. Arciero J, Harris A, Siesky B, Amireskandari A, Gershuny V, Pickrell A, et al. Theoretical Analysis of Vascular Regulatory Mechanisms Contributing to Retinal Blood Flow Autoregulation. Invest Ophthalmol Vis Sci [Internet]. 2013 Aug 1 [cited 2021 Aug 28];54(8):5584–93. Available from: www.iovs.org
- 706. Jani B, Rajkumar C. Ageing and vascular ageing. Postgrad Med J [Internet]. 2006 Jun 1 [cited 2021 Sep 7];82(968):357–62. Available from: https://pmj.bmj.com/content/82/968/357
- 707. Ribera-Casado JM. Ageing and the cardiovascular system. Zeitschrift für Gerontol und Geriatr 1999 326 [Internet]. 1999 Dec [cited 2021 Sep 7];32(6):412–9. Available from: https://link.springer.com/article/10.1007/s003910050138
- 708. CV Physiology | Factors Regulating Arterial Blood Pressure [Internet]. [cited 2021 Dec 30]. Available from: https://www.cvphysiology.com/Blood Pressure/BP022
- 709. Tejero J, Shiva S, Gladwin MT. Sources of Vascular Nitric Oxide and Reactive Oxygen Species and Their Regulation. Physiol Rev [Internet]. 2019 [cited 2021 Sep 16];99:311–79. Available from: www.prv.org

13. Appendices

13.1. Appendix a. Ocular Surface Disease Index (OSDI)

	All of the	Most of	Half of	Some of	None of		
	time	the time	the time	the time	the time		
1. Eyes that are sensitive to	4	3	2	1	0		
light?							
2. Eyes that feel gritty	4	3	2	1	0		
3. Painful or sore eyes?	4	3	2	1	0		
4. Blurred vision?	NA						
5. Poor vision?	NA						

Have you experienced any of the following during the last week:

Have problems with your eyes limited you in performing any of the following during the last week:

	All of the	Most of	Half of	Some of	None of	
	time	the time	the time	the time	the time	
6. Reading?	4	3	2	1	0	NA
7. Driving at night?	4	3	2	1	0	NA
8. Working with a computer or	4	3	2	1	0	NA
bank machine (ATM)?						
9. Watching TV?	4	3	2	1	0	NA

Have your eyes felt uncomfortable in any of the following situations during the last week:

	All of the	Most of	Half of	Some of	None of	
	time	the time	the time	the time	the time	
10. Windy conditions?	4	3	2	1	0	NA
11. Places or areas with low	4	3	2	1	0	NA
humidity (very dry)?						
12. Areas that are air	4	3	2	1	0	NA
conditioned?						

13.2. Appendix b. Dry Eye Questionnaire (DEQ-5)

1.	Ques	tions abo	ut EYE DISCO	OMFORT:				
	a. During a typical day in the past month, how often did your eyes feel discomfort?							
	0	Never						
	1	Rarely	7					
	2	Somet	times					
	3	Freque	ently					
	4	Consta	antly					
	b. W	hen your	eyes felt disco	omfort, hov	v intense was t	his feeling o	of discomfort at the	e end of the day, within two
	h	ours of go	oing to bed?			_		
	Ν	ever	Not at all				Very	
	h	ive it	Intense				Intense	
	0		1	2	3	4	5	
2.	Oues	tions abo	ut EYE DRYN	ESS:				
	a. D	uring a ty	pical day in th	e past mon	th, how often d	id vour eves	feel drv?	
	0	Never	1 5	1	,	5 5	5	
	1	Rarely	7					
	2	Somet	imes					
	3	Freque	ently					
	4	Consta	antly					
	b. W	hen your	eyes felt dry, h	ow intense	was this feelin	g of dryness	s at the end of the da	y, within two hours of going
	to	bed?						
	Ν	ever	Not at all				Very	
	ha	ive it	Intense				Intense	
	0		1	2	3	4	5	
3.	Oues	tion abou	t WATERY E	YES:				
	Durii	ng a typic	al day in the pa	st month, I	10w often did v	our eyes loo	k or feel excessivel	v watery?
	0	Never	<i>v</i> 1	,	2	5		
	1	Rarely	7					
	2	Somet	imes					
	3	Freque	ently					
	4	Consta	antly					
			5			Sco	ore: $1a + 1b + 2a + $	2b + 3 = Total
							++	+=

13.3. Appendix c. Patient Examination Sheet

Patient ID: [FS dd mm yyyy] Examination date: [dd mm yyyy]
First name: Date of birth: <u>[dd mm yyyy]</u> Surname: Sex: [f m]
Ethnicity: ()Asian ()African ()Caucasian ()Chinese ()Latino ()Oriental ()South-Asian ()Other
Phone: [] Email: []
Optometry
Intraocular pressure left: 1. [] 2. [] 3. []
Intraocular pressure right: 1. [] 2. [] 3. []
Blood pressure
Systolic blood pressure: [] Diastolic blood pressure: []
Pulse per minute: []
Lifestyle
Smoking: ()never smoked ()smoked for [] years
()been smoking for [] years, [] cigarettes/day
Physical activity: [] times a week for [] hours with [low medium high] intensity Alcohol: [uk units/week] Healthy diet: [yes more-or-less no] Sleeping: [] hours per night
Medical conditions []Allergies []Eczema []Arthritis []Meningitis []Constipation []Cardiovascular disease
[]Uveitis []Anaemia []Epilepsy []Faintness []Hypertension []High blood pressure
[_]Headache [_]Stroke [_]Glaucoma [_]Depression [_]Heart attack [_]Shortness of breath
[]Diabetes []Asthma []Joint pain []Migraines []High cholesterol []Low blood
Other: 1. [] 2.[]
3.[]
Medications
1.[] 2.[] 3.[]
Family history of medical conditions [F M]Allergies [F M]Anaemia [F M]Arthritis [F M]Asthma [F M]Cardiovascular disease [F M]Constipation [F M]Depression [F M]Diabetes [F M]Eczema [F M]Low blood pressure [F M]Faintness [F M]Headache [F M]Heart attack [F M]Glaucoma [F M]High cholesterol [FIM]Loint pain [FIM]Meningitis [FIM]Migraines [FIM]Hveitis
[F M]Shortness of breath

[F M]Stroke
[F M]Epilepsy
[F M]Hypertension
[F M]High blood pressure

Other: 1. [F M] [] 3. [F M] []
2. [F M] [] 4. [F M] []	-
Anthropometric measurements Height: [cm] Body fat percentage: [0-100] Muscle mass: [kg] Bone mass: [kg] Weight: [kg] Body water percentage: [0-100]	
Endothelix aTR: [] TR: [] TMP-AUC: [] aTMP-AUC: []	
Bloods Total cholesterol: [mmol/L] Glucose: [mmol/L] HDL cholesterol: [mmol/L] Triglycerides: [mmol/L]	
Checklist DVA: [] Body fat%, muscle mass%, BMI: [] Visualis: [] Information consent signed?: [] Bloods: [] Overnight fasting? Endothelix: []	

www.nature.com/scientificreports

SCIENTIFIC REPORTS

natureresearch

OPEN Microvascular function and oxidative stress in adult individuals with early onset of cardiovascular disease

Hala Shokr¹, Irundika H. K Dias² & Doina Gherghel^{1*}

The current study aims to investigate retinal vascular function and its relationship with systemic anti-oxidative defence capacity in normal individuals versus those with early hypertensive changes according to the current ESC/ESH guidelines. Retinal microvascular function was assessed in 201 participants by means of dynamic retinal vessel analysis. Blood pressure, lipid panel, oxidized (GSH) & reduced glutathione (GSSG) were also evaluated for each participant. Individuals classed as grade 1 hypertension demonstrated higher retinal arterial baseline diameter fluctuation (p=0.0012), maximum dilation percentage (p=0.0007), time to maximum constriction (p=0.0003) and lower arterial constriction slope (p=0.0131). Individuals classed as high normal and grade 1 hypertension also demonstrated higher time to maximum dilation than individuals classed as optimal or normal. GSH levels correlated negatively with SBP, DBP and MBP values in all participants (p=0.0010; p=0.0350 and p=0.0050) as well as with MBP values in high normal and grade 1 hypertension (p=0.0290). The levels of GSSG correlated positively with SBP, DBP and MBP values in all participants (p=0.0410; p=0.0330 and, p=0.0220). Our results point to the fact that microvascular alterations can be identifiable at BP values still considered within normal values and go in parallel with the changes observed in the level of oxidative stress.

The assessment of the microvascular function represents an important part in establishing the pathophysiology but also the risk stratification of cardiovascular disease (CVD)¹. Indeed, endothelial dysfunction, one of the main culprits for the development of atherosclerosis, occurs much earlier at the microvascular than at the macrovascular level²³. Dynamic retinal vessel analysis (DVA) was identified as a useful measure of early changes that signal endothelial dysfunction at the microvascular level. This method can also be used to identify risk for future cardiovascular pathologies in individuals at risk for⁴⁻⁹ or already suffering from CVD^{10,11}. This is generally possible due to the fact that the retinal microvascular response to flicker provocation is, in part, dependent on nitric oxide (NO) release¹², and compromised NO homoeostasis is known to be a key factor in endothelial dysfunction at all vascular levels.

The assessment of retinal microvessels is used in research but also in clinical practice for diagnosis and follow-up of various CVD, including hypertension^{13,14}. Nevertheless, the overwhelming majority of protocols that look at retinal microvessels in the course of various CVD, use static imaging to detect abnormalities associated with various degrees of pathology. This is, however, useful only when CVD already established itself and not at earlier, pre-clinical stages. In order to determine the risk at earlier stages to allow preventive measures to be adopted, using the assessment of the retinal microvascular function represents a better alternative to structural imaging because it provides integrated and dynamic data to help establishing possible CVD risk.

In certain conditions that accelerate degradation of NO, such as high oxidative stress, microvascular dilation can be severely impaired¹⁵. In order to counteract such effects, the human body uses various anti-oxidative mechanisms including glutathione. Therefore, any condition associated with low levels of circulating glutathione result in a higher rate of oxidative reactions that contribute towards low NO bioavailability and, consequently, to an impaired microvascular function^{5,8,16,17}. We have already shown that retinal microvascular dilation and

¹Vascular Research Laboratory, School of Life and Health Sciences, Aston University, Birmingham, UK. ²Aston Medical Research Institute, Aston Medical School, Aston University, Birmingham, UK. *email: d.gherghel@aston. ac.uk

SCIENTIFIC REPORTS | (2020) 10:4881 | https://doi.org/10.1038/s41598-020-60766-0

www.nature.com/scientificreports

Check for updates

scientific reports

OPEN European Society of Cardiology/ European Society of Hypertension versus the American College of Cardiology/American Heart Association guidelines on the cut-off values for early hypertension: a microvascular perspective

H. Shokr & D. Gherghel[™]

The aim of this study was to investigate retinal and peripheral microvascular function in asymptomatic individuals that fall into different BP groups when using either the ESC/ESH or the ACC/AHA guidelines. Retinal and peripheral microvascular function was assessed in 358 participants by means of dynamic retinal vessel analysis and digital thermal monitoring, respectively. Blood pressure and lipid panel were also evaluated. Retinal vascular function measured in all groups belonging to the ACC/ASH classifications were within the normal values for age-matched normal population. Individuals classed as grade 1 hypertension according to the ESC/ESH guidelines, however, exhibited a significantly decreased artery baseline (p = 0.0004) and MC (p = 0.040), higher slope_{AD} (p = 0.0018) and decreased vein MC (p = 0.0446) compared to age matched normal individuals. In addition, they also had significant lower artery baseline, artery BDF, MD and MC than individuals classed as stage 1 hypertension based on the ACC/ASH guidelines (p=0.00022, p=0.0179, p=0.0409 and p=0.0329 respectively). Peripheral vascular reactivity (aTR) was lower in ESC /ESH grade I compared to those graded ACC/ASH stage I hypertension (p = 0.0122). The conclusion of this study is that microvascular dysfunctions is present at multiple levels only in individuals with ESC/ESH grade 1 hypertension. This observation could be important when deciding personalised care in individuals with early hypertensive changes.

One of the main differences between the latest guidelines published in 2018 by the European Society of Cardiology/European Society of Hypertension (ESC/ESH) and those published in 2017 by The American College of Cardiology/American Heart Association (ACC/AHA) is the cut-off for what is considered elevated blood pressure (BP) and the first stage of hypertension (HTN) diagnosis¹. As it can be expected, such difference is bound to have a high impact on clinical diagnosis and management of HTN. By using the ACC/AHA guidelines, the number of patients diagnosed with having this disease increased significantly and so did the pressure on the health system and economy. In addition, such change in practice has also a significant impact on the patients' physical and psychological wellbeing¹. Nevertheless, it would seem sensible that, in individuals with higher risk for cardiovascular disease (CVD), early diagnosis and interventions are applied at lower BP values. Indeed, both ECC/ESH guidelines and ACC/AHA have recognised this fact and recommended considering treating patients with high CVD risk at a BP threshold lower than their current cut-off for grade/stage 1 HTN². In this way,

Vascular Research Laboratory, College of Health and Life Sciences, Aston University, Birmingham B4 7ET, UK. [™]email: d.gherghel@aston.ac.uk

Scientific Reports	(2021) 11:3473	https://doi.org/10.1038/s41598-021-83096-1	natureportfolio

Acta Ophthalmologica

Аста Орнтнаемогодіса 2021 —

Dry eye disease is associated with retinal microvascular dysfunction and possible risk for cardiovascular disease

Hala Shokr,^{1,2} James S. Wolffsohn,² Sonia Trave Huarte,² Emily Scarpello^{1,2} and Doina Gherghel^{1,2}

¹Vascular Research Laboratory, Ophthalmic Research Group, College of Health and life Sciences, Aston University, Birmingham, UK

²Optometry and Vision Sciences Research Group, College of Health and life Sciences, Aston University, Birmingham, UK

ABSTRACT.

Purpose: To explore the presence of microvascular endothelial dysfunction as a measure for early cardiovascular disease in individuals diagnosed with dry eye disease (DED) as compared to age-matched normal controls.

Methods: Systemic blood pressure, Body Mass Index, intraocular pressure, blood levels of glucose (GLUC), triglycerides, cholesterol (CHOL), high-density lipoprotein cholesterol (HDL-C), and low-density lipoprotein cholesterol (LDL-C)] as well as retinal and peripheral microvascular function were assessed in twenty-five 35–50 year olds with diagnosed with DEDa (using the TFOS DEWS II criteria) and 25 age and sex-matched controls.

Results: After controlling all the influential covariates, individuals diagnosed with DED exhibited significant lower retinal artery baseline (p = 0.027), artery maximum diameter (p = 0.027), minimum constriction (p = 0.039) and dilation amplitude (p = 0.029) than controls. In addition, the time to reach the vein maximum diameter was significantly longer in the DED patients than in normal controls (p = 0.0052). Only in individuals diagnosed with DED, artery maximum constriction correlated statistically significantly and positively with HDL-C blood levels (p = 0.006). Similarly, artery slope_{AD} correlated positively with T-CHOL and LDL-C (p = 0.006 & 0.011 respectively). Additionally, artery baseline diameter and maximum constriction were significantly and negatively correlated to T-CHOL/HDL-C ratio (p = 0.032 and p = 0.013 respectively) in DED individuals only.

Conclusions: Individuals with positive diagnosis of DED exhibit abnormal retinal microvascular function and possible higher risk for CVD.

Key words: cardiovascular disease - dry eye disease - microvascular function - retinal vessels

Acta Ophthalmol.

© 2021 The Authors. Acta Ophthalmologica published by John Wiley & Sons Ltd on behalf of Acta Ophthalmologica Scandinavica Foundation. The is no necessar particle under the terms of the Creative Commons Attribution License, which exemits use

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, listribution and reproduction in any medium, provided the original work is properly cited.

doi: 10.1111/aos.14782

Introduction

Dry eye disease (DED) represents a, multifactorial, chronic and debilitating pathology of the ocular surface characterized by loss of homeostasis of the tear film and accompanied by ocular symptoms. Tear film instability and hyperosmolarity, ocular surface inflammation and damage as well as neurosensory abnormalities play etiological roles in this disease (Craig et al. 2017).

In addition to other risk factors, DED has previously been associated with dyslipidaemia, a group of metabolic abnormalities characterized by any or a combination of the following: raised low-density lipoprotein cholesterol (LDL-C), raised total cholesterol (TC), raised triglycerides (TG) and low high-density lipoprotein cholesterol (HDL-C) (Musunuru 2010). Indeed, as lipid homeostasis is important for the stability of the tear film, the association between dyslipidaemia and DED is entirely justified. Moreover, disruptions of cholesterol biosynthesis are also associated with sebaceous/ Meibomian gland (MG) dysfunctions (Bu et al. 2019), another cause of tear film instability and dry eye. Dyslipidaemia also represents a significant risk factor for cardiovascular disease (CVD) especially due to its contribution in the pathogenesis of atherosclerosis in medium-sized and large arteries but also at the microvascular level (Pereira 2012; Padró, Vilahur & Badimon 2018). This affects not only the anatomy of these vessels but, most importantly, their function. Indeed, at the functional level, it impairs endothelium-dependent vasodilation because of defects on nitric oxide (NO) bioavailability (Padró, Vilahur & Badimon 2018). This has catastrophic effects on the balance between the physiological vascular dilatory and constrictory states, which, in turn, will also affect other important circulatory

1 -





Oxysterols and Retinal Microvascular Dysfunction as Early Risk Markers for Cardiovascular Disease in Normal, Ageing Individuals

Hala Shokr¹, Irundika HK Dias^{2,*,†} and Doina Gherghel^{1,*,†}

- ¹ Vascular Research Laboratory, College of Health and Life Sciences, Aston University, Birmingham B4 7ET, UK; Shokrh@aston.ac.uk
- ² Aston Medical Research Institute, Aston Medical School, Aston University, Birmingham B4 7ET, UK
 ⁴ Correspondence: diashki1@aston.ac.uk (LH.D.); d.gherehel@aston.ac.uk (D.G.);

Abstract: The aim of the present paper is to assess the relationship between oxysterol levels and

- Correspondence: diashki1@aston.ac.uk (I.H.D.); d.gherghel@aston.ac.uk (D.G.);
 Tel.: +44-(0)121-204-4678 (I.H.D.); +44-(0)121-204-4120 (D.G.); Fax: +44-(0)-121-204-4220 (D.G.)
- + Both senior authors contributed equally to this work.
- Both senior authors contributed equally to this w

check for updates

Citation: Shokr, H.; Dias, I.H.; Gherghel, D. Oxysterols and Retinal Microvascular Dysfunction as Early Risk Markers for Cardiovascular Disease in Normal, Ageing Individuals. *Antioxidants* **2021**, *10*, 1756. https://doi.org/10.3390/ antiox10111756

Academic Editors: Nicola King and M.-Saadeh Suleiman

Received: 26 September 2021 Accepted: 30 October 2021 Published: 3 November 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). retinal microvascular function in individuals of various age groups, free of clinically evident diseases. Forty-two apparently healthy individuals were included in the present study (group 1: 19–30 years, group 2: 31–50 years, and group 3: 51–70 years). Retinal microvascular function was assessed using the dynamic retinal vessel analyzer (DVA, IMEDOS GmbH, Jena, Germany). Fasting plasma was obtained from all subjects and quantification of monohydroxy and dihydroxy oxysterols assessment was performed using LC-MS/MS following reverse phase chromatography. A Griess assay was used to evaluate the Nitric Oxide (NO) concentration in all individuals. The glutathione redox ratio was also analyzed by means of whole blood glutathione recycling assay. In all participants, the levels of 7-Ketocholesterol, 25-hydroxycholesterol and 7 β -hydroxycholesterol correlated significantly and positively with the time to maximum arteriolar dilation. In addition, 25-hydroxycholesterol and 7 β -hydroxycholesterol negatively correlated to the percentage of maximum arteriolar dilation. A negative correlation was observed for 27-hydroxycholesterol and 7 β -hydroxycholesterol with microvascular arteriolar constriction. These results suggest that, with age, abnormal oxysterol levels correlate with early changes in microvascular bed function. This relationship could signal early risk for cardiovascular diseases (CVDs) in an ageing population.

Keywords: oxidative stress; oxysterols; cardiovascular disease risk

1. Introduction

The assessment of microvascular function gained a lot of clinical interest since endothelial dysfunction, the main culprit for CVDs, was found to occur earlier at the microvessel level than the macrovessel level [1]. Accordingly, many methods were developed to evaluate functional microvascular responses to biological stimuli in different vascular beds [2]. One of these methods, the dynamic retinal vessel analysis (DVA), was highlighted as one of the most sensitive and reliable methods to measure early vascular changes that point out early endothelial dysfunction and future risk of cardiovascular pathologies in individuals with and without overt clinical symptoms [3–5]. Moreover, it has also been shown that retinal microvascular function is modified by age [6]. Nevertheless, although this is a consistently demonstrable effect, little is known about what factors in particular directly determine this outcome. We have previously demonstrated that high levels of circulating reactive oxygen species (ROS) have a direct effect on retinal microvascular function [7,8]. Indeed, the evidence suggests that oxidative stress-induced cellular senescence is a key influencer of age-induced mechanical and structural vascular dysfunction [9–12] as well as of the development and progression of CVDs [13–17]. Additionally, high levels of

Antioxidants 2021, 10, 1756. https://doi.org/10.3390/antiox10111756

https://www.mdpi.com/journal/antioxidants

MDPI